		•			
•	Nation a l Library of Canada	Bibliothèque nationale du Canada	CANADIAN THESES	THÈSES CANADIENNES SUR MICROFICHE	
	.				
		1			
AME OF	AUTHOR NOM DE L	AUTEUR LEONHAD	HENRY KING	· · · · · · · · · · · · · · · · · · ·	·
ITLE OF	THESIS / TITRE DE LA	THESE Hn allest	ional analysis c	f student	
		achievement 1	related behavior	and the	
		expectancy	effect.		
JNIVERS	NTY/UNIVERSITÉ	Unicersity of	Alberta		<u></u>
GREE GRADI	FOR WHICH THESIS W	AS PRESENTED E THÈSE FUT PRÉSENTÉE	Doctor of Philos	crhy	
EAR TH	IS DEGREE CONFERRE	D-ANNÉE D'OBTENTION DE CE G	RADE 1979.	· · · ·	
AME O	F SUPERVISOR/NOM (DU DIRECTEUR DE THÈSE	Dr. A MacKay		
			/		

Permission is hereby granted to the NATIONAL LIBRARY OF CANADA to microfilm this ties is and to lend or sell copies of the film,

-43448

The author reserves other publication rights, and neither the thesis nor extensive extracts from it may be printed or otherwise reproduced without the author's written permission. L'autorisation est, par la présente, accordée à la BIBLIOTHÈ- QUE NATIONALE DU CANADA de microfilmer cette thèse et L de prêter ou de vendre des exemplaire: du film.

L'auteur se réserve les autres droits de publication; ni la thèse ni de longs extraits de cella-ci ne doivent être imprimés ou autrement reproduits sans l'autorisation écrite de l'auteur.

DATED/DATE_May 17, 1979_SIGNED/SIGNE_____

PERMANENTLADDRESS/RESIDENCE FIXE Churchlands College Pearson Street Churchlands AUSTRALIA W.A. 6018

National Library of Canada

Cataloguing Branch Canadian Theses Division

Ottawa, Canada K1A 0N4

.

NOTICE

The quality of this microfiche is heavily dependent upon the quality of the original thesis submitted for microfilming. Every effort has been made to ensure the highest quality of reproduction possible

If pages are missing, contact the university which granted the degree

Some pages may have indistinct print especially if the original pages were typed with a poor typewriter ribbon or if the university sent us a poor photocopy

Previously copyrighted materials (journal articles, published tests, etc.) are not filmed

Reproduction in full or in part of this film is governed by the Canadian Copyright Act. R.S.C. 1970, c. C-30 Please read the authorization forms which accompany this thesis

THIS DISSERTATION HAS BEEN MICROFILMED EXACTLY AS RECEIVED

Bibliothèque nationale du Canada

Direction du catalogage Division des thèses canadiennes

AVIS

La qualité de cette microfiche dépend grandement de la qualité de la thèse soumise au microfilmage. Nous avons tout fait pour assurer une qualité supérieure de reproduction

S il manque des pages, veuillez communiquer avec L'université qui a conféré le grade

La qualité d'impression de certaines pages peut laisser à désirer, surtout si les pages originales ont été dactylographiées à l'aide d'un ruban usé ou si l'université nous a fait parvenir une photocopie de mauvaise qualité

Les documents qui font déjà l'objet d'un droit d'auteur (articles de revue, examens publiés, etc.) ne sont pas microfilmés

La reproduction, même partielle, de ce microfilm est soumise à la Loi canadienne sur le droit d'auteur, SRC 1970, c. C-30. Veuillez prendre connaissance des formules d'autorisation qui ac ompagnent cette thèse

LA THÈSE A ÉTÉ MICROFILMÉE TELLE QUE NOUS L'AVONS RECUE

THE UNIVERSITY OF ALBERTA

AN ATTRIBUTIONAL ANALYSIS OF STUDENT ACHIEVEMENT-RELATED BEHAVIOR AND THE EXPECTANCY EFFECT

Ъy



LEONARD HENRY KING

A THESIS

SUBMITTED TO THE FACULTY OF GRADUATE STUDIES AND RESEARCH IN PARTIAL FULFILMENT OF THE REQUIREMENTS FOR THE DECREE OF DOCTOR OF PHILOSOPHY

DEPARTMENT OF EDUCATIONAL ADMINISTRATION

EDMONTON, ALBERTA

FALL, 1979

THE UNIVERSITY OF ALBERTA FACULTY OF GRADUATE STUDIES AND RESEARCH

The undersigned certify that they have read, and recommend to the Faculty of Graduate Studies and Research, for acceptance, a thesis entitled An Attributional Analysis of Student Achievement-Related Behavior and the Expectancy Effect submitted by Leonard Henry King in partial fulfilment of the requirements for the degree of, Doctor of Philosophy in Educational Administration.

Supervisor

, Thail

Murray Smith

E mklos

External Examiner

Date May 16, 1979....

ABSTRACT

This study focused upon the attributions of student achievement-related behavior as a function of the expectancy effect. The purposes of the study were to identify the characteristics of students' overt and covert behavior during the learning of mathematics, to gain some insight into the conceptions of self-performance which students appear to develop in classroom achievement situations and, from an attributional perspective, to discover some of the phenomena which characterize the teacher expectancy effect.

Ten consecutive mathematics lessons were videotaped in a Grade 6 classroom. These were used for follow-up stimulated recall interviews. Four target students reported their interactive thoughts and feelings pertaining to self-performance in mathematics. The teacher also reported her interactive thoughts. Transcripts of each student's and the teacher's reported thoughts and feelings were prepared. The videotape recordings of the lessons also facilitated later observation and analysis of teacher and student overt behavior.

Both student behavior data and teacher-student dyadic interaction process data were gathered. Through the development of category systems for analyzing student behavior data, student overt behavior and students' interactive thoughts and feelings were categorized, quantified, and described. Other student covert behavior data, including causal explanations of behavior, were analyzed by qualitative means. At an observable level, teacher-student dyadic interaction process data were coded using a low inference observation system, while teacher interactive thoughts pertaining to students and

iv

students' performances were-submitted to a qualitative analysis.

For each of the four students a characterization of their conception of self-performance was interpreted from the student behavior date. The students were observed to be mainly on-task and reported self-oriented thoughts more frequently than subject matter oriented thoughts. Causal explanations of student behavior were discerned from identified clusters of chains of comments and from basic underlying ideas and beliefs. These indicated that a student's approach to learning seened to be a function of the expectancy of success or failure, the perceptions of tasks, the affective state, the perceptions of self-performance in relation to others, and the success of the teacher's impressions of him or her. Variations within these findings were recognized between the two "successful" students and the two "less successful" students.

The study established tentative relationships among and between the students' conceptions of self-performance, their causal perception of success and failure, their achievement-related behavior, the teacher's causal perceptions of the students' performances, the teacher's performance expectations of the students; and the teacher's behaviors toward the students. The widespread acceptance of effort attribution as the criterion of success or failure and the pervasiveness of the motivation of the report card seemed to act as underlying mechanisms for variations between the achievement-related behaviors of the successful students and the less successful students.

Implications for teaching arising from the findings were presented, along with some recommendations for further research.

ACKNOWLEDGEMENTS

Q.

The development and completion of this research project was dependent on the help, advice, and support of many people. I wish to acknowledge my appreciation and indebtedness to them:

Dr. D. A. MacKay who, as Supervisor, so generously gave considerable helpful advice and inspiration to the inquiry and provided extensive and warm encouragement.

Dr. M. Horowitz who so readily supported the project and provided such insightful comments and suggestions; Dr. M. Smith whose enthusiasm and waluable ideas contributed significantly to the project; Dr. E. Miklos and Dr. L. Stewin for their constant interest and helpful reactions.

Dr. T. Good, Director, Center for Research in Social Behavior, University of Missouri, for serving as external examiner.

To the teacher of the classroom in which the research was conducted for her considerable assistance given in such a professional manner, and to the students of the classroom for their willing gooperation. To the teachers and students of the school in which the pilot study was conducted for their tolerance and generous help, and to the principals of both schools for their hospitality.

To my wife, Heather, for her loving support, understanding, and encouragement, and for her professionalism in the typing of this thesis, and to my children, Philip, Susan and Andrew, who provided love and consideration. They have all shared in so many ways in the completion of the project. I dedicate this thesis to them.

vi

TABLE OF CONTENTS

1

-

`~

CHAPTER		FAGE
I.	INTRODUCTION AND STATEMENT OF THE PROBLEM	1
	Introduction	1
	Need for the Study	5
	Purposes of the Study	?
4 k.	Research Problems	9
	Significance of the Study	:0
	Definitions	
	Outling of the Study	
ΙΙ.	REVIEW OF RELATED LITERATURE AND RESEARCH	14.
	Werview	• E • _
	The Fhilosophical Setting of the Study	+ C + /
1	Research in TeachingAn Overview	18
	Classroom Social Interaction ResearchThe Classroom Climate Tradition	ר ²²
	The Early Studies	22
	The Studies of the 1950's and 1960's	20
	The Current State of Classroom Interaction Research	28
	Summary	34
	Teacher Expectation Research	34
	An Historical Overview	32
	The Emerging Patterns in Expectation Studies	35
	Classroom Interactions and the Expectancy Effect	41
	Summary	46

CHAPTER

u

HAPTER	PAGE
Attributional Analyses of Achievement Behavior	. 47
Attribution Processes	. 47
Attributional Model of Achievement Motivation	. 49
Teachers' and Students' Attributions of Students' Performance	• 59
The relationships between students' attributions of their performance and their achievement-related behavior	· ()()
The relationships between teachers' attributions of students' performances and their behavior toward the students \dots	. t.
The relationships between teachers' behavior toward the students and the students' attributions of their performance	•
The significant other	• • • •
Summary	
Summary	
III. DESIGN AND PROCEDURES	$\frac{1}{2}$
Overview	75
Design	75
The Subjects	77
Selection of the Teacher and Student Subjects	77
The Lesson Context	80
Specific Problems	82
Assumptions	54
Limitations	85
Data and Data Sources	85
Videotapes of Lessons	86

.

	CHAPTER					PAGE
	Interviews • · · · · · · · · · · · · · · · · · ·		•	•		86
•	The Low Inference Observation System: A Revision of the Expanded Brophy-Good Teacher-Pupil Dyadic Interaction					د
•	Classroom Observation System		•	•		57
	Field Notes	• •		•		29 .
	Research Methodology		•	•	•••	34
	Stimulated Recall Methodology	• .			. .	·大)
	The Observation-Interview Methodology					9. '
	Validity of the Research Data					(30)
	Prases in the Study \ldots \ldots			•		1 4 h a
	Pilot Studies		•			19.24
	Fimiliarization Prase					101
	Data Collection Phase	• •		•		• J 3
	"Sumary		•	• •		::0
	IT. DATA ANALYSIG	• •		• •	•	4 4 4 4 4 1
	Overview	• •		• •	•	
	Analysis of Student Overt Behavior	•••	•	• •		4 4 V 4 4 4
	The Development of the Category System					112
	Procedures for Coding Student Overt Bena	vio	r	• •		116
	Reliability					1:6
	Content Analysis System of Student Interactive Thoughts				÷	N.: 17
	The Development of the Category System	•••		• •		:19
	Reliability	· .				122
	Attributional Analyses of Student Covert Behavior Data					4 , 4 4

.'

	,	
CHAPTER		FAGE
	Gaugal Explanations of Behavior	127
	The Underlying Covert Behavior of Students	129
	The Characterization of Student Conception of Self-Performance	131
	Analysis of Dyadic Interaction Data	131
	Interview Data	132
	The Observation Data	134 ~
	The Summary Analysis	135
V. CA	SE STUDIES : TINA AND MIKE	138
	Organization of the Case Studies	:39
	TINA Biographical Information	140
	Quantitative Analysis of Behavior	142
	Overt Behavior	142
	Covert Behavior 🔉	147
	Summary of Quantitative Analysis of Behavior	152
	Attributional Analysis of Self-Performance Covert Behavior	153
	Causal Explanations of Tina's Behavior	153
	Approach to learning	154
	Task difficulty	155
	Specific performances	157
	Public participation	158
	Perceptions of other students' behavior	159
	Tina's Basic Ideas and Beliefs	160
	Reasons for success	160

. X

CHAPTER	•	F	AGE
	Perceptions of her own mathematics ability	•	162
	Tina's perception of the teacher's performance expectations of herself		103,
	Perceptions of other students' performances	•	164
	Perceptions of tasks and activities	•	165
	A Characterization of Tina's Conception of Self-Performance		106
	Analysis of the Dyadic Interactions Between Tina and the Teacher	•	168
	Analysis of the Overt Behaviors	•	169
	The Overall Analysis of the Dyadic Interactions		169
Ċ.	Summary Analysis	•	:75
	MIKE Biographical Information		177
	Quantitative Analysis of Behavior	•	179
	Overt Behavior	•	179
	Covert Behavior	•	183
	Summary of Quantitative Analysis of Behavior	•	188
	Attributional Analysis of Self-Performance - Covert Behavior		189
•	Causal Explanations of Mike's Behavior	•	189
*3	Task perceptions and behaviors	•	189
	Specific performances	•	191
•	Public participation	•	192 •,
	Perceptions of other students' behavior	•	193
4 .	Mike's Basic Ideas and Beliefs	•	194
	Self-expectations		194

٩

.

`、

CHAPTER

-

.

PAGE

.

,

٠

Mike's perceptions of the teacher's performance expectations of himself	104
	196
Perceptions of other students' performances	196
Percaptions of tasks and activities	19 7
A Characterization of Mike's Conception	198
Analysis of the Dyadic Interactions Between Mike and the Teacher	200
Analysis of the Overt Behaviors	200
The Overall Analysis of the Dyadic Interactions	● ¥+
Summary Analysis	207
•A Summary Comparison of Tina and Mike	• ೧೪
VI. CASE STUDIES : LISA AND GREG	210
LISA. Biographical Information	2:0
Quantitative Analysis of Behavior	212
Overt Behavior	212
Covert Behavior	216
Summary of Quantitative Analysis of Behavior	222
Attributional Analysis of Self-Performance Covert Behavior	222
Causal Explanations of Lisa's Behavior	222
Approach to learning	223
Task perceptions and behaviors	225
Specific performances	227
Public participation	228
Perceptions of other students', behavior	231

i

.

.

.

.

•

Lisa's Basic Ideas and Beliefs	2 3 2
erceptions of her own mathematics ability	2 3 2
Lisa's perceptions of the teacher's performance expectations of herself	235
Perceptions of tasks and activities	237
A Characterization of Lisa's Conception of Self-Ferformance	238
Analysis of the Dyadic Interactions Between Lisa and the Teacher	240
Analysis of the Overt Behavior	240
The Overall Analysis of the Dyadic Interactions	244
Summary Analysis	291
GREC Biographical Information	254
$_{\alpha}$ uantitative Analysis of Behavior $\ldots \ldots \ldots$	257
Overt Behavior	257
Covert Behavior	it i
Summary of Quantitative Analysis of Behavior	260
Attributional Analysis of Self-Performance Covert Behavior	260
Causal Explanations of Greg's Behavior	267
Approach to learning	7ھ
Task perceptions	269
Specific performances	27:
Public participation	۲ <u>4</u>
Perceptions of other students' behavior	275

CI	HA	£,	Τ	El	R
----	----	----	---	----	---

•

.

•	
	FAGE

.

Greg's Basic Ideas and Heliefs	27É
Self-expectations	276
Greg's perceptions of the teacher's performance expectations of himself	.278
Perceptions of other students' performances	279
Perceptions of tasks and activities	280
A Characterization of Greg's Conception of Self-Performance	281
Analysis of the Dyadic Interactions Between Greg and the Teacher	284
Analysis of the Overt Behavior \ldots	284 ·
The Overall Analysis of the Dyadic Interactions	2 3 8
Summary Analysis	L 94
A Summary Comparison of Lisa and Greg	294
VII. SUMMARY, CONCLUSIONS, IMPLICATIONS, AND RECOMMENDATIONS FOR FURTHER RESEARCH	29É
Summary of the Study	29
Conclusions	290
Implications for Teaching and Learning	3 2.8
Recommendations for Further Research	334
Student Interactive Thoughts	334
Teacher-Student Dyadic Interactions	33c
The Teacher's Interactive Thought Processes	337

APPENDICES		
A. THE INTELLECTUAL ACHIEVEMENT RESPONSIBILITY SCALE	355	
B. GUIDELINES FOR STIMULATED RECALL INTERVIEWS	360	
C. PRE-TEST AND POST-TEST INTERVIEW SCHEDULES	366	
D. SUMMARY OF CATEGORIES IN THE REVISION OF THE EXPANDED BROPHY-GOOD TEACHER-PUPIL DYADIC INTERACTION CLASSROOM OBSERVATION SYSTEM	3 08	
E. POST-OBSERVATION OPEN-ENDED STATEMENT COMPLETED BY THE TEACHER	374	
F. SAMPLE OF LESSON OVERVIEW	376	
G. CONTENT ANALYSIS OF STUDENTS' INTERACTIVE THOUGHTS (CASSIT)	379	

.

3

a--

LIST OF TABLES

TABLE		PAGE
1.	Summary of Objective Type Information Used in Selection of Target Students	81
2.	Distribution of Grouping Modes Used When Interviewing Target Students and Decoy Students During the Data Collection Phase	107
3.	Intercoder and Intracoder Measures of Reliability in the Use of the Category System for Analyzing Student Overt Behavior	118
4.	Intercoder and Intracoder Measures of Reliability	125
Ş.	Intercoder and Intracoder Measures of Reliability in the Use of the Low Inference Classroom Observation System	:36
έ.	Fercentage Distributions of Student Overt Behaviors in Each Lesson Situation Category for Each Lesson : Tina	143
7.	Percentage Distributions of Means for Student Overt Behaviors Within Each Category of Lesson Situations : Tina	1444
5.	Fercentage Distributions of Student Overt Behaviors for Total Lesson Time : Tina	145
9.	Percentage Distributions of Student Thoughts Over CASSIT Categories for Each Lesson : Tina	148
:0.	Within-Category Percentage Distributions of Sub-categories of Selected Student Thought Categories of CASSIT : Tina	150
4 . 2 . *	Frequency Distribution of Dyadic Interactions Between Tina and Teacher	170
:2.	Forcentage Distributions of Student Overt Behaviors in Each Lesson Situation Category for Each Lesson : Mike	180
13.	Percentage Distributions of Means for Student Overt Behaviors Within Each Category of Lesson Situations : Mike	181

14.	Percentage Distributions of Student Overt Behaviors for Total Lesson Time : Mike		•	•	•	•	182
15.	Percentage Distributions of Student Thoughts Over CASSIT Categories for Each Lesson : Mike		•	•		•	184
16.	Within-Category Percentage Distributions of Sub-categories of Selected Student Thought Categories of CASSIT : Mike	,	•	•	•	• .	186
17.	Frequency Distribution of Dyadic Interactions Between Mike and Teacher		•		•	•	201
18.	Percentage Distributions of Student Overt Behaviors in Each Lesson Situation Category for Each Lesson : Lisa	•	•	•	•		213
19.	Percentage Distributions of Means for Student Overt Behaviors Within Each Category of Lesson Situ a tions : Lisa	•	•				214
20.	Percentage Distributions of Student Overt Behaviors for Total Lesson Time : Lisa	•		•	•		215
21.	Percentage Distributions of Student Tho ughts Over CASSIT Categories for Each Lesson : Lisa		•	•	•	•	217
22.	Within-Category Percentage Distributions of Sub-categories of Selected Student Thought Categories of CASSIT : Lisa	•	•	•	•	•	219
23.	Frequency Distribution of Dyadic Interactions Between Lisa and Teacher	•	•	•	•	•	241
24.	Frequency Distribution of the Duration of Student-Initiated Private Dyadic Interactions Between Lisa and the Teacher						244
25.	Percentage Distributions of Student Overt Behaviors in Each Lesson Situation Category for Each Lesson : Greg	•	•			•	258
26.	Percentage Distributions of Means for Student Overt Behaviors Within Each Category of Lesson Situations : Greg			•		•	259
27.	Percentage Distributions of Student Overt Behaviors for Total Lesson Time : Greg	•				•	260

TABLE	-	PAGE
28.	Percentage Distributions of Student Thoughts Over CASSIT Categories for Each Lesson : Greg	262
29.	Within-Category Percentage Distributions of Sub-categories of Selected Student Thought Categories of CASSIT : Greg	264
30 .	Frequency Distribution of Dyadic Interactions Between Greg and Teacher	285
31.	Collation of Summary Analyses of the Case Studies	299

•

۲

4

•

.

.

.

۶

.

xviii

.

.

LIST OF FIGURES

FIGURE	•	PAGE
1.	Weiner's Classification Scheme for Perceived Determinants of Achievement Behavior	51
2.	An Attributional (Expectancy-Value) Model of Achievement Motivation	52
3.	Schematic Representation of Research Results of Differential Effects of Success/Failure on High/Low Achievement Need Persons	57
4.	A Model for Expectation Communication and Behavior Influence	66
5.	Framework for Recording Categories of Student Overt Behavior in Relation to Categories of Lesson Situations	115
5.	Representation of the Interrelationships Among and Between Different Sources of Data for an Interaction Episode	133
7.	Nature of Public Dyadic Interactions Between Tina and Teacher and Their Proportions	171
8.	Nature of Private Dyadic Interactions Between Tina and Teacher and Their Proportions	172
9.	Nature of Public Dyadic Interactions Between Mike and Teacher and Their Proportions	202
10.	Nature of Private Dyadic Interactions Between Mike and Teacher and Their Proportions	203
11.	Nature of Public Dyadic Inter actions Between Lisa and Teacher and Their Proportions	242
12.	Nature of Private Dyades Interactions Between Lisa and Teacher and Their Proportions	243
13.	Nature of Public Dyadic Interactions Between Greg and Teacher and Their Proportions	286
14.	Nature of Private Dyadic Interactions Between Greg and Teacher and Their Proportions	287

Chapter I

INTRODUCTION AND STATEMENT OF THE FROBLEM

Introduction

Research in teaching appears to be on the threshold of change. While the intention of this research has always been to investigate and identify the criteria of teaching effectiveness, the directions have ranged across such areas as teacher characteristics, teacher tehavior correlates of student achievement and, currently, the mental life of teachers. But now the research seems to be characterized by a more profound change than one of focus of interest. Seeking to understand the mental life of teachers seems to have emerged at the same time as researchers have increasingly turned toward and embraced some of the conceptualizations and methodologies of research in related social behavior disciplines.

The motive for change in the research in teaching seems to have been provoked by a general call for more consequential findings than those derived from the research to date. When reviewing this research, writers such as Dunkin and Biddle (1974), Rosenshine (1976), and Bennett (1977) all commented on the inadequacy of conceptualizations of variables and on the diversity of methodological issues. Concerns such as these seemed relevant, given that much of the research until recently had been limited to one facet of social behavior--that which was observable.

The present research concern with teacher thought processes seems typical of a wider move by researchers and theorists such as Doyle (1978), Winne and Marx (1977), and Yinger (1978) to reconceptualize the research on teaching. As Winne and Marx (1977:076) state, "the considerably richer literature of research on human learning and information processing see, too germane to research on teaching to be ignored." From another perspective, the bidirectional nature of classroom influences (Fiedler, 1975; Winne & Marx, 1977) and the impact of student aptitudes on teacher effects (Frophy & Evertson, 1970; Lockheed, 1975) introduce what would appear to be other relevant and significant variables. As a consequence, the very recent trend for research in teaching to incorporate the study of the mental life of students as well as of the teacher has prompted a renewed awareness by researchers of relevant variables normally associated with such disciplines as child development, learning psychology and social psychology.

Alongside this reconceptualization of research variables there seems to have been a growing willingness by researchers in teaching to adopt different research methodologies. With research in teaching being increasingly undertaken within the context of regular functioning classrooms, the phenomena under investigation often become the behavior and actions of people in their normal everyday lives. Not surprisingly, therefore, research designs, observational techniques, and data gathering processes which are usually employed in such social behavior disciplines as anthropology, social psychology, and sociology, are now being reconsidered and often used in recent research in teaching.

One domain of the research in teaching which reflects this pattern of change is that of the teacher expectation studies. In accord with many writers, Schmuck and Schmuck (1975(25) and Frephy and Evertson (19765:48) point out that expectations characteristically are defined as predictions of how the self and others will behave. These expectations become translated into teacher expectancy effects when, as Dusek (1975:660) states, "functional relationships [exist, between teachers' own self-generated expectances and atminents' academic achievement."

As definite trends in the findings from the teacher expectation studies have emerged, so the focus of interest in the research has changed. Reviews of research evidence by Frophy and Good (1974), H. Gooper (1975) and Dusek (1975) reveal that expectancy effects are reality in most classrooms. Furthermore, Brophy and Good (1974) indicate that the nature and extent of these expectancy effects in classrooms seem to depend on individual differences among teachers. Of recent significance, however, are studies that attempt to explain the dynamic processes that underlie the expectancy effects.

Recently, Braun (1976), Brophy and Good (1970), H. Cooper (1975), and Lockheed (1975), among others, have conceptualized models which hypothesize a relationship between process expectation effects, that is, the nature and extent to which the teacher varies his behavior when relating to or working with individual students, and product expectation effects, that is, the differential student academic achievements resulting from the teacher-held expectations. Such models postulate the origins of teacher expectations, the Interaction phase between the teacher and students, and the reactions of students to the expectations. Hence, the primary focus of many recent studies has been on the process expectation effects.

As researchers review their conceptualizations in the search for an explanation of the dynamic processes underlying the expectancy effect, certain patterns have evolved. The recognition of the significance of the cyclical two-way nature of classroom interactions has resulted in many expectation studies being focused on student learning variables as much as teacher behavior variables. Given that students tend to be striving constantly for self-esteem and selfrespect, there seems every reason to believe that the interpersonal relationships in a classroom exert considerable influence on the academic performance of those involved. Consequently, expectation researchers have turned to the fields of educational psychology and social psychology for variables which will help in inderstanding some of these educational and interpersonal processes.

Attributional analyses of classroom social cognitions, that is, the thoughts and feelings of the teacher and students about self and interpersonal relationships in the classroom, is one such contemporary focus of interest in which the theory and research of social psychology is viewed as a resource for possible explanations of classroom interaction behavior. The educational implications of Weiner's (1974) attributional model of achievement motivation seem to provide a promising basis for exploring some of the behavioral mechanisms underlying the teacher expectancy effects. Seeking to

understand the causal perceptions and behaviors of students and teachers in the social environment of the classroom is the primary goal of this study.

Need for the Study

This study was prompted by the need for a closer analysis of the processes by which a teacger's performance expectations of a student are formed and maintained, how they are "communicated" to the student and how they influence the student's behavior. Such an analysis of process expectation effects should reflect appropriately the intrapersonal, interpersonal, and environmental character of the typical classroom and thereby take adequate account of the bidirectional nature of classroom interactions. Only by an in-depth examination of the social cognitions of classroom participants can research hope to gain some fundamental insights into the underlying "mechanisms" of the expectancy effects.

From the individual student's perspective, a number of questions and issues seem pertinent to this analysis which, as yet, remain largely unanswered or unresolved. Does the student enter the learning situation with thoughts and feelings related to his expected performance? If so, what is the nature of these thoughts and feelings? To what can the student attribute their formation? Do they affect his performance in the learning situation? By what "mechanisms" do they affect performance? Also, because they derive from perceptions, these thoughts and feelings would likely be susceptible to some inappropriate or unrealistic interpretations and inferences. Beyond this analysis,

a major concern of the student's thoughts and feelings is probably the extent to which they influence student behavior and interact with the teacher's behavior and expectations.

While many questions and issues surrounding the performance expectations the teacher forms and holds about students have been the focus of considerable inquiry, the recent momentum of changing trends in research into teacher expectations has raised a number of other concerns. Many of these pertain to the teacher's thoughts and feelings regarding the student's successes and failures. For example, in what way do a teacher's thoughts and feelings about a student's performance determine his behavior toward the student? How does a student's achievement-related behavior affect the teacher's thoughts and feelings about that student's performance?

In addition, the continuing concern with the extent to which teacher expectations are always appropriate is an issue still to be resolved. While there is every reason to believe that teachers hold performance expectations about their students which in some form may be unrealistic in their formation and maintenance, and/or detrimental/inhibitory to student growth, not a great deal is known about their respective consequences. Inappropriate expectations are commonly encountered human behaviors which, to some extent, are unavoidable. However, as Good and Brophy (1973) point out, teachers who become aware of some of these processes are likely to hold expectations and adopt behaviors which are more appropriate to successful student learning.

The kinds of questions and issues emanating from both the

teacher's and the student's perspective involve interdependent processes. The expectation processes of a classroom need to be researched in the context of ongoing interactions. However, as a consequence of teacher expectations research concentrating on the teacher's domain of the expectancy effect, little is known about the dynamics and influences of the expectancy effect from the student's perspective. As Bar Tal (1978b;23) suggests, "an attributional model of achievement behavior provides a useful analysis for understanding pupils' achievement behavior and teachers' behavior toward pupils." This study, which focused upon the overt and covert behavior of students and included some analysis of the overt and covert behavior of the teacher, therefore, should contribute information needed for expectation researchers to gain an overall understanding of the expectancy effect.

To date, few studies of an observational nature have been undertaken in the naturalistic setting of the regular classroom to investigate the nature of the expectation processes from an attributional perspective. There is a real need for a detailed study employing a cluster of observational techniques which will facilitate a description of the classroom and its social processes "the way it really is."

Purposes of the Study

This study was designed with four major purposes in mind. The first purpose of the study was to investigate some of the intrapersonal thoughts and feelings of students, especially those of

an achievement-related nature, during and after a learning activity. More specifically, this aspect of the study examined individual students' thoughts and feelings about their performance, about the nature of the learning task itself, and about other persons involved in the learning situation. In the course of participating in learning situations a student develops his own ideas, views, and beliefs about self-performance which can be represented as his conception of selfperformance. This conception of self-performance may be regarded as an evolving personal system of interrelated ideas, beliefs, views, emotions, and lines of reasoning concerning self-performance which direct and control his achievement-related behavior in the classroom.

122

Second, the study examined and described the interaction expectation process as it occurred in one classroom. By combining observations of the interactions between the teacher and individual students which occurred in the classroom with the information about individual students' thoughts and feelings, some insight should have been gained as to how the performance expectation, held by both the teacher and the individual student were formed and maintained, how expectation effects were communicated, and how these may or may not have influenced student performance.

The third purpose of the study was to analyze the use of the introspective technique of stimulated recall with elementary school students. Studies reported by Clark and Yinger (1977), Conners (1978a), and MacKay and Marland (1978) have used stimulated recall in naturalistic settings with teachers, but only a few studies, such as

recall methodology with elementary school students. Consequently, the effectiveness of this methodology as a technique for observing covert behavior of classroom participants can only benefit from continued evaluation.

This study varied from most other expectation researches because it included the combination of being classroom based, focused on the student, employed an attributional perspective, and used introspective procedures for observing covert behavior. Another purpose of the study was to present the findings and to consider their implications so as to facilitate their possible later examination and comparison with the evisc be available from published studies. The explanations of, first, the communication of f we expectations and their influence on student behavior and, so we processes of self-attribution of performance as they related student achievement motivation may well be the richer for the evidence obtained in this study.

Research Problems

In this study four general questions were investigated.

- 1. What are the characteristics of students' behavior during a lesson?
- 2. What are the conceptions of self-performance that students appear to develop in classroom achievement situations?
- 3. From an attributional perspective, what phenomena characterize teacher expectancy effects?
- 4. How offective are stimulated recall procedures when used as a

research technique for observing the covert behavior of classroom participants?

Significance of the Study

Teacher expectations comprise an aspect of teaching behaviors worthy of study. Researchers view them as natural and unevoidable components of human functioning which, according to Good and Brophy (1973:50), can be converted to appropriate teaching behaviors. By holding expectations that are both realistic and flexible, teachers can prove to be more effective, which in turn should have a positive impact on the quality of learning.

At a more specific level, the dynamics and processes underlying the expectancy effects in natural classrooms are yet to be understood clearly. This study could contribute information which might aid to such an understanding. An examination of the attribution behavior of students in achievement-related situations may well provide an explanation of the expectancy effects. Findings of this nature could have an impact on the kind of teacher behaviors most appropriate for maintaining positive student achievement motivation in the learning situation. Hence this study could have implications for improving the quality of instruction in classrooms and for associated teacher education programs, as well as contributing toward a develoring theory of the expectancy effect.

Until now, nearly all teacher expectation research has been tased on observable behaviors of student-teacher interaction. The implication has always been that covert behavior is a vital determinant of expectation-related behavior and certainly, in his study of teacher interactive thought processes, Marland (1977:185) found significant evidence to support this contention. By examining and analyzing both covert and overt interactive behavior of classroom participants, this study may have confirmed/disconfirmed the

portance of what appears to be a more complete representation of classroom behavior.

This study was of an exploratory nature. The reported findings were not intended to be generalizable except in the heuristic sense that they might suggest what others may look for in future research. Students' achievement-related behavior as a function of the expectancy effect is merely one domain of inquiry emerging amidst the complexities of students' mental lives in classroom learning. As these complexities are researched, better theories of classroom learning and teaching should evolve.

Finally, the study continued a trend of research within the Centre for Research in Teaching at the University of Alberta which focuses upon the use of stimulated recall methodology as a research technique. Research by Conners (1975a), C. Cooper (1978), Marland (1977), and Nolan (1978) all involved the use of the stimulated recall technique in attempts to observe the cognitive thought processes of teachers or students. This study was the first to concentrate upon some social-psychological aspects of learning.

Definitions

.

In this study the following definitions of major terms were used:

- <u>Teacher Expectations</u>: Inferences teachers make about the performance potential of children.
- <u>Student Self-Expectations</u>: Inferences a student makes about his performance potential.
- Teacher Expectancy Effects: Functional relationships between the teacher's own naturally formed expectancies and students' academic performance and achievement.
- Attribution Processes: The processes by which a person explains the causes of events.
- <u>Self-Attributions</u>: The explanations of causes for event outcomes in one's own behavior.
- <u>Self-Evaluation</u>: An individual's ideas of his own at ties developed from comparing his performance outcomes with performance standards.
- <u>Student Self-Performance</u>: The performance on a task by the student as viewed by the self. Student covert behavior pertaining to self-performance is the thoughts and feelings of self in relation to performance on an achievement task.
- Significant Other: A person whose evaluative opinions matter to the student, one whose performance evaluations are likely to be accepted by the student and used as a basis for forming his self-expectations.

Interpretations: Inferences that teachers make about the cognitive and affective conditions of learners during instruction. As Marland (1977:111) notes, "teachers made inferences, based on their perceptions of students, about--

- 1. what students were thinking;
- 2. what students' needs, motives, and desires were;
- 3. student motivational states; and
- 4. student feelings."

Interactive: The phase of teaching when the teacher is involved in

instructional activity with students in the classroom.

Other major terms used during the study are defined in context.

Outline of the Study

This chapter has established the context and direction of the research. The study considered some attributions of achievement-related behavior of students as a function of the expectancy effect. The major purposes of the study, the research problems, and the significance of the study were presented along with the definitions of the major terms.

The initial section of Chapter II briefly focuses on some philosophical notions underlying the research approach which was adopted in this study. This is followed by a review of some of the related literature and research. A brief overview of research in teaching and an historical review of social interaction research, especially the classroom climate tradition, precedes an extended discussion of teacher expectations and their relationships to diassroom interactions. Because of their significance to this study, some substantive aspects of attribution processes research are presented prior to a description of some attributional approaches related to classroom social behavior.

In Chapter III the research design is described. The subjects, the specific problems, the assumptions, the limitations, the data and data sources, the research methodology, and the phases in the study are presented.

In Chapter IV the methods for analyzing the data are described and the reliability information presented.

The four case studies have been selected so that the first pair of case studies, in Chapter V, describe the achievement-related behavior of successful students and the second pair of case studies, in Chapter VI, describe the achievement-related behavior of less successful students.

Chapter VII contains a summary and discussion of the students' achievement-related behaviors as a function of the expectancy effect as they relate to the research questions. Some implications and recommendations for further research are also presented in this final chapter.

Chapter II

REVIEW OF RELATED LITERATURE AND RESEARCH

Overview

The review of related literature and research which follows is designed to accomplish two main purposes. First, a philosophical, historical, and conceptual background to the investigation describes the base upon which the study is set, places the study in a time frame, and provides some underlying rationale for the research direction that has been adopted. The philosophical setting of the study section, the research in teaching section, and the classroom social interaction research section focus upon this purpose.

The second purpose of this review is to present the theory and research relevant to the study. The teacher expectation research section and the attributional analyses of achievement-related behavior section contain the theoretical approaches and research evidence which should facilitate the interpretation of the phenomena under investigation.

The Philosophical Setting of the Study

Underlying much of the "new" research on teachin. Marx, 1977; is a philosophy of science which serves to reeducational research. Magoon (1977:654) cites Gronbach's that "researchers [should] simply but radically reverse true

they have traditionally placed on building generalizations about effects of variables and give careful attention to particular cases first." Harre and Secord (1972) also call for an explanation of social behavior by intensive study of individual cases. They state (1972:133) that "the generative mechanisms at work in social life" can only be discovered by accepting the notion that the processes which are productive of social behavior occur in individual people--"it is there that the vitally important dimensions of spontaneity and idiospheracy occur." But this need for a consideration of individual cases is embedded in a deeper spectrum of research thought, notably the phenomenology tradition and the orientation referred to by Eay (1975) and Van Manen (1975), among others, as an interpretive science.

In Van Manen's (1975:t) terms, the interpretive orientation refers to all inquiry "which has as its main concern a systematic search for a 'deep understanding' of the ways in which man subjectively experiences . . . the social world." Fay (1975:79, views an interpretive social science as one which:

. . . attempts to uncover the sense of a given action, practice or constitutive meaning: it does this by discovering the intentions and desires of particular actors, by uncovering the set of rules which give point to these sets of rules or practices, and by elucidating the basic conceptual scheme which orders experience in ways that the practices, actions, and experiences which the social scientist observes are made intelligible, by seeing how they fit into a whole structure which defines the nature and purpose of human life.

The constructivist notions for inquiry are elaborated further by Harré and Secord (1972). They advocate the use of collection and analysis of participants' accounts of social behavior. These
thoughts, feelings, and social perceptions enable the discovery of the rules, plans, conventions, images and so on underlying the social behavior of individuals. Harré and Second (1972:151) consider that "social behavior is the result of conscious self-monitoring of performance by the person himself + + + Hand, the presentation of an appropriate social self is one of the important products of the self-monitoring of social performance."

These notions are based on a model of man in which people are treated for scientific purposes as if they were human beings as we know and understand them in everyday life.

The Anthripomorphic model of man conceives of the satject of social investigation as a tiplogical individual whose characteristically human actions are generated by the conscious self-monitoring of its performance in accordance with certain sets of rules which it represents to itself in the course of making anticipatory and monitoring commentaries upon its performance, and which at subjects to critical appraised in retrospective commentaries. (Herré & Second, 1972:93.)

Wilson (1977), in a discussion of the uses of ethnographic techniques in educational research, points out that the phenomenological tradition requires that the researcher learn the meaning structures that determine each individual's behavior. This is achieved by the researcher experiencing and interpreting the behavior partly from the perspective of the participant and partly from the perspective of an external observer. As Wilson (1977:250) states, "the researcher must develop a dynamic tension between the subjective role of participant and the role of observer so that he is neither one entirely."

These approaches to scientific inquiry underlie grounded theory as advocated by Glaser and Strauss (1967). From using a strategy for handling descriptive and interpretive data in research, theoretical conceptualizations are generated. Yinger (1975) considers such a method of theory construction to be very appropriate to the theory development and research in teaching. Ethnographic studies, such as those undertaken by Fremme (1976), Doyle (1977), Jackson (1955), and Smith and Geoffrey (1968), provide support to the usefulness and effectiveness of conducting this type of extensive descriptive and interpretive line of scientific inquiry. Many other descriptive type researches in teaching, while not ethnographic in character, nonetheless in teaching, while not ethnographic in constructivist or interpretively. approach to science.

Given the intention and direction of thic study, a constructivist or interpretive approach to research, as described generally by Edy (1975), Magoon (1977), and Van Maden (1975), and specifically by Harré and Second (1972), seemed the more appropriate to adopt. The context for the phenomena of social behavior under examination was a normal classroom. By describing and seeking to interpret and analyze individuals' covert and overt social behavior, nopefully the "generative mechanisms" in classroom social life which pertain to expectancy effects might be discovered.

Research in Teaching--An Overview

After several decades, research in teaching is still in a state of evolution. While conceptualizations of research variables have continued to be modified and/or refined, and a variety of methodologies have been tried, few definite and pertinent findings

have emerged. Yet research in the area continues to proliferate. If the major research efforts in the field have occurred since 1960 and it is possible to discern a number of changes in the pattern of studies up to the present time. Research in teaching, therefore, is in a dynamic state.

Hecently, Rosenshine and Furst (1973) and Durkin and Piddle (1974) have contributed to organizing the direction of research in teaching. Rosenshine and Furst's descriptive-correlationalexperimental loop suggests that research into a domain of teaching should proceed from a descriptive, exploratory phase of identifying variables, to a correlational phase of establishing significant relationships between and among the variables, to an experimental indust in which hoperully cause and effect relationships between and among the variables, to propose a paradigm for research into classroom life. By grouping the relevant variables under four larger classes, notably presage, context, process, and product variables, Dunkin and Hiddle have provided a model and accompanying rationale which has served as a theoretical framework for a number of researches into teaching.

General reviews of research on teacher effectiveness by Lunkin and Biddle (1974), Gage (1963), Rosenshine (1971), and Travers (1973), and smaller scale reviews such as those by Bennett (1978), Fedigan (1978), King and Smyth (1978), and Rosenshine (1977), all indicate that some conclusions of a seemingly consistent nature have emerged amidst a maze of tentative, inconclusive, and often pessimistic findings. As Brophy and Evertson (1977:79) comment,

"the number of teaching behaviors established as correlates of teaching effectiveness is still limited to a handful." Many of these findings may not be generalizable to all teaching and learning situations, for often they are derived from descriptive and correlational studies involving a particular subject area, with it as a certain grade level who possess certain characteristics. Furthermore, the findings tend to pertain only to those teacher behaviors which are observable. Hence, failure to take account of Gage's (1903:132) comment that "the programs that teachers carry around in their heads need analysis" may explain the relative lack of return on the investment of research effort to date.

Not surprisingly, some redirection of research effort has now evolved. The National Institute of Education's Panel & Report (1975) expressed a commitment in future research to studies which seek to understand the mental life of teachers. As the report states, "What teachers do is directed in no small measure by what they think. Moreover, it will be necessary for any innovations in the contex¥, practices, and technology of teaching to be mediated through the minds and motives of teachers" (1975:1). The report is adamant that the relationship between teacher thought and action cannot be overlooked. Clark and Yinger (1977), Conners (1978a), Marland (1978) and Marliave (1976) all report research which studied this covert mental behavior of teachers.

Winne and Marx (1977) also describe a reconceptualization of the research in teaching. Foremost in their discussion is an extension of the research effort to embrace the unobserved and 0~

Ð.

previously undescribed internal events in the teaching learning environment. "Specifically, we see the mental life of both teachers and students in classrooms as critical items to be studied if we are to understand the process by which teaching influences students' learning" (Winne & Marx, 1977:670). Few studies in the research in teaching have concentrated upon the covert behavior of students. Doyle (1978:31) recently emphasized that "the study of teaching involves considerably more than the study of teachers."

While suggesting there is a demand for a greater research concern with student processes in classreems, boyle (19¹⁰⁾ nevertheless sees several problems to be encountered when attempting to conceptualize student covert behavior in teaching effectiveness:

- 1. Little information is available regarding student processes in the classroom because previous research has induced largely on teacher variables.
- as time-on-task, and aptitude variables are known accut students.
- 3. Most of the existing research on student mediating processes is laboratory oriented.

The need for comprehensive descriptive studies of student covert behavior in the naturalistic setting of regular classrooms seems considerable in order to explore the relevant variables as well as the relationships between and among them.

Some studies have attended to this concern. Erlwanger's (1974) study into the covert behavior of elementary level students

.

as they learn mathematics, and Nolan's (1978) investigation of student thought processes while involved in story writing are two resurches which have attempted to conceptualize student mediating processes in teaching effectiveness. Only a few studies, such as that by Weinstein and Middlestadt (1975), have investigated student covert behavior related to social interaction and various affective aspects of learning. The research to be reported in this study, similar to that of Weinstein and Middlestadt's, is intended to contribute further to the evolution of variables in struent social cognitions as they portain to teacher effectiveness.

Classron, Social Interaction Research -- The Classron Clinate Tradition

Ine Exercy Studies

While research in teacher expectations seement stem mainly income Research and Caccison's (905) study, as reported in Pygnalion in the Chaogroom, interest in social-psychological variables relative to teaching effectiveness in general, and to some antecedents of teacher expectations in particular, can be traced back to the 1940's. Appropriate substantive issues for investigation gradually emerged as researchers relined continually their conceptualizations of variables. Likewine, methodological issues gradually clarified as research. designs and late gathering teorniques were developed appropriate to the phenomena under study. This line of research is popularly referred to as the diasgroom climate tradition.

According to Withall and Lewis' (1903) historical perspective, the earliest studies in social interaction as related to teaching effectiveness focused on identifying the teacher traits, qualities, and behaviors thought to facilitate learning. In a sense, the studied by Morrison, Naminga, wickman, Barr, and others, cited by Withall and Lewis, have continued but, as Ryans (1900) found, few consistent findings have emerged. In Dunkin and Fidule's terms (1974: 3t-4d), such a result in the research effort was hardly surprising. Trying to establish a relationship between presage and product variables without considering the very injortant set of process variables, often reterred to as "the black-box," was basically unrealistic.

A tearnique of data gathering that had a definite in teacher early classroom social interaction research was Morenc's (1994) schloretric device for measuring the relationships within groups. The tearnique inspired a line of research which, for the first time, attempted to assess the interpersonal relationships among students by rathering data about social relationships in the classroom from the students themselves.

About the mid-thirties, teachers' behaviors in the classroom theorem the focus of analysis and the classroom climate tradition was established. This represented the first real attempt to research. Dankin and Eiddle's (1974:44, process variables. The major studies of this kind were the landmark works of H. H. Anderson and his colleagues which culminated in miles of Applied Psychology Monographs (1940). They studied the efficiency on student classroom performance of teaching behaviors classified as "dominative" versus teaching behaviors classified as "integrative." These categories of teaching behaviors reflected opposing ends of some continuum in teachers' use of language -3

and use of control techniques. The Anderson studies seem to demonstrate that a teacher's classroom personality and behaviors influence the behavior of the students taught in that class.

Noamwhile, the Iowa studies of Lewin, Lippitt, and White (1934) resulted in a troadening of the social interaction research perspective. They investigated the interpersonal interactions of children in differing social climates by observing the impact of subserving, democratic, and laissen-faire leadenship styles on group sprale, or a particly flow, and group productivity. When applied to the entire complex of relationships in learning groups.

The matter of the affective states of learners possibly mediating their overall performances and achievements became a substantive issue of preminent concern. Shoup dynamics opened the way for research in classroom social interaction to consider what Withall and Lewis (1963) refer to as the "psyche" φ r 1; and the "social group of any classroom group. This tasis for organisity group social interaction data, notably the accommodation of personal needs and institutional or task needs through group processes, has persisted in social interaction research (Setzels and Thelen, 1950).

Withull's (1949, attempt to measure the social-emotional climate of classrooms was part of a major attempt at the University of Chicage to develop a percent theory of instruction based on affective as well as cognitive aspects. He saw the social-emotional climate as a group phenomenon determined mainly by the teacher's verbal behavior as representative of his total behavior. Withall's well known Climate

Index represents the ratio of learner-centered statements to teachercentered statements. One of many early users of the Climate Index was Flanders (1949) who found teacher-centered climates to be associated with more negative achievement-related student behaviors, and learner-centered climates to be associated with more positive achievement-related behaviors.

Another treni in the Chicago studies, as reported by Withall and Lewis (1903:099), was to examine:

. . . the dynamics of social interaction at the overt level of teachers' and learners' observable behaviors and at the same time trying to relate these behaviors as observed and interpreted by the "actors" themselves to the inferred covert dynamics.

In essence, this trend of research dipears to be a generalized antecedent of the study reported in this dissertation and represents a significant broudening of investigation into Dunkin and Biddle's' (1774) process variables. Singletary's study, cited by Withall and Dewis (1403:7307, which sought to investigate teachers' and sumministrators' perceptions of students for each other, seems to be one of the earliest to refer to the concept of expectations in classroom research.

By the early 1950's researchers were only coming to realize the complexities involved with classroom processes in the learning situation. They seemed to recognize that research would have to tocus on the teacher's actual behaviors in the classroom and his understanding of the student's self and social perceptions, on the student's perceptions of the instructional activities, and on the group life context in which the teacher and student interact. Yet researchers were without adequate instrumentation and conceptualization of these process variables to handle the direction of research. The subsequent widespread concern with the implications of findings in social psychology for classroom groups reflected more than just an awareness in the importance of the social context of learning. Fossible means for dealing with the complexities of the classroom. were also being considered by investigators. The character of the classroom climate tradition of research was still evolving.

The Studies of the 1950's and 1960's

Among the numerous studies in the field of classroom social interaction undertaken in the 1950's and 1960's are some which pertain directly to substantive issues related to the later expectation research. Jenkins' (1951) study typified a line of research which established the interdependent nature of the student-teacher relationship. Other studies by Gage and Suci, Jenkins and Lippitt, and Bush, all cited by Withall and Lewis (1963:703), and ty Hallworth (1962) investigated the nature of interpersonal perceptions held by teachers and students, but they yielded mixed results. Withall and Lewis (1903), in reference to other studies again which developed further the use of sociometry, noted that students' relationships with each other seemed to be determined by teacher behavior factors. However, the many studies of the differential effects on teaching/ learning of opposing styles of teacher behavior seem to have the most significance for the later expectation research.

Ever since 1950, classroom social interaction research into styles of teacher behavior and their effects on student performance

seems to have been characterized by conceptual confusion, inappropriate methodology and unsophisticated design. The continuum along which teacher behavior is hypothesized to vary has remained substantially the same, yet new terminology, modified definitions, and refined instruments keep appearing. Attempting to establish relationships between Dunkin and Biddle's (1974) process and product variables was fraught with problems.

Withall's (1949) seven categories of styles of behavior along the teacher-centered -learner-centered continuum protably formed the Easis of Flanders' (1967) reconceptualized teacher behavior continuum of direct influence - indirect influence and its nine categories. The upsurge of classroom based observational studies, using Flanders' Interactich Analysis Categories system, modifications of it, or some other observational instrument, has yielded a maze of evidence. Dunkin and Biddle (1974) reviewed the evidence from the perspective of two not necessarily related concepts -- indirectness and warmth. Given the conceptual confusion of these notions and the problems of their valid and reliable measurement, it is not surprising that Dunkir and Hiddle (1974:132) were only able to conclude that some relationships between the two concepts and student growth were tentatively established, even though such relationships were based on weak evidence. Twenty years of vigorous research effort seemed to constitute a necessary growing phase in the classroom climate tradition.

The Current State of Classroom Interaction Research

In the 1970's a great deal of classroom social interaction research has focused upon the quality and quantity of actual specific interpersonal interactions. Several studies showing a relationship between specific teaching behaviors and student achievement have emerged from the findings. The more fruitful results in the search. for process-product relationships probably reflect the greater use of descriptive and correlational classroom based studies, more sophisticated instruments, and the more refined, better defined research variables. Rosenthal's (1974; typology of teacher interaction behaviors -- classroom climate, teacher input, teacher output, reeaback -- forms a useful basis for reviewing this research, even though the typology was originally designed to categorize teacher expectancy effects. References to socio-economic status (JES) of students will appear frequently. Many of the research findings are qualified according to this dimension of student individual differences. There is a definite tendency in the research on teaching for SE3 level to be used as a proxy measure for other dimensions of student individual differences such as ability, knowledge, motivation, and performance expectations.

Research suggests that teachers who create a warmer socioemotional <u>climate</u> will be generally more effective. Marliave (1970:25, reports that "a family-like atmosphere is one of the most crucial instructional variables for enhanced student learning." Joyce (1975) found the variables of classroom cohesiveness and affirmativeness to be important elements of a positive classroom climate which is most conducive for learning. Good and Grouws (1977:123) also found that more effective teachers were those who established successfully a more relaxed atmosphere by employing a number of interaction teaching behaviors known to be associated with gains in student learning. Erophy and Evertson (1977) and hosenshine (1977) concluded that expressions of teacher affect were associated with the achievement of low SES students, but not associated with the achievement of high SES students.

Verbal <u>input</u>, as a factor of classroom climate, pertains to the amount and kind of presentation of new and difficult material to students. Research as reviewed by Bennett (1970), Frophy and Good (1974) and Rosenshine (1977), suggests that the more effective teachers are those who monitor appropriately the verbal input in aboundance with levels of student achievement and motivation. Such a finding has major implications for teacher expectations, as observed by E. Cooper (1975) and Brophy and Evertson (1975).

Vertal <u>output</u> refers to those teacher interaction behaviors which provide appropriate action opportunities for students to respond and ask questions. Frequency of occurrence and staying with students are considered by researchers to be two dimensions of ertal output. Staying with a low SES student seems to be important for a different reason than staying with a high SES student. As a general rule, Frophy and Evertson (1977:67, found that, when asking factual questions of lower elementary students, the more effective teachers tended to give the answers if no $r_{\rm c}$ inse had followed. They would only stay with low SES students in this situation if concerned about

۱.

the personal participation of the student. Erophy and Good (1974) and studies cited by H. Cooper (1975) found that staying with high SEC and also upper elementary students was an effective teacher behavior when the student failed to respond to a thought process type question.

High SES students seem to create more output opportunities for themselves and these were more accepted by the effective teacher. Evidence reported by Good and Greuws (1977:125), Frophy and Evertson (1977:37), and Frophy and Good (1974) contributed to this overall finding. Meanwile, in terms of teacher initiated contacts, research has found considerable diversity in their effectiveness.

Teacher feedback appears to be related closely to stilent addievement. According to Permett's (1977:25) review of the recent research, feedback in the form of praise, to have any effect on whieveness, should be contingent on performance, should reflect genuineness on the part of the teacher, and should be employed with regard to individual differences. Stallings and Maskowitz (1974), Sour (1973), and brophy and Eventson (1974) all found teacher price - to be positively related to the academic achievement of low SES students with higher correlations being recorded for teacher praise of academic responses in comparison with global praise of student benavior. Brothy and Evertson (1974) found for low SES students that teacher criticism of incorrect student responses was negatively related to academic achievement. Conversely, brophy and Evertson utserved that for high SES students teacher vertal graise failed to correlate with student achievement, but teacher criticism of poer work was justively related to achievement gains.

Academic and behavioral feedback has also been linked with student achievement. Good and Grouws (1974-123), state that highly effective teachers are good providers of process feedback. Brophy and Evertson (1970a) and Sour (1973) both found for low SES, lower elementary grade students, immediate corrective feedback was important for student achievement. In reviewing a number of studies, Marilave (1977-114) concluded that "the provision of feedback and explanation according to student need, was important for high r learning subcomes."

Graphy of interaction involves numerous kinds of interpersonal translations and relationships. The research in teaching has indicated that teacher interaction behaviors are closely appled to with statent learning. However, appreciated by the pample of studied reviewed at we, the nature of the relationship is a very complex one depending on public crucial variables as subject matter, the teaching strategies being employed and, most important, the individual attributes of the persons involved. Nevertheress, the classroom climate tradition of research appears to have "come-of-age."

Throughout this review of recent classroom interaction research, many of the conclusions drawn will need to be qualified in terms of staient individual variables and their associated effects on the performance expectations held by teachers. Furthermore, the conclusions have been derived from studies of observable aspects of behavior and do not reflect the growing research concern in investigating teacher and student covert behavior. The relationships between process and product variables presented in the review, therefore, should be viewed as tentative and accepted with caution.

Summary

In this section of the review, an historical overview of the research literature in social interact. It was presented as background to the current research effort into teacher expectancy effects. Whereas studies on teacher expectancy effects have occurred largely in the last it years, many of the conceptual areas and issues relevant to the research evolved anists the global studies of the classroom climate tradition. The importance of affective aspects of behavior in learning, the character of interpersonal perceptions, the influence of teacher rehaviors on students' perceptions of each other, and the logact of teacher-student interactions on student achievement are all probably significant components of the expectancy effect.

Teacher Expectation Research

Teacher expectation research has a comparatively short history. In just it years a proliferation of studies, some experimental and others descriptive, some laboratory based and others classroom based, have brought into some perspective the pervasive effects of teacher expectations on classroom interaction behaviors and student achievement. This review of teacher expectation research consists of a trief overview of the early studies and a more comprehensive review of current emerging patterns in expectation. studies, especially those pertaining to process expectation effects.

Ar. Historical Overview

Nearly all the literature on teacher expectation studies points to Rosenthal and Jacobson's (1968) <u>Pygmalion</u> in the <u>Classroom</u> as the seminal study from which much of the later researches are derived. Yet the notion of self-fulfilling prophecy, which was the primary phenomenon under investigation in that study, was well established in the social psychology literature before 1968. Robert Merton (1949:179), who is often credited as naving introduced the self-fulfilling prophecy, makes reference to an early sociologist, W. I. Thomas, who in turn observed, "If men define situations as real, they are real in their consequences." Merton pointed out that predictions or prophecies of a situation can become an integral part of the situation and so affect subsequent developments.

Remential and Jacobson (1968) were probably the first researchers to attempt to show how the self-fulfilling prophecy might work in the classroom. They hypothesized that teachers devoted approximately equal time to students regardless of expectation, but that expectation affected the quality of teacher-student interaction. By providing teachers with false information about the intellectual capabilities of some of the students, - experimenters found they were able to manipulate teachers' experimentars such that significant changes in student intellectual performance subsequently accrued.

Several reviewers of the Rosenthal and Jacobson study, such as Elashoff and Snow (1971), Jensen (1969), and Thorndike (1968), have criticized the experimental design and statistical procedures of the study for not being sufficiently rigorous to justify the generalizations about teacher expectancy that were made. However, this criticism had little effect on "neestigators' attempts to continue or replicate the research. Studies which attempted to replicate Rosenthal and Jacobson's findings have occurred both in the classmoom and the laboratory. Some of the early studies which have provided evidence to support the expectancy hypothesis are reported by Anderson and Rosenthal (1968), Erophy and Good (1970), Conn, Edwards, Rosenthal, and Crowne (1969), Meichenbaum, Howers, and Ross (1969), Rothbart, Dalfen, and Barrett (1971), and Rutovits and Miehre (1971). Nevertneless, a number of studies, such as those by Claiborn (1909), Fleming and Antonnen (1971), and José and Cody (1971), report no supporting evidence.

Once research had established tentatively the existence of the expectancy effect, investigators began to examine the variations in teacher behavior that might communicate such expectations. This change emphasized less concern for product expectation effects and a channelling of the research effort mainly into the study of process expectation effects. As Haigh (1974:98) points out, this change seemed necessary if investigators were "to determine the success of their attempts to modify expectations, or understand the means by which teacher expectations produce change in pupil behavior." Some of the early evidence on the character of verbal interaction in the classroom, as reported by Hudgins and Ahlbrand (1970) and by Jackmon and Lahaderne (1967), indicated uneq distribution of teacherstudent interactions. However, many of the early studies have shown that teacher expectancy may be a major underlying cause of variations

in interaction. The experimentally induced expectations studies of Kester and Letchworth (1972), Meichenbaum, Bowers, and Röss (1969), Rothbart, Dalfen, and Barrett (1971), and Rubovits and Maehre (1971), and those studies which investigated existing, unmodified teacher expectations, such as Brophy and Good (1970) and Haigh (1974), all found definite evidence to the effect that teacher expectations influenced the quantity and quality of teacher behavior when interacting with students.

The early research on teacher expectations raised many issues for further research and, predictably, these have provided the focifor much of the research in the last five years. The pros and cons of investigating manipulated or induced expectancy effects as quainst naturally contributions manipulated expectancy effects, now performance expectations might be communicated in the classroom, and now these communications influence student performance seem to form the bases of the continued research effort into teacher expectations.

The Emerging Patterns in Expectation Studies

From the review of current expectation research, several major trends seem to have emerged in recent pears. Some of these provide important background to this research.

One trend of research that has gained greater credence recently has been the recognition by research that expectation formation and communication involves a two-way process. As Brophy and Evertson (1972), Fiedler (1975), and Winne and Marx (1977) all report, research on classroom interaction generally, and expectancy

effects specifically, has in the past tended to focus conceptually and methodologically on adult/teacher as teing active and causal and the child student as being reactive and passive. Int now, as brophy and Evertson (1974:9) point out, there are significant individual Altherences among students in the opportunities and problems which they present to the teacher. H. Cooper (1975), Lusek (1975), and Luckneed (1975) recognized the significance and function of student characteristics as contributing to the expectancy effect. Braun (1970) states that student self-expectations constitute the Received tenavioral cycle. The bidirectional nature of the scient state of seens to be well surmed up by H. Cooper's (1970: cyclical process of mitual influence seems must of purt of literature." The two-way process emphasizes the contraction . . . of expectations, with students' behaviors and self-electronic te viewed as major influences on the development and major the one · xpectation effect.

A second trend to emerge from the research is the recognition of the complexity of teacher and student individual differences. Muttart (1977:*, cites a number of studies which demonstrate that only some teachers treat students differently on the basis of expectations. As well, Brophy and Good (1974, have recognized for some time that teachers vary in their susceptibility to the expectation effect, namely, proactive, reactive, and over-freactive teachers.

Proactive teachers appear to be undeterred by their Expectations for low achieving students, so that they spend more time interacting with lows than highs. Reactive teachers simply allow existing differences between high and low students to unfold, so that highs, due to their own initiative and ability,

¥.

come to dominate public classroom life. . . . A third class of teachers overreacts to student differences (in supplying qualitatively and quantitatively superior treatment to highs), thus exacerbating differences between students. (Brophy and Good, 1974:303.)

1

likewise, Braun (1970:195) discusses what he calls "teachers' differential susceptibility to input factors," and concludes that teacher information and personality are the prime factors underlying the respective positions.

Only very recently have individual differences in students , ecce into the limelight with expectation research. Therefore, the following studies are significant as they report information about student attributes or characteristics which are associated with contrasting teacher expectations, attitudes, and beliefs.

Dusek and O'Connell (1973), Lockheed (1975), and Willis and brophy (1974) all found st don' achievement to correlate highly with teacher expectations. Alams and La Voie (1974) and Brophy and Eventson (1976), found considerable sex differences especially in the elementary grades. According to the latter study, teachers hold nuch nore positive views of their female students, and these were transferred into their tenaviors more during private contacts than during public sessions. The studies by Gooper, Baron, and Lowe (1975), Finn (1971), Lockheed (1975), Long and Henderson (1974), Rosenthal, Barat: and Hall (1974) and Rubovits and Maehr (1973) all report generally no race effect on teacher expectations. Other studies that investigated the effects of physical structiveness and social class seem to provide no consistent pattern.

Another trend that has emerged over a longer time period has

been the move toward naturalistic studies, both in terms of the setting for research and the kind of expectation effects to be observed, that is, the non-manipulated as against manipulated effects. Rosenthal (1970, notes that non-latoratory studies of this kind have risen from 10 percent prior to 1970 to 41 percent between 1971 and 1975. Brophy and Evertson (1970), justify the move away from the latoratory on the grounds that more accurate and more generalizable findings can to derived from natural classroom research. Braun (1970), H. Cooper (1970), and Lockneed (1975) state that the move is justified because much of the supportive evidence for the expectatory effect cones from hon-manipulative studies such as those by Prophy and Good (1970), Orano and Mellon (1975), Dusek and O'Connell (1973), Good and Prophy (1972, deter and Davis (as reported in Jeter, 1975), MeDonald and Elias (1970), and C'Jonnell, Dusek, and Wheeler (1974).

A fourth recent trend in the expectancy effect research is the concern with developing an understanding of how teacher performance expectations of students are translated into particular student behaviors and achievement. Braun (1976), Brophy and Good (1970), H. Cooper (1975), and Lockherd (1975, all propose models which endeavor to explain the process expectation effects and which are still multipleing researched.

Brophy and Good (1970:365-366) posited a model conceptualizing the processes underlying the self-fulfilling hypothesis:

(a) The teacher forms differential expectations for pupil performance;
(b) He then begins to treat children differently in accordance with his differential expectations;
(c) The children respond differently to the teacher because they are being treated differently by him;
(d) In responding to the

teacher, each child tends to exhibit behavior which complements and reinforces the teacher's particular expectations for him; (e) As a result, the general academic performance for some children will be enhanced while that of others will be depressed, with changes being in the direction of teacher expectations; (f) These effects will show up in achievement tests given at the end of the year, providing support for the "self-fulfilling prophecy" notion.

Lockheed's model of the expectation process reflects an incluence of expectation states theory (Berger, Gonner, and Fisek, 1974). A crucial aspect of the model is the use of the teacher as a source of a student's self-evaluation, which in turn determines a student's self-evaluation. Lockheed's (1975:5) model of the respectation process focuses upon:

. . . the student characteristics of achievement, status, and tenavior and personality as determining teacher expectations and teacher behaviors, that teacher exjectations determine student academic achievement and teacher tehaviors, that student characteristics are correlated with academic achievement, and that teacher behaviors affect student academic achievement.

Furthermore, student self-evaluations are determined by status **F** characteristics, evaluations of others, and achievements. This model implies that student individual differences are crucial to the explanation of expectation effects.

Braun (1976:192-208) proposed a model of the process of experiation effects which attempts to explain in more social psychological terms the behavioral cycle between teacher input and learner output. He traces the process through a number of stages in the cycle: sources of teacher expectancy--input factors, differential susceptibility to input factors, transmission of expectancy cues-teacher output, learning response to output cues, learner selfexpectations--pupil output. Braun cites a great deal of evidence in support of each stage. The interesting features of this model include the role of student self-expectations--similar to Lockheed--and the cyclical nature of the process.

H. Gooper's (1978) model for expectation communication and behavior influence focuses on an attributional explanation. The causal sequence of teacher cognition, teacher behavior, student cognition, and student behavior is set within the interaction context of a classroom. Again, evidence is cited to support some of the features of the model. Cooper's model is presented in a later section of this review of literature.

Each of the models tend to trace the "communication" of expectations from the time teachers form them, to the way teachers' behavior toward individual students varies according to these expectations, to the ifferential way in which the individual students behave and react accordingly and, finally, to how students' performance and achievement are affected. The supporting research underlying these models indicates clearly that the relationship between teacher behaviors and student achievement is far from being a simple one; that as a consequence no definite cause and effect mechanisms have yet been established and, as already indicated, the direction of influence regarding performance expectations is probably of a two-way kind between the teacher and the student concerned.

These four trends generate a fifth. Research in teaching seems to be increasingly taking account of the uniqueness of any one classroom group. As Brophy and Evertson (1976b:10) state, "Students in one classroom may only have chronological age and presence in the

classroom in common." How the individual members of any one classroom group think and behave, react and interact, are phenomena unique to that group. Consequently, research has moved toward a phenomenological perspective. The study of attributional processes has evolved from this kind of reality--trying to make sense out of the everyday experiences of human behavior. Because attribution processes are related to expectations, it seems logical that an attributional perspective might be used to view the classroom in "the way it really is."

Classroom Interactions and the Expectancy Effect

The early studies in teacher expectation research established the need to examine classroom interactions as the possible means by which teacher performance expectations of students are communicated in the classroom. A great deal of process expectation research has now been undertaken investigating the differential treatment of students based on different performance expectations. The quantity and especially the quality of interactions seem to serve as the medium of communicating the expectations. In the review that follows, Rosenthal's (1974) typology of teacher behaviors, namely, climate, input, output, and feedback, provides a useful basis for organizing the evidence which has accumulated from the recent research.

<u>Climate</u>. There is no paucity of evidence to support the fact that a teacher's high expectations for a student's performance gives rise to a <u>climate</u> of warmth, attention, and emotional support. Teachers appear to spend more time and interact verbally in more

Positive and supportive ways with high expectation students than with low expectation students. In addition, as reported by Lockheed (1975:4), teacher expectations may be communicated by such indirect teacher behaviors as general seating arrangements, use of ability groups, and use of superior ability students to assist less able students. H. Cooper (1978:9) cites researches by Chailkin, Sigler, and Derlega, Kester and Letchworth, and Page which found that many non-vertal teacher behaviors associated with positive emotional attraction were displayed by teachers most frequently in interactions with higher achieving students.

Input. Research evidence suggests that teachers vary the amount and kind of vertal input they use with different students. The rever of activities teachers select to correspond with their expectations is a critical example of verbal input. H. Cooper (1970:10) cites research by Corntleth, Lavis and Button, Jeter and Davis, and Mendoza, Brophy and Good, which indicates that slow students have less difficult material taught to them. Doyle, Hancock, and Himer (1971) report findings that teachers monitor the learning opportunities for a group according to their perception of that group's potential. Brophy and Good (1974) report that teachers generally not only expect and demand less of low expectation students, but that less difficult material is given to those students. Research by Beez (1970), Carter (1969), and Pidg. (1970) indicates that slower learners receive fewer opportunit. The learn new material than students e suggests one reason, therefore, why perceived as bright. The the quantity and quality ______ ictions between the teacher and low

expectation students are likely to be quite different from those between the teacher and high expectation students.

Output. Evidence is also available which indicates that teacher verbal output behaviors contribute to differences in the quantity and quality of interactions with students for whom differential expectations are held. Brophy and Good (1974) report that, overall, low achieving students generally receive fewer teacher contacts than high achieving students. In terms of staying with students, Good and Brophy (1973), Rosenthal (1973), and Rowe (1974) found that teachers' encouragement for the cutput of student responses was much greater with high achieving students. Teachers would wait longer for unswers, would prompt and prote more by reparasing and giving helpful hints, and would ask harder questions of bright students. Likewise, as Rothbart, Dalfren, and Farrett (1971, and Cornelett., Lavis, and Button (1974) Yound, teachers interacted verbally in more publicle ways with high achieving students by accepting and using their ideas more and, overall, spending more interaction time with them than with low achieving students.

Not surprisingly, high ability students are likely to initiate more academic teacher-student interactions than low ability students. This finding was reported by Brophy and Good (1974) after reviewing a number of studies. However, this may well be a function of other teacher interaction behaviors such as the use of feedback. In an experiment, H. Cooper (1977:426) found that when teachers were asked to stop criticising lower ability students the number of studentinitiated interactions increased to a significant level. Entwisle and

Webster (1474), in another study, found that when teachers increased the amount of praise the frequency of student hand raising also increased.

Research into teacher-initiated contacts has revealed no consistent pattern. H. Cooper (1970:12) suggests that "the teacher contact-initiative may be a function of teaching strategy differences" and cites a number of studies, such as Firestone and Frody's (1907), to support this view.

<u>Fuedrack</u>. Teacher feedrack tehaviors have teen the subject of considerable expectation research inquiry. Must studied, especially those which are classroom based, provide evidence which indicates pullity of interaction differences exist between the teacher and " students for whom differential performance expectations are held. Teacher feedback comprises two components--an evaluative component, usually in the form of praise or criticism, and an academic corrective or explanatory components.

Expectation research has revealed that differential doe of praise and criticism is particularly sensitive in communicating teacher performance expectations to students. Even (1970), Brophy and Good (1974), Gooper and Baron (1977), Dalton (1909), Firestone and Frody (1975), Meichenbaum, Bowers, and Ross (1909), and Rubovits and Maetr (1971) all report evidence that bright students tend to receive more praise than their lower achieving counterparts. Also, lower achieving students were reported to receive more criticism, especially in failure situations. Smith and Luginbuld (1970:270) concluded that "there existed a tendency to accord the student of greater ability more evaluative communications (encouragement and criticiss"." As an explanation, they suggest that low expectation students may receive less evaluative feedback because teachers feared that it would contribute to a negative label effect.

Teacher reinforcement behaviors found by brophy and load (1974), which are thought may transmit the differential effects, include rewarding inappropriate behavior of low expectation students, criticizing low expectation students more than high expectation students in parallel situations, failing to praise low expectation students in situations where other students are typically probably and failing to give low expectation students feedback concerning examplements. A study by willis (1370), which indicated that teachers tended to ignore contents from lower achieving students more than from higher achieving students, and a study by Jay (1975), which showed that here positive teacher feedback was given to higher achieving student responses, provide evidence which is consistent with teacher feedback behaviors and the expectancy effect.

While the accumulated evidence indicates clearly that teacher isocicach behaviors contribute significantly to a differential quality of interaction between the teacher and different students, Dunkin and Hiddle (1974:130) alert researchers to the need for some caution when interpreting the findings, given the existence of some conflicting evidence. However, as Smith and Luginbuhl (1975:206) suggest, feedback can be considered an important mediator for all student performance. Hughes' (1973) experiment, which showed that praise was a major determinant of student learning, is just another study which. attests to the potency of teacher feedback behaviors.

Summary

As From (1970) (1) suggests, "it appears that the anomal and quarity of pupil teacher interaction offer crucial clues to the systemy of possible expectancy dues that the learner internalizes." does is the presise underlying a number of the models which have teach postulated to explain the "communication" of teachers' performance expectations to students. But the use of the notion of communication has little relevance for the expectancy phenomenal mises toleaced consideration is given to the "receivers of the messages." Therefore, what is the relationship to tweet differential teacher below is toward black the '' environ and student' proceptions and tonaviore, expectably performances, in the classroom learning situation. This seems to be the crux of the expectancy effect--exploring more fully the process expectation effects.

While the importance of the consequences on student tenavior of the expectancy effect has been relterated by Braun (1974), Brophy and Jood (1974), H. Cooper (1975), and the National Institute of Education's Fanel & Report (1975:34), research to this effect has concentrated almost solely on product variables, namely, performance bitcomed. Student perceptions of the processes which are thought to underlie the effects of teacher expectations are just becoming the focus of research effort. Much of this effort is linked directly to the application in the classroom of attributional analyses of numan behavior.

A tributional Analyses of Achievement Behavior

The linking of expectancy effect research with attributional analyses of achievement-related behavior has teen a recent development. The National Institute of Education's Panel & Report (1975:31) suggested such a linkage to be a worthwhile research domain. However, as evidenced by recent studies, this seems to have motivated more interest in process expectation effects involving teacher behavior than those involving student behavior. Meanwhile the application of attribution processes to achievement-related behavior has generated to antiderable research interest, including studies which are classroom hased.

Attribution processes are phenomena incryday life, including numan behavior in classrooms. The instantial aspect of tenavior and need to be recognized as such. Joursequently, a brief review of some of the research pertinent to attribution concepts is presented. This will be followed by a consideration of some attributional explanations of student achievement-related behavior.

Attribution Processes

Attribution theory, as initially projected from the realms of naive psychology by Heider (1958, and developed further by Jones and Davis (1965), Kelley (1967), and Weiner, Frieze, Kukla, Reed, Rest, and Rosenbaum (1971), offers a number of concepts which can be applied to this study. As borko and Shavelson (1978:271) point out, the theory "deals with the processes by which people integrate information to arrive at causal explanations for events." According to Heider (1958),

Kelley (1967), Shaver (1975), and others, the overall attribution process consists of three phases: an observation of an action; a judgement of intention; and the making of a dispositional attribution either to the person or to the environment.. The process, in Shaver's (1977:137) terms, makes up the cognitive phase of social perception, that is, the perception of the social behavior of a person.

Harvey, Ickes, and Kided (1976, 1978) differentiate between the personal and interpersonal consequences of attribution. However, two fundamental but related concepts arise from a review of the research. on both sets of consequences. First, studies such as those by Eradley (1978), Joethals (1970), Jones and Davis (1975), Kelley (1967), Sonroeder and Linder (1975), Shaver (1975), and Steele and Woods (1977), provide a surprising amount of evidence on the nature and extent to which attributions are open to bias, distortion, or defensiveness--upually to fulfill self-protective or self-serving purposes of the perceiver or observer. Second, studies by Wortman (1970), Snyder (1976), Harvey (197t), and Storms and McQaul (1970), and others, suggest that there is a considerable effort made by the stimulus person or actor to maintain a sense of perceived personal control and freedom in his behavior in any situation. As an example, of this Snyder, Stephan, and Rosenfield (1978) refer to attributional egotism which is the motive to take credit for success and deny blame for failure in order to enhance or preserve set - entered eem. This concept of perceived personal control bears a close relationship to Goffman's (1959) self-preservation in social situations.

A considerable number of studies have detended from the

433

seminal work by Jones and Nisbett (1971/1980). They maintain that "there is a pervasive tendency for actors to attribute their actions to situational factors while observers tend to attribute the same actions to statle personal dispositions of the actor." Fell, Wickland, Manko, and Larkin (1972), Bretam and Aderman (1977), Balper (1975), Hanse and Stonner (1977), Lowe and Hansen (1976), Miller (1976), Honse and Stonner (1977), Lowe and Hansen (1976), Miller (1976), Monson and Snyder (1977), Regar and Totten (1975), and Ross (1976) are just some of the many studies which investigated aspects of, and found evidence supporting, the Jones and Nistert thesis. The implication for the classroom for most for the teacher who observes ids student in action may interpret that remark r differently to the student. The studies suggest that this tendency stems primarily from differences in available information and differences in the processing of its information.

The concepts of attribution processes seem to have a direct relevance for the kind of inquiry intended of the phenomena under examination in this study. This relevance is accentuated when these concepts are applied to the cognitive (attribution) model of achievement motivation as developed by weiner et a_{\pm} . (1971). As Andrews and Debus (1970:194, point out, "the key to this model is the assumption that causal beliefs about success and failure experiences have important consequences for subsequent feelings, expectancies and behavior."

Attributional Model of Achievement Motivation

Attributions "have a special significance when applied to

achievement-related events. These are events for which the outcome " . be judged to be either successful or a failure or some mix of the two" (Smith, 1977:2). Weiner's (1974) attributional model of achievement motivation seeks to examine the effects of attribution on student achievement motivation, and thereby performance. According to the model, causes of success and failure can be represented largely along two dimensions. One dimension is the internal-external locus of control continuum of causes, with ability and effort comprising the properties internal to an individual, and task difficulty and luck being external causes. A second dimension categorizes the same causes according to stability and instability. Ability and task difficulty are the stable causes, while effort and luck are unstable. Other possible causes and dimensions have been suggested by the studies of weiner et al. (1971), Frieze (1976), and Cooper and Burger (1978), but research tends to emphasize the four valual factors as being the " is general and salient of the causes of achievement motivation. The two main dimensions of success and/or failure causality are represented in Figure 1.

Weiner's (1974) attributional model of achievement motivation is presented in Figure 2. A number of antecedent cues act as the bases from which causal ascriptions for success and/or failure are inferrely. These cues include: specific information such as past success \$\u00e4istory, social norms, patterns of performance, time spent on task; causal schemata, which is the relationship perceived between an observed event (effert and the causes of that event; and student individual difference characteristics.



4

Stability	Locus of Control		
	Internal	External	
Statly	Atility	Task Difficulty	
Unstatle (Eifort	Luck	

Figure 1. Weiner's Classification Scheme for Ferceived Determinants of Achievement Behavior.

(Source: Weiner, 1974:52.)

51

J

•

,

	- Mark -	·.	
Behavioral Consequences		indices of juriormance	1
Dimensional Consequences	F - Tancy 1 stifts	Attiont J	
Gausal Dim-nsions	Stahllity	Lorus of control ·→	
Causal Ascriptions		difficulty [7] Luck Others	
Antroordents	Specific cues) Causal schemata	Individual differences Others	

•

1

Figure 2. An Attributional (Expectancy-Value) Medul : Achievement Mativation.

Courses weiner, 1975, 1944, -

٠

• 5~
Attributing an outcome to one or a combination of ability, effort, task difficulty, or luck, according to Weiner, will influence the future expectancy of success and the affective consequences of achievement performance. As Weiner (1970:18.2) points out, this is an integration of attribution processes with Atkinson's (1964) expectancy-value conception of motivation.

As Figure 2 indicates, Weiner's original conception of the consequences of attributing an outcome to a particular cause or causes suggested that the locus of control dimension influences the affective reactions of pride and share. In a chlure situation, when students attribute their performance to an internal cause such as lack of achility or lack of effort they are likely to experience greater encorrupsment and shame than if they attribute the same performance to the external causes such as task difficulty or luck. In a success situation, an attribution to an internal cause results in greater pride and self-satisfaction, whereas success attributed to good luck or ease of task results in less pride.

The affective consequences of success and failure ascriptions have been the subject of considerable scrutiny in recent years. The discussion appears to have derived from Weiner's (1974) position that effort attributions are a more potent source of affect than attributions to ability. Nicholls (1976) and Sohn (1977) conducted studies which indicated that Weiner's statement was in need of some qualification. Nicholls (1976) found that the affective intensity of an effort attribution may well be greater than an ability attribution in the immediate achievement situation, but that over the long term the intensity of affective consequences is reversed--more affect follows an ability attribution than an effort attribution. Sohn (1977) reports that relative affective intensity depends on the emotion involved. An effort attribution seems to generate more "morally unneutral affects" such as pride and shame, whereas ability attributions probably generate more "morally neutral affects" such as happiness and unhappiness. Weiner, Russell, and Lerman (1978:62) have also investigated Weiner's (1974) original position and concluded that "it appears that affects often (but not necessarily always) are directly tied to the causes, without locus of control serving a mediating role." The findings thus far suggest that a problem has arisen in how to conceptualize locus of control within Weiner's attributional approact to motivation, for undoubtedly the internalexternal dimension of causality has a considerable influence on human behavior.

As shown also in Figure 2, the stability dimension is considered to relate directly to cognitive changes in the expectancy of future outcomes following success or failure. When success is perceived by students to be caused by ease of task, the resulting expectancy is one that success probably will occur in the future, given that task difficulty is a stable cause. Attribution of failure to task difficulty results in low expectancy for success. Success attributed to the other stable cause of ability will also result in high expectancy for success, since ability normally is viewed to be constant. Likewise, failure attributed to lack of ability results in low expectancy for success in the future. When success is attributed

-54

to good luck, students might expect failure in the future, since luck is believed to be an unstable cause. Similar expectations are found for attributions to bad luck in a situation of failure. Attributions to the other unstable cause of effort in a failure situation results in higher expectancy for future success since effort is under volitional control.

As Bar-Tal (1978a:203) points out, the types of causes students use to explain their successes or failures tend to have a direct influence on their achievement-related behaviors, such as desire to approach or avoid a task, persistence of behavior, intensity of performance, and choice of tasks. Students identified as persons high in achievement needs will tend to approach achievement situations differently to students low in achievement needs because of differential attributions for outcomes with previous similar achievement situations. As Smith (1977:4) states:

. . . people who consistently experience success are people with high achievement needs. They value success, strive to attain it, and experience it frequently. On the other hand, people who more often experience failure are defined as having low achievement needs. It is likely that these people would prefer success but for whatever reasons have come to accept failure at least in specific situations.

Students with high achievement needs are more likely to attribute success to ability and effort, will therefore experience pride and reward for their successful efforts, and will be more likely to approach subsequent achievement situations. Students with low achievement needs tend to attribute success to external causes, experience little pride for their success, and subsequence are likely to attempt to avoid achievement situations.

-

Because students with high needs for achievement usually attribute failure to effort, which is changeable, they will persist longer. In contrast, because students low in achievement needs usually attribute failure to ability, their persistence in a failing situation will be much lower. Finally, high achievement need students, who believe that success or failure on a task is greatly determined by effort, perform with greater intensity than low achievement need students who perceive outcom s to be attributable to causes other than effort.

A summary of the relationship between bausal attributions and performance on achievement related tasks, as interpreted by Smith (1977), appears in Figure 3. This presents the protable consequences following attribution of the causes of bacceco failure by high and low achievement need students. As stated previously, the consequences reflect both an expectancy and an affective outcome. These, in turn, determine largely the achievement motivation of the student to perform. Ames (1978) found a similar relationship between causal attributions and performance when students were categorized on a high self-concept, low self-concept basis.

Another pritical factor determining the achievement motivation. of a student in an achievement situation is the student perception of tasks. Smith (1977:10-17) analyzes the nature of tasks as they relate to motivation. He considers how a student might define a required task in terms of perceived structure. "The student either consciously or otherwise sees it as structured as a skill task, a chance task, or some combination with two" (Smith, 1977:10). The task will be





Success/Fillure on High/Low Achievement Need Fersons.

Smith, 1977.) (Source:

defined as chance when the student feels the outcome of events is controlled by external factors. That is, his success or failure will depend on factors other than ability and/or effort. Alternatively, when the student defines the task as skill structured, the outcome will to controlled by the internal factors of effort and, or ability.

A second basis by Which tasks in achievement situations can be analyzed in the extent to which a student perceives a task as easy or difficult. Students with nich achievement needs more often choose tasks of intermediate difficulty than do students with low achievement needs. Such a level of difficulty, according to research cited by Shith (1977:13), aroused the greatest emotional or affective reaction in these students and, as well, provides maximum information feedback as at a student's capability and means to improve performance.

Some interesting multional student perceptions of tasks have been reported by Doyle (1975). He was able to conclude that, for a student, the achievement of ightharpoondown task represented a performance-gradeexchange. As far as the students were concerned, the most importantgoal was to acquire the performance capabilities which have thegreatest likelihood of receiving a positive evaluation by the teacher,hence a favorable performance-grade exchange. Success, therefore, wasdependent upon students finding the solution strategies foraccomplishing a particular task which were both reliable andproduction efficient. Enlwanger's (1974) study, however, showed thatstudents may not always use the most efficient solution strategy, solong as success is achieved. Doyle's (1975) dispussion seems tounderscore the differential consequences facing students who attribute

success and perception of task structure to internal causes, such as their own ability and effort, and those students who attribute their success and perception of task structure to external causes.

As Bar-Tal (19756:5) points out, the attributional analysis of achievement related behavior can apply in the classroom situation to teachers' perceptions and behaviors as well as to those of students. Teachers' causal perceptions of their students' successes or failured, using the same attributional structure, probably are as algoritheant as students' causal perceptions, especially in the extent to which they match or mismatch. As the teacher's causal perceptions are thought to underlie the performance expectations held for student, an attributional perspective seems particularly appopriate for attempting to contribute to the explanation of the expectatory effect. In the following section, a review of some of the interrelationships tetween the teacher's causal perceptions and behaviors will be presented.

leachers' and Students' Attributions of Students' Performance

Ear-Tal (19755:L) presents a useful framework for understanding the perceptions and behaviors of students and teachers in terms of Weiner's attributional model of achievement-related behavior. He presents three specific questions pertinent to this study:

- i. how students' causal perceptions of their successes and failures determine their achievement-related behavior;
- 2. Low teachers' causal perceptions of their students' successes and failures determine their behavior toward students; and

3. how teachers' behavior influences students' causal perception of successes and failures.

The review that follows presents evidence in support of these inter relationships between the students' and teachers' causal perceptions and tenaviors.

The relationships between students' attributions of their performance and their achievement related ignavior. In a review of redearches which focused on classic prospheres of students, For fail (1975) described three studies which demonstrated that correctary students attribute academic successes and failures to a variety of reason. However, the most common causes were effort, stilling, mod, whit task difficulty. While two of the studies indicated little or n life rence is the attribute to denote the studies indicated little or n life rence is the attribute to held to attribute success name to extern a causes and failure mainly to internal causes of their successes and failure mainly to internal causes of their successes and failures in academic situations, although more research of a naturalistic kind is needed before any definite patterns of findings can be claimed.

A number of studies have investigated the relationship between students' equival perception of success and failure and achievement related behavior. Many of these seem to have focused on the consequences of failure attributions, investigating the prenomenon described by Seligman (1975) as "learned helplessness." As Brustein (1970:5) reports, learned helplessness pervades the classroom where "failure-prone children perceive their behavior as having no impact on their environment; they are powerless to control the outcomes of events. These children perceive their failure as independent of their actions." Such effects, as H. Gooper (1975:20) points out, result from a feeling of little or no perceived personal control, that is, in Eakla's (144) terms, little effort-outcome covariation. Dweck and Josts (1975:157) categorize students as helpless on the basis of their tendency to disregard effort as a determinant of failure. Students who tend to perceive effort as a determinant of failure are categorized as mustery oriented.

In linking of deterioration in performance following failure has been well established by the researches of Andrews and Debus (1975), Diener and Dweck (1975), Dweck (1975), Dweck and Repucci (1975), hilper (1975), an Seligman (1975). These studies revealed that helpless students tended to attribute failure to lack of ability. Equally estiving evidence has been provided by the studies of Andrews and Debus (1975), Dweck (1975), and Dweck and Repucci (1973) regarding the paccessful use of training procedures to redirect the attribute is of poor performance, normally to effort. As Brustein (1975;5) states, "the success of 'attribution retraining' lies in the fast that the children were supplied a strate. The control the trainatic experience of failure." The intention and direction of these cognitive retraining procedures would appear to resemble closely the training programs developed by deCharms (1975).

deCharms' (1955, 1975) work on personal motivation incorporates attributional aspects. His origin-pawn concept is a specific application of attribution of internal causality to oneself. As

 $\mathbf{0}$

decharms (1970:5) states, "to help a person to be an origin is not to determine his goals but to help him develop a commitment and purpose so that he can reach his own goals more effectively." The basic elements in the conditions which promote origin behavior, according to deCharms (1976), involve self-study, internal goal setting, planning and goal-directed behavior, and personal responsibility. That these conditions can be translated into a training program in the school setting was successfully shown by deCharms (1976). 62

The studies presented seem to indicate that the kinds of Scases students use in explaining successes and failures will determine their achievement related behavior. The evidence consistently shows that students who attribute failure to unstable causes, such as low effort, are more persistent in their behavior than students who attribute their failure to a stable cause which is perceived to be unchangeable. Consequently, intervention programs designed to alter student attributions of failure to personally controllable, distable causes have met with success.

The relationships between teachers' attributions of students.' performances and their behavior toward the students. The interrelationship of causal perceptions by teachers and their effect on behavior toward students has been the subject of extensive research. The National Institute of Education's Panel 6 Report (1975:36) urged researchers "to determine what attributions are made by teachers for the performance of their pupils, to determine the consequences of these attributions for pupil behaviors, and to discover methods for changing attributional decisions." After citing some research evidence, Far-Tal (1978b:1.) concludes that teachers of elementary level students tend to attribute the students' successes to student effort and interest, their own quality of explanations, and in part to the students' home conditions. Teachers tended to attribute failure to students' lack of effort, difficulty of the material, and inappropriate home conditions. These research results also found support for Snyder, Stephan, and Rosenfield's (1976) notion of "attributional egetism." A definite tendency prevailed for teachers to chars the credit is reaccess with the students but distribute blane for staient failure to chases external to the teacher.

Teachern' classi perceptions of their students' performances probably are associated with their renaviors towards students through the mediating influence of teacher performance expectations of students. Most reviewers of research into the formation of teacher performance expectations of students, such as Braun (1970), Brophy and Evertson (1976b), and H. Gooper (1970), showed that teachers particularly noted student past achievement, intelligence of the student, and sex of the student. A related line of research has been concerned with how information concerning a student influences teachers' performance expectations of students. This has evolved from weiner's (1974:21) statement that:

ascriptions of an outcome to stable factors produce greater typical shifts in expectancy than do ascriptions to unstable factors. A typical shift is defined as an increment in the subjective expectancy of success following a success experience and a decrement in expectancy after a failure.

Some evidence is available which indicates that teachers do form differential expectations on the basis of mainly stable causes.

Weiner and Kukla (1970:5) showed that student teachers tend to reward more favorably students who are successful, who are of high ability, and who are highly motivated. Bar-Tal and Saxe (in press) found that, from the causes provided regarding a student's performance, teachers formed their expectations mainly from ability (the greatest effect, effort, and luck.' Bar-Tal (1978b:14) cites research undertaken by Darom and Bar-Tal (1977, which showed that "the more success or failure was attributed to stable causes such as ability, easy subject matter or easy test, the more future outcome was expected to be similar to the prior outcome."

Teachers' performance expectations for students have been shown to influence teachers' differential behavior toward students., Most of the research evidence for this relationship was reviewed in an earlier section, but a further model for "expectation communication" has been postulated by H. Jooper (1975). This model is significant for it focuses upon some attributional determinants for differential teachers' behaviors toward students. Attribution research suggests that many persons endeavor to maintain a sense of perceived personal control and freedom in their behavior in any situation. Cooper postulates the thesis that this concept might provide the basis of the classroom expectancy effect.

In many respects, Cooper's still largely unresearched model contains similar features to earlier models that seek to explain how teacher expectation effects are communicated to students. However, the distinguishing feature of this model is the hypothesized vesting of the causal sequence in the teacher's motive to retain a sense of

perceived personal control over classroom interactions and student performance. According to H. Cooper's (1978:15) model, "expectations and context characteristics influence teacher perceptions of control over student performance."

The model, as represented in Figure 4, proposes:

1) that teachers form differential expectations for students' performances; 2) that expectations, in conjunction with the interaction context, influence teacner perceptions of control over student performance; 3) that control perceptions influence teacher feedback information and the socio-emotional climate of the classroom; 4) that feedback contingencies influence student beliefs concerning the importance of effort 1r. producing personal outcomes; and 5) that effort-outcome perceptions influence the quality of student performance. (M. Copper. 1975; 14)

Differential expectations are formed such that low expectation students and high expectation students are identified. But teacher perceptions of control over student performance are largely context based. Consequently, H. Cooper suggests that some situations of classroom interaction provide a greater sense of control for the teacher than others. Teacher-initiated interaction as compared to student-initiated interaction, and private interaction as compared to public interaction settings are two buch situations. According to the model, teachers are considered to feel the greatest personal control over student performance in teacher-initiated private interaction settings, and the least personal control in studentinitiated public interaction settings. The crucial variables of these situational factors are content of the interaction, timing of the interaction, and duration of the interaction.

H. Cooper (1978:17) states that these situational factors are likely to interact with student characteristics. High expectation t5



.



(Source: H. Gooper, 1978)

students are likely to succeed regardless of situational factors. Low expectation students' success, on the other hand, should depend to some extent on difficulty of the task and on the amount of time available to bring about the success, that is, contest factors. Therefore, by manipulating some of the context variables, Cooper and Seymour (1977) found that teachers may feel they can exert some control over low expectation students' performance. Cooper's model suggests that, as they find student-initiated public interaction settings least controllable, teachers might be expected to take steps to avoid such situations with low expectation students.

H. Cooper (1978:15) explains that teachers who have a need for a sense of perceived personal control could manipulate classroom interactions with low expectation students through differential patterns of back and the kind of classroom climate established. As found by H. Cooper (1977), by using feedback and expression of effect, teachers can inhibit low expectation student-initiated public interactions and thereby retain a sense of perceived personal control over these students. A study by Entwisle and Webster (1972), involving teacher use of praise, realized similar findings. The feedback and affect used with high expectation students will be more often based on the specific quality of the performance, such as effort. The need for a sense of perceived personal control over their interactions would appear from the model to be less of a matter for concern.

Finally, the relationship between teacher feedback and effort contingent outcomes for students links Cooper's model with that of

67

. 1

Weiner's (1974) attributional model of achievement motivation. Low expectation students will perceive little effort-outcome covariation in their performance. For them, this represents very little personal control, but for the teacher the situation reflects considerable perceived personal control. Quantitative and qualitative aspects of feedback and affect by the teacher means the low expectation student is likely to attribute his success or failure to external and/or unstable causes. There seems every likelihood that such teacher behavior will influence the student's self-image.

H. Cooper's model of experiention communication and performance influence intrudes into the review of literature and research pertaining to the relationship between teachers' behavior and students' cancel perceptions. Given the nature of the expectancy effect, expectations formed on the basis of causal perceptions of students' performances do influence teachers' behaviors. These, in turn, probably affect students' causal perceptions of successes and failures. The next section reviews the research on this relationship.

The relationships between teachers' behavior toward the students and the students' attributions of their performance. Students' causal perceptions of success and failure seem to be under the influence of teacher behavior. Bar-Tal (1978b:16) cites a classroom-based study by Blumenfeld, Hamilton, Wessels, and Falkner (1977) which suggests that teachers transmit to students three kinds of messages, all of which influence students' causal perception:

1. Explicit attributional feedback which refers to causes contributing to success or failure; emphasized mainly ability

and effort for achieving positive outcomes.

- 2. Explanations for behavioral expectations and evaluative feedback; were of an extrinsic nature such as referred to authority, rules, or circumstances.
- 3. General statements referring to social or procedural expectations; consisted more of threats and punishments than rewards

However, the most notable evidence regarding the relationship between teachers' behaviors and student cause rceptions seems to be provided by the sex differences stu AAT bution studies in social rsychology, such as those une Deaux (1976) and Deaux and Farris (197 noted that a tendency exists for males to attribute succes lity and failure to unstable causes, and for females to attri success to luck and failure to ability. Dweck and her associates have investigated sex differences in attribution as they function in the classroom. Dweck and Goetz (1978:165-173. concluded that the differential causal perception of boys and girls in the classroom is due to dimential teacher behaviors. Dweck, Davidson, Nelson, and Enna (1978:274) found that teachers were overall more critical of boys and used more negative evaluation in both academic and non-academic matters. Consequently, boys more easily sattributed negative feedback to the attitude of the teacher rather than to the intellectual quality of their performance. When boys did view the negative feedback to be contingent on the quality of their work they braned lack of effort, for teachers had tended to attribute

boys failures to lack of motivation. With girls, teachers provided

largely positive assessments and only used negative feedback for girls' intellectual failures, without emphasizing motivation as a cause, Girls tended to attribute negative feedback to the intellect on quality of their work and perceived lack of ability as the reason for failure.

Regarding positive evaluation, the reverse effect, while being consistent, was less pronounced. Because girls received overall more positive feedback in both academic and non-academic matters, they attributed teacher praise contingent on the quality of work more to the teacher's favorable attitude or non-intellectual aspects of their performance. Positive feedback for boys tended to be specific for academic performance, hence empuraging an attribution by toys to ability. This differential feedback of teachers influences directly boys' and girls' causal perceptions of success and failure.

Other studies, such as those of H. Cooper (1977) and Entwise and Mebster (1972) which were reviewed earlier also indicate ways in which teachers' differential behaviors are related to students' causal perceptions of performance. Weinstein and Middlestadt (1978:21) found, in a classroom-based study, that differential teacher treatment is perceived by students and, as a consequence, fudents learn a great deal about the achievement hierarchy of the classroom. This finding led Weinstein and Middlestadt (1978:22) to suggest that "the expectations of classroom peers for a student's performance can be as critical to the student's own developing self-expectations as those of the teacher." Such a suggestion raises the need to consider briefly the impact of significant others on students' self-attributions and

S in

.∩1 -

self-expectations.

The significant other. The notion of significant others, according to Webster and Sobieszek (1974), was initially suggested by H. S. Sullivan in 1947. He defined significant other as "one whose opinions 'matter' to the individual, someone whose approval he desires, and whose censure he wishes to avoid" (cited by Webster & Sobieszek, 1974:419). Additional notions have been proposed to this original concept. For instance, a significant other's "performance evaluations are likely to be accepted by the actor and used as a basis for forming his expectations for himself and other actors" (Webster & Solieszek, 1974:124). As well, the significant other is viewed as having the competence to evaluate, both in terms of his greater ability and his formal status. Kash and Horier. (1975:14) differentiate between what they refer to as a significant other and a salient other. To them the significant other is "an individual selected and unconditionally valued by the developing self as a source of self reflection and an interveter of the behavioral dialogue." The salient other, on the thir hand, is a person who functions as a significant other only r specific events and is valued for a specific reflection of the self. Whereas the teacher appears as a significant other to the student, a hockey club coach may fulfill the role of a salient other.

ē

In the classroom learning situation, significant others for a student may be the teacher, certain peers, perhaps the principal, and probably the parents. Two studies have examined the influence of significant others in modifying a student's self-evaluation, and hence his self-expectation. Wetster and Entwiste (1974) and Rappaport and Rappaport (1975) both demonstrated the effect with small samples of disadvantaged students. The influence of significant others should be a relevant notion in research dealing with self-expectations of students.

Summary

Attribution theory and its application to achievement-related situations in general, and to classroom learning situations in farticular, constitutes a rich line of literature and research. Recaise attributions are phenomena common in everyday existence, their validity as research variables for human behavior seem more of the Also, because the research tradition of attributions nellets, in Glaser and Strauss' (1967) terms, more of a grounded theory developmental process, the research area seems very applicable to descriptive, exploratory studies of students functioning in their regular classroom environment.

That attribution processes can prove to be relevant to the study of the expectancy effect has been established in this section of the review of related literature and research of the study. The attributional analyses of behavior probably facilitate the most specific and probing inquiry yet undertaken.

Summary

This chapter reflects the blending of the two domains of literature and research underlying the investigation. One domain, teacher expectation research, is a relatively recent area of inquiry derived from the classroom elimate tradition of research in teaching. The other domain, attributional analyses of orders' and teachers' behavior, has evolved from attribution the ry, a rapidly developing area of inquiry in social psychology. The dynamics underlying the expectancy phenomena may well be described, interpreted, and explained by these two related research domains.

٠.)

A theme recurring throughout the expectation research and the research in teaching generally is one of reconceptualizing and retining the research variation. The recognition of indequacies of research variables has resulted in a surge of interest with the mental life of classroom participants, and very recently this change has need adopted increasingly in the research on teacher expectations. Two outcomes of this expansion of variables to incorporate the mental lives of students and teachers have been, first, the increased hopes for finding explanations of the dynamics or mechanisms underlying the expectancy phenomena and, second, the application of attribution theory and research to assist in this search for explanations.

Access to the thoughts and feelings of students and teachers should help in understanding the interrelationships between and among teachers' perceptions of students and their behaviors, teachers' causal perceptions of students' performances, teachers' behavior, and students' perceptions of teachers' behaviors. Hopefully, the dyna of the process expectation effects will be explained by this approxim

This study is one of very few researches undertaken thus far which have examined in depth students' perceptions of the processes thought to underlie the expectancy effect. Weinstein and Middlestadt

(1777:..), who undertook such a study, point cot that it is imperative to approach process expectation effects from the student's position for a number of reasons. First, performance outcomes of achievement, with one or two exceptions, have been the only student variables considered so far in classroom expectancy research. Second, research on student perceptions into classroom climate factors have, in the just, used classroom means of student observations, thereby discounting the important effects of individual perspectives A third reason for examining student perspectives on the expectancy effect is the recognition of the differential perspectives of the student and the tracher involved. This reason is underlined by the significance of the Jones and Nisbett (147. Thesis of actor-conserver discrepancies in cause, attributions of an electric However, as Harvey, Ickes, and Hidd (1977:153 point out, mayte this reaction flects a wider research discrepancy that only now is mbeing handled, namely, the virtual -absence of research attention given "to the interactive nature of inference making in social relationships and to the special problems that arise when one person makes attributions about another's attributions." This concern is related to Weinstein and Middlestadt's (1975:+) fourth reason that "student perceptions may serve as intervening variables which mediate the effects of teaching behaviors on achievement."

Given the dearth of research evidence on students' thoughts and deelings in the learning situation, this study--a classroom-based research. When a contribute to the growing body of knowledge underlying the teaching/learning process. 74+

Chapter III

DESIGN AND PROCEDURES

Overview

This chapter begins with a description of and a rationale for the design of the study, a description of the subjects, the specific problems to be investigated, the assumptions underlying the research, and finitations of the study. In subsequent sections there are accounts of the data collected in the study, the major research methodologies used, and the phases of the study which include descriptions of the research procedures followed.

Design

The research project was planned as a set of four case studies designed to facilitate the exploration and description of the expectancy phenomena as viewed by students. The case study method was considered an appropriate research design for two main reasons. The first reason derives from the fact that students' conceptions of self-performance in classroom achievement situations are basically unknown. In order to provide a data base, intensive studies of. individual cases should be undertaken to establish some understanding and explanation of what Harré and Secord (1972:133) referred to as the generative mechanisms of social behavior.



- ----

The case study method catered to the exploratory intention of the research. In Kerlinger's (1973:405) terms, such an exploratory study should attempt "to discover significant variables in the eld situation, to discover relations among variables, and to lar groundwork for later, more systematic and rigorous testine of hypotheses." In this way the study would be conforming to first phase of Resenshine and Furst's (1973) descriptive-correlationalexperimental loop for research in teaching.

The second main reason for adopting the case study approach eentered on the fact that the research was set in the context of a regular classroom during normal teaching/learning sessions. This naturalistic setting comprized an extremely complex aggregation of intrapersonal, interpersonal, and environmental processes over and above the expected occurrence of the learning processes. The case study method, by its approach to the phenomena and its informal techniques for data gathering, seemed to offer the most effective means for studying the realities of the classroom. Stake (1970:7) underscores the case study in these situations when he states that:

. . . most case studies feature: descriptions that are complex, holistic, and involving a myriad of not highly isolated variables; data that are likely to be gathered at least partly by personalistic observation; and a writing style that is informal, perhaps narrative, possibly with verbatim quotation, illustration, and even allusion and metaphor. Comparisons are implicit rather than explicit. Themes and hypotheses may be important, but they remain subordinate to the understanding of the case.

Similarly, MacDonald (1978) noted that practice-based research avoids the constraints of Formal research because it is concerned essentially with trying to find insight into existing reality rather than

developing some internal reality.

While the intensive small-scale nature of the research facilitated the desired micro-level study of the phenomena under investigation, it also accommodated certain practical constraints such as manpower, resources, and time.

The Suljects

<u>Selection</u> of the Teacher and <u>Student Subjects</u>

The choice of school for the research was determined largely by a teacher at the required grade level who volunteered to participate. The school was a typical large elementary school located in a residential community in a large uncan center. The Grade & class of 25 students was accommodated in a temporary classroom.

The selection of Grade t level of elementary students was based upon findings in the pilot study which indicated that: 1, younger students were less comprehensive and articulate in vertalizing thoughts and feelings about self-performance; and 2, peers seemed to exert a greater influence on Grade 6 students' behavior, thereby sharpening an awareness of self-performance in the learning situation.

Selection of students. From the class, four target students were selected from the 21 students whose parents consented to their participation. These four target students are the four case studies of this research. Two of these were high achievement/high motivation students and two were low achievement/low motivation students. The research literature suggested that differences exist in the perceptions and behaviors of these two categories of students. The balance of the students were used as decoy students, four of whom were considered as "back-ups" to the target students.

Target students were identified by use of pre-observation data of the students obtained from five sources. First, the Intellectual Achievement Responsibility Scale (TAR) developed by Grandall, Eatkovsky, and Grandall (1965) provided student measures on an internal-external locus of control scale. The scale is reported to assess students' beliefs that they, rather than other people, are responsible for their intellectual-academic success and failure. Of particular significance, as Grandall et al. (1965:93) point out, the scale "limits the source of external control to those persons who most often come in face-to-face contact with a child, his parents, teachers, and peers." The scale also limits students' beliefs in locus of control to intellectual-academic achievement situations. The scale, With some accompanying psychometric data, is presented in Appendix A.

The second source of data used to assist in identifying the target students was the Projected Academic Performance Scale (PAPS,, a student self-expectation scale recently developed by Chapman and Boersma (1975). The scale measures student self-expectations in school subjects on both academic and affective criteria. For each of the school subject areas of spelling, reading, language arts, mathematics, social studies, and science, students were required t. predict for the current year, and over the long term, their selfexpectations of ability to learn the subject matter, of enjoyment of the subject matter, and of performance in relation to other students. Psychometric data on the PAPS are in the process of being published but, according to Chapman (1979), acceptable external validity and test-retest.reliability have been established.

The students' levels of achievement in mathematics, the subject area chosen for this study, comprized the third source of data. These were obtained from the students' cumulative records, where the levels of achievement were presented as percentiles for all Grade 6 students within the school district.

A fourth source of data used in the identification of target students consisted of the teacher's comments and ratings of the students' performances. Prior to the observation phase, the researcher adopted the role of "counselor" in Cooper and Paron's (1977) study by discussing with the teacher the relative progress and performance of each student in the subject area of mathematics. In the course of her comments, the teacher rated the students' achievements in mathematics as either above average, average, or below average.

Finally, the researcher made an assessment of the suitability of potential target students and this data contributed to the final selection. Factors considered by the researcher, following interviews with the students, included ease of establishing rapport with the student, student's "verbalme o," and student's clarity of voice and speech as revealed by audio-cassette recordings.

The 10 students on the short list were observed carefully during the familiarization phase of the study which included three rehearsals of the research procedure. Students who bes, fitted the criteria and who proved to be consistent during the pre-observation phase were chosen as the four target students. Table 1 provides a summary of the information about the target students.

The Lesson Context

٠.

ι

Mathematics lessons were chosen as the context for the study. The formal, disciplined nature of the subject matter, as described by Lovell (1962), was believed to accentuate the achievement-related nature of the situation. All the lessons involved a study unit on motion geometry from the introduction of the unit through to the end-of-unit test. The study unit followed the material prescribed in the curriculum guides for Grade 6 mathematics.

While the potential uniqueness of teacher and student behavior in mathematics might well be attributable to the characteristics of the subject matter, research by Evertson, Anderson, Edgar, Minter, and Brophy (1977) would suggest that this is not so. In the first extensive study following Shavelson and Dempsey-Atwood's (1976) statement on the application of generalizability theory to classroom. behavioral data, Evertson et al. (1977) found consistency of behavior of the same student when observed in junior high school mathematics and English classes. The stable variables were classified ty Evertson et al. (1977:54) in four ways:

. . . the extent to which a student functioned as an <u>independent</u>, <u>task oriented worker</u>; the extent to which the <u>student initiated</u> <u>his or her involvement in interactions</u>; the consistency of the <u>quantity and quality of teacher-student contacts</u>; and <u>the tone</u> of the interactions.

The results of this study were judged to have relevance to the current study given the proximity of grade level of the students.

Ŷ

	1 0	ł				1	•
	Teache r 's	Rating	High	High	Low	Low	
	Maths	Achievement Percentile	. 22	89	5	6	
Selection	ations)	Total	149	145	100	147	128.8
Table : Summary of Objective Type Information Used in Selection of Target Students	PAPS (Self-expectations)	Maths Subscore	25	25	17	54	23.3
Tat e Type Ir of Target	irol)	Total	26	24	24	26	24.3
objectiv	cus of Control)	Internal Failure	12	6	۲۹ ۴٦	0	11.3
Summary o	IAR (Locus	Internal Success	14		C I	16	13
		Age	11	11	1	11	
		Sex	ĹŦ.	¥	[x.,	Σ	F.
		Name	Tina (T)	Mike (M)	Lisa (L)	Greg (G)	Class Mean

•

Specific Problems

The specific problems investigated related to four major areas.

1. What are the characteristics of students' behavior during a lesson?

What categories of student overt behavior occur during a lesson? What is the frequency of the various student overt

What are the relationships between student overt behaviors and lesson situations?

What categories of student covert behavior are reported to occur during a lesson?

What is the frequency of the various student covert behaviors reported by students? What are the characteristics of a student's reported

behaviors?

What are the characteristics of a student's reported covert behavior which pertain to self-performance? What comparisons can be drawn between the reported covert behavior related to self-performance of high achievement/ high motivation students and low achievement/low motivation students?

2. What are the conceptions of self-performance that students appear to develop in classroom achievement situations? What sorts of ideas, beliefs, views, emotions, lines of reasoning make up a student's conception of selfperformance? What sort of relationships exist among the ideas, beliefs, views, emotions, and lines of reasoning? What seem to be the origins of these ideas, views, beliefs, emotions, lines of reasonings of a student's conception of self-performance? What role does a student's causal perception of success and failure appear to fulfill in a conception of self-performance? What, if any, are some of the characteristic qualities of a student's conception of self-performance?

3. From an tributional perspective, what phenomena characterize teacher expectancy effects?

How do a student's causal perceptions of his successes and failures determine his achievement-related behavior? How does a teacher's behavior influence a student's causal perception of his successes and failures? How does a teacher's causal perception of her students' successes and failures determine her behavior toward students?

What is the nature and extent of consensus between the teacher and student regarding the causal perceptions of the student's successes and failures?

4. How effective are stimulated recall procedures when used as a research technique for observing the covert behavior of classroom participants?

What difficulties or problems are likely to be encountered

in the use of stimulated recall with the teacher and the students?

What appears to be the potential of the stimulated recall methodology for investigating student interactive thoughts?

Assumptions

This investigation of the students' perceptions of the processes underlying the expectancy phenomena was based on three substantive assumptions:

- That the student entered and participated in achievementrelated situations with thoughts and feelings about his self-performance.
- 2. That the student's conception of self-performance influenced his achievement-related behavior.
- 3. That the students, in some form and to some extent, were influenced by the expectancy effect phenomena. The following two assumptions in the study pertained to methodological issues:
 - Introspective reports produced under conditions of stimulated recall from videotape recordings enabled effective and legitimate data from the teacher's and the students' covert behavior to be obtained.
 - 2. An adequate characterization of each student's conception of self-performance and its relation to learning experiences was derived from all the sources of data used, including the

introspective reports.

Limitations

Y The major limitations of the study were:

1. The sample of school, teacher, students, and lessons was small.

2. Random sampling was not used.

3. Classroom observation was only of a short time period and in only one subject area.

- 4. The problem of asking elementary school students introspective and reflective questions and receiving reliable usable answers was one that was difficult to overcome.
- 5. By not including teacher self-expectations, a significant component of the interactive expectation process may have been excluded.
- 6. While essentially a naturalistic study in character, certain observer obtrusive effects probably distorted the behaviors observed.

These limitations precluded the making of generalizations beyond the individual teacher and students involved in the study. However, the findings may suggest what other researchers might look for in similar or related situations.

Data and Data Sources

The kinds of data for considering the stated research problems and the sources of, or procedures used to obtain the data are

outlined below.

Videotapes of Lessons

Ten mathematics lessons of 45 to 60 minutes duration were recorded on videotape. These videotapes were used: 1) to stimulate the recall of the teacher's and students' interactive thoughts; and 2) as a data source for coding student overt behavior and classroom interactions between the teacher and target students.

During the recording of the videotapes, a lesson overview was prepared which described the sequence of the 1-sson activities, the development of mathematical ideas, and the occurrence of teacherstudent interactions.

Interviews

Interviews with the teacher and the students yielded several kinds of data depending on the type of interview conducted.

Stimulated recall interviews with the teacher and the students were conducted on the same day as the lessons were recorded. These interviews yielded the teacher's and students' interactive thoughts. The interviews, ranging from 30 to 60 minutes with the teacher, 20 to 40 minutes with the target students, and 10 to 30 minutes with the decoy students, were all audiotaped. Transcripts of all interviews were then prepared. Procedures and techniques for structuring some aspects of the stimulated recall interviews with the teacher and the students were developed from those used by Marland (1977) and C. Cooper (1978) and are presented in Appendix B.

Interviews intended to yield data on student self-performance

took two forms. First, a pre-lesson interview and several postlesson interviews were conducted with target students during the observation phase. Also a 10 minute pre-test interview and a 10 Inute post-test interview were conducted with each of the target students. Interview schedules were developed for these interviews (see Appendix C). All interviews were recorded on audio-cassette tape-recorders and transcripts prepared.

A second form of interview designed to yield self-performance data was integrated with the stimulated recall interviews. As opportunities arose, such as particular incidents in the lesson, pertinent interactive thoughts reported by the student, or relevant non-interactive comments made by the student, the researcher manipulated the interview to inquire into student self-performance phenomena. A rationale for, and procedures used with, the integrated interviews are described later in this chapter.

Pre-observation phase interviews were conducted with the teacher and the students primarily to obtain data which would assist in the selection of target students. These interviews provided a variety of introductory data, some of which were relevant to the expectancy effect.

The Low Inference Observation System: A Revision of the Expanded Brophy-Good Teacher-Pupil Dyadic Interaction Classroom Observation System

This instrument is a detailed low-inference observation system. The system provides a means of recording naturally occurring sequences of teacher-pupil interaction in elementary classrooms as

well as every dyadic interaction between the teacher and individual students. Also, the instrument takes account of some of the contextual differences in classroom interactions and, as explained by Brophy and Good (1969, 1970) and Brophy and Evertson (1973), is
Dased on real and psychologically meaningful units. The authors report that it is possible to train coders to reach an 80 percent agreement criterion using a strict definition of agreement.

The instrument yields both quantitative and qualitative data. Each video recording of the lessons provided the data source from which the dyadic interactions between the teacher and each of the target students were coded. The resulting data of teacher-student interactive overt behavior contributed to the attempted analysis and explanation of the expectancy effect phenomena.

Because the teacher in this study used a cordless microphone, the researcher was able to record private dyadic interactions between the teacher and the student, that is, those interactions normally intended only for the hearing of the two participants, such as qufetly conducted desk supervision. The existing system of Brophy and Good's was not adequate to categorize these interactions in detail. Therefore, the researcher redesigned the sections dealing with student initiated dyadic teacher-pupil contacts and teacher afforded dyadic contacts, employing the kinds of categories Brophy and Good used in the public response opportunities. This set of additional categories facilitated a much sharper and more detailed analysis of interaction behavior between the teacher and individual students. The revised low inference system is fully outlined in Appendix D.
Field Notes

A variety of field notes were recorded throughout the familiarization and observation phases of the study. These nates recorded general contextual information including community, school, classfroom, the subject area, the unit of study, and extra-curricular matters that influenced classroom behavior, such as team soccer. Hallowe'en, Remembrance Day, school photographa, visits of art exhibitions, etc. Also recorded in the field notes were specific data relating to the phenomena being researched. Target student behaviors and related teacher behaviors not observed during the interviews or recorded on videotape were noted whenever they were observed to occur. Such field notes were informal, anecdotal, and recorded at the moment.

Research Methodology

This research required the investigator to study the mental life of students and the interactive thoughts of the teacher. For this purpose, a refinement of the observation-interview method was selected as the major data gathering methodology. The observationinterview method is based on observations, discussions, and interviews with participants over a period of time. Stimulated recall methodology offered the possibility of condensing these three components of the observation-interview method into a more intensive and analytical interview approach.

Stimulated Recall Methodology

According to Conners (1978b:1): -: . stimulated recall is a branch of introspective methodology in which audio and/or visual cues are presented to facilitate a subject's, recall of the covert mental activity which occurred simultaneously with the presented cue or stimuli.

At the classroom level, stimulated recall methodology tends to involve the recording of a lesson or event on audiotape or videotape. Classroom participants, either the teacher or the students, at a later point view and/or listen to the recording. The stimulus tends to assist the person in recalling the thoughts and feelings experienced in the actual lesson or event. The expression of these interactive thoughts is then recorded on audiotape.

Marland (1977:37) points to three uses of stimulated recallin studies of teaching and learning, in medical education, and in psychotherapy and therapeutic counselor education--but as a research tool it does not have a long history. This lack of extensive use can largely be attributed to a tradition of research skepticism held toward the use of introspective techniques in the behavioral sciences generally. However, as Marland notes, the commonly expressed opposition, which tends to focus on the potential for error in introspection, has been countered in part by recent developments in research, including the rise in interest of stimulated recall. A gradually changing research attitude seems to be prevailing, as if in a response to the call by writers such as Harré and Secord (1972:151, 154) who have advocated a return to the use in social behavior research of participants' accounts of realities and phenomena. Foth Conners (19755) and Marland (1977), in their use of stimulated recall methodology, attended to the precautions to be taken to enhance the probability of accurate recall of thoughts, which was the critical major objection to the technique. These precautions include the interviewer's use of skilfall questioning techniques, the small re-lag between recall of thoughts and the actual event (according to Fluor, 1953, this should be less than 45 hours, and the appropriate preparation of the interviewee.

The problem of accuracy of recall of thoughts raises the issue of their volidity. As Conners (1975a:L94) notes, three linked problems are the get to underlie accuracy of recall, namely, "can the individual remember, secondly, is the individual reporting his actual throughts, or is he distorting therefor some reason, and thirdly, is as hey selectively reporting his inclusion. And the second of the individual theorem is no single way to establish the validity of the individual theorem, Mariand (1977.227) reports that a logical consistency was perceived between interactive thoughts and events on videotape. Conners (1976a:293) conducted a number of checks which, according to Saier (1954), enable a researcher to infer the validity of the individual's reports. As with Mariani, Conners was allo to infer that acceptate validity was established. Therefore, as Mariand (1977:227, concludes, "validity and reliatility can be assumed out not demonstrated or guaranteed."

0

Must classroom researchers who use stimulated recall methodology express positive comment about the quantity and quality if data which is yielded. In general, the data is described as rich and interesting, and seems to be providing new and useful insights

into the covert behavior of classroom participants.

}-

1.

The promising potential of the technique is being recognized gradually. Most classroom investigators who have used this technique, such as Clark and Peterson (1976), Conners (1978a), C. Cooper (1978), Marland (1977) and Nolan (1978), point out that subjects must be willing and able to verbalize thoughts and feelings as accurately and as completely as possible. Stimulated recall methodology is : used on the assumption that such reports about these thoughts and feelings are reasonably accurate representations of that behavior.

The Observation-Interview Methodology

Normally the observation-interview method consists of three components, namely, an observation of an individual's action or segment of behavior, discussion with the individual and possibly other involved participants about the action or segment of behavior, and an interview with at least the individual concerned. The method is open to the use of participant or non-participant observation or, as used in this study, to the condensing of the components through the use of Y2

stimulated recall methodology. For case study research designs the observation-interview method facilitates extensive data gathering of all forms of overt behavior as well as gaining access to individuals' thoughts, feelings, conceptions, social perceptions, and so on.

1

The observation-interview method has been the subject of some criticism. Criticisms about the cost factor in terms of time and effort, and that the data are difficult to analyze and interpret are protlems which researchers using the method tend to accommodate. Of more significance, though, are the criticisms which claim that the data obtained lacks reliability, validity and objectivity. However, as Erlwanger (1974:314) points out, the validity of these criticisms may well depend on the purpose of the interview. The criticisms may te valid for interviews in which the emphasis is to assess the knowledge of subject matter acquired by students in order to arrive at conclusions or generalizations. But where the emphasis is on attempting to study and understand the phenomena underlying students' tehavior, and the concern is more on the value of interviews as a nears of gaining access to students' ideas rather than with its limitations, then the criticisms of interviews seem to be less valid. Erlwanger's clarification is particularly salient for the use of interviews in this study.

In using the observation-interview method, this study was concerned with the student's ideas underlying his behavior rather than with specific aspects of his behavior. The bases of the interviews were mainly observations made during the lesson and which appeared on the television monitor, points arising from the student's interactive

and non-interactive comments, and matters recorded in the field notes.

The interviews were structured and sequenced such that the students perceived the sessions to be concerned largely. with the recall of interactive thoughts. Certainly this was the impression conveyed by the interviewer when preparing the students for the interviews. Hence, stimulated recall techniques not only yielded valuable data on student interactive thoughts but served as a trigger for in-depth investigation of students' underlying ideas, beliefs, lines of reasoning, and so on. During the student's recall of thoughts or viewing of the lesson the interviewer was alert for clues. When these appeared, the interviewer would manipulate the interview to prote. Some clues led nowhere. On other occasions, the interviewer missed a great deal because he failed to grasp the significance of some of the students' responses. Care was taken by the interviewer to avoid a systematic approach to probing which could have reflected preconceived intentions. Likewise, suggestions and leading questions were minimized. As Piaget (1960:38) points out, though, suggestive questions do have a place: "Naturally, these last questions, which are suggestive, must be kept to the end, that is to say till the moment when the child cannot be made to say anything of itself." Erlwanger (1974:329) used suggestive questions on occasions "to assess a child's ideas or to confirm a hunch about the child's ideas." In this study, given the intent of the interviews, suggestive and leading questions were sometimes used.

An important component of the observation-interview method is the relationship between the interviewer and the student. A positive

rapport in the interview situation reflects a considerable amount of time and effort being expended by the interviewer in a variety of situations with the student prior to the interviews.

In this study the interviewer was conscious of the need to gain student confidence and trust in order to enhance the likelihood of valid data being obtained. The interviewer expected that students would be curious to know what was the purpose of the research, that they would be concerned about anonymity, and so on. At all times fairness and honesty was intended and maintained, with explanations cast in the form of generalities rather than as deceptions. Trust was only ensured when students became aware that the interviewer did not "talk."

However, trust alone was found to be not the only requirement for students to feel free to reveal their own understanding of selfperformance. Most students were not accustomed to discussing ideas or expressing points of view about their self-performance. This seemed to be compounded further by the student having to talk in this way with an adult figure. Therefore, the establishing of the personal relationship between the interviewer and the students also concentrated on this unforeseen problem.

Some limitations of the observation-interview method are recognized. Only a small number of students can be adequately observed and interviewed at any one time period. Data is obtained in , an informal manner but, most of all, success of the method depends on an adequate relationship being established between the interviewer and students. This relationship, in part, determines the success of the

interviews, while other factors include the preparation and perceptiveness of the interviewer. A final limitation pertains to the ethical responsibility of the interviewer. In this study the . interviewer had to monitor carefully just how far he could go in probing the private thoughts and feelings of the students.

Validity of the Research Data

A number of major validity issues had to be considered in the gathering of data for this study. Those that pertain directly to the particular research methodologies used in this study have already been discussed. However, other validity issues also presented a problem when designing the study.

This study was investigating aspects of the expectancy effect. phenomena in a natural classroom setting. In order to preserve any naturally occurring expectancy effects, it was essential that the teacher and student be unaware that the expectancy effect phenomena were being investigated and that target students were being used. The research strategy, therefore, meant that all students whose parents had consented were interviewed. Data was collected on decoy students but these were much less extensive. Because of time constraints, decoy students were often interviewed in pairs and/or were interviewed for shorter time periods. Target students were interviewed alone until one target student asked when could she be interviewed with a friend. Target students were interviewed each day, whereas most decoy students missed an occasional day. To enhance the screening of use of target students, all possible scrambling strategies were used to avoid the setting of any pattern. The researcher believes that all

students and the teacher remained unaware of the use of target students. This is verified by a post-observation open-ended statement completed by the teacher (see Appendix E).

The obtrusive effects on the classroom participants of observations and interviews and videotape recording equipment raise another class of error sources which could invalidate some of the data. Webb, Campbell, Schwartz, and Sechrest (1966:12-21) describe some of the kind of effects which influence people's behavior. The awareness of being tested, the selection of a particular role, the effect of initial observations upon subsequent observations, and the adoption of particular response sets were all considered to prevail in the classroom. A two week period of familiarization was employed in an attempt to reduce these effects. This appeared to diminish the novelty effect of students seeing themselves on a television monitor.

As Webb et al. (1966:21) note, certain interviewer effects also may prevail in an interview. The investigator had little control over these effects other than to attempt to maintain a neutral inquiry role.

A real concern for the interviewer was his position as perceived by the students. For them there existed a potential role conflict between the interviewer as teacher/adult (someone who told them what to do) and the interviewer as researcher. It interviewer sought to harmonize this role conflict. The students varied markedly in the extent to which they were able to discuss ideas and express views to the interviewer. This appeared to depend on what his role happened to be to the students. Those who perceived the interviewer

in the teacher/adult role tended to provide more formal responses saying what they thought was required. Students who perceived the interviewer more as a researcher tended to provide more personal responses by relating aspects of their private and personal ideas and views. As the observation phase of the study progressed, more and more students perceived the interviewer in the researcher role.

Phases in the Study

The study proceeded through three phases. The first phase consisted of two pilot study periods which the investigator used to develop adequate technical competency with the equipment, to finalize decisions regarding the research design and procedures and use of instruments, and to provide experience with interviewing techniques. The second phase was the period of familiarization in which the investigator completed a number of pre-observation tasks. The third phase was the period of observation in which data was collected.

Pilot Studies

Two rilot studies, each of one week's duration, were undertaken in April 1978 and June 1978 at a different school to that used in the research. They were undertaken to fulfill five main functions:

 The research was based on one particularly salient assumption, namely, that students enter and participate in a learning situation with thoughts and feelings about their selfperformance. A primary function of the pilot studies was to verify whether this assumption could be justified.

- 2. The projected research design consisted of pre-lesson interviews as well as extended post-lesson interviews. This and other features of the design were to be checked for practical viability.
- 3. The use of videotape recording equipment in a classroom necessitated some field experimentation and testing in order to gain the best possible visual and audio recordings.
- 4. The researcher had no previous experience at interviewing students for research purposes. The interviewer tried a number of different strategies in order to evaluate his technique in terms of the quality and quantity of data efficited.
 - 5. He researcher was interested in determining from which grade level of student the most relevant and comprehensive data might be obtained. For this reason, the pilot studies involved both Grade 4 and Grade 5 level students and their teachers. One Grade 6 class and two Grade 4 classes comprized the sample. The subject area for the lessons was mathematics.

Results of the pilot studies. Analysis of the audiotape recordings of the interviews justified the assumption upon which the study was based, namely, students do enter and participate in a learning situation with thoughts and feelings about their selfperformance. Effort and task difficulty attributions were cited mainly as explanations for success or failure. Also, achievementrelated behavior was very much under the influence of peer effects.

The research design and procedures were found to be workable. Stimulated recall methodology was judged to be a useful research tool with students of elementary school level. Doubts had arisen, however, about the effectiveness of pre-lesson interviews. These were intended originally to provide data on target students' thoughts and feelings pertaining to self-performance about the forthcoming lesson, but the data appeared to become repetitious after successive days. The decision whether or not to retain pre-lesson interviews was held over till the familiarization phase.

The pilot studies raised the very real need to adopt the strategies which would screen the use of target students in the st However, such strategies were expected to be costly in terms of timing.

From the pilot studies the researcher gained valuable experience in using the videotape recording equipment. Audio problems were overcome sufficiently to ensure effective recordings.

The researcher gained essential experience with interviewing. Later evaluation by the researcher and another graduate student enabled feedback on interviewer style, especially with questioning techniques. In terms of the most effective interviewing strategy, the pilot studies indicated that an integration of stimulated recall methodology to elicit interactive thoughts with probing questions to obtain self-performance thoughts and feelings yielded the more relevant data.

The teacher and the students were all able to recall interactive thoughts, suggesting that the procedures and the role definitions for interviewer and interviewees, as summarized in Appendix B, were functional and sufficient. Some students were less verbal than others. While this may reflect inadequate establishing of rapport by the interviewer, it alerted attention to the need to select students who were able and willing to verbalize thoughts and feelings.

The pilot studies indicated that Grade 6 students were better able to express thoughts and feelings about self-performance once good rapport was established. Maybe self-performance is of greater concern to these students. Most certainly they seemed more conscious of peer influences and this factor seemed to affect greatly their achievementrelated behavior in the classroom.

Familiarization Phase

For two weeks immediately prior to the data gathering and observation phase a variety of familiarization and pre-observation procedures were undertaken in the classroom.

1. The researcher introduced himself as a visitor to the classroom a few days before the two week familiarization period. The importance of the initial contact with the teacher and the students was fully recognized. The teacher volunteer had consented to participate before the study was introduced to the students. The researcher spoke to the students in terms of being ipterested in what students and the teacher thought as well as what they did during the teaching/learning process. The use of videotape recordings and interviews was explained prior to distribution of parent consent forms.

L

- ... The researcher familiarized himself with classroom routines, schedules, resources.
- 3. The researcher familiarized himself with the students in terms of memorizing names, noticing characteristics, and identifying potential target student candidates.
- 4. The researcher introduced the class to videotape recordings, both with the actual recording process and with the playback. The teacher and students viewed themselves on videotape during mathematics lessons on three occasions before interview rehearsals using stimulated recall methodology were introduced. The awareness of being observed effects were prominent during the first two or three days but the distraction of "being on T.V." diminished progressively following this initial phase. Technical concerns with the videotape recordings were overcome at this time as well.
- 5. The researcher orientated the teacher and the students to the interview situation. Outlining the roles of the teacher and students in the interview setting was deemed essential for yielding effective data.
- 6. The researcher administered the IAR and the PAPS scales, and consulted the students' cumulative records to obtain data to aid in the selection of target students.
- 7. The researcher conducted initial pre-observation interviews with the teacher and with the students for the purpose of obtaining data which would contribute to the process of selecting target students.
- "The researcher rehearsed the research procedure on three

consecutive occasions. These rehearsals involved recording the mathematics lessons on videotape and conducting the post-lesson interviews with all students. A number of final decisions were made during these rehearsals. First, the target students were finally selected. Second, the pre-lesson interviews with the target students were dropped from the research design because of timing difficulties and because elements of these interviews were accommodated within the post-lesson interviews. Third, the use of screening strategies involving all students were found to be viable and they were retained.

9. The researcher developed the vital personal relationship with the teacher and students. This relationship seemed to underlie the effectiveness of the entire study. The researcher spent considerable time interacting and talking with the teacher and students in a variety of situations and venues.

In the judgement of the researcher, classroom life was less affected by the obtrusion of the research after two weeks of familiarization. The videotape recording equipment still attracted some attention but the distractions were short-lived. The researcher generally felt accepted as a person in the classroom group.

Data Collection Phase

The process data were collected by the researcher over a two week period in early November, 1978. The research design consisted of observing and collecting data during 10 consecutive mathematics lessons. These mathematics lessons occurred in the first hour of the school day, leaving the balance of the day for previewing the videotapes and interviewing.

The research procedures involved with videotape recording, previewing the videotapes, and interviewing are detailed in the following sections.

Videotape recording of the lessons. The entire mathematics lesson on each day was recorded on videotape. One Sony Videocamera CVC 2100A and a half-inch reel-to-reel Javelin Videotape Recorder VTR 2000 were used for recording the lesson. The picture being recorded was viewed on a Sony 110 Television Monitor. Two microphones were used to record sound. A multi-directional Sony ECM 150 Microphone was placed above the blackboard and a Lectrosonics Wireless Microphone M30R/R3: was worn by the teacher. The sound from these microphones was directed into a Snure audio mixer and then recorded on videotape. The researcher, who operated all the recording alone, monitored the incoming sound through the use of a set of Sharpe H-A-10 Mark II Headphones. The audio mixer allowed the researcher to localize the recording of classroom interactions when desired.

For best visual effects for the purposes of this study, the recording camera was located near the front and away to one side of the room. By regularly panning the camera within a 45 degree are the front or side profiles of the teacher and all the students were recorded.

While recording the lessons the researcher took care to avoid selective concentration of the recorded picture and/or sound on the target students, hence no zooming for close-ups was used. A regular panning of the camera across the class in general was maintained to facilitate later coding of student overt behavior. Frief pauses in the panning only securred when a dyadic interaction between the teacher and any student was being sustained.

During preserved by the sequence of happenings throughout the researcher. This described the sequence of happenings throughout the lesson. Progressions of teaching ideas, changes in lesson activities, significant or intriguing questions or comments from either the teacher or student, and every dyadic public or private interaction between the teacher and the student - neerned were recorded against a VTR clanter realing and clot the time on the clock (see Appendix) for the less of part of a lesson overview). Extential stimulus points for the lesson presence interaction interaction interaction interaction interaction interaction interaction comments from the clock (see Appendix) for the lesson overview). Extential stimulus points for the lesson, were also indicated.

Freviewing the viagotageg. Inhediately following the interview room, the researcher previewed the videotage recording in the interview room, therearcher previewed the videotage recording in the interview room, therearly a storeroom in the main school fullaling. The preview facilitated the confirmation of appropriate features in the resson which would serve as useful stimulus points in stimulated recall discussions. These features had been first noted on the lesson overviewe-the lesson description completed by the researcher during the reporting of the resson. In addition, ideas to be discussed and questions to be raised with the teacher and the students were often determined while previewing the videotages. The researcher was ally to preview the videotages in the period before merning recess.

The interviews. M st of the interviews with the students

'consisted of their reporting interactive thoughts through use of stimulated recall. Integrated with these reports were the interviewer ' manipulated segments of the interview which probed for selfperformance data. The interviews with the teacher only employed stimulated recall methodology.

The target students were, with one exception, interviewed individually, were interviewed daily, and were never interviewed in the same order from one day to the next. The decoy students were interviewed either individually or in pairs, and this varied for each student on a day by day basis. Due to timing difficulties, it was not possible to interview each decoy student every — However, as Table 2 indicates, with an average daily interview rate of 10 decoy students, a reasonable coverage was achieved. A random order of interviewing also was maintained with the decoy students.

Generally, the first students to be interviewed in any one day were decoy students. The researcher often used these initial interviews to clear up any doubts or misconceptions regarding the content or progression of the lesson. As well, if a particular self-performance notion had come this mind the researcher would often explore the relevancy and meaningfulness of that notion with decoy students. At another level the researcher would seek to refine this questioning technique, such as practising the wording or phrasing, and observing the style most readily understood by the student.

The order of a typical day's interviewing would likely consist of the following:

1. The first two interviews, with decoy students, would occur

Tahlu J

•

Distribution of Grouping Modes Used When Interviewing Target Students and Decoy Students During the Data Collection Fhase

•

					Lessons	10				
Jrouping mode		2	~	-==	5	9	6	ω	6	0.
Target studentsindividual	+	4	1	. 	4	, ش	t	4		4
Target studentspairs	I	I	ł	ı	I		ı	I	ı	ł
Decoy studentsindividual	ti	$\tilde{}$	4 .	• •	ı	I		9.	. 	2:
Decoy studentspairs	ω	ဆ	4	C.	ω	۲۰.	œ	I	I	I
Total students interviewed	<u>.</u> 9	5:	† .	• •	7	• •	£:•	50	ω	5.

Mean number of students interviewed per lesson = [4.]

immediately following morning recess. Normally, a target . student was also interviewed before lunch.

- 2. During the lunch hour, either the teacher was interviewed or, when this was not possible, decoy students who were available and willing were interviewed.
- 3. Throughout fternoon an intermingling of decoy and target students we interviewed.
- 4. After school, the teacher was interviewed on those occasions when she had not been available for a lunch hour interview.

A student or pair of students would be withdrawn from regular class either by the researcher or the preceding interviewee(s). The role definitions for the interviewee (as obtlined in Appendix F) were repeated on most occasions, though this seemed unnecessary after the initial interviewing periods. The interviewer controlled the VTR, but selection of stimulus points was also available to the student interviewees. On numerous occasions they selected points on the recording for reporting interactive thoughts. The interviews with target students often commenced and/or concluded with an inquiry into self-performance thoughts and feelings relative to the lesson of the day or the forthcoming lesson on the next day. These inquiries were similar in direction and intention to the former pre-lesson interviews which were dropped as a data gathering source.

The stimulated recall interviews with the teacher, in many respects, were similar in procedures to those undertaken with the students. The interviewer was less verbally active during interviews with the teacher. As well, she was able to nominate more stimulus points than the interviewer for reporting interactive thoughts.

The ninth lesson of the ten lessons which comprized the data collection period was given over to an end-of-unit test. This was perceived by the researcher to be a critical opportunity to gather student self-performance data in a particularly sensitive achievementrelated situation. As a consequence, pre-test and post-test interview schedules (as presented in Appendix C) were developed. These were based on the format and structure of the former pre-lesson interview schedules which were trialled and modified during the pilot study. The pre-test interview was conducted with every student on the day before the test. Likewise, the post-test interview was conducted with every student on the day after the test--the day the students received the results of the test. For the target students the pre-test and post-test interviews were conducted in conjunction with the regular interviews. The accoy students' pre-test and post-test interviews were conducted on an individual basis with no accompanying stimulated recall interview because of timing difficulties.

All interviews with the teacher and with the students were audiotaped on a Sony Cassette Tape-recorder, Model TC-1:0E.

<u>Classroom overt behaviors</u>. Student overt behavior data and teacher-student dyadic interaction data were obtained from the video recordings after the data collection phase. The researcher developed a coding system for recording student overt behavior data from observation of students' behaviors. The coding system is described in Chapter IV. Student overt behaviors were recorded once every minute on a time sample basis.

The teacher-student dyadic interactions were also coded from the video recordings using a low inference system based on the Expanded Brophy-Good Teacher-Pupil Dyadic Interaction Classroom Observation System (Brophy and Evertson, 1973).

Summary

This study was designed to examine some of the phenomena underlying the expectancy effect in a regular functioning classroom. Ten consecutive mathematics lessons were videotaped in a Grade 6 classroom for use with follow-up stimulated recall interviews. Four target students reported their interactive thoughts and described their thoughts and feelings pertaining to self-performance in mathematics. The teacher also reported her interactive thoughts. The videotape recordings of the lessons also facilitated later observation and analysis of teacher and student overt behavior. Such a research design was intended to yield data which provided insights into the attributional behavior of classroom participants. These insights may well contribute to an explanation of the teacher expectancy phenomena. 110

÷.,

Chapter IV

DATA ANALYSIS

Overview

18

Two main sets of data were gathered in this study: 1) student behavior data; and 2) teacher-student dyadic interaction process data. For the purposes of data gathering and analyses of that data, student behavior while learning mathematics in the regular classroom was divided into overt and covert aspects. The teacher-student dyadic interaction process data also incorporated both overt and covert aspects.

This chapter outlines the development of category systems for analyzing student behavior data. Student overt tehavior and students' interactive thoughts and feelings were categorized and quantified to provide a description of student process behavior. Other student covert behavior, such as non-interactive thoughts about the lesson situations and causal explanations of behavior, were analyzed by qualitative means.

Teacher-student dyadic interaction process data at an observable level were coded using a low inference observation system, while teacher interactive thoughts pertaining to students and students' performances were submitted to a qualitative analysis.

The two sets of data, when analyzed, were used to conduct a summary qualitative analysis which provided information about the phenomena underlying the expectancy effect.

Analysis of Student Overt Behavior

A category system for analyzing student overt behavior during classroom learning was developed by inductive means. The students' behavior during 10 mathematics lessons was recorded on videotape and the examination of these videotapes by the researcher led to the building of the category system.

The Development of the Category System

The initial examination of the videotapes indicated that a time sample of recording student overt behavior was the most feasible. In order to pick up all students' and the teacher's behaviors in a systematic manner, the researcher found it necessary to pan the classroom regularly when operating the videocamera. A time sample of one minute intervals was judged to be practical and to be efficient in reflecting specific behaviors and behavior changes.

The development of the category system stemmed from observation of the target students' behaviors. At one level students appeared to be either on-task or off-task. On-task behaviors were defined as those which appeared to indicate that the student was concentrating on the progression of the lesson, on the learning task required of the students, and/or on the subject matter directly under consideration by the teacher and students. Generally the student would be observed to be attending or listening. On some occasions the student might have his head on the desk, be fidgeting, or doing work in relation to the task but, so long as the researcher was able to judge that the student was mainly attending or listening, he was categorized as being on-task.

Off-task behaviors seemed to take a number of forms of self-involved distractions. These behaviors included just staring aside as if daydreaming, watching others, fidgeting in or on his desk, doing work not related to the task before the class, and head on desk. where the researcher judged the student's intention to be off-task then the behavior was categorized accordingly.

At times students interacted with other students or were out of their desks. Because it was difficult to always assess which of these were on-task or which were off-task, the researcher categorized these behaviors separately. Some of the students' interactions with other students seemed work-related while others seemed non-workrelated. Other interactions seemed to involve analyzing or deliberately distracting elements. Students were out of their desks for a number of reasons. Some of these included approaching the teacher, moving to be with other students, just moving around the classroom for no apparent purpose, being absent from the room, or moving to procure materials.

A category of behavior judged to be on-task, yet treated separately from the attending-listening category, was those occasions when the student was being supervised by the teacher either at the student's desk or elsewhere in the room. This supervision consisted of either the teacher merely observing the student's work or

interacting verbally with the student. Such instances of behavior always occurred during seatwork. Being supervised was treated separately because the behavior involved a dyadic interaction and was considered to provide support data to the low inference observation system which was used to code classroom dyadic interactions.

Three broad categories of lesson situations were observed to occur during the mathematics lesson:

- 1. Teacher-led whole class instruction.
- ... Student demonstration, whole class instruction.
- 3. Seatwork.

Teacher-led whole class instruction consisted of teacher exposition or lecture, teacher controlled discussion, and teacher demonstration using the blackboard, overhead projector or other apparatus and equipment. Student demonstration in a whole class instruction situation referred to these occasions when the teacher called upon a student to demonstrate an idea, reinforce a teaching point, or attempt a discovery of an idea, either at the blackboard, with the apparatus, or using the overhead projector. Seatwork included those activities in which the students were assigned to work independently on worksheets, practice with apparatus, and so on.

The categories of student overt behavior in relation to the categories of lesson situations are summarized as a two-dimensional framework in Figure 5.

		Categor	Categories of Student Overt Behavior	ert Behavior	
Categories of Lesson Situations	Om-Task	Off-Task	Interaction with Students	Out of Desk	Supervised
Teacher-led whole class instruction					
Student demonstration, whole class instruction					
Seatwork		,			
Figure 5. Frame Relat	work for Rec ion to Catego	ording Catego ortes of Loss	Framework for Recording Categories of Student Overt Behavior in Relation to Categories of Lesson Situations.	ert Behavior in	

Relation to Categories of Lesson 31

L)

115

,

Procedures for Coding Student Overt Behavior

From viewing the ideo recordings of the 10 mathematics lessons, the researcher was able to observe and record the overt behavior of all four target students in this study at the same time. At minute intervals (accurate to the nearest 5 seconds) each target student's behavior was coded by the researcher onto the framework of categories of student overt behavior in relation to categories of lesson situations presented in Figure 5. Coding was sustained throughout the lesson except for transition periods at the beginning of the lesson, during the lesson, and at the end of the lesson.

Following coding, the data for each target student was analyzed on a lesson by lesson basis and percentage distributions derived.

Reliability

Reliability coefficients for intercoder and intracoder reliability of coding from the videotapes were calculated. A student in a doctoral program in the Department of Educational Administration at the University of Alberta acted as coder. He experienced a two-hour training period during which the categories of the system were discussed and the coding process practised. A sample of two lessons was coded completely in the reliability check. Intracoder reliability was undertaken two months later with another sample of two lessons.

Because the researcher anticipated a high frequency of coded decisions would be obtained for one of the categories--on-task

behaviors--consideration, in terms of a reliability estimate, was given to what Johnson and Bolstad (1973:13) refer to as the base-rate problem, "that is, the obtained percent agreement figure should be compared with the amount of agreement that could be obtained by chance." Scott's formula, as described by Holsti (1969:140), adjusts for the occurrence of chance and, therefore, was used in this study to calculate the reliability coefficient. Scott's formula is:

$$R = \frac{Po - Pe}{1 - Pe}$$

where Io represents the percentage observed agreement between coders, and Pe the percentage of expected agreement by chance.

Scott's formula is considered to provide a stringent estimate of reliability, yet because the complexity of the category system used to analyze student overt behavior was considered not great, 0.8 was established as an acceptable level of reliability. Tatle 3 contains a summary of the results of the reliability checks. Reasonable measures of reliability in stability of researcher coding and in the category system were obtained.

<u>Content Analysis System of Student</u> <u>Interactive Thoughts</u>

A category system for analyzing student process covert behavior, that is, the student's interactive thoughts and feelings during the classroom learning of mathematics, was also developed by inductive means. The Content Analysis System of Student Interactive Thoughts (CASSIT) was developed in accordance with the principles and

Table 3

•

, **.**

Intercoder and Intracoder Measures of Reliability in the Use of the Category System for Analyzing Student Overt Behavior

• • • • •		
	Sample Coded	Coefficient of Reliatility
Intorcoder reliability	\mathcal{L} lessons (Net 4 and Net 8	v.~~ •
Intracoder reliability	2 lessons (No. 3 and No. 6)	ບ.ສາ

118

12

.

requirements for developing a content analysis system provided by Hulsti (1965).

Holdti (1909:14 described content analysis as a technique used "for making inferences by systematically and objectively identifying specified characteristics of messages." Objectivity refers to the need for precise descriptions of categories, while the systematic component of content analysis ensures the application of olear guidelines. Such notions stress non antiguity. The third element of Holdti's description emphablies generality effindings must have the retical relevance.

The process of coding is used to approach data such as contribut in transcripts of interviews. The late, in Helsti's (interview, terms, "are systematically transformer and approxated interview, terms, "are systematically transformer and approxated into units which permit precise description of relevant content characteristics." This is achieved by two fundamental coding steps. First, the data source is segmented into units which are reaningful to the research problem. Second, the units are placed into one of several discrete, clearly defined categories; the categories bing relevant to the researcher's theories.

ine Development of the Sategory System

The inductive development of GALUTE involved repeated examinations of the transcripts containing the data before tells finalized. The phases of development were:

 The differentiation of interactive phenomena from non-interactive phenomena. As well as students' interactive thoughts and feelings, the transcripts contained such. non-interactive verbal reports as causal explanations of behavior, preactive thoughts, post-active thoughts, and descriptions of what they were doing.

- 2. The establishment of the unit of analysis to be used. For this study, similar to that adopted by Conners (1975a) and Marland (1977), a single thought or idea embodied in a single word, a part of a sentence, or an entire paragraph was chosen as the unit of analysis.
- J. The classification of the many different units of interactive thoughts and feelings into distinguishable groups. Each group consisted of a set of thought units which shared one or nore distinct characteristics. These groups were the categories of student interactive thoughts and feelings. They were, therefore, mutually exclusive, were exhaustive in that all data were classifiable, and they reflected the purposes of the research.

A full description of CASSIT is presented in Appendix G. In Frief, the main components of the system include a set of nineguidelines for distinguishing between interactive and non-interactive data, a unit of analysis as being based on a single thought or idea, and nine categories of units of interactive thoughts. A brief description of these categories is as follow

43

Subject Matter: Units in which a student's substance to used specifically on the content and skills of mathematics.

Cognitive irocesses: Units in which a student reports a thought

process involved in learning the subject matter.

Behavioral Moves--Self: Units in which the student reports his thoughts about an action he was performing, had performed, or was considering performing in relation to the learning process. Such an action must reflect a personality orientation.

<u>Hehavioral Moves--Student</u>: Units in which the student reports his thoughts and feelings about an action involving other students. Hehavioral Moves--Teacher: A unit in which the student reports his

thoughts and feelings about an action involving the teacher. Sold -Ferformance--Thoughts: Units in which the student is thinking about his or her performance behavior and outcomes. The essential element of these work-related units is self. Sold -Ferformance--Heelings: Units in which the student reports an emotion pertaining to his or her performance behavior and outcomes.

Rectings: Units in which the student reports an emotion experienced during a lesson which is not associated with self-performance. Non-Task-Related: Units of thought which do not pertain to the

learning of the subject matter.

Jiven the purpose of this study in student classroom behavior in general, and self-performance thoughts and feelings in particular, the thought units in six of the categories of the content analysis system, namely, behavioral moves--self, behavioral moves--student, behavioral moves--teacher, self-performance--thets, selfperformance--feelings, and feelings, were analyzed further. Subcategories within each of these categories were clearly defined and established.

Reliability

With most content analysis systems concern with reliability is usually directed at the coding processes of unitizing and of categorizing. The CASSIT used in this study introduced an additional reliability concern. The coding process which is used to distinguish interactive data from non-interactive data should also be subjected to a reliability check. Conners (1978a:107) notes that reliability checks should be undertaken with all three coding steps and he used three different coefficients of reliability accordingly. The researcher employed the same procedures of establishing reliability for the coding processes in this study.

To establish reliability coefficients for the coding of data into interactive and non-interactive categories, which Holsti (1969:138) refers to as dichotomous decisions, the ratio of coding agreements to the total number of coding decisions was used:

Holsti (1969:440) explains that, "in this formula M is the number of coding decisions on which the two judges are in agreement, and N₁ and N₂ refer to the number of coding decisions made by judges f and 2 respectively."

Reliability coefficients for the coding process of unitizing reflect the consistency with which coders segment the data into a similar number of units. Conners (1978a:107) cites Guetzkow's contention that unitizing reliability can be established by "expressing the difference between two coders as a percentage of the sum of the number of units obtained by each coder."

$$U = \frac{O_1 - O_2}{O_1 + O_2}$$

In this formula 0_1 is the number of units established by the first coder and φ_2 is the number of units established by the second coder. For this formula, a ratio of zero indicates perfect agreement between coders.

To establish reliability coefficients for the coding process of categorization, use was again made of Scott's formula. Holsti (1969:140) reports that this formula takes into account "the extent of intercoder agreement which may result from chance." Hence, the formula corrects for the number of categories in the content analysis system and for the probable frequency with which each is used.

Scott's formula is:

Reliability
$$= \frac{Po - Pe}{1.00 - Pe}$$

In this formula, Fc represents the agreement between two observers and Pe represents the agreement between two observers which occurs by chance.

Reliability checks. Both intracoder and intercoder reliability checks were undertaken. Intracoder reliability, an index of stability of the researcher's coding, was calculated on two separate occasions, two months apart, using a sample of 15 segments from the stimulated recall transcripts of all four target students on each occasion. Intercoder reliability, an index of system reliability, was determined using the same sample of 15 segments as with intracoder reliability. A student in a doctoral program in Educational Administration at the University of Alberta acted as a coder. Training of four hours was undertaken and consisted of discussion of all elements of CASSIT and practice in coding.

According to Holsti (1969:142), "defining an acceptable level of reliatility is one of the many problems in content analysis for which there is no single solution." He suggests, therefore, that in content analysis research the determination of an acceptable level of reliability needs to be determined in relation to the complexity of the content analysis system used and the significance of the research problem involved.

Marland (1977:85) established a reliability coefficient of 0.70 as an acceptable level of reliability. Given the complexity of the content analysis system used, and that Scott's formula was described by Holsti (1969:141) as one "which produces a conservative estimate of reliability," a reliability coefficient of 0.70 was considered to be an acceptable level of reliability for this study.

The results of the reliability checks are presented in Table 4. Both intracoder and intercoder reliability coefficients for the categorization process of coding were greater than the 0.70 set for acceptable reliability, that is, the most significant reliability check of the three used for the content analysis system. This indicated that CASSIT seemed to be a reliable instrument.
Table 4

125

,

.

.

Intercoder and Intracoder Measures of Reliability in the Use of CASSIT

	Coefficient of Reliability
Intercoder Reliability	
Interactive versus non-interactive data	0.73
Litization	0.12
Categorization	0.79
Istracoder Reliability	
Interactive versus non-interactive data	0.86
Unitization	0.02
Categorization	0.89

.

The indices of reliability for the coding processes of distinguishing interactive from non-interactive data and of unitization are also acceptable. Intercoder reliability estimates are close to the pre-established levels of adequacy. Maybe more training time to enhance proficiency in these coding processes would have resulted in higher reliability being achieved. Examination of the coding decisions of the coder who undertook the reliability check indicated a practice effect occurring in the initial begments of data. The slightly depressed results do emphasize the need to connect reliability checks for these processes, for error and lack of consistency with one of the coding processes must compound the reliability of the other occurs.

Measures of intraction reliability were acceptable and suggested researcher stability in coding the data over at least the short term of two months.

Attributional Analyses of Student Sovert Behavior Lata

The transcripts of the post-lesson interviews contained more student covert behavior than the student interactive thoughts derived irom stimulated recall. Among the non-interactive data were an array of ideas, views, beliefs, emotions, and lines of reasoning which served to explain student interactive thoughts and realings and student overt behavior. Some of the data were elicited incidentally in association with student reports of interactive thoughts. Other data were obtained in a more direct manner from struct and questioning, such as in the pre-test and post-test interviews. Much.

...

of these data pertained to student self-performance. And these of these data were undertaken for each target student in order to describe each student's conceptions of self-performance.

The analyses of the data pertaining to student self-performance sought to identify two sets of phenomena, namely: 1) the causal explanations of a student's expression of thoughts and feelings about self in the learning situation; and 2) the characteristics of a student's inderlying ideas, views, belief's, emotions, and lines of reasoning, that is, the base which seems to influence the causal derivations of cenavior.

The distinction between the two sets of phenomena may be parely archtrary. However, the researcher approached the phenomena in this exploratory study as if they consisted of some dynamic process aspects and more enduring aspects. Each of the aspects were considered from an attributional perspective.

Jausal Explanations of Behavior

A framework for analyzing the data to obtain causal explanations of behavior was developed by inductive means. The researcher examined all transcripts and noted patterns of statements which seemed to be related contextually. Where an interactive thought or feeling which contained a cue about self-performance had teen expressed by the student, the interviewer had often pursued the point by using probing questions to elicit further information about the student's mental life. This questioning tactic tended to result in a chain of student comments, usually linked in the following pattern:



The student's report of an initial cognition with or without an emotional consequence often referred to an aspect of selfperformance. Sometimes the student had included a reason for his behavior but more often the interviewer had elicited the reason by asking such questions as: Why did you think that?

Why did you do that?

How come that you . . .

The reason provided by the student tended to be related directly to the initial cognition, so usually the interviewer probed further with such questions as: Why do you think (the reason given)

made you think or do (the initial cognition or emotion). Student responses to these questions tended to disclose belief systems, lines of reasoning, and points of view underlying a student's thoughts and feelings about his self-performance.

The following examples of chains of student comments include the emotional consequence linkage.

When I got \rightarrow I was so \rightarrow cack my r.appy test paper	I had worked → hard on the worksheets and always listened	to show on my report	I nardly heard anything she said when she went through the test. (Tina,
When $I \rightarrow I'm \text{ scared} \rightarrow$ need help to ask the teacher	think I	I worry \rightarrow about what the teacher thinks of me	

Other chains of student comments do not **cont**ain an emotional consequence linkage.

 \rightarrow I know I only \rightarrow If I don't --- Will check with I don't the teacher as understand understand got it right something I'm to how it is it through really the guesswork not atraid to uiter school. ask (Tina)

Not all chains of stuient comments included the consequent behavior linkage. Consequent behaviors tended to become the initiating tenavior for another chain of student comment.

After collating chains of student comments which more or less matched the evolved framework or pattern, the researcher then placed into clusters those chains which contained, referred to, or reflected common initial coefficients, perceptions, and/or motives of tehavior. From these clusters, evidence of consistencies in the causal explanations or clusal perceptions of a student's behavior were discerned. As a consequence, a dynamic view of a student's covert behavior pertaining to self-performance in the learning situation began to emerge.

The Underlying Covert Benavior of Students

In the process of examining the data for causal explanations of tenavior, the researcher became aware of numerous student thoughts which seemed to represent the very base from which originated all other thought and action pertaining to self-performance. When these generic-like ideas, beliefs, views, emotions, and lines of reasoning were identified and recorded for each student, a composite array of phenomena were revealed which contributed greatly to the description of a student's conception of self-performance.

- Some of these student ideas and beliefs had developed into a kind of implicit theory about successful achievement as indicated in the following example:

0 1

- T: He and I are about the same. We are pretty good at math.
- I: why do you think you're pretty good at math?
- I: Well, I think I try harder and I get my homework done all the time. You know, if you just use your brains you can do something you want. Some of them, they...you know, you can tell, like they are smart kids but they just don't use their brains.
- I: What about other kids you think are not smart?
- T: Well they are smart. Like, you know, just not as good as some others. Because they don't try, they don't like the subject, they don't bother trying. And since they don't get things right they hate it even more, you know. (Tina)

Other underlying thoughts of students reflected a single basic

idea, telief, or point of view, as in the following examples:

- Mike: Hids who do well should be recognized in front of the class.
- Lisa: The teacher will remember even the smallest trings months later.
- Greg: I prefer half the worksheet to be easy and half of it to be hard, because the easy work helps you to understand, and so lets you try better on the hard work which makes you think more.
- Tina: Boys do better in maths because girls like more the language arts.

Descriptions of the phenomena of underlying studen* - vert benavior yielded a number of insights into student causal perceptions of success and failure, of task structure and difficulty, of other students' performances, and of the teacher's performance expectations for them. The phenomena seemed to influence greatly the causal explanations and perceptions students expressed in relation to their behavior. While the phenomena of students' underlying thoughts were not static forms of covert behavior in the strict sense, they seemed to be reasonably stable during the period of observation : this study.

The Characterization of Student Conception. of Self-Performance

The attributional **an**alyses of student covert behavior pertaining to self-performance has involved a descriptive approacn. Commonalities among the chains of thoughts and among the underlying tase of ideas, views, beliefs, emotions, and lines of reasoning have teen derived inductively from examination of the transcripts of the target students' verbal reports. From a description of these interrelated commonalities the researcher has been able to infer and "piece together" some of the more significant characteristics of each student's conception of self-performance during the learning of mathematics.

Analysis of Dyadic Interaction Data

A major purpose for building up a characterization of a student's conception of self-performance was to attempt an explanation of the expectancy effect. The expectancy effect is thought to involve mainly process elements or variables, especially those pertaining to teacher-student interactions. Data from different sources were gathered on the dyadic interactions between the teacher and each of the target students. The main data sources were:

1. video recordings of the lesson from which overt behavior of

the interaction participants was coded by use of the low inference classroom observation system;

- 2. transcripts of post-lesson interviews using stimulated recall methodology with the teacher which contained verbal reports of teacher interactive thoughts on some of the interactions; and
- 3. Finings are it student behavior and student conceptions of solf performance which had been analyzed previously.

The method of analysis used for these various data was based on the suggestion by Webb, Campbell, Schwartz, and Sechrest (1966:174) about the triangulation of different data as a means of validating evidence. The inter-relationships of the different sources of data for an interaction episode are illustrated to gure -. This method of analysis involved an examination of all the ata collected, that is, the interview data and the observational data

Interview Data

Students. The students' covert behavior data, as obtained from interview situations, were already analyzed, resulting in the characterization of students' conceptions of self-performance.

<u>Teacher</u>. The teacher's verbal reports of her thoughts and feelings during the observation lessons, as derived from the use of stimulated recall methodology, were recorded and transcripts were prepared. These transcripts were examined and sections of interactive and non-interactive thoughts relative to the target students, and especially the teacher-student dyadic interactions, were extracted and



Figure : Representation of the Interrelationships Among and Between Different Sources of Data for an Interaction Episode analyzed. These thoughts tended to consist of causal perceptions of a student's performance, performance expectations for that student, thoughts about the student's mental processes in learning and understanding, feelings about the student, and perceptions and interpretations about student behavior in general.

The Observation Data

Dyadic interaction data were coded through use of the low inference classroom observation system. The system yields both quantitative and qualitative data. Frequency of interaction between the teacher and each target student, frequency of student-initiated as against teacher-initiated contact, and frequency of private interactions as against public interactions provided data of a quantitative nature. Fatterns of use of various teacher reactions or feedback to student responses to questions, or to student initiated contacts, enabled some qualitative data of teacher tenavior in the interactions to be obtained.

The video recordings of the mathematics ressons provided the data source for coding the dyadic interactions. Coding was conducted in accordance with the procedure outlined in Brophy and Good's (1959) manual for coding classroom behavior.

Intracoder and intercoder reliability checks were undertaken. Reliability was calculated using a formula proposed by Brophy and Evertson (1973). The formula is: Percentage agreement - number of coding decisions made by both coders and agreed upon, divided by itsel: plus the number of coding decisions not agreed upon plus the number of codings made by the first coder but not the second plus the number of codings made by the second coder but not the first.

An 60 percent agreement criterion was set as an adequate measure of reliability for both intracoder and intercoder checks. A student in a doctoral program in the Department of Educational Administration at the University of Alberta acted as a coder. This student had previously used the low inference observation system. After two hours of retraining, which involved discussion of the observation system and practice conducted the video recording, he was able to code the interactions between the teacher and each of the four target students during two complete lessons. An intracoder reliability check was conducted two months after the original coding. In each case the measure of reliability was within the agreement criterion of do percent which had been recommended by Brophy and Eventson (1973 - Table 5 presents a summary of the measures of reliability.

All the data pertaining to the dyadic interaction - ;.sodes between the teacher and each of the target students were carefully examined. Selections were made from all these data sources to provide overall descriptions of the episodes.

The Summary Analysis

Student overt and covert behavior data, the characterization of student conceptions of self-performance, and the dyadic interaction data provided a detailed descriptive base upon which to explore the expectancy effect phenomena. The information base was examined for the presence of any relationships among and between student

Table 5

÷

••

Intercoder and Intracoder Measures of Reliability in the Use of the Low Inference Classroom Observation System

	Coefficient o: Reliability
Intercoder Reliability	
Lesson 1	74.62
Lesson 2	83.37
Attracoder Reliability Student :	82+61
Student 2	3 8. 3 0
	90.18
Student 3	901-20

130

 \bigcirc

conceptions of self-performance, student causal perceptions of success and failure, student achievement-related behavior, teacher behavior towards the student, and teacher's causal perceptions of student performance. The finding of any suggested relationships was considered to contribute toward an explanation of the expectancy effect as it existed with the teacher and each of the four students in the learning of mathematics in a regular functioning Grade 6 classroom.

.

Chapter V

CASE STUDIES : TINA AND MIKE

Organization of the Case Studies

The case studies of the four Grade 6 students illustrate how typical students in a classroom perceive and behave in achievementrelated situations, in this case the learning of mathematics. Each case study consists of a unique and idiosyncratic description of a person and his, her behavior, suggesting a considerable diversity in now individual students view the complexity of the same learning setting.

An analysis of these descriptions, however, proceeds on an individual basis in order to develop as complete as possible a profile of each student's behavior and conception of self-performance. Each case study is described in a way that best describes the student's conceptions. An attempt is made to illustrate how the student's conception appears to function as a relatively stable cohesive system that controls and directs his thinking in the learning of mathematics.

Chapter V contains the case studies of two "successful" students--Tina and Mike. These are the students who were classified as high achievement, high motivation students. Chapter VI contains the case studies of two "less successful" students--Lisa and Greg--who were classified as low achievement, low motivation students.

The profiles are presented in the following way:

- 1. A short selective biography of the student is presented. The sketch focuses on school-related aspects and is based upon information gathered from interviews with the teacher and the student and from objective sources such as the two scales relating to student self-expectations of performance and school cumulative records.
- A quantitative analysis of the student's overt and covert behavior during the learning of mathematics is presented and discussed.
- 3. The causal perceptions and explanations of student behavior, as reported by the student, are examined. From the information available a characterization of the student's conception of self-performance is presented.
- 4. Student behavior while interacting with the teacher is examined. This description includes toth a quantitative and qualitative analysis of the interactions as observed at the overt level. Also, an analysis is made of the covert tenavior of toth the student and teacher during some of the interactions.

In the profiles frequent reference is made to higher or lower ability, higher or lower self-expectations, and higher or lower teacher's performance expectations for a studer. ⁹Unless qualified otherwise, "higher" refers to above average for lower class, while "lower" refers to below average for the class.

biographical Information

Tina was the second of three girls in her family. Her elder sister attended senior high school, while her younger sister was in Grade 3. The father was not living with the family in their apartment unit in the neighborhood of the school. Her first four years of life were spent in a rural community before the family moved to their Iresent home.

Tina was aged 11 years and reported tike-riding and reading to te her main interests. She enjoyed watching television, especially movieus. The family had travelled to british dolumnia and to the horth-west areas of the United States of America.

As part of her involvement in school life, Tina was a litrary melper, a traffic patrol warden, and also acted as a helping manu in a lower grade classroom during after-school nours. She participated in all the usual extracurricular activities such as sport, music, and so on.

In the classroom Tina judged mathematics, health, and social studies as her best subjects. Her preference for mathematics was based on a liking for the challenge at working with numbers. Tina had one close friend in the classroom but reported she liked most, if not all students.

The teacher described Tina as a girl of average to slightly above average performance in mathematics, but with the potential for much better achievement. Tina was perceived by the teacher to be a

TINA

good, happy-goolucky student who was easily satisfied, and who did not seem overly concerned with a refined work performance just so long as the work was done.

Teacher: You normally expect a girl's book to be quite neat, but not Tina. Like, she'll do all her math in a felt pen. She won't use a pencil because she thinks it neat to use a felt pen.

In response to the question of "why do you come to achoral", line stated that she needed the education. This would enable her to got a good job which would mean she would not have to work too meru! line also pointed out that she liked school.

Ine formal descriptive data on line supported the teacher's comment. A "A percentile for achievement in mathematics at the insist of even, as indicated on the school contrative records, represented a higher achieving student. The FAPS total expectation score of the was one of the highest scores recorded by the students in line's class and reflected a high degree of expectancy of success in performance across six subject areas. Her mathematics sub-score of appoint of an suggested that her expectations for this subject area were also relatively high. Examination of her responses indicated that Tina expected to always enjoy math, expected to learn all there was to learn, and expected to be better at math than most kids, but not one of the best in the class.

Tina's self-reporting of internal responsibility for success or failure on the IAR revealed that out of 34 situations she attributed 25 to internal causes and 5 to external causes. Because her internal responsibility for success score of 14 was close to being equivalent to her internal responsibility for failure score of 14, it can be concluded that Tina assumed credit for causing good things to happen in much the same way as she accepted blame for unpleasant consequences. Examination of her responses supported this conclusion, but with one notarie exception. With items pertaining to task difficulty, success on easy tasks was attributed to external causes, whereas failure on difficult tasks was attributed to internal causes.

quantitative Analysis of Fehavior

Tina's behavior during the 10 lessons on mathematics was divided into covert and evert aspects.

WHIT Behavior

Tima's overt tenavior was reached on the category system for analyzing student overt tenavior which was described in Chapter IV. The number of recorded observations of her overt behavior in each of the the categories in the three main lesson situations for each lesson was expressed as a percentage of the total number of recorded observations of Tima's overt behavior identified in that lesson. The results of this analysis are shown in Table 1.

Within-category of lesson situation percentage distributions of means of overt behaviors are presented in Table 7, while the condensing of Table 7 according to categories of student overt behavior is shown in Table 5.

The majority of Tina's time was spent in the lesson situations of teacher-led whole class instruction and in seatwork; these situations accounting for approximately 4c percent and 45 percent, respectively, of the learning time. In all lesson situations a

Φ	
he Le	
يا م	
F-1	

Percentage Distributions of Student Overt Behaviors in Each Lesson Situation Category for Each Lesson & Tina

Categories of Student Overt Behaviors Within					ressons	ons					Fercentage	Within- Category Sum of
Lesson JIUATIONS		~	3	t-		0	c	ω	0	10	TPAU	Percentage Means
Teacher-Led Whole Class Instruction												
Attending-listening Not attending-listening Interacting with	:6.7 20.8	3.3	22.8 12.3	39.3	25.8	6.9 7	22.0 30.0	58.5	11.6	87.5 9.4	36.7 4.9	
other students Dut of desk) 	2	6		1 p 1 1	2.4	† 4 1 1		<u>.</u>	47.0
Student Demonstration Whole Class Instruction				_								
Attending-listening Not attending-listening	25.0		~~~		ا ک -	- - -	0. 	6. -	1 1	1 1	6.C	
Interacting with other students Out of desk a		э • • •	1 I † 1	1 1		+ 1 + +	; ;		: :		6.2	2. 2
Seatwork Attending-listening Not attending-listening Interacting Utth	12.5	26.2 3.3	t-2-t	50.0	52.5	بخ ن ب	0.0 5.0 5.4	26. 8 	83.7	4 I I I	38.9 9.9	
other students Out of desk Supervised by teacher	t - 1	δ. + ! :			α 	€ 6			· • • • • • • • • • • • • • • • • • • •		6 a a'	5 .

143

.

Tatle ?

iercentage Distributions of Means for Student Overt Behaviors Within Each Category of Lesson Situations : Tina

.

Categories of Lesson Situation	Categories of Overt Behavior	WithIn-Category Fercentage Distributions of Means ^a
Teacher-led whole class instruction	Attending-listening Not attending-listening Interacting with other students Out of desk	
Student demonstration whole class instruction	Attending-listening Not attending-listening Interacting with other students Out of desk	0.1. 0.1. 0.1.
Seatwork	Attending-listening Not attending-listening Interacting with other students Out of desk Supervised by teacher	87 20 20 20 20 20 20 20 20 20 20 20 20 20

^aMeans for student overt behaviors derived from Table 6.

•

•

Table 8

2

Fercentage Distributions of Student Overt Behaviors for Total Lesson Time : Tima

Advertised to set to develop					Lessons	suos			•		Percentage
Overt Behavlors		~		±	5	9	~	ω	σ	0	Mean
Attending-listening	5+-2	86.9	51.9	89.3		92.3 75.5	5. S	3. Ś	95.3	.5.9	5.62
Not attending-listening	41.7	6.6	15.0	1	6 .	11.3	1tó.0	7.3	1	4.6	1+.0
Interacting with other students	!	ć.6	I I	2	 5	ę.:.	1	t V	;	3.:	3.2
Out of desk	1 1	4 †	12.3	3.6	3.5	}	ý•3	l	¢.4	I I	\$. \$
Supervised by teacher	4.2	! ?	1 1	1 1	I I	6.7	:	t ÷,	¦ .	1	Ċ.

. • ۰. . 5 ł . ; ; ;; 1 1 1 2 \$ • • • 1. K. -

,

\$.

significant amount of Tina's behavior was attending and/or listening, that is, on-task. As Table ? indicates, for teacher-led whole class instruction, Tina was observed to be on-task for a mean across the 10 lessons of approximately ?? percent of the time, and for student demonstration whole class instruction she was observed to be on-task for a mean of approximately 50 percent of the time. At seatwork, Tina was observed to be attending for a mean of approximately 50 percent of the time. Table 8 indicates that attending-listening behaviors were observed to account for a mean of nearly 80 percent of Tina's overt behavior across all mathematics lessons.

Most of the balance of Tina's overt behavior was coded as not attending-listening directly. However, Tatle t shows that not attending-listening behaviors were observed to occur for a mean of approximately 10 percent of the time with teacher-led whole class instruction, 47 percent of the time with student demonstration whole class instruction, and 2 percent of the time with seatwork. Overall, as shown in Tatle 8, Tina was observed to be not attending-listening for a mean of 14 percent of the time. Relativaly for observations of "Tina's over: behavior were recorded in the other bategories of interacting with other students and out of desk. During seatwork, Tina was observed to be out of her desk for a mean of 6 percent of the time.

In quantitative terms, Tina was observed to be on-task much of tractime during most lessons. This feature of her behavior was particularly prominent during seatwork. She was most likely to be not attending when other students were demonstrating an idea to the 140

While class. Tables t and t indicate that lessons 1 and 7 were days when Tina was coded at not ittending or listening more than 40 percent of the total lesson time. In lessons 3 and 6 she was observed to be not attending for it and 11 percent, respectively, of the total lesson time. The daily variations were deemed to be significant but can only be explained from an analysis of Tina's thought processes.

Covert Fravior

Time's reported covert tenavior during each lesson was fivided late interactive and non-interactive thoughts and feelings. Only the interactive teleants and feelings--the actual thoughts and feelings experienced during the lesson--were quantified. The content analysis system of student interactive thoughts, CASSIT, was used for this purpose. The number of Time's interactive thought units in each of the nine categories for each lesson was expressed as a percentage of the total number of thought units identified in that lesson. In Table 2 the results of this analysis are presented.

Most of the thoughts reported by Tina were usually behavioral moves--self, behavioral moves--student, self-performance--thoughts, and inveltance. Cognitive processes, non-task-related, and behavioral moves--teacher were categories of thoughts reported on an average of priority percent of all thoughts. Relatively few thoughts in the pategories of surject matter and self-performance--feelings were reported.

For Tina, the interactive thought is categories were further analy cording to sub-categories : what the thoughts were about. Five a se categories had the highest frequencies of

Table 9

iercentage Distributions of Student Thoughts Over CASSIT Gategories for Each Lesson : Tina

٠

Categories of Student						Lessons	IS				Phr-
Govert Behavior	•	64	3	t-	5	0		X	6	10	centage Mean
Subject Matter	1	88	i t	•	$\dot{}$	a∵ .⇒	с. ;	1	1	1	6•;
Cognitive Processes	· 9.	2:•6	0.v	÷	Э.: •:	ų.,	9.2	1	3.7	1	6·0
Behavioral MovesSelf	2:.9	ю.:	20.0		6.	2t	28.9	22.6	7.4	42	:9.8
Behavioral MovesStudent	15.6	27.0	0.51	:0·3	18.7	4	10.5	3.2	14.8	2: 2	13.6
Behavioral MovesTeacher	:5.6	5.4	ر. ۲	•••••	9.4	6.	:0.5	6.7	14.3	3.0	9.5
Self-PerformanceThoughts	4.6	6.8:	28.0	37.4	5:0	35.7	5.8	29.0	33.3	27.3	26.0
Self-PerformanceFeelings	1	5.2	0.4	ł	ı I	4.8		6.7	र ⊐ • र •1 र •		1 1
Feelings	:8.7	5.4	0·9;	3.4	4.6	4.8	2.9	1941 1	1.0	9.1	10.2
Non-Task-Related	ີ. • •	2.3	0. v	р. С	t /* *3	a. 7	•	3.2			8.5
				-							

148

•

occurrence, while self-performance--feelings was sub-categorized for purposes relevant to the direction of the study and for purposes of later comparison with other students. The results of this more intensive analysis are shown in Table 10.

behavioral moves--self. Behavioral moves--self, the thought units in which the student reports her thoughts about an action she was performing, had performed, or was considering performing in relation to the learning process, were further analyzed in terms of the kinds of motives and mental behaviors which prevailed while the student was learning. Table 10 reveals that, of her schavioral moves, Tina's reported thoughts were frequently focused on attendinglistening as well as the motives for attending or for avoiding attending. Tina's interactive thoughts of seeking or avoiding public participation accounted for 9 percent and nearly c percent, respectively, of all her behavioral moves--self thoughts.

Behavioral moves--student. Thought units in which the student reported her thoughts about an action involving other students were sub-categorized in an attempt to identify the nature and extent of student concern with peers during learning. Table 10 shows that Tina was aware primarily of what other students were doing. She was also ifrequently concerned about and aware of other students' successes and failures. To a lesser degree, Tina reported thoughts in which she was inferring or interpreting what other students were thinking.

Behavioral moves--teacher. Sub-categories were also used for those thought units in which the student reported her thoughts about

Table 10

Within-Category Percentage Distributions of Sub-categories of Selected Student Thought Categories of CASSIT : Tina

Sub-categories of Selected CASSIT Categories of Student Thoughts	Within-Category Percentage Distributions
Behavioral MovesSelf	
Seek public participation	0.01
Avoid public participation	.9.01
Attending-listening	7.58
Not attending-listening	36.55
Motive to attend	1.52 18.18
Motive to avoid attending	
Motive to use a technique	18.18
Motive to av id use of a technique	~ . 58
Behavioral MovesStudent	
Perceptions of other students' performances	31.82
Inferences of other students' thoughts	15.91
Interactionpositive	6.82
Interactionnegative	2.27
Berceptions of other students' behavior	43.18
Setavioral MovesTeacher	
fercentions of teacher's instructional moves	25.81
Perceptions of teacher's reactions	12.90
Interactions	
Perceptions of teacher's overt behavior	16+1 3 +5+16
Self-PerformanceThoughts	
Self-assessmentsuccess	16.67
Self-assessmentfail re	11.90
Perceptions of task difficulty	21.43
Perceptions of task structure	28.57
Self-attributions	
Self-expectations	2.38
Significant Other	11.90 7.14
Self-:erformanceFeelings	
Anxiety	7.14
Morally neutralpositive	35.71
Morally neutralnegative	14.29
Morally unneutralpositive	· · · · · · · · · · · · · · · · · · ·
Morally unneutral negative	42.86
eelings	2 8 4
Positive	68.75
Negative	31.25

an action involving the teacher. These were intended to enable observation of some of the kinds of thoughts about teachers which students experience during learning. Table 10 indicates that Tina was aware of what the teacher was doing and also reported thinking about the teacher's instructional moves. Hewer of Tina's reported thoughts pertained to interactions with the teacher and an awareness of the teacher's reactions to situations in the lessons.

<u>Sulf-performance-thoughts</u>. These interactive thought units represented the primary focus of the study. They refer to thought units in which the student is thinking about <u>his</u> or <u>her</u> performance tehavior and outcomes. Time's reported thoughts pertaining to her successes and failures accounted for nearly 19 percent of her selfperformance interactive thoughts, with the successes being considered more frequently than the failures. Perceptions of tasks, in terms of difficulty and of structure, accounted for 50 percent of her selfperformance thoughts. Thoughts involving self-expectations were expressed less frequently, yet they were consistently present across must lessons.

<u>Self-performance-feelings</u>. Thought units in which an emotion related to self-performance was expressed by a student were analyzed further according to anxiety, morally neutral, and morally unneutral sub-categories. Of the few self-performance feelings reported by Tina, Table 10 reveals that morally neutral positive feelings, such as happiness and pleasurable surprise, and morally unneutral negative feelings, such as feelings of shame, occurred most frequently.

151

· 🖍

<u>Feelings</u>. Thought units in which feelings were expressed in relation to the class, the lesson, and other general matters, were considered to reflect a student's overall view of the learning situation and indirectly contribute to her behavior and her thoughts acout self-performance while learning. Approximately two-thirds of Tina's expressed thoughts revealed positive emotions such as enjoyment, pleasurable surprise, and amusement, while one-third of her expressed thoughts in this category revealed negative emotions such as worry, shame, annoyance, anger, and toredon.

<u>Surmary of quantitative Analysis</u> of Behavior

The quantitative analysis of Tina's overt and covert behavior during the lessons provided some description of her behavior during learning. Most of the time she was observed to be on-task, especially during seatwork. Not attending-listening behaviors, when coded, mainly occurred during lesson situations which required her to watch or listen. The majority of Tina's reported interactive thoughts focused upon the self, namely, her own performance and her behavior in the learning process. New feelings related to self-performance, were reported. Most of her reported self-oriented thoughts referred to task-related matters of how she perceived the task and her attention to tasks. These findings support the strong trend for the on-task overt behavior noted earlier. Self-assessment and selfexpectation thoughts were frequently reported to occur as well.

Other frequently occurring interactive thoughts which Tina reported included an awareness of what other students and the teacher

Attributional Analysis of Self-Ferformance Covert Fehavior

A significant proportion of Tina's mental activity during the marning of mathematics pertains to components of self-performance thoughts and feelings. An analysis of the qualitative aspects of this obvert behavior revealed the relationships between the thoughts expressed and also pointed to the base of ideas, views, beliefs, emotions, and lines of reasoning which were perceived to underlie these relationships. The relationships tended to be causal in nature, hence an attributional analysis of these thoughts and feelings was deemed appropriate.

Causal Explanations of Tina's Behavior

The transcripts of the interviews with Tina contained several chains of comment which included causal explanations of behavior and causal perceptions of performance. Her report of an initial cognition, which sometimes included an emotional consequence, often referred to

an aspect of self-performance. Tina sometimes provided or readily disclosed reasons for these thoughts and usu described consequent behaviors which stemmed from such cognitions. In collating these chains of comments five clusters were discerned which referred to common cognitions, perceptions, and motives of behavior, namely, approach to learning, task difficulty, specific performances, public participation, and perceptions of other students' performance. Each cluster of Tina's chains of corment is described with examples and unecdotes included.

Approach to learning. Tima's pattern of benevier when challenged with new or difficult material seemed quite consistent. The pattern was exemplified ty:

. . . and then the first answer was right but I still den't get it. It was just a guess for the first one. So I'm going to talk to her about it tomorrow. (lesson .)

Tina's approach to learning can be described as mastery-oriented and persistent under these circumstances. When the teacher was introducing new work or the work was hard she always listened.

Because usually I listen when I'm...like something new 1 always listen, I know that. (Lesson 1)

But I listen when it comes to trings that are hard, but I just go on if it's something I know. (Lesson 7)

Where Tina did not understand a discussion point or worksheet problem then she was motivated to call upon the help of the teacher:

Let's say I don't understand something...then I'm not miraid to ask. I'll even put up my hand to tell the wrong answer so that I'll find out how to do it. (Lesson 3)

During the unit test, Tina became worried when she couldn't complete one problem. Aware that some answer was possible she persisted, even anier the increased stress of the situation, until the solution was found.

Tina's approach to learning challenging material seemed systematic and nightly motivated. Realizing that a new or difficult task had to be learned, she was able to increase the effort input in order to gain a sense of satisfaction in inderstanding the material ritisk.

Task divisedty. The made Prepart reference to the ease of the surflect matter to be bearned and to the simple worksheets and exercises to be done. Often these connents were absorpticled by the expression of find in or discust and by references to raving learned the material in previous years.

I feit...you know it seems like we were in a kindergarten shou or cometning or a lower grade class because it's all review from those years and my little sister just to k that last grade. And it's easy. It was easy for here It's all review from other years. (Lesson 7)

- T: It [the worksheet] was protty simple.
- I: Did you have time to finish it:
- T: Yes. I have handed it in already.
- I: Why do you think you found it pretty simple:
- T: Well, see, last year, that was something we were very good at because I never took unything like that before, and I really enjoyed that section of maths. And that's how I can remember it...it was so good. (Lesson 3)

When the lesson situation was tearner-led whole class instruction,

Time mode statements such as:

Sometimes things are so boring that I feel like doing something, so I eat. Sometimes I draw, but that draws away my attention from the board and then I might miss something. (Lesson 5)

Like, we took turns and slides and motion...motion geometry, we go took all that last year. And I didn't really have to listen that good. I'd wake up thad been dozing--aftermath of Hallowe'en] and then she would be saying something else and I didn't even know what she was talking about because I just remembered what I heard before. (Lesson 3)

On other occasions when Tina perceived the discussion to to contered on easy material she would spend time completing other written work or engage in intellectual leisure retreats such as crosswords. Tina nevertheless tended to keep taned in to the lesson sufficiently to follow the drift of the lesson, queing to what one learned in previous years for enecking progress of the discussion. On some occasions she reported different tenaviors: "Sometimes I look at the teacher...I hear what one says but I'm not really listening. Die thinks I'm concentrating" (Lesson t ...

Time's typical reaction to easy or familier worksheets was to start immediately and not wait to follow the teacher's step by step "lead-in" to the pask. Again she reported review as the main reason for finding the material simple, even stating on one obtaint: "... actually it was a worksheet I got last year so I knew all the "howers" (Lesson 4). Sometimes Tina initially perceived the task to the easy, yet found it more difficult once she "got into it": "I felt dumb because, I mean, I thought this was going to be easy and then get into it and it's not so easy after all" (Lesson z_1 . At other times the reverse applied when she found hard looking worksheets to be easier than expected.

In general, when Tina perceived task difficulty to be easy, the cognition seemed to be accompanied by feelings of boredom and a sense of disgust. She generally attributed the easy level of material to the external factor of review and, as a consequence, engaged in a variety of alternative behaviors. A number of underlying points of view and beliefs were associated with these chains of comment but these will be described later.

Specific performances. Tina made frequent reference to specific successes and failures and often accompanied these with an expression of an emotion. \blacksquare

- T: I wasn't sure whether it was right or not and I got it wrong, and I was mad.
- I: Why do you think you got it wrong?
- I: I didn't check it--I just did it with my eyes. I didn't bother checking it. (Lesson b)

In this example lack of effort was cited as the primary reason for failure, though she also added that the diagram on which she was working was poorly printed.

In the few occasions when Tina had difficulty in perceiving task structure she tended to become confused and often experienced temporary failure. Where this involved another student, feelings of embarrassment were experienced: "Tom was right and he kept on saying, 'You're wrong but I'm right.' And then I found that I was wrong. I felt stupid" (lesson 6). Instances of temporary failure did not seem to give rise to intense feelings. Causal explanations tended to revolve around insufficient effort on Tina's part which she readily counteracted given her mastery-priented approach to learning.

Intense feelings were reported to have been experienced when Tina received back her result on the unit test.

- I: How did you get on in the test?
- T: I did actually very good, I would say. I was pleased with. myself. I expected that.
- I: Why do you think you did so well?

T: Well, because I listened. You know I learned and I did my worksheets pretty good. I thought I did a good job and it should get me a good mark on my report card so my Mom can spoil me or brag about me. (Lesson 10)

This item sample provided some clue to the reason for the intensity of feeling and also indicated clearly Tina's view about the major success having resulted from her input of effort. Incidental successes during the learning of mathematics in this unit were attributed by Tina to ease of learning material, as described earlier.

<u>Public participation</u>. Time generally preferred not to volunteer a response during discussion or offer to demonstrate an idea in front of the whole class.

I just don't like to answer questions. I know what it [the discussion] is about. I just don't want to...I guess it gives someone else a chance who doesn't know who wants to find out. I am atraid I'd get laughed at if I got it wrong. When I know the answer, I just don't like to tell. It's just me. (Lesson 3)

This and other item samples suggested that the influence of peer effects played an important role in relation to participating publicly in front of the class. Tina tended to avoid placing herself in situations in which she would feel embarrassment in front of her peers. "Sometimes I say the stupidest answers and I think they are right at the time. But when I find out the right answer I think I'm so stupid you know" (Lesson 5).

On some occasions Tina chose to put her hand up and, as already described, volunteer a wrong response in order to find out the correct answer. On one occasion she reported that she felt like volunteering all the time on that particular day. In paraphrasing this one chain of comments, Tina felt glad but surprised that the

teacher had selected her to respond, for she had perceived that the teacher usually asked the same persons over and over again. Tina's explanation for this pattern of the teacher's behavior was:

Sometimes I think that she asks people because she can tell... like if they don't know something, she can tell, and then when they don't know she can explain it. And it seems that she just knows when you don't know. And then some, she knows that they know it and that they can show the others. (Lesson 5)

Perceptions of other students' behavior. While Tina was disinclined to demonstrate ideas in front of the class herself, she tended to diserve critically other students' efforts in this regard. A mixture of emotions seemed to accompany these perceptions.

In general, where other students' motives to demonstrate ideas in front : the class were perceived by Tina to be mainly non-task focused her thoughts and feelings tended to be negative. She experienced annoyance when she perceived students going to the front of the class in order to be the focus of everyone's attention. Motives of this kind conflicted with her own reasons for volunteering or not volunteering to publicly participate.

Where other students were perceived by Tina to be not attending sufficiently, negative feelings of a different order were experienced. On watching a less successful student trying to demonstrate an idea which was impossible, she obviously experienced feelings of derision toward the peer:

Oh Lisa, you're nuts, ...you're stupid!" 'Cause...it was so simple, I mean, my little sister would have known that. (Lesson 2)

In another case, Tina considered another student to be "kind of dumb" for working at the most difficult item and not noticing the most

obvious point. Again, other students' lack of attention to the task seemed to conflict with Tina's personal reasons for publicly.

Tina's Basic Ideas and Beliefs

Underlying the causal explanations and perceptions of Tina's self-performance related behavior in the learning of mathematics were a number of ideas, beliefs, views, emotions, and lines of reasoning. These case froughts seemed to be interrelated with and influence significantly Tina's behavior which would be apparent to the teacher and the other students in the class.

Heasons for success. One central belief of Tina's which pervaded all interviews was her direct association between student success and the amount of effort which students expend in the learning process.

- I: You know if you just use your brains you can do something you want. Some of them, they...you know, you can tell, like they are smart kids but they just don't use their brains.
- I: What about other kids you think are not smart?
- T: Well they are smart. Like, you know just not as good as some others. Because they don't try, they don't like the subject, they don't bother trying. And since they don't get things right they hate it even more, you know. (Lesson 5)

Tina believed she broke this perceived circle by trying harder.

A couple of years ago I used to hate maths, now I really like it. Just because, I guess, I'm trying harder and I get a lot right and hardly any wrong. (Lesson 5)

These notions surfaced readily in discussions about the unit test and other student performances. She felt disappointed that some class members had scored low: "I mean, that's real bad for Grade 6. He
[a low scoring student] should be paying attention" (Lesson 10).

• As one explanation for this performance level, Tina reported that these students have "lots of brains but they don't bother trying." Another explanation provided by Tina was:

- T: They still can't understand it but they are dumb. They won't go up and tell. They won't understand. I'm not afraid to go up there because she helps you.
- I: Why do you think they were afraid to go up?
- T: Well...they are afraid they will get embarrassed in front of the class. And that's their biggest problem because "Who cares what the other kids think, really." Let them think what they want to, that is what I say. (Lesson 10)

Time's rationalization was expressed at a time of Personal success, yet it was consistent with her approach to learning described earlier. However, the thought seemed to conflict with her own reluctance to publicly participate in class for fear of being laughed at by the other students if she was wrong.

The often referred to instances of keeping one ear tuned on What the teacher was saying during teacher-led whole class instruction while at the same time concentrating on some off-task activity. This was her mode of handling lesson content which she perceived to be review. "I don't believe I have to listen if I know it already" (Lesson 7). When questioned as to what she would do if the teacher suddenly asked her a question. This replied: "Well...I was prepared to take the risk. You know, I know symmetry and she...I would just ask what the question was again and she would tell me and then I would know" (Lesson 7). Tina made conscious attempts to give the impression of listening to the teacher by sometimes putting up her hand without knowing the teacher's question or by occasionally looking at the teacher yet without listening to what was being said.

ũ

A fina's belief that success in classwork was related to how hard a student worked or listened was reinforced by comments comparing her efforts to remain tuned in to the teacher while involved with off-task activities and other students who became totally distracted and who did not lister at all. Even with easy worksheets she was critical of students who did not work on ahead but who preferred to waste time listening to the teacher's preamble when they already knew what had to be done.

<u>Furcertions of her own mathematics ability</u>. Tina perceived , hersel: to be a good student at mathematics. "I think I'm pretty good at math because I try harder and I get my homework done all the time" (Lesson 5). She disclosed similar thoughts in the interviews which embraced the unit test situation. Before the test she expected to do well ty answering correctly most, if not all the questions. Tina felt that she had worked hard and that it would be just a matter of accuracy. On the test day she reported that she had not studied for the test because, "I didn" think I would be in a matter of beliefs was provided by the test results from which she concluded: "I did better than some of the girls that I thought were smarter at math than me" (Lesson 10).

Tina's higher performance self-expectations in mathematics seemed to be perceived clearly and these were attributed largely to the reasonable amount of effort she expended in the learning situations. But Tina also felt the teacher's performance expectations of her were correspondingly high.

<u>Tina's perception of the teacher's performance expectations</u> of herself. Tina's belief that the teacher held higher performance expectations for her seemed to be based or a number of perceptions and observations which had led her to conclude:

Well I don't think she thinks I'm bad at it [mathematics] but I \square not one of the best. She probably thinks I'm pretty good I think. (Lesson 6)

Tina's belief regarding the teacher's performance expectations of her were readily disclosed during interviews involving the unit test situation. The day before the test she expected the teacher was "going to think pretty well of her after the test." When the results of the test were announced, Tina's comments were:

Well she will think I'm pretty good...she won't be surprised though...She probably thinks that I am a little bit better maybe than I think. (Lesson 10)

Occasionally Tina referred to aspects of her behavior iron. which the researcher surmized that she was concerned with preserving or maintaining the teacher's performance expectations held towards her. The attending behaviors of providing the teacher with an impression that she was listening to the teacher-led whole class instruction while mainly being off-task confirmed this interpretation. On one occasion Tina lost her mathematics book and spent much of the teacher-led oral discussion searching in her desk. She was worried lest the teacher perceived her to be not listening. "She wouldn't really be mad but she would be kind of disappointed" (Lesson 1). Tina explained that the teacher would be surprised to find her not listening, whereas with other students who regularly did not listen she would not be surprised and so would get upset with them.

•

j.

Early in the year, with review type mathematics, Tina found herself not succeeding. This worried her because "maybe she [the teacher] must think I was crappy at first" (Lesson c). Tina was concerned with the early impressions which the teacher formed of her.

While Tina's concern with the performance expectations held by the teacher for her "surfaced" on a number of occasions, she did express a point of view which seemed to deny the notion of concern.

It matters what I think about myself, and what she thinks, I don't...like, about my work, if she thinks I'm not trying Well maybe I'm'not. But if she thinks I'm not trying and I am, well then I just don't care what she thinks. Because I know what I think. It doesn't matter that really...it matters what I think...I know when I'm not trying. (Lesson 6)

A later comment by Tina suggested that this reasoning stemmed from the model provided by the mother. Tina explained that if she did poorly in the unit test her mother would be her if she knew that Tina "had not tried." But if Tina "had tried" the mother would be disappointed but would.not hassle. Tina found the position which her mother adopted to be very helpful.

<u>Perceitions of other students' performances</u>. Tina perceived clearly the relative mathematics performances of other students in the class. She was able to place most students into a mathematical hierarchy in relation to herself. Their mathematical competencies as revealed by their work and by the way they answered questions in class contributed to the clear perception. Tina was very aware that the teacher spent a greater proportion of time with some students who had problems.

During the unit test Tina reported being unconcerned with the

performances of the other students. Those who finished the test early were perceived "to be real good at maths or that they weren't even trying." When the test results were announced she guessed which students performed poorly. Tina chose not to share her result with a student who scored low "because it would make her feel bad." She expressed surprise at which boy had scored highest, but noted that boys do better in mathematics because girls like more the language arts.

The teacher had preceded the handing tack of test results with an explanation of the distribution of marks, the average mark for the class, and why students who scored very well would be recognized. Tink approved these procedures because they helped in identifying her relative standing in the class. They also publicly rewarded those students who had been listening and attending during the learning because of the mathematics unit.

<u>Ferceptions of tasks and activities</u>. This believed the listening and the worksheet facets of learning mathematics to be mutually significant. She pointed out that the listening parts of lessons, such as teacher-led whole class instruction, were important because they enabled the worksheets to be completed. The worksheets were viewed as the parts of the lessons where she worked hardest and were the parts she remembered most about lessons.

In terms of difficulty of tasks, Tina preferred hard worksheets.

- I: Why would you prefer to have hard work?
- T: I like straining my brain. I don't know why. 'Cause then I feel good. (Lesson 6)

165

She seemed unconcerned that hard worksheets could increase her chances of failure.

I will just have to see if I don't understand it and just check it over. It doesn't worry me because that's how you learn...ty your mistakes. (Lesson 6)

These comments reflected Tina's mastery-oriented approach to learning.

Finally, on numerous occasions Tina referred to "getting a good mark on her report card." As an underlying motive for much of her behavior, this thought appeared to be salient in that it pervaded her thoughts about attending, her views about tasks, and her teasons for achieving well on the unit test. The teacher protaily a second Tina as the significant person in the classroom to realize the mark on the report card. Tina's motive for generated at the kind of performance expectations the teacher med and maintained toward her can be interpreted readily in the evidence described in this attributional analysis of behavior.

A Characterization of Tina's Conception of Self-Performance

Tina became the focus of this study because was perceived to be a successful student. The results of the attributional analysis of her behavior, together with some of the relevant findings of the quantitative data, have enabled an interpretation to be made of how Tina might have perceived herbelf and her self-performance in the learning of mathematics. The major elements of this characterization of her conception of self-performance were:

1. Tina's approach to learning can be described as masteryoriented, that is each engaged in learning behaviors which resulted in immediate or eventual success. She believed that success was the function of effort expenditure, hence perceived a clear effort-outcome covariation.

- 2. Tina was undaunted by what she perceived to be hard worksheets. She preferred difficult material because this enabled her to work hard, that is, to apply considerable effort which would result in eventual success.
- 3. Tina found success which had, been mined by considerable effort expenditure to be satisfying and pleasurable. Feelings under these circumstances were more intensive. Success attributed to ease of material to be learned gave rise to negative feeling states such as boredom and storn.
- The found failures which were short-lived analying and sometimes embarrassing. However, these feelings were of shallow intensity and temporary. Because Tima did not experience major failures, no evidence was available as to the kind of feelings which would accrue. However, there were indications of deep-seated feelings of fear if she ever performed badly.
- 5. Tina was reluctant to participate publicly. She avoided placing herself in situations which might prove embarrassing.
- Tina seemed unhappy when she perceived other students' behaviors as reflecting negative or non-academic motives which conflicted with her own.

8.

}

7. Tina perceived clearly the classroom achievement hierarchy in relation to herself. She approved of the procedures followed

5 2

ly the teacher when giving back test msults.

3

- 8. Tina believed the teacher held higher performance expectations for her, and expended effort to preserve these favorable expectations.
- Tina valued as the ultimate goal of classroom behavior the 9. gaining of a good mark on her report card. She clearly perceived the expenditure of effort to be the means of achieving this goal. The teacher and parent apparently had transmitted successfully to Tina the societal value of effort. But the faining of a good mark on her report was viewed by Tina not only as self-rewarding but as rewarding the mother, Tina's most significant love figure. Tina perceived the teacher to be the person who would provide the reward, hence the mediating role of teacher expectations. By caring about the kind of performance expectations the teacher formed and maintained about her, and maintaining a high profile on effort expenditure, Tina appeared to fulfill her desires of gaining social approval from the two most similant others--the teacher and the mother-as well as retaining a higher degree .of self-esteem.

Analysis of the Dyadic Interactions Between Tina and the Teacher

Both the covert and overt behavior of Tina and the teacher were observed and analyzed for each dyadic interaction episode. The low inference classroom observation system was used to code the observable behaviors and stimulated recall methodology was used to

gather the participants' reported covert behavior.

Analysis of the Overt Behaviors

The results of the analysis of the interaction sequences between Tina and the teacher are reported in Table 11 and in. Figures 7 and α_{\pm} Table 11 shows that Tina and the teacher interacted dyadically 12 times during the 10 mathematics lessons. The majority of these dyadic interactions were student initiated and private in nature and occupied a total duration time of 59 seconds.

Figure 7 indicates that the four-public dyadic interactions were teacher initiated, This volunteered on each occasion to respond, that she always answered the questions correctly, and that the teacher's feedback terms to be brief and neutral. Figure & shows that the private dyadic interactions were mainly student initiated, to which the teacher responded in a more extended manner than in the public interactions. However, a substantive analysis of the interactions revealed that only two of these private dyadic interactions consisted of sustained content information exchanges. Overall, the quality of interactions between Tina and the teacher were characterized by trief, positive, and neutral communications.

The Overall Analysis of the Dyadic Interactions

Some of the dyadic interactions between Tina and the teacher were analyzed by an overall examination of the observable data, the teacher's covert behavior and the student's covert behavior. The purpose of this analysis was to attempt to describe in detail the process behaviors during dyadic interactions which might have

Table 11

Frequency Distribution of Syadic Interactions Between Tina and Teacher

Type of					Lee	Lessons					
Interaction		2	\sim	4	5	\$	6	ω	6	;;	Total
Teacher initiated public interaction	8	,	I	I	•	ı	• •	I	1	ı	÷
Student initated public interaction	i	I	1	ı	I	+	J	ı	,	1	ı
Teacher initiated private interaction	I	ı	I	ŀ	I	. 4	ı	I	ı	I	÷ ·
Student initiated private interaction ^a	• •	ı	ŀ	. '	I		T	• •	\sim	ı	•
Total:) m	ł	1			.J.			~ ~		

^aDuration of the student initiated private interactions was 77, 4, **2**, 77, 5, 3, and 10 seconds, respectively, for a total time of 89 seconds.

.

 $(\cdot$

Process 0.25 - Volunteer :.CC - Correct :.00 - Teacher 1.00 - Product 0.50 - Volunteer :.00 - Correct :.00 - Public 0.33 Public 0.33	Answer Reedback
Teacher 1.00 Product 0.50 Volunteer :.00 Choice 0.25 Volunteer :.00	
Teacher 1.00 Product 0.50 Volunteer 1.00 Correct 1.00 Correct 1.00	- Correct 1.00 Affirm 1.00
Student Choice 0.25 Volunteer :.00 -	:.00 Affirm :.00
Student	:.00 - Frocess :.00
7	

-





¢

۱

ì



contributed to the "communication" of expectations. With Tina, the surprisingly small number of interactions meant that only a sample of behaviors could be described. Given this limitation, they are reported as items of relevant data derived from four dyadic interactions between Tina and the teacher.

During lesson i Tina initiated an interaction with a content comment. The teacher sustained the interaction by giving a clue, to which Tina said "Oh!" in such a way as to imply that she understood. Obviously the teacher, in reading her mind, assumed Tina ' had worked out the correct answer for she affirmed the as yet unstated correct response by verbalizing what she presumed was in Tina's mind. In this way the teacher provided Tina with the answer, while Tina, who had not known the answer, as learned from the later interview, had provided the teacher with the impression that she had caught on to the slightest of clues. Consequently, Tina had preserved the teacher's performance expectations held towards her.

Lesson & contained a similar situation. Tina initiated an interaction with a content question. The teacher provided explanations and worked through an example with Tina, at the conclusion of which she asked:

Teacher: And do we have ____? Tina: Nods Teacher: Hmm Hmm!!

1.

0

The affirmation by the teacher reflected the teacher's assumption that Tina understood for she left Tina at that point. Tina later stated she was quite satisfied with the outcome of the interaction. This example, together with that in lesson i, raised the possibility that by presuming Tina understood, and not checking whether this was the situation, the teacher conveyed the impression of holding higher performance expectations to Tina. Tina meanwhile used noncommittal or non-verbal messages to convey the impression that she understood to the teacher, thereby preserving the teacher's impression of her.

At an earlier point, the teacher had reported that Tina had asked for help.

I thought at the time then it's rather unusual because she doesn't usually ask for help. She is usually an independent sort who goes right ahead and does...but she asked, so it could be that she wasn't paying attention [referring to Tina with her head on the desk]. (Lesson 4)

This comment suggested that the teacher seemed happier at finding an excuse for Tina's need for help in the form of lack of attention or effort. An attribution to lack of ability presumably would have been inconsistent with the teacher's performance expectations for Tina.

In lesson 4 Tina had asked the teacher if what she had done was correct.

- T: I asked her if that was okay.
- I: And how did she respond?
- T: She went "Aaargh!"
- I: Why do you think she said it that way?
- T: Because I don't think she hardly even looked. She knew I'd get it right I guess. (Lesson 4)

Tina's interpretation of what was coded on the low inference observation system as an affirmation type terminal teacher feedback, and recorded as a brief "Hmm Hmm," indicated an awareness and acceptance of teacher performance expectations.

An episode during the unit test in lesson 9 yielded

Interesting interaction information. Tina initiated two interactions, close wogether in time, which pertained to whether or not a test problem could be answered. Brief, noncommittal replies by the teacher left Tina searching for the problem and, at the same time, thinking: "I knew she was looking at me and I thought, 'What is she thinking?'." Meanwhile, the teacher was watching Tina intently:

Teacher: I was hoping that I would see like the light bulb above her desk-kind of go "ping"...a look of recognition or discovery that she had got it.

These comments can be interpreted as the teacher apparently waiting for confirmation of her expectations, while Tina might have been concerned that the maintenance of the teacher's performance expectations of her could be at stake to some extent.

The items of dyadic interactions described above seemed to constitute a very limited range of the interaction process behaviors between Tina and the teacher. Yet the sample of 4 interactions from a total of 11 dyadic interactions across the 10 mathematics lessons appeared to contain elements of performance expectation behavior on the part of both Tina and the teacher. This might indicate a relatively high incidence of performance expectation behavior across a wider sample of lessons and subject areas.

A plausible interpretation of the evidence presented seemed to confirm that the teacher held higher performance expectations for Tina. Her behavior toward Tina included a number of subtle yet non-deliberate process behaviors which appeared to convey to the student that the teacher expected her to succeed. The teacher's verbal comments also contained strongly suggestive evidence for the existence of these expectations. Tina's verbal reports indicated clearly that she was aware of the teacher's performance expectations

f

of her. She seemed to adopt subtle interaction behaviors which appeared to be motivated by the desire to preserve the higher performance expectations held by the teacher of her.

Summary Analysis

The results of describing Tina's behavior during the learning of mathematics have established some tentative relationships among and between Tina's conceptions of self-performance, Tina's causal perceptions of success and failure, Tina's achievement-related behavior, the teacher's behavior towards Tina, and the teacher's causal perception of Tina's performance.

- 1. Tina valued the gaining of a good mark on the report form as the goal of classroom behavior.
- 2. Success in achieving this goal was perceived by Tina to be dependent upon the expenditure of effort.
- 3. The teacher was perceived by Tina to be the person who granted the reward of a good mark on the report firm.
- 4. The teacher was perceived by Tina to approve of effort expenditure as the main criterion for success.
- 5. The teacher was perceived by Tina to hold higher performance . expectations for her.
- The perceived these teacher performance expectations for her were based on her effort-oriented behavior.
- 7. Tina was viewed as a student who was mainly on-task.
- 8. The teacher held higher performance expectations for Tina based on attribution to ability and effort.

- 10. Tina appeared to consciously engage in behaviors which were designed to preserve or maintain the performance expectations held by the teacher toward her, that is, to be perceived as a successful student who expended considerable effort in herwork.
- 11. Tina was a successful student because she was mastery-oriented in her approach to learning.

MIKE

Biographical Information

Mike was the only child of a family living in a private home in the neighborhood of the school. He was born in a mining town in Ontario and had lived in Edmonton for the past five years.

Mike was aged 11 years and reported bottle collecting and reading to be his main interests. He watched some television, especially "Disneyland" and "House on the Frairies." The family had travelled extensively in Canada and during 1979 were expecting to move for a one or two year stay in South America.

As part of his involvement in school life, Mike was a library helper, a team soccer referee, and a patrol warden. In the classroom he especially enjoyed social studies, but reported liking most subjects. In mathematics he felt he was doing well but stated that "it gets boring if we repeat work done before." Mike had two close friends whom he noted were similar to him. They apparently shared experiences and activities during recess and after school. In response to the question "Why do you come to school?", Mike replied briefly, "I like to and I have to."

The teacher described Mike as a student of above average ability. While his work was perceived to be messy, she noted that he worked very hard. The school cumulative records indicated that Mike had gained an achievement percentile of 89 in mathematics, which suggested he was a high achieving student.

The PAPS total expectation score of 145 reflected a high degree of overall expectancy of success. His mathematics subscore of . 25 out of 28 suggested that his expectations in mathematics were consistent with the overall pattern. Examination of Mike's responses to the mathematics subsection of the FAPS indicated that he expected to always enjoy math, he expected to learn all there was to learn, and he expected to be better at math than most kids but not one of the best in the class. Mike did expect to do very well in math when he got older.

Mike's self-reporting of internal responsibility for success or failure on the IAR revealed that out of 34 situations he attributed 24 to internal causes and 10 to external causes. The subscores for internal responsibility for success of 15 and for internal responsibility for failure of 9 suggested that Mike assumed credit for causing good things to happen in a different way to which he accepted blame for unpleasant consequences. Examination of his responses on the IAR indicated some possible areas of differential

. **X**.

responses. Where effort or attention by Mike was pitted against teacher's explanations, the need for help, or other students' distraction, Mike claimed responsibility at times of success, but placed the blame on external factors at times of failure. This pattern of response on the IAR was almost non-existent when effort or attention by Mike was pitted against task difficulty and significant others' thoughts about Mike. His internal responsibility for success was close to being equivalent in these circumstances.

Quantitative Analysis of Behavior

Mike's behavior during the nine lessons which he attended was divided into covert and overt aspects for analysis purposes.

Overt Behavior

The results of the analysis of Mike's overt behavior are shown in Tables 12, 13, and 14. Table 12 indicates that the majority of Mike's time was spent in the lesson situations of teacher-led whole class instruction and in seatwork; these situations accounting for approximately 45 percent and 47 percent, respectively, of the learning time. A considerable amount of Mike's behavior in all lesson situations was attending-listening on-task behavior. As Table 13 indicates, Mike was observed to be on-task for a mean across the nine lessons of approximately 86 percent for teacher-led whole class instruction, 75 percent for student demonstration whole class instruction, and 93 percent for seatwork lesson situations. Attending-listening behaviors, as shown in Table 14, accounted for about 88 percent of all Mike's learning time in the mathematics Table 12

•

•

Percentage Distributions of Student Overt Behaviors in Each Lesson Situation Category for Each Lesson : Mike

Categories of Student Overt Behaviors Within				•	Lessona	BUROS	, ,				Percentage Mean	Within- Category Sum of
Lesson Situations	1	2	. 3	4	5	9	2	ω	6	10		Percentage Neans
<u>Teacher-Led Whole</u> Class Instruction												
Attending-listening Not attending-listening	33.3	55.7 8.2	28.6	t. 6.4	32.7	26.4	30.0 20.0		10.6	81.2	38.3 6.0	
Interacting with other students Out of desk	4.2		: ;		: ;		11		•		0.5	8. ¶;
Student Demonstration	•	•						A			•	
Attending-listening Not attending-listening	41.7	1.6	1.8 1.8		5.8	: :	4.0 12.0	സമ	; ;	: :	6.1 2.0	
interacting with other students Out of desk) (1 1	; ;	; ;	<u> </u>	EI Z	1 1		: :	Ω Ω
<u>Seatwork</u> Attending-listening	16.7	26.2	57.1	39.3	59.6	7.17	€	₽	4.68	ł	43.8	
Not attending-listening Interacting with other		3.3				 	4			;	4.0	
students Out of desk Supervised by teacher		t.9	5.4	14.3	1 1 1	: : :	: : :			1 1 1	0.2 0.2 0 2.5 0 2.5	47.1

180

•

•

~*

.

Table :3

1

.

Percentage Distributions of Means for Student Overt Behaviors Within Each Category of Lesson Situations : Mike

•

Categories of Lesson Situation	, Categories of Overt Behavior	WithIn-Category Percentage Distributions of Means ^a
Teacher-led whole class instruction	Attending-listening Not attending-listening Interacting with other students Out of desk	85.5 13.4
Student demonstration whole class instruction	Attending-listening Not attending-listening Interacting with other students Out of desk	· 75.3 24.7
Seatwork	Attending-listening Not attending-listening Interacting with other students Out of desk Supervised by teacher	69 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.

•

٠

^aMeans for student overt behaviors derived from Table :2.

Table :4

3

,

.

:**,

.

.

,

.

Percentage Distributions of Student Overt Behaviors for Total Lesson Time : Mike

	ý		01 701	AVTU : ANTI MOSSAN TOTAL TOT			1)				
Categories of Student					Les	Lessons					Percentage
Overt Behaviors		2	9	+	5	9	2	ω	6	10	Nean
Attending-listening	91.6	/ 91.6 83.6 87.5 85.7 98.1 98. 1 68.0	87.5	85.7	98.1	98.1	68.0		100.0	81.2	88.2
Not attending-listening	4.2	11.5	2.1	;	1	6.:	X .0	κ eq	ł	18.7	4. 8
Interacting with other students	4.2	6.4	1	1	;	;		ഗക	ł	;	0.1
Out of desk	1	!	5.4	14.3	4	:		X	1	i I	2.2
Supervised by teacher	:	1	;	;	1.9			Ч	1	1	0.2
							7				

٠

,



lessons.

Not-attending or listening behaviors accounted for most of the balance of learning time. While in seatwork, not attending-listening behaviors were virtually not observed; they were more prominent in the whole class lesson situations--13 percent for teacher-led and 25 percent for student demonstration. Overall, Mike was observed to be not attending-listening for about 8 percent of his learning time. Relatively few observations of Mike's overt behavior were recorded in the other categories of interacting with other students and ovt of desk.

A prominent feature of the quantitative analysis of Mike's overt behavior during most lessons was the high proportion of time spent attending or listening. Tables 12 and 14 indicate that lessons 2, 7, and 10 were days when Mike was coded as not attending or listening 11 percent, 32 percent, and 19 percent of the time, respectively. He was most likely to not attend or listen during teacher-led whole class instruction lesson situations.

Covert Behavior

In Table 15 the results of analyzing the percentage distribution of Mike's interactive thoughts across each of the nine lessons are presented. Most of the thoughts reported by Mike were those classified as cognitive processes, behavioral moves--self, behavioral moves--student, and self-performance--thoughts. Behavioral moves--teacher, non-task-related, feelings, and subject matter were categories of thoughts reported on an average of 5 to 10 percent of all thoughts. Relatively few thoughts in the category

Table 15

•

٩

•

Percentage Distributions of Student Thoughts Over CASSIT Categories for Each Lesson : Mike

Categories of Student						Lessons	IS			1	Per-
Covert Behavior		5	~	±	5	۰	~	ω	6	10	Rean
Subject Matter	9.1	2.7	13.6	•	6	<u>.</u>	4.5		;	;	5.9
Cognitive Processes	9.1	11.5	4.5	16.6	· · · ·	3.1	.8.2		9.5	23.5	6.11
Behavioral MovesSelf	9.1	11.5	4.5	27.8	22.7	15.6	9.:-	۲	23.8	5.9	4.41
Behavioral MovesStudent	13.6	5:15	13.6	5.6	ç•E:	:5.6	:3.6	μ μ	:4.3	5.9	6.11
Behavioral MovesTeacher	9.1	3.8	4.5	5.5	.t	21.9	5.6	0 60	6.4	5.9	8.7
Self-PerformanceThoughts	22.7	38.5	7.9	33.3	3: .8	6.52	27.3	Ż	23.8	41.2	3: -:
Self-PerformanceFeelings	1	1	1	I I	l (1	1	F	5.4	17.6	2.5
Feelings	13.6	ı I	4 .5	5.5	;	0.2	18.2		9.5	1	4.9
Non-Task-Related	13.6	154	2.8.	5.6	9.:-		;		:	;	7.2

•

.

184

•

r

of self-performance--feelings were reported.

For Mike nearly 18 percent of all thoughts were reported for the combined categories of subject matter and cognitive processes, which are the categories considered to be academically pertinent to the learning of mathematics. However, for purposes of this research, six other categories of thoughts were further analyzed. These categories were behavioral moves--self, behavioral moves--student, behavioral moves--teacher, self-performance--thoughts, selfperformance--feelings, and feelings. The results of this more intensive analysis are presented in Table 16.

Benavioral moves--self. Table 15 shows that, of his tehavioral moves, Mike's reported thoughts were frequently focused on attending-listening actions, including the motives for attendinglistening. At no time did Mike express thoughts that he was not listening or attending which could be categorized as behavioral moves--self. A few interactive thoughts of seeking to participate publicly were reported by Mike.

Behavioral moves--student. The category of behavioral moves--student interactive thoughts accounted for nearly 12 percent of all Mike's reported thoughts. As Table 16 indicates, 56 percent of these thoughts were perceptions of other students' behavior and the balance were perceptions of other students' performances and inferences of other students' thoughts. These percentages indicated that Mike was aware of what other students were doing in the class and was concerned about their successes and failures and about what they

Table 16

.

•

Within-Category Percentage Distributions of Sub-categories of Selected Student Thought Categories of CASSIT : Mike

Behavioral Moves-Self ?. Seek public participation ?. Attending-listening ?. Motive to attend ?. Behavioral Moves-Student ?. Terceptions of other students' performances Inferences of other students' thoughts Ferceptions of other students' behavior %. Behavioral MovesTeacher Terceptions of teacher's instructional moves 42. Interactions Ferceptions of teacher's behavior %. Self-PerformanceThoughts Self-assessmentsuccess Self-assessmentfailure 8. Ferceptions of task difficulty 2?. Perceptions of task structure Self-expectations Significant others 4. Self-PerformanceFeelings Anxiety Morally neutralpositive Morally neutralnegative Feelings Positive	tegory
Seek public participation 7. Attending-listening 7. Motive to attend 7. Behavioral Moves-Student 7. Terceptions of other students' performances 7. Inferences of other students' behavior 7. Behavioral MovesTeacher 7. ierceptions of teacher's instructional moves 42. Interactions 5. Ferceptions of teacher's behavior 52. Self-assessmentsuccess 16. Self-assessmentfailure 8. Ferceptions of task structure 25. Self-expectations 14. Significant others 4. Self-PerformanceFeelings 4. Anxiety 25. Morally neutralne	
Attending-listening71-Motive to attendBehavioral Moves-StudentTerceptions of other students' performancesInferences of other students' behaviorBehavioral Moves-TeacherTerceptions of teacher's instructional moves42.InteractionsFerceptions of teacher's behaviorSelf-Ferformance-ThoughtsSelf-assessmentsuccessSelf-assessmentfailure8.Ferceptions of task difficultyPerceptions of task structureSelf-attributionsSelf-expectationsSignificant othersMorally neutralpositiveMorally neutralnegativeFeelings PositivePositiveSelf-ings PositiveSelfings PositiveSelfings PositiveSelfings PositiveSelfings PositiveSelfings PositiveSelfings PositiveSelfings PositiveSelfings PositiveSelfings PositiveSelfings PositiveSelfings PositiveSelfings PositiveSelfings PositiveSelfings PositiveSelfings PositiveSelfings Positive	:4
Motive to attend.1Behavioral MovesStudent Terceptions of other students' performances Inferences of other students' thoughts Terceptions of other students' behavior.4Behavioral MovesTeacher Terceptions of teacher's instructional moves Terceptions of teacher's behavior42.Behavioral MovesTeacher Terceptions of teacher's instructional moves Terceptions of teacher's behavior42.Self-PerformanceThoughts Self-assessmentsuccess Self-assessmentfailure Terceptions of task difficulty Self-attributions Self-expectations Self-expectations Terceptions16.Self-PerformanceFeelings Anxiety Morally neutralpositive Positive25.Feelings Positive92.	
Behavioral MovesStudent	
<pre>ierceptions of other students' performances</pre>	•)
<pre>ierceptions of other students' performances</pre>	
Inferences of other students' thoughts Ferceptions of other students' behavior 50. Behavioral MovesTeacher ierceptions of teacher's instructional moves 42. Interactions 5. Ferceptions of teacher's behavior 52. Self-PerformanceThoughts Self-assessmentsuccess 16. Self-assessmentfailure 8. Ferceptions of task difficulty 27. Perceptions of task structure 25. Self-expectations 14. Significant others 4. Self-PerformanceFeelings Anxiety 25. Morally neutralpositive 25. Morally neutralnegative 50. Feelings Positive 92.	30
Jerceptions of other students' behavior5t.Behavioral MovesTeacher ierceptions of teacher's instructional moves42.Interactions5.Ferceptions of teacher's behavior52.Self-PerformanceThoughts16.Self-assessmentsuccess16.Self-assessmentfailure8.Ferceptions of task difficulty27.Perceptions of task structure25.Self-expectations14.Significant others4.Self-PerformanceFeelings4.Anxiety25.Morally neutralpositive50.Feelings Positive92.	
Behavioral MovesTeacher ierceptions of teacher's instructional moves 1nteractions 5 Ferceptions of teacher's behavior Self-FerformanceThoughts Self-assessmentsuccess Self-assessmentfailure Herceptions of task difficulty Perceptions of task structure Self-attributions Self-expectations Self-PerformanceFeelings Anxiety Morally neutralpositive Morally neutralnegative Selfings Positive 92.	JU
ierceptions of teacher's instructional moves42.Interactions5.Perceptions of teacher's behavior52.Self-PerformanceThoughts52.Self-assessmentsuccess16.Self-assessmentfailure8.Ferceptions of task difficulty27.Perceptions of task structure25.Self-attributions3.Self-expectations14.Significant others4.Self-PerformanceFeelings4.Morally neutralpositive25.Morally neutralnegative50.Feelings Positive92.	
Interactions5.Perceptions of teacher's behavior52.Self-PerformanceThoughts52.Self-assessmentsuccess16.Self-assessmentfailure8.Perceptions of task difficulty27.Perceptions of task structure25.Self-attributions3.Self-expectations14.Significant others4.Self-PerformanceFeelings4.Morally neutralpositive50.Feelings50.Positive92.	
Perceptions of teacher's behavior52.Self-PerformanceThoughts Self-assessmentsuccess16.Self-assessmentsuccess16.Self-assessmentfailure8.Ferceptions of task difficulty27.Perceptions of task structure25.Self-attributions3.Self-expectations14.Significant others4.Self-PerformanceFeelings Morally neutralpositive Positive25.Feelings Positive92.	
Self-PerformanceThoughts16.Self-assessmentfailure8.Ferceptions of task difficulty27.Perceptions of task structure25.Self-attributions3.Self-expectations14.Significant others4.Self-PerformanceFeelings25.Morally neutralpositive25.Morally neutralnegative50.Feelings92.	
Self-assessmentsuccess:6.Self-assessmentfailure8.Ferceptions of task difficulty27.Perceptions of task structure25.Self-attributions3.Self-expectations:4.Significant others4.Self-PerformanceFeelings25.Morally neutralpositive25.Morally neutralnegative50.Feelings92.	3
Self-assessmentsuccess:6.Self-assessmentfailure8.Herceptions of task difficulty27.Perceptions of task structure25.Self-attributions3.Self-expectations:4.Significant others4.Self-PerformanceFeelings25.Morally neutralpositive25.Morally neutralnegative50.Feelings92.	
Self-assessmentfailure8.Herceptions of task difficulty27.Perceptions of task structure25.Self-attributions3.Self-expectations14.Significant others4.Self-PerformanceFeelings25.Morally neutralpositive25.Morally neutralnegative50.Feelings92.	: ٦
Ferceptions of task difficulty27.Perceptions of task structure25.Self-attributions3.Self-expectations14.Significant others4.Self-PerformanceFeelings25.Morally neutralpositive25.Morally neutralnegative50.Feelings92.	
Perceptions of task structure25.Self-attributions3.Self-expectations14.Significant others4.Self-PerformanceFeelings25.Morally neutralpositive25.Morally neutralnegative50.Feelings92.	
Self-attributions3.Self-expectations14.Significant others4.Self-PerformanceFeelings25.Morally neutralpositive25.Morally neutralnegative50.Feelings92.	
Self-expectations14.Significant others4.Self-PerformanceFeelings25.Anxiety25.Morally neutralpositive25.Morally neutralnegative50.Feelings92.	
Significant others4.Self-PerformanceFeelings25.Anxiety25.Morally neutralpositive25.Morally neutralnegative50.Feelings92.	
Self-PerformanceFeelings25.Anxiety25.Morally neutralpositive25.Morally neutralnegative50.Feelings Positive92.	
Anxiety25.Morally neutralpositive25.Morally neutralnegative50.Feelings92.	
Anxiety25.Morally neutralpositive25.Morally neutralnegative50.Feelings92.	
Morally neutralnegative50.Feelings Positive92.	
Feelings Positive 92.	00
Positive 92.	00
Positive 92.	
	31
Negative ?.	69

186

.

were thinking.

Behavioral moves--teacher. Of all Mike's reported interactive thoughts nearly 9 percent involved behavioral moves--teacher. Table 16 shows that Mike was aware of what the teacher was doing and also reported thinking. about the teacher's instructional moves.

<u>Self-performinge--thoughts</u>. About 31 percent of all Mike's interactive thoughts were reported to be self-performance--thoughts. Of these, 16 percent were focused on successes and another 8 percent were focused on failures. Perceptions of tasks, in terms of difficulty and of structure, accounted for 53 percent of Mike's self-performance thoughts. Thoughts involving self-expectations accounted for another 14 percent of Mike's reported interactive covert behavior.

Self-performance--feelings. Very tew expressions of self-performance--feelings were made by Mike. It those reported, 75 percent occurred in the lesson during which the unit test results were announced. Because Mike was disappointed at his result. Table 16 presents a spuriously high percentage of morally neutral--negative feelings.

Feelings. The majority of Mike's feelings about the class in general were positive in nature. Emotions such as pleasure, amusement, pleasurable surprise, positive hopes, and being unworried characterized his covert behavior. The small percentage of "megative feelings was directed solely to expressions of boredom.

Summary of Quantitative Analysis of Behavior

• The quantitative analysis of Mike's behavior provided some descriptive trends in how he functioned during the learning of mathematics. Most of the time Mike was observed to be on-task in all lesson situations. This was concluded from an analysis of both Mike's overt and covert behavior. Not attending-listening behaviors mainly occurred during lesson situations which required him to watch or listen.

The majority of Mike's interactive thoughts : bused upon the self, both in terms of his own performance and his behavior in the learning process. Most of his self-oriented thoughts referred to task-related matters of how he perceived the task and his attention to task. This partern of covert behavior, together with the reasonably frequent occurrences of learning thoughts such as in the categories of cognitive processes and subject matter, supports the strong trend for on-task overt behavior. Self-assessment and self-expectation thoughts comprised a significant set of self-oriented thoughts which reflected a more personal aspect of behavior.

Other frequently occurring interactive thoughts which Miss reported included an awareness of what other students and the teacher were doing and thinking, how other students were performing, and mainly positive emotions about the classroom generally.

Attri) ational Analysis of Self-Performance Covert Behavior

Causal Explanations of Mike's Behavior

The transcript: of the interviews with Mike contained several chains of comment which included causal explanations of behavior and causal perceptions of performance. In collating these chains of i comments four clusters were discerned which referred to common cognitions, perceptions, and motives of behavior. These clusters were task perceptions and tehaviors, specific performances, public participation, and perceptions of other students' tehavior. Each cluster is described with examples and anecdotes included.

Task perceptions and behaviors. Mike often referred to the easy material which the class was required to learn. His commonly reported reason for finding the subject matter and learning tasks to the easy was that it tended to be review from Grade 5.

- I: How did you feel you a . on with it?
- M: I thought it was pret wasy...yes, you know, it was pretty easy. I had done this last year. (Lesson 4)
- M: It was pretty easy.
- I: Wry was it easy?
- M: Well, it's only review, like, well it is just a review of what we did this year [early half of 1975] and we did it last year too [1977]. (Lesson 5)

Mike expressed disappointment in lesson 4 that the unit topic was proving to be all review. He had expected to learn a great deal more new material in Grade 6. His justification for this was a preference to experience the challenge offered by new work and so that he could learn all that there was to be learned.

Consequent behaviors varied according to lesson situations. In addition to attending or listening during whole class instruction, Mike often occupied himself with other classroom assignments. During one mathematics lesson he wrote a story for language arts, on another occasion he completed a word search task for spelling, and during a third lesson completed all mathematics worksheet assignments set for the day when he was atsent. While engaged in these off-task behaviors, Mike nevertheless kept an ear tuned to the teacher-led or student demonstration whole class instruction.

During seatwork, which normally involved completing worksneets, Mike endeavoured to finish the assignment quickly. He often went on anead, not walting for the teacher's directions or instructions.

- I: Okay Mike, as you first looked at the worksheet what were you thinking:
- M: Well, to try and get some things done in a hurry, to get a head start on them. (Lesson 2)

Mike's reasons for going ahead with the worksheet were that he understood what had to be done anyway and also that he might have somfree time at the end in which he might do some reading. Because the worksheet material was perceived to be review, he expected to get the worksheet mainly correct, if not all correct.

One aspect of the worksheets did prove annoying to Mike. "I was thinking, like, in some of what we are asked to check, it is a lot easier to use...looking of it you can tell. Using a Flexiglas is quite tad" (lesson 4). Where Mike perceived the answers in the worksheets on motion geometry to be self-evident from visual inspection he saw little point in going through the hassle of checking the work technically. His annoyance with this procedure led him to , i

comment to the teacher in lesson 4 that the thickness of the Flexiglas induced errors anyway. In lesson 2 Mike found he could not achieve sufficient precision in his drawing so left that example and went on with the next. Having understood fully the central ideas and relationships, Mike was disinclined to allow mere technicalities to slow his progress.

<u>Specific performances</u>. Mike expected to succeed on all his mathematics worksheets and on the unit test.

- I: How well did you do [on a worksheet]?
- M: I got that one all correct.
- I: Can you remember your thoughts and feelings at that time when you looked at the sheet?
- M: I expected to get it all right. (Lesson 5)

With the worksheets he expected them to be easy because they were just review. He accepted that from time to time he might get one or two wrong but would only be surprised if he got many wrong.

- I: How would you feel then?
- M: I don't know.
- I: If you did get any wrong, what would you see as some reasons why you got it wrong?
- M: Well, I might have been rushing too much or never understood it. (Lesson 5)

Of interest in this hypothetical situation is Mike's attribution of failure to lack of effort.

The implications of Mike not knowing how he would feel if he failed underscore the typical level of achievement of this student. Yet Mike dii not experience total success in the unit test.

- I: How did you get on in the test Mike?
- M: I had 87 percent.
- I: Were you surprised at that result?
- M: Yes. I was more or less. I thought it was a bad mark.
- I: What reasons would you give for not getting as good a

mark as you hoped for? M: Maybe I was careless. (Lesson 10)

Mike explained that the mark was not too bad seeing as he had lost $4\frac{1}{2}$ marks on minor technicalities due to lack of precision. But he was disappointed that he had not scored higher.

- I: What were your thoughts when you heard her announce your mark and you walked up to get it?
- M: I was sort of disappointed that I only got 87. I was feeling bad.
- I: But was it still a good mark?
- M: To me it's not really a good mark. I am so used to higher marks, I guess. (Lesson 10)

Mike attributed his lower than expected performance to lack of effort, yet in a sense he also attributed the relative failure in part to the rigorous marking system.

Each one was something small and it happened in a different part. I got four marks off for four different questions and then a half mark off for another one. As she said, the half marks added up...On one of them I forgot on the parallel lines, I got marked off 1 mark out of 2 marks because of no label.

The teacher's constant reminder of the need for precision and Mike's disinclination to concern himself with detail during the other learning sessions of the unit probably accounted for Mike's initial internal attribution of perceived failure. This reaction was also consistent with the earlier described hypothetical situation.

<u>Public participation</u>. Mike readily volunteered and frequently found he was called upon by the teacher to demonstrate an idea in front of the whole class or to answer a question during discussion. On most of these occasions Mike experienced no emotion, reporting that he was not really conscious of other students watching him. I: When you are up at the front are you aware that all the kids are looking at you?
M: A little bit.
I: Does it worry you at all?
M: No.
I: Why do you think it doesn't worry you?
M: Well...I usually don't worry or get embarrassed. (Lesson 5)

Mike stated repeatedly that he did not worry about being in front of the class. He could not recall ever experiencing failure in that situation.

In general, Mike was not surprised to be picked by the teacher to contridute to the discussion or publicly demonstrate an idea.

I: Why do you think she picked you?
M: I don't know. Well, she might have thought I hadn't had a turn for a while or something like that.
I: Do you find that you often get called upon by her to answer questions?
M: Quite often.
I: Why do you think that is?
M: Well, I guess most people get called upon quite often when they put up their hand.
I: Do you think you get called on more than other people?
M: No. (Lesson 5)

These comments by Mike suggested he felt relaxed when contributing publicly to the lesson and apparently perceived his amount and kind of participation to be normal for class members.

٢.,

<u>Perceptions of other students' behavior</u>. While being aware of other students' public contributions to the class, Mike seemed only slightly concerned about why they participated or how they performed. When Lisa was in front of the class demonstrating an idea, Mike commented:

- M: I was thinking she probably couldn't do it, you know. Then she would get all embarrassed.
- I: Okay.
- M: I was just watching, 'cause I knew she couldn't do it anyway, so I started doing my story again. (Lesson 2)

On another occasion Mike pointed out that he sat up and took more notice when another student whom he perceived to be smart began challenging what appeared to be a fact. "He [the other student] usually picks out little mistakes. And he usually debates quite a bit" (Lesson 6). Mike did report that he knew most people's mannerisms and habits. During the unit test Mike hardly considered how other students were doing. He did make one comment: "Some people I think are uptight, you know!" (Lesson 9)

Mike's Basic Ideas and Beliefs

<u>Self-expectations</u>. Mike seemed to hold high performance expectations for himself in the learning of mathematics. This pervading notion was discerned across a variety of situations. When he first studied newly assigned worksheets, Mike expected to do well because he perceived them to be mainly review. Consequently, Mike nearly always "got it all correct" or close to being all correct.

During situations in which he participated publicly, Mike generally believed his contribution to be correct.

- I: What were you thinking?
- M: I was thinking about the math thing, you know, when she asked the question and it just came into mind, the answer.
- I: Then you put your hand up.
- M: Yes, 'cause I had it right. I was pretty sure I was. (Lesson 5)

Mike seldom experienced any form of failure in these situations and hence appeared at ease personally and always confident that he understood clearly the subject matter being learned.

In terms of the unit test, Mike also expected to do well and believed he would feel "pretty good about this."

- I: How do you expect you will get on in the test?
- M. I think I'll probably get on pretty good.
- I: What do you mean by pretty good?
- M: Well I might get it all right or nearly all correct. It's hard to tell.
- I: Have you ever got all correct?
- M: Yes, quite often. (Lesson 8)

In an earlier lesson Mike commented in a similar way about the upcoming unit test, noting that the test would be easy because it was mainly review. However, Mike also added a further reason for the expected success: "Well, I guess I usually do good in most things at school on math or so" (Lesson 6).

The comment regarding Mike's perception of his own ability was supported by another statement from the pre-test interview. In response to the question, "Do you think they [the other students] wonder how well you are doing?", Mike replied.

Maybe. I sometimes think that they might very well find me better than anyone else, something like that. I sometimes get the thought.

A clear indication of Mike's self-expectations was revealed in comments when the teacher was discussing the distribution of test scores prior to handing back the tests. Mark expected to be at the top--he had no expectations at all for being below half marks: "I was wondering what I got. I sort of felt that I might be one of the best in the class."

Mike also volunteered the information that the previous year the Grade 5 students were streamed for mathematics and that he was in the top maths group. The direct evidence and interpretations of a variety of his comments lend support to the conclusion that Mike believed he expected to do well because he perceived himself as a student of higher ability. Ease of task probably served to reinforce the high self-expectations.

Mike's perceptions of the teacher's performance expectations

of himself. While Mike often referred to his high performance selfexpectations in mathematics, he rarely expressed comment as to how he thought the teacher perceived him. During the interviews involving the unit test, Mike did comment:

- M: I don't think she would be surprised if I got it nearly all or all correct. I think she would be surprised if I did real bad.
- I: Why do you think she would be surprised if you did real bad?
- M: Well, in most of the worksheets I did pretty good. (Lesson ϑ)

Following the announcement of the test results Mike felt that the teacher wouldn't be too surprised with his lower than expected score. Mike believed the teacher's thoughts about him in mathematics more or less matched his own thoughts of his performance. The inference is drawn that, because Mike perceived himself to be a student of higher or above average performance, the believed the teacher perceived him in a similar manner.

<u>Perceptions of other students' performances</u>. Mike seemed to perceive clearly the achievement hierarchy of students in the classroom. When asked how would he know who were the top five cr six students in mathematics, Mike commented:

Well they sort of seem to always know what's happening and whenever they are asked a question they get it right or at least mostly, you know. And they can explain things you know. And some of them like, have been in my math class for a couple of years so I know. (Lesson 5)

He perceived that students who were not sure about the work tried not
to get asked questions so these students would be the lower students. As well, Mike reported noticing that the teacher supervised some students more than others because they were people whom the teacher thought might have had more problems.

Mike was able to determine how other students performed in the unit test by means of the teacher's discussion about the distribution of marks. However, he was basically unconcerned with the performances of other students, except to note that his mark was all above the average. Mike seemed to hold no negative attitude towards students of lower ability than himself. Likewise, here noncomplete towards the teacher recognizing publicly those students who had performed well in the test. While this may reflect Mike's lower than expected score, it was also consistent with a wider attitude of lack of real concern or interest in how other students performed. He did express disappointment when he realized another student had gained the top mark in the unit test, but only because it deprived him from achieving that success.

Perceptions of tasks and activities. Mike held differential . views regarding the most important facets of lessons:

Well, if it is new stuff then you listen to the teacher so that you can get your worksheet right. If it's review, doing the worksheet is the most important. (Lesson 8)

Throughout the lesson series he would have preferred new work for the challenge it offers. When asked as to what was the most important thing in mathematics, Mike replied, "Learning new stuff."

In terms of difficulty level of worksheets, Mike preferred those which contained some exercises which caused him to struggle, yet which he was able to manage all right. However, if only a short period of work time was available, Mike preferred easy worksheets:

If we don't have all that much time we usually like an easy worksheet, but usually we like it half and half....Sometimes I like doing a hard one, you know, like star questions or think questions. I sort of like doing those ones. (Lesson 7)

The importance of the unit test was emphasized by Mike.

I: Do you see the end-of-unit test as an important thing.
M: Yes. I think so, especially since the report cards are based on that. (Lesson 7)

When asked during the pre-test interview if he cared about tests, Mike reiterated the same point:

Because, you know, if I don't care, I'll just do all the answers without thinking. Like, you know, it will reflect on my report card, and I don't want that. (Lesson 3)

The report card featured as one underlying motive in Mike's behavior and performance in the classroom and a good performance on the unit test was perceived as the necessary component. Mike stated in lesson 7 that the lesson situation of listening counted most toward doing well on the unit test because worksheets were viewed mainly as review.

A Characterization of Mike's Conception of Self-Performance

Mike was selected for study because he was judged to be a successful student. A characterization of his conception of selfperformance was interpreted from the available evidence. The major elements of this characterization were:

 Mike's approach to learning can be described as masteryoriented. Success was always perceived by Mike to be obtainable through a combination of ability and effort. His self perception of higher ability seemed to be the major factor underlying his high performance self expectations. Any pertial failure was readily accounted for by insufficient effort.

- A' Mike perceived all subject matter and lesson tasks to be of a low level of difficulty. He understood all concepts and all operations in the unit topic being studied and attributed this clear perception of task structure mainly to review.
- 3. Mike revealed few emotional reactions to his performance. Successes were viewed as inevitable confirmations of his self-expectations. Where his performance fell short of his expectations, Mike tended to experience morally neutral feelings of disappointment, but these were of temporary duration and of shallow intensity. No evidence was available as to the kind of feelings Mike would experience if he ever performed poorly.
- 4. Mike readily volunteered to participate publicly. When in front of the class Mike experienced little or no emotion, always appearing relaxed. He expected to succeed and his behavior reflected his high degree of self-confidence.
- 5. Mike tended to be unconcerned with the motives and behaviors of other students, yet he was aware of what they were doing.
- 6. Mike perceived clearly the classroom achievement hierarchy in relation to himself. He perceived himself to be one of the best students in the class and he considered other students shared this point of view of his performance. Mike

Judged the other students' relative positions by how they behaved and performed during the learning situation.

- 7. Mike believed the teacher head higher performance expectations for him.
- 8. Mike perceived as an important goal of classroom performance the gaining of a good mark on his report card. Because he held high self-expectations based on his self-perceived higher ability and because Mike perceived the teacher held higher performance expectations of him, Mike readily believed he would achieve a good mark on his report. Mike did not expect to fail. He possessed a high degree of self-esteem based, mainly on his successful behavior and performance in learning mathematics. Mike's perceptions that his peers acknowledged his ability contributed to his high level of self-esteem.

Analysis of the Dyadic Interactions Between Mike and the Teacher

Analysis of the Overt Behaviors

٩,

The results of the analysis of the interaction sequences between Mike and the teacher are presented in Table 17 and in Figures 9 and 10. Table 17 shows that Mike and the teacher interacted dyadically 13 times during the nine mathematics lessons. Approximately half of the interactions were teacher-initiated and public, while most of the balance were student-initiated private interactions. These student-initiated private interactions occupied a total duration time of 102 seconds, although one such interaction of 55 seconds pertained to a non-mathematical concern. Tatle :7

۲

Frequency Distribution of Dyadic Interactions Between Mike and Teacher

Type of					r.ee	Lessons					
Interaction	•••	61	\sim	+	5	ò	2	æ	ড		Total
Teacher initiated public interaction	••	1	• •	I	· 4		()		1	1	٤-
Student initiated public interaction	ł	•	I	'n	ſ	,	,	< д	,		
Teacher initiated private interaction	Ţ	I	,	,	ı	ı	i	പ	, I	,	
Student Initiated private interaction ^a	·	ı	V	. •		•	•	R (-1	•-4	•	~ \
Total :	• •	1	-	•••	~ ~	0	7			• •	:

Duration of the student initiated private interactions was 3, 22, 20, 55, and 2 seconds, respectively, for a total time of :02 seconds.

١

y



Type of		Interaction	Teacher	
Interacti n	rulated by	Related to	Reaction	Student Reaction
Ň	Teacher			
Private 0.38		Jontent com 0.40		
	Student :.00	irocedm. 00	Trocess U.JU	
		Truced. q. C.20		
			TOCESS . UU	
		•		
Figure :0.	Nature of irivate Dyadic Inter Freportions		wen Mike and Teacher and Their	Their

.

Figure 9 indicates that in the seven teacher-initiated public interactions Mike volunteered on most occasions to respond, that he answered correctly the process questions and that the teacher tended to give process feedback by repeating Mike's response and outlining the reasoning he used to arrive at that response. In four of the seven interactions the teacher incorporated praise in her reaction to Mike's answers. On the one occasion that Mike was initially judged to respond in a partially correct manner the teacher provided a brief clue which enabled Mike to arrive at the answer.

The private dyadic interactions as shown in Figure 10 were all student-initiated, to which the teacher responded in an extended manner in three of the five interactions. A substantive analysis of these three interactions revealed that each concerned trivial aspects of the learning involved and not one dealt with subject matter content information. Overall, the quality of private dyadic interactions was characterized by non-significant issues, and by trief and neutral communications.

The Overall Analysis of the Dyadic Interactions

Some of the dyadic interactions between Mike and the teacher were analyzed by an overall examination of the observable data, the teacher's covert behavior, and the student's covert behavior. However, the number of interactions was very small and, consequently, the items of data described were only samples of actual behavior. They may not constitute generalizations of phenomena prevailing whenever Mike and the teacher interacted dyadically.

In lesson : Mike had volunteered to demonstrate an idea on the blackboard in front of the class. The teacher commented that she was concerned that previous attempts to explain the idea had proved misleading or had sidetracked the discuspion.

Teacher:	Now when a child doesn't answer what you expect, you
	have to suddenly make a snap decision as to what
	direction your lesson is going to go from there.
	I knew that our time was running, and I said I hope
	Mike does do itthat there are no mistakesthat
	he does the method that I was thinking of.

- I: Was that one of the reasons why you called Mike up in that case?
- Teacher: Probably yes. I know that sometimes I do that. When I ask a question you can read the answer in the child's "ace and you know that they understand or not. I probably did call Mike up at this point to say... let's move along...we've got to get on with the worksneet. (Lesson 1)

Mike viewed the episode in a rolline manner. His thoughts were on the question at hand and he saw no particular reason why he was picked and not someone else. He expected to demonstrate the point correctly and commented as he returned to his desk, "I knew that I had done it right." Mike further recalled that as he returned to his desk he was watching a desk dispute between two boys and that his mind was not really focused on the teacher's discussion of his demonstration to the whole class.

Earlier in the same episode a student perceived by the teacher to be less successful had demonstrated what seemed to be an ingenious idea.

Teacher: I was actually superised...coming from Clare. Now in Duncan or Mike or someone like that had come up and done it, I would have thought "Yes" because they are the ones you sort of think of as "quick thinking."

The evidence provides further confirmation of the teacher's

perceptions of Mike as a successful student and that this success is attributed to his ability.

Some interactions involved trivial concerns of Mike. While they reflected an intellectually stimulated mind they nevertheless were viewed as probably unnecessary.

Teacher: I was aware of Mike behind me [while working with another student, at this point and I was thinking, "Is it going to be another one of yesterday's comments?" [an earlier innane comment]. I gave him a funny look when I stood up and I looked at rim. You know, "What is it now?" He noticed the thickness of the Plexiglas and compared it to the thickness of the line and he was saying it may create error...At least it was a bit more of an intelligent comment.

On another occasion during the learning of mathematics Mike was particularly concerned with the number of library books ne had read in a period of time and felt that the number had been incorrectly recorded. While the teacher acknowledged Mike had completed his work, she was concerned that this reflected an off-task behavior which probably conflicted with her expectation of Mike. Mike, meanwhile, had expressed concern with the inaccurate recording is is read as early as the first five minutes of the lesson. The dyadic interaction involved a total of 55 seconds of time.

A plausible interpretation of the evidence presented suggests that the teacher held higher performance expectations for Mike. Mike was one of the students used by the teacher when she wanted to change the pace of the lesson or redirect the discussion along pre-planned lines. The teacher felt sufficiently assured of Mike's understanding and contribution to the class to call upon him whenever she posed process questions involving higher order thinking. She tended to

provide praise in her feedback to Mike as well as integrating his correct responses into the overall discussion. Mike was aware that the teacher held higher performance expectations of him but little or no evidence was available to suggest he manipulated situations to enhance the expectations held.

Summary Analysis

The results of describing Mike's achievement-related behavior during the learning of mathematics have established some relationships among and between aspects of his thoughts and his behavior:

- 1. Mike held high performance self-expectations which included gaining a good mark on the report form.
- 2. His expected success was perceived by Mike to be based on higher ability.
- 3. Mike perceived the teacher and other students held higher performance expectations of him.
- 4. Mike perceived these teacher and other students' performance expectations were based on causal perceptions of higher ability.
- 5. Mike was perceived by the teacher as a student who was mainly on-task.
- 6. The teacher held higher performance expectations for Mike.
- 7. The teacher's causal perception of Mike's success was mainly to ability and to some effort.
- 8. The teacher appeared to communicate these expectations through interaction process behaviors such as selective use of Mike's

volunteered responses to pace the lesson, by selective use of Mike to respond to higher order questions, and by the use of praise when giving feedback to Mike's correct responses.

- 9. Mike performed successfully in all lesson situations, reflecting his mastery-oriented approach to learning. He hardly knew failure other than those accredited to minor lapses in effort.
- 10. Mike's successful performance served to sustain the high performance expectations held for him by the self, the teacher, and the other students.

A Summary Comparison of Tina and Mike

This chapter has described the achievement-related behavior of two "successful" students, Tina and Mike. Both students were observed to be on-task most of the time, especially during seatwork. Not attending-listening behaviors were more likely to be coded during teacher-led and student demonstration whole class instruction. In quantitative terms, the interactive covert behavior of the two students was similar in pattern except for one basic difference. Mike's thoughts tended to be more content or task-oriented than Tina's, while Tina's thoughts tended to be more self-oriented than Mike's.

Tina and Mike subscribed mutually to the importance of the report card grade, their approaches to learning were mastery-oriented, and their performance self-expectations were perceived by them to be higher or above average. They usually experienced success with positive affective consequences. However, where Mike attributed his success largely to ability, Tina attributed her success largely to effort. As a consequence, Mike seemed uninvolved emotionally and unconcerned about his public image, which reflected his task-oriented self-conception. In contrast, Tina was concerned about her public image, hence her self-conception seemed to involve more personal elements.

Both Tina and Mike seemed aware that the teacher held higher performance expectations toward them, as conveyed by some of the teacher's instructional moves. Tina perceived the teacher's higher performance expectations were effort-based and hence she sought to maintain or enhance this image. Mike perceived the teacher's higher performance expectations of him were ability-based. The teacher was viewed by both students as espousing effort as the criterion of success.

Chapter VI

CASE STUDIES : LISA AND GREG

LISA

Biographical Information

Lisa was the youngest of three girls in her family. Both of her older sisters were teenagers attending junior high school and senior high school, respectively. The father was not living with the family in their home in the neighborhood of the school. Lisa had lived much of her early life in a northern Alberta community before the family moved to their present urban home.

Lisa was aged 11 years and reported swimming and running to be her main interests. She enjoyed watching television, especially "Bewitched" and sometimes the news. Her travel experience extended as far as Vancouver and northern Alberta.

Lisa stated reading, spelling, and language arts in general to be her best subjects. In terms of maths, she reported being in the bottom class in Grade 5 where the classes were streamed for that subject area. Lisa had one close friend in the classroom. Her involvement in school life included being a traffic patrol warden, a library helper, and a helping hand in a lower grade classroom during after-school hours. When asked as to why she came to school, Lisa

replied, "To learn. The teachers can explain it better than anyone at home."

The teacher viewed Lisa as a student whose ability was slightly below average. The teacher detected early in the year Lisa's problem with processing new information: "I noticed several times, something I have just gone over and over and over, she'll ask a question about, as if she hasn't heard anything." On examining Lisa's cumulative record, the teacher found that the problem had been investigated and was a factor to be considered by the teacher when working with Lisa.

The school's cumulative records showed that in Grade 5 Lisa nad gained an achievement percentile of 5 in mathematics. This represented a low achieving student. Her FAFS total expectation score of 100 was the lowest recorded in the class and reflected a low degree of expectancy of success in performance across most school subjects. Her mathematics sub-score of 17 out of 28 indicated that her expectations for that subject area were correspondingly low. Examination of Lisa's responses in mathematics showed that she expected to learn only some of what would be taught, expected that most other kids would be better than her, and expected to like math a little.

Lisa's self-reporting of internal responsibility for success or failure on the IAR revealed that she attributed 24 of the 34 situations to internal causes and 10 to external causes. Because her internal responsibility for success score of 13 was close to being equivalent to her internal responsibility for failure score of 11, it can be concluded that Lisa assumed credit for causing good things to happen in much the same way as she accepted blame for unpleasant consequences. Examination of her responses tended to support this conclusion.

.

quantitative Analysis of Behavior

Lisa's behavior during the learning of mathematics was analyzed quantitatively according to both the overt and covert aspects.

Overt Behavior

Not attending-listening behaviors were observed to account for a very small proportion of Lisa's learning time. Table 20 indicates that these behaviors accounted for barely 3 percent of the Table : $\hat{\partial}$

Percentage Distributions of Student Overt Behaviors in Each Lesson Situation Category for Each Lesson , Lisa

												élthin-
Gategories of Student Overt Behaviors Within Lesson Structions					Lessons	SUO					řercentage Nean	Category Sum of
STID TA 87A TA 110990A		<u>ب</u> ې	3	•	5	<u>،</u>	~	ന	6	0;		Percentage Means
<u>Teacher-Led Whole</u> <u>Class Instruction</u> Attending-listening Not attending-listening	10.0		3:4	8 8	9	25.0 25.0	0.0 + 0 +	6.10 4.90	8.7	87.5	4 6 1 1	
Interacting with other students Out of desk	; ; ; ;			÷ : ;		<u></u> С	1 1	5. 1	ž.2	÷ : :	(۲ د) ••••	τ τ τ
Student Demonstration Whole Class Instruction				·	-							
Attending-listening Not attending-listening Interaction with Other	1.0	···	3.9	1) 1 1		: :	0 * 7 *	5 1 1	 		50	
students Out of desk	 	4 F t 1	1 1	1 1		-	I I I I	1 1 1 1			+ 1 1	r. r
Seatwork Attending-listening Not attending-listening	0				· · · · · · · · · · · · · · · · · · ·	53.3	<u>्र</u> ्	· D • 1 • 1	62.6 .5		e E E	
Interacting with other students Out of desk Supervised by teacher		1.00	5.5 5.5 5.5	.				- : : : 0 : : : : : : : : : : : : : : : :			2 2 4 ·· M	43.9

Fercentage D. #1th1n	Furcentage Distributions of Means for Student Overt Fenavious Within Each Category of Lesson Situations : Lisa	a A
Categories of Lesson Situation	Categories of UVert Behavior	Hithin-Category Preentage Distributions of Means ^a
Teacher-led whole class instruction	Attending-listening Not attending-listening Interacting with other students Out of desk	6. 6. 1. 0. 0. 0. 0. 0. 0. 0. 0. 0. 0. 0. 0. 0.
Student demonstration Whole class instruction	Attending-listening Not attending-listening Interacting with other stulents Out of desk	
Seatwork	Attending-listening Not attending-listening Interacting with other students Out of desk Supervised by teaster	
-		

^AMeans for student event behaviors derived from Tatle 18.

•



Tatle 14

¢

Jable .

•

٩

Fercentage Distributions of Student Overt Benaviors for Total Lesson Time : Lisa

.

•

Catedories of Student					Lessons	ons					rercentage
Overt Behaviors	-	2	\sim	t		\$	6	ω	5	ري • •	Mear
Attending-listening	0. %		7 3.3 84.3 36	ж	ين ئە	a) a) a)	0· 06	90.0 BC.5	9: . 3		ۍ. جن
Not attending-listening	1 1	1	5	ı I	I k	Ŭ.	•}	6.4		3.1	9·ĉ
Interacting with other students	;	£.£:	6.0	£ • ? ;		(I) ب	ر ۲	e. 2	8	5. 5	4) • •L/
Out of desk	ł	6.3	5.0		•	;	1	t U	7.7	4	4. Y
Supervised by teacher ') • •	6.3	3.4		1	v) 	I I	-4 -4	1	1	3.2
			en								

.

3

. .

•

415

.

-

¢

lesson time. Not attending-listening behaviors were not observed in
+ of the je lessons.

A noticeable proportion of the learning time when hiss when recorded as not being on-task was absorbed by Lisa interacting with other students, especially during seatwork. Table how reveals that more than a percent of lisa's overt behaviors were coded as interacting with other students, with the seatwork lesson situation accounting for more than two-thirds of that percentage distribution. As Table 14 shows, interacting with other students was could to percent of lisa's seatwork behaviors. She was being supervised by the teacher for another 7 percent of her seatwork time. Out of desk behaviors were diserved to occur infrequently.

While Lisa was recorded as teing on-task most of the time, Table if shows that lessons 2 and 7 were days on which Lisa's time during seatwork was divided significantly between on-task and other tenaviors. In lesson 2 seatwork, Lisa was on-task [3 percent of the total lesson time and engaged in other behaviors about (9 percent of the total lesson time). The percentage distributions for on-task and for other behaviors during seatwork in lesson 7 were (5 percent ad to be the lesson situation in which Lisa was most likely to be engaged in the votors other 'han attending-listening.

Covert Betavior

The percentage distributions of lisa's interactive thoughts across 11 lessons are presented in Table 21. Because the final * rehearsal lesson contained valuable information about Lisa's Ť 41 . · · · 1

•

incontage Mistricutions of Student Thoughts Over CANNIT Categories for Each Lesson : Lina

.

Covert Behavior			1			Lessons					:	Percentage
	æ			\sim	.+	····				5	0;	Muan
Subject Na tter	r I	5 2		1	;	2.2	t t	:	g.)	1		C- - 4
Cognitive Processes	5.5	1. D	÷.		I 	1, J	1) -1	÷.		1	ł	5.0
Behavioral MovesSelf	· 0 ·	в. 57	5. £.	<u></u>	Ţ,	۔ ب		28.5 20.8	5.5	3. 6	(r)	2.52
Behavioral MovesStudent	0 C :	4.2	÷.	:4	r ·		ج. <i>ب</i>		€. E	5.5	3.) • • •	0 • 0 •
Behavioral Moves-Teacher	7.5	! }	5.6	, , ,	а. 	•••	6.6	1	$\tilde{\mathbf{c}}$	1	20.C	6.2
Self-FerformanceThoughts	6.72	0	ε. ζ.	:-67	4.63	-5-6	33+3	×0++ 1;0+2	. E. 7	37.5	J.C.	26.7
Self-Performance-Realings	.)•o.	1		, . , 1	5.4	(°.)	i ŧ		$\hat{\mathbf{x}}$	- 7-2	2.2.25	5.0
Feelings	0.01	77.0	• •	$\frac{1}{2}$	I I	(`. (`	т. 1		? .	:	5.5	10.0
Non-Task-Related	ي د ب	्र - -		l i	1 	1	Ç.9		, v.çz	1	1	6.6

117

conception of self-performance, the data from the lesson were also analyzed. About half of Lisa's reported thoughts pertained to selforiented aspects. An average of about 27 percent of all thoughts across all lessons were categorized as self-performance--thoughts, while 23 percent were coded as behavioral moves--self. The self-focused character of Lisa's covert behavior during learning was more pronounced with the inclusion of self-performance--feelings; these accounted for another 8.5 percent of all thoughts.

lisa's feelings about classroom matters in general were reported frequently and averaged 10 percent of her interactive covert tenavior. Another 9 percent of Lisa's thoughts were oriented toward tenavioral moves of other students. The categories of cognitive processes, behavipral moves-teacher, and non-task-related thoughts each contained between 5 to 10 percent of Lisa's reported thoughts, while she made very little reference to subject matter thoughts when learning mathematics.

Six of the categories of interactive thoughts were subcategorized in order to further analyze those thought units which were presumed to pertain to Lisa's conception of self-performance. Those categories were behavioral moves--self, behavioral movesstudent, behavioral moves--teacher, self-performance--thoughts, self-performance--feelings, and feelings. Table 22 presents the results of the more internave analysis.

Behavioral moves--self. As Table 22 indicates, over half of Jisa's thoughts coded as behavioral moves--self were concerned with attending-listening and with the motive to attend. This finding

Table 22

Within-Category Percentage Distributions of Sub-categories of Selected Student Thought Categories of CASSIT : Lisa

Sub-categories of Selected CASSIT Categories of Student Thoughts	Within-Categor Fercentage Distributions
Behavioral MovesSel:	
Seek public particitation	7.40
Avoid public participation	. :9.40
Attending-listening	38.80
Motive to attend	:4.93
Motive to avoid attending	11.94
iretending to attend	7.40
Eehavioral MovesStudent	
Ferceptions of other students' performances	3 14
Inferences of other students' thoughts	
Interactions	ى) نى بۇلىم
Serceptions of other students' becalar	3
Hetavioral MovesTeacher	
Frierceptions of teacher's instrumional moves	5.25
Interactions	54.53
lerceptions of teacher's behavior	· · · · · · · · · · · · · · · · · · ·
Sel:erformenceThoughts	
Sel: -assessment success	14.10
Self-assessmentfailure	25.04
ierceptions of task difficulty	7.59
Perceptions of task structure	21.79
Self-attributions	3.85
Self-expectations	10.26
Significant others	10.07
Self-lerformanceFeelings	34.29
Anxiety Morally neutralpositive	17.8L
Morally neutralnegative	35.71
Morally unneutral positive	3 .77
Morally unneutral negative	3.57
Morally unneutral negative	$\mathcal{I} \in \mathcal{I}$
Feelings/ Fositive	43.33
Negative	56.62

supports the trend noted in the analysis of her overt behavior. However, a trend not discerned in the overt analysis was the 19 percent of thoughts which were concerned with motives to avoid attending and pretending to attend. Table 22 suggests that Lisa's thoughts about volunteering or contributing publicly in the class were more often oriented toward avoiding participation than seeking participation.

Behavioral moves--student. Table 22 shows that Lisa was aware of what other students were doing and how well they were performing. Foth sub-categories contained about 32 percent of Lisa's behavioral moves--student thoughts. The overt analysis had indicated that Lisa spent a significant amount of her seatwork interacting with other students and this is matched by the 25 percent of Lisa's behavioral moves--student thoughts which were concerned with interactions. Only rarely did Lisa report thinking about what other students were thinking.

<u>behavioral moves--teacher</u>. Of the small percentage of Lisa's interactive thoughts which she reported about the behavior and actions of the teacher, most were concerned with the interactions which she shared with the teacher. However, as Table 22 also shows, Lisa was aware of what the teacher was doing. Examination of this sub-category of Lisa's thoughts revealed that approximately half were perceptions of how the teacher would react if Lisa initiated an interaction for a particular reason.

Self-performance--thoughts. Table 22 indicates that most of Lisa's self-performance--thoughts during the learning of mathematics

were concerned with self-assessment. Of the nearly 40 percent of all thoughts in this category, 26 percent referred to failure and 14 percent referred to success. Also very much on Lisa's mind were perceptions of task structure which accounted for another 22 percent of thoughts in this category. Lisa also appeared to give consideration to significant others, a sub-category which was coded 17 percent of all self-performance--thoughts. A further 10 percent of these thoughts referred to Lisa's self-expectations. Examination of the transcripts containing the data used in this analysis showed a generally negative trend prevailing in most sub-categories of Lisa's self-performance--thoughts.

Self-performance--feelings. Most of Lisa's self-performance-feelings contained negative elements. Table 22 reveals that 39 percent of all her expressions of emotions reflected anxiety such as fright and worry, 36 percent reflected morally neutral--negative feelings such as disappointment and despair, and 3.5 percent reflected morally unneutral--negative feelings such as jealousy. Of the remaining 21 percent of self-performance--feelings expressed, most referred to morally neutral--positive emotions such as pleasure and hojing for success.

<u>Feelings</u>. Table 22 shows that more expressions of negative affect were reported by Lisa than positive affect. Negative feelings expressed included anger, boredom, disappointment, dislike, envy, embarrassment, and fear. Of the positive emotions expressed, amusement, excitement, pleasure, and pride were most often reported.

Summary of Quantitative Analysis of Behavior

Some patterns in Lisa's behavior can be discerned from a quantitative analysis of the available data. At the overt level, Lisa was observed to be a student who was on-task for the whole class instructional situations and mainly on-task during seatwork. The covert analysis would seem to suggest, however, that Lisa was viewing instructional events through a screen of largely negative thoughts and feelings.

Elements of this negative learning screen identified from the quantitative analysis of Lisa's behavior included trends for avoidance of public participation, some intentions for not attending-listening, self-assessments of failure, perceptions that success or failure on tasks was more likely to be controlled by external factors than by her own level of competence, awareness of the influences of other people on her performance, and a collation of emotions considered generally to suppress learning effectiveness. The majority of Lisa's interactive thoughts, therefore, were self-oriented, both in terms of her own performance and her behavior in the learning process.

Attributional Analysis of Self-Performance Covert Behavior

<u>Causal Explanations of Lisa's</u> <u>Behavior</u>

Numerous chains of comment which included causal explanations of behavior and causal perceptions of performance were discerned in the transcripts of the interviews with Lisa. Five main clusters of these chains of comments reported by Lisa were approach to learning,

perceptions of tasks and behaviors, specific performances, public participation, and perceptions of other students' behavior. Each cluster is described, with examples and anecdotes included. '' "Lesson R" after some examples pertains to the final rehearsal lesson.

...

Approach to learning.

1:	How did you feel at that time?
L:	I just didn't know what it was and I was just going to
	stay there I have what it was and I was just going to
	stay there. I was frightened because I thought if she
_	
1:	
L:	Because everybody else has their hand up, they knew the answer and I didn't.
	another the second state of the second
1:	You would fuel bad?
L:	Yes, because of what the other kids might think about me. (Lesson B)
	me. (Lesson \mathbf{R})
	Me. (Lesson H)

This sample of a chain of comments illustrates the kind of thoughts which pervaded Lisa's mind during the learning of mathematics. She frequently made reference to the cognition that she did not understand the material to be learned. In most cases this cognition was accompanied by a negative emotion. Very often the cause of this emotional accompaniment of the initial cognition was perceived by Lisa to be what other students and the teacher might think of her if and when they became aware of her failure.

- L: I thought I would get worried if I had to do a worksheet
- I: Yes...
- L: Get all confused...
- I: Why would you get worried if you had to do a worksheet?
 I: Because I wouldn't want to do it and I would be scared to ask the teacher because then she would think that I wasn't listening.
- I: What happens when you are worried like that?
- L: I'll ask other kids. I ask Andrew. (Lesson R)

Lisa's plight was even more sharply kposed in comments such as the following:

I was stuck on one question, but I asked Clare, 'cause I didn't want to bother the teacher. I thought maybe she [the teacher] would think I was dumb because I didn't get it. (Lesson 2)

These samples of several similar reports of Lisa's covert behavior illustrate her reactions to failure. They also suggest that Lisa attributed her failure to a combination of lack of ability and lack of effort. Her approach to learning, therefore, appeared to contain a number of the characteristics of learned helplessness. The frequent comments such as "I don't understand it" or "I needed help" implied that Lisa perceived success or failure on tasks to be more likely controlled by external factors than by her own level of competence or her input of effort.

The helpless approach to learning was observed in a number of circumstances. Once Lisa reported:

When she [the teacher] was at the tlackboard, I'd swear I was wrong, and I thought, "Why should I lister...I know I'm wrong and why pay attention." I said, "Carol, can I use your comit" and here I was combing it and pretending I was listening. I didn't feel like listening because it was getting toring. (Lesson R)

In another occasion Lisa asked the teacher for help. Lisa perceived from the teacher's voice that she had become annoyed with her because she said, "You've done that before!" Lisa's comment in reaction to this was: "That's why I hate asking her and why I ask other kids." While these examples reflect a student's distortion of what really happened, it nonetheless illustrates elements of a student's view of the learning situation; a student whose approach to learning could be described as learned helplessness. Task perceptions and behaviors. During the interviews, Lisa often referred to the nature of tasks either in terms of their level of difficulty or how well she understood them. As described in the previous section, she frequently commented that she did not understand the material to be learned or that she needed help with the mathematics. Accompanying these cognitions were expressions of negative emotions ranging from anxiety to boredom. While she tended to attribute these perceptions of failure to internal factors, her consequential behaviors varied from situation to situation.

Lisa quite often pretended to listen when she did not understand the lesson discussion:

- I: Were you thinking about the lesson at this time?
- L: No, not really. I was just doing that 'demonstrates nodding of the head' so she didn't think I wasn't paying attention.
- I: Eut were you listening to what she said.
- I: No.
- I: What were you thinking then at that time?
- 1: I don't know. I can't remember. But I remember me shaking my head 'cause I saw her look at me and I went like that [demonstrates] and I thought if she saw me I'd get into trouble. So I wasn't paying attention, I was daydreaming. (Lesson R)

While daydreaming was one distracted behavior, others reported by Lisa in the interviews were thinking about Hallowe'en, the weekend and recess, passing notes to a friend, annoying other students seated near her, and planning a play she was organizing for a coming party.

Lisa's perceptions of the tasks were sometimes positive:

- L: I enjoyed it. I'd had so much fun.
 - I: Why did you find it fun and enjoy it?
 - L: Because it was interesting. I learn geometry better than math--the other kind of math. (Lesson 5)

On other occasions she reported that the worksheets were easy. With

the unit test, her **ini**tial feelings of threat became feelings of relief when she first looked at the sheets.

I was looking at it and I thought the first page was easy, the second page was getting a little hard, the third page was pretty easy. The second page was the hard one. (Lesson 9)

As a consequence, Lisa did not wait to listen to the pre-test instructions and words of advice but rather went on ahead.

A commonly expressed feeling toward tasks was that of boredom. Lisa reported boredom during some lesson situations in which she did not understand the material to be learned. She was also likely to report these feelings when she perceived the material being learned to be review:

- I: Were you thinking about maths? L: No. I: What had happened to the maths at this point? L: Boring. I: Why was it boring?
- L: Eecause I knew it already. It was getting turing so I began to play with my desk. (Lesson 2)

Lisa even became bored when the teacher discussed the teat questions following the announcement of the test results. The conclusion is drawn that reporting boredom served as a surface expression for when Lisa "turned off" maths. Once a whole class discussion involving material perceived by Lisa to be easy became prolonged; she became bored. Alternatively, when she perceived the discussion to be beyond her understanding, she became bored. The post-test discussion occurred when Lisa was fraught with intense negative feelings. These feelings also had the effect of her "turning off" mathematics so that she became bored. Specific performances. Lisa's approach to learning was

accentuated when reference was made by her to specific performances during the learning of mathematics. From an initial cognition of success or failure an emotional consequence tended to be expressed. Some of these expressions of emotion seemed to reflect intense

feelings:

- I: Can you remember your thoughts and feelings at that time a public dyadic interaction ??
- L: Actually the question...answering it, right. And I was right. And I was proud too.
- I: Why were you proud?
- L: Because I got it right and I don't usually get it right.
- I: And the teacher answered you in a special way referring to praise.
- L: Wow, I was listening for that! (Lesson 5,

On receiving tack a worksheet, Lisa rejorted:

- 1: I knew I didn't do too well and I got three wrong...I just felt kind of bad because I thought I might have got it.
- I: Why do you think you got three wrong?
- L: Well, I didn't look hard at the bottom when I started it and I go "I don't know how to do this," and she 'the teacher put, "Then ask!" and I did ask and she didn't help me. So I got mad. (Lesson 5)

While Lisa's feelings during the lessons ranged from jubilation at experiencing a minor success to feelings of annoyance with herself for what she described once as a "measy mistake," through to intense feelings of anxiety for significant failures, the most profound emotions of dejection were recerned about the happenings involving the unit test.

- I: How did you get on in the test?
- L: Horrible. I did rotten [she scored 54 percent, the class average was 75 percent]. I don't like it. And I am not taking it home either because I'm going to get a licking if I do.

While these comments alluded to a possible underlying motive for the

negative reaction, Lisa did perceive internal causes for the performance.

oh, I was mad. I said it was al. fault. I should have done it right. And if I do it again, I'm going to get a hundred...I swear. (Lesson 10)

Earlier Lisa reported that she wasn't surprised at the result for she raw expected worse: "I had expected worse. Harvey got the lowest though. I am not trying though. I want to try but I can't...I know I am going to flunk this year" (Lesson 10).

lisa perceived little or no effort-outcome covariation and consequently she seemed to attribute her performance in the unit test as much to a lack of ability as to lack of effort. On other occasions lisa attributed her poorer performance solely to lack of effort:

I had one little measy mistake. I knew the answer. It would not have been so had if I would have not known the answer. But I knew the answer and I was so careless. I was mad at myself. (Lesson 2)

- I: How did you get on in math yesterday
- 1: Well, I got a little bit right and a little bit wrong. I just didn't understand quite a bit of it.
- I: Why do you think you didn't understand quite a bit of it
- L: I don't know. Maybe I wasn't listening. (Lesson 1)

The general pattern, therefore, seemed to be one of attribution to effort for minor failure, and attribution to mainly atility and some effort for what Lisa would have deemed significant failure. It was her self-performance covert behavior during situations involving perceived significant failure which contained many of the characteristics of a learned helpless approach to learning.

Public participation. Lisa registed frequently comments about seeking or avoiding public participation such as contributing to the

class discussion or volunteering to demonstrate an idea to the whole class. On some occasions she tried to avoid being selected. These tended to be times when she did not know or was not sure of the correct answer or idea. Under these circumstances Lisa attempted to avoid conveying an impression to other students or to the teacher that one either did not know or was incorrect.

When Lisa volunteered to demonstrate an idea at the tlackboard offer was equally as concerned for the same reason even though she felt sure at being correct.

I: How do you meet when you go up to the toard like that :: Nervous I: Why are you nervous? L: Because I don't know if it's right or wrong, and if it's wrong I'll get laughed at from the kids. 1: Whit about whethou get it right'
1: I feel happy () (soon R) on recasions Lisa voil and to demonstrate an idea in front of the class when she was in sur doubt about the answer: 1: I was thinking, "Could you or could you not do it," and I debated, "Well, I don't think so, but maybe I could." So you gave it a try. Ι: L: Yes. I knew I was wrong. When the teacher picked you, how were you feeling then? Ι: L: Proud. I: Why were you feeling proud? L: Because I finally got up there. It's the second time 1.7 got up there in a whole year. (Lesson _) wever, once Lisa perceived she couldn't demonstrate the idea she felt upset. She did not want the other students to see or know that she had failed, and consequently she wanted to escape the situation by sitting down. This incident raised the question of why lisa should want to risk public failure and its consequences when she was always

uncertain of the accuracy of her contribution.

.

P . ' 4

A chain of comments reported by Lisa provided some insight into why she was likely to risk public failure. After trying repeatedly to contribute correct responses during a class discussion, lisa schieved her nope at a transition point in the lesson:

- 1: She asked me...I had put up my hand...she goes, "Yes?"...and I go, "Is a bridge parallel?" because I knew it was. I just wanted her to get my attention. So I said, "Is a bridge" parallel?" and she goes, "Yes it is."
- I: How did you feel then?
- L: I felt good. (Lesson 1)

Reasons for Lisa wanting to get the teacher's attention carr. the identified from this specific transcript item but, as contributed part of an overall pattern, she was probably engaging the of the designed to enhance the teacher's overall impression of the teacher's overall impression over a specific over other over

When liss believed she knew the answer to a line of the line of the she became and the became an

I telt mad, because I put up my hand for nearly every setting and she didn't pick meas (lesson 1)

- I: Now you have got your hand up.
- 1: I know that...it ain't fair. I felt sad in a way tocause I had my hand up first. I want to say something to her about that...I was getting man at her...She picks me about once a year maybe if I'm lucky.
- I: Well, once every couple of days anyway. (Lesson 7)

Lisa perceived a number of children were asked frequently, whereas she felt she was often ignored by the teacher. While these views were distortions of what really happened, they nonetheless portray the view held by one student. Interestingly, the students named by Lisa whom she believed were asked to respond to questions frequently were all successful students.

Lisa's thoughts and behavior regarding public participation in the class appeared at first to be confusing. However, a balance

3

3. A

of motives between seeking the teacher's attention and avoiding public failure seemed to suggest some kind of dilemma for Lisa. The desire to impress the teacher, whom Lisa perceived to be an influential is significant other, had to be "weighed" against the risk of sensitive embarrassment if she failed in front of her peers.

Perceptions of other students' behavior. Lisa viewed the behaviors of other students with mixed feelings. Where she found terself being helped in her work, either directly or indirectly, by " the efforts of other students her reactions were generally favorable:

- I: What were you thinking at that time!
- 1: Trying to think now to do that question. I didn't know new to do it and then when she 'another student up at the clackboard, did it, I remembered. (Lesson R.)

lisa frequently received help from Andrew, a student whom she perceived usually knew the answer. He was seated next to her and readily provided assistance when she required help. She often felt encouraged when she noticed that Andrew was experiencing difficulty or was progressing slowly with the same work that she was doing.

With some students Lisa experienced feelings of a different order. When a successful student when Lisa perceived to be coastful made an error at the blackboard, she reported feeling good because he had to be helped by the teacher while in front of the class. On another occasion Lisa commented that, when students she perceived to be successful went up to the blackboard and correctly demonstrated an idea which she was unable to answer, "I feel like I'm dumb and you are smart" (Lesson R). Once Lisa reported that she wished she could have been in the position of another student who was successfully capturing the limelight when out in front of the class.

Lisa was very aware of other students' performances in the unit test. The students who scored well had their marks announced putlicly. Lisa did not approve of the teacher's procedure of recognizing those students who achieved high marks. As she stated, "I knew I wasn't going to get my name called out" (Lesson 10) and, a little later noted that, "It gets you jealous." However, Lisa appeared to be aware of the students who scored low, noting that three or four other students scored lower than her mark. Because one student, whom she perceived normally achieved lower marks, gained a higher mark, Lisa speculated on whether or not he had cheated. She had noted this student had scored the same mark as a successful student who sat next to him. Lisa's behavior in relation to her perceivel students' performances in the test revealed something of the intensity of feeling and the inpact on her mind of what she deemed to be a significant failure.

While Lisa once reported that she was unconcerned with how other students were performing during the regular learning sessions, she nevertheless seemed very aware of how well her peers were achieving.

Lisa's Basic Ideas and Beliefs

Perceptions of her own mathematics ability. Lisa generally was concerned with her ability in mathematics. She seemed to believe that her lower performance was attributable to lack of ability, as inferred in a number of her comments.
I usually don't get it right 'referring to her response to a public question;. (Lesson 5) Last year I was in the low class (referring to the streaming of students for mathematics in Grade 5]. (Lesson 2)

!

I have never got a maths test back that I felt happy with. (Lesson 10)

I: How do you expect you will get on in the test?

L: Rotten!

I: What do you mean by rotten?

L: Hopeless!

I: Why do you think you won't do well?

L: I don't like it, and I'm not trying any more. (Lesson c)

lisa often disclosed. sither directly or indirectly, that she wanted to try but found fittle reward for her efforts in the onscing daily lessons.

Overall she perceived some improvement in her mathematics Pertormance since she first started school: "I get a C last year. That was herricle. On the first year I got a D^{*} (bessen 5). Consequently, Lisa was hoping for an even better grade this year.

I hope I get a better mark. If I do, I get a ouji game...I just love it. And my Mom goes, if I get a...some...no less than a "E", I get a ouji game. And they are expensive. They are about \$5. (Lesson 4)

However, these comments were expressed prior to the unit test. Because Lisa identified the unit test result with her report card, after the test she concluded:

I know I'm going to flunk this year. I am going to flunk social too. I know. I'm going to flunk. I told my Mom that too, and she goes, "You'd better not or you'll get a whipping." So I'm hoping. (Lesson 10)

Lisa appeared to be aware of her ability in relation ${f R}$ other students' performances. She often referred to situations where she perceived "everyone else knew the answer and I didn't" (Lesson R).

These perceptions had developed into a mind-set for Lisa. Whenever she thought of an answer which she perceived to be different to "everyone else," Lisa assumed she was wrong.

I was thinking how to do it because I thought there was only 12 out of the whole thing, and then everybody...that person came up with 14, Clare. And I couldn't think. I said, "Oh., I'm gone completely!" (Lesson R)

Watering successful students' performances in front of the class contributed to Lisa's basis for comparison between herself and other students.

•••

The transcripts contained many instances of list attempting to screen from other students the impression that she was a lower achieving student. She was certain that if other students pergeived her to te wrong they would laugh at her. Likewise, when she needed help from the teacher, Lisa reported a preference to go to the teacher rather than have her come to the desk:

- I: How do you feel when the teacher is with you like that [referring to desk supervision]]
- 1: Okay. I felt better because I knew I'd get it right... I prefer to go and find her, but she won't let me.
- I: Why do you prefer to get out of your seat?
- L: Because notody can hear what I say [to the teacher]. (Lesson 2)

Lisa seemed very concerned with concealing her perceived lack of atility in the learning of mathematics.

An aspect of Lisa's ability which appeared to have some significance in her self-performance was illustrated by her comment "I had forgotten already." On several occasions Lisa reported forgetting an idea or forgetting a word.

- L: I didn't know what that one word said, intersect_____ I still can't say it...And as soon as she came and told me I knew what it was.
- I: How did you feel then?
- L: I felt stupid with myself, because I should have known that word. We talked about it so much. (Lesson 1)

She even expressed surprise in lesson 9 when the unit test was being distributed for she claimed to have forgotten that as well. Given that this learning difficulty was recorded in Lisa's cumulative school record, the evidence from this study would confirm the influence upon her learning of the difficulty in immediately processing certain information.

Lisa's perception of her own ability reflected an acceptance of lower self-expectations, yet contained elements of hoping for more success. However, with her perceptions of little effort-outcome covariation, Lisa's hopes appeared to be forlorn and there were indications that this was recognized by Lisa.

Lisa's perceptions of the teacher's performance expectations c: nerself. 'Lisa often reported that she was concerned about the teacher's impressions of her. She did worry about what the teacher thought about her and this often resulted in her asking other students for help. Lisa endeavoured to avoid conveying the impression to the teacher that she was not succeeding because she had not listened or because she was "dumb."

- I: Why do you feel bad when you get it wrong? [Lisa had volunteered to respond to a public question.]
- L: I don't like to...It's just that I don't like her getting to know...so that, like, the teacher thinks I'm doing better. (Lesson 5)

Alternatively, Lisa engaged in behaviors calculated to convey

a good impression of her performance to the teacher. At times the teacher would have observed Lisa to be attending, when actually she reported that she was only pretending to attend. Lisa also adopted some attention-seeking behaviors and often was willing to risk public failure in an attempt to gain recognition from the teacher. Yet underlying these behaviors was Lisa's belief about how the teacher perceived her performance:

- I: Do you think the teacher knows how well you are doing in maths?
- L: Oh yes.
- I: And what do you think she thinks about you and maths?
- I: I'm horrible.
- I: Why do you think she thinks that about you in mathematics?
- L: 'Cause I know. I just think that. (Lesson 1)

Lisa's comments about her achievement in the unit test further exemplified her perceptions of the performance expectations held by the teacher toward her.

- I: What do you think the teacher thinks of you after that test?
- L: Rotten.
- I: Do you think she's surprised about you and maths?
- L: No.
- I: Do you think the teacher has the same idea about you and math as your ideas about you and math?
- L: I doubt it. I'm going to ask her what she thinks about me after school...I'm scared! (Lesson 10)

Perceptions of the teacher's performance expectations of her were not confined to success and failure behaviors. Lisa perceived the teacher differentiated in the selection of students for responding to public questions. She also reported a belief that she was "picked on" for misbehaviors when c^{-1} or students had often provoked her, but would not tell the teacher sing accused in error for fear of being accused of answering back. However, the success concerned by these incidents for she felt sure that the teacher remembered even small happenings some months later.

Lisa was sensitive to perceived differential teacher behavior, especially following the unit test: "Hers [Mary's] was the same thing as mine and she got it right and I got mine wrong. I think she favored her. I think she likes Mary" (Lesson 10). These perceptions of teacher bias in favor of other students probably served to accentuate Lisa's belief that the teacher held lower performance expectations for her.

<u>Ferceptions of tasks and activities</u>. While Lisa often found she was unable to understand the material to be learned, her reported comments did suggest that she felt more uncomfortable in the listening whole class situations than with the seatwork. This notion was supported by her response to the question, "What was the most important part of the lesson?":

- L: Doing the worksheet.
- I: Why do you find it most important?
- L: Because it's more fun, it's not important, and when she just talks she doesn't really give us a chance. (Lesson 7) Lisa claimed she preferred hard worksheets to those which were easy

or about 50/50 because they made her concentrate more.

These views seem somewhat difficult to understand when examined as specific elements but they reflected Lisa's outlook in relation to self-performance. Perhaps she believed that striving to succeed on difficult tasks would convey an impression to the teacher of high effort input--a quality which she perceived was valued by the teacher. The unrealistic expectation of success also seemed consistent with Lisa's hopes for a higher grade in mathematics. Such views would serve only to compound an already complex set of elements and relationships.

A Characterization of Lisa's Conception of Self-Performance

Lisa became the focus of this study because she was perceived to be a less successful student. The results of the attributional analysis of her behavior, together with some of the relevant findings of the quantitative data, have enabled an interpretation to be made of how Lisa might have perceived herself and her self-performance in the learning of mathematics. The major elements of this characterization of her conception of self-performance were:

- 1. Lisa's approach to learning can be described as learned " helpless. She perceived little effort-outcome covariation, believing that failure tended to be uncontrollable by any action on her part.
- Lisa perceived effort to be the criterion of success as espoused by the teacher, the home, and peers.
- 3. Lisa frequently perceived a task to be structured such that success or failure was likely to be controlled by external factors rather than by her own effort and ability.
- 4. Lisa found success to be satisfying and pleasurable. These feelings were the more intense since success for Lisa tended to be infrequent and minor in nature.
- 5. Lisa was concerned with most failures. Significant failure and public failure occurred frequently and aroused intense

feelings of anxiety and disappointment. Minor failures were annoying to Lisa and often frustrating if they robbed her of experiencing any kind of reward for success.

- b. Lisa attributed significant failure mainly to lack of ability and to some lack of effort. Minor failures were attributed mainly to effort.
- 7. Lisa was in a dilemma regarding public participation. Her first inclination was to avoid public failure situations, yet she seemed to want to contribute for purposes of impressing the teacher.
- 5. Lisa often perceived other students' success in front of the class as confirming her belief of lack of ability. She did not approve of the procedures followed by the teacher when giving back test results.
- Isa believed the teacher held lower performance expectations for her. While she expended some effort to enhance these expectations, she retained the belief that the teacher perceived her as an unsuccessful student.
- 10. Lisa hoped for a good mark on her report but perceived the goal to be virtually unobtainable. She seemed very aware of her lack of ability but sought to conceal this from herself, her peers, and the teacher during the daily learning sessions. By employing self-protective tendencies such as misattributing lack of success to lack of effort, mood, and the nature of the tasks rather than to lack of ability. Lisa was able to sustain a partially acceptable level of self-esteem and to retain the

socially approved expectation of better grades. At the same time she engaged in a series of behaviors designed to enhance the teacher's performance expectations held towards her. The teacher was perceived to be the main element in obleving the better grade which, in turn, would appease the threatened pressure from Lisa's home. The fragile behavior network which Lisa seemed to have established for coping with her failureprone performance was shattered by the reality of the unit test. Her perception of helplessness, her attribution to lack of anility and her intense negative emotions all resurfaced with the realization of her perceived failure.

Analysis of the Dyadic Interactions Hetween Lisa and the Teacher

The dyadic interactions between Lisa and the teacher work analyzed according to their overt and covert aspects.

Analysis of the Overt Behavior

The results of the analysis of the interaction sequences between Lisa and the teacher are presented in Table 23 and in Figures 11 and 12. Table 23 indicates that Lisa and the teacher interacted dyadically 54 times during 11 mathematics lessons. Of these interactions, 44 were student-initiated, of which 32 were private in nature and 12 were public. The student-initiated private interactions occupied a duration time of 15 minutes 33 seconds across the 11 lessons. Table 24 shows the frequency distribution of the duration of these private dyadic interactions. Table 23

Frequency Distribution of Dyadic Interactions Between Lisa and Teacher

•	:											
Type of						Lessons	ions					F
Interaction	ec.		5	~	t	5	\$	~	α	6	0,1	TPACT
Teacher initiated public interaction	\sim	~ 1	e	* *	t	• •	١		I	ı	I	5
Student initiated public interaction	C1	\sim	I	€ ∀	ı.	• •	4	3	ı	ł	• •	12
Teacher initiated private interaction	١	ı	I	- •	ı	ı	ı	ų	ı	r	ı	
Student initiated private interaction ^a	6	• •	\sim		t	5	ي م	I	Ś	(N	Ś	ĸ
Total:	2	t	9	6	t	~	6	m	m	8	t	ま

^aDuration of the student initiated private interactions totalled 15 minutes 33 seconds and ranged from 2 seconds to 2 minutes 20 seconds in length.

•



Process 0.58 - Volunteer :.00 Teacher 0.43 + iroduct 0.25 - Volunteer :.00 < Public 0.39 Public 0.39 Public 0.39 Volunteer 0.00 Volunteer 0.00	
Teacher 0.43 iroduct 0.25 Volunteer blic 0.39 Choice 0.17 Volunteer Question 0.63 Volunteer	<pre>/ + 0.13irocess / + 0.13irocess / + 0.13irocess / Negate</pre>
<pre>ublic 0.39</pre>	<pre>< *** 0.67 </pre> Affirm
<pre>blic 0.39</pre>	• :.00 . New Q.
Volunteer	Rel :.00 C Brief
Comment 0.17 Call 1.00	. Rel :.00 : Integ

24.

•

		TIL TO COL	ILEACT 101		
		Behavior	- 00 · : = ·		
	, /		++	0.02	
			II T	0.08 Ask	k q. 0.67
			0 /	0.02	ж.: Э
		Cont o j the	Afflrm	0.16 - Ackrow	C.09 KTION 0.18
		;	Negate	0.04 Com	lent
Frivate 0.61			Frocess	0.36 ASK	o
	ł		Give clue	• 0.16	
	A.		· New q.	Common Comm	lent
				ASK	64. 0.13 0.83
			р У	1	Comment 0.17
	Student 0.97		0		J.C
		Cont a. C.c.	Frocess		17. 14. 14. 14.
		4	Give Ans		
			Give Clue	0.29	• •
			Kep ss	0.06	1.00
	-	· iroc c. U.U9 -	i Tocess	:.00	
			1	Ū.: >	
		iroc q. i.20 +		0.25 Ask q.	с ч. 1.00
			/ New J.		Comment 1.00

Table 24

. •

.

Frequency Distribution of the Duration of Student-Initiated Private Dyadic Interactions Between Lisa and the Teacher

•

ration of Interaction in Minutes and Seconds	Frequency
0:01 - 0:10	·, ·,
0:11 - C:LV	Ł
C:.: - 0:30	•
0:31 - 0:40	3
0:41 - 0:50	
0:5: - ::00	:
1:01 - 1:10	-
1:11 - 1:20	:
1121 - 1130	-
1431 - 1440	,
1:41 - 1:50	-
<u>1151 - 2100</u>	:
2:01 - 2:10	-
ZI II - 2120	

244

Figure :: indicates that, of the teacher-initiated public interactions, Lisa volunte-red to respond to all question types. Most of the teacher questions were process type questions to which Lisa made incorrect responses. The teacher's reactions to these incorrect responses included negations, is and repeating of student statements. With toth product and process type teacher questions to which Lisa volunteered correct or partially correct responses, the teacher provided mainly process type feedback. Student-initiated public interactions were characterized by Lisa calling out relevant questions, to which the teacher provided mainly crief feedback.

The nature of the private dyadic interactions between line and the teacher is presented in Figure 12. Apart from one teacherinitiated contact directed at correcting Lisa's behavior, these interactions were all initiated by Lisa. Most of the dyadic contacts initiated by Lisa concerned the mathematical content of the learning. The teacher reacted with a variety of feedback behaviors, with offirmations, giving of clues, and, or rephrasing the question, and process explanations as the most frequently used. Where teacher feedback involved sustaining behaviors, Lisa's reactions varied with porrect statements, asking of additional questions, and the contributing of comments as the most frequently given responses.

Figure 1. also indicates that about 29 percent of the contacts with the teacher which were initiated by Lisa involved procedural matters. The teacher tended to react to these concerns with affirmations or process explanations. A feature of all the private dyadic interactions was the trend for the teacher to use more

criticism than praise with Lisa. Across the 11 lessons criticism was coded on eight occasions, whereas praise was coded once.

In summary, where Lisa volunteered incorrect responses in public interactions, the teacher's reactions tended to be brief and neutral both in terms of corrective and evaluative feedback. When Lisa volunteered correct or partially correct responses in public interactions, the teacher tended to affirm or provide process feedback. Private dyadic interactions were characterized by numerous sustained content information exchanges.

The overall Analysis of the Dyadic Interactions

Some of the dyadic interactions between liss and the teacher were analyzed by an overall examination of the observable data, the teacher's covert behavior, and the student's covert behavior. This analysis provided additional descriptive information about the quality of the interactions concerned.

During a whole class discussion, the teacher asked for examples of parallel lines in everyday life. The teacher was confronted with "a sea of hands" and, after hearing examples from some of the students, moved onto the negative section of the lesson. Lisa was disappointed that she had not been invited to contribute, so initiated a public interaction at a transition point in the lesson a short timelater.

- L: Mrs T____.
- T: Yes.
- L: Is a bridge an example of a parallel line?
- T: It probably is. (Lesson :)

As illustrated in the attributional analysis, Lisa's intention for persisting with the idea was, "I just wanted her to get my attention." The teacher's comments about the incident as reported in the postlesson interview included:

I probably should have pursued that with ner and asked her to draw a diagram to check out...to see what exactly she was thinking of. That is kind of interesting. I notice with some children you can kind of put them off and they eventually forget about what it is, but not Lisa. She'll...that's typical of her...she'll come back to keep after you till she gets her question answered.

The teacher's report contained little or no awareness of Lisa's need for attention, yet the persistent behavior apparently was not unexpected. However, the report did indicate some pumperent about hisa's thought processes. Given the teacher's performance expectations for float, including an awareness of her problem with lisk of immediate recall, the teacher's publicment could reflect a sensitive reaction to this aspect of Lisa's ability.

On another obcasion, Lisa had volunteered to demonstrate an idea. The teacher asked, "Does anyone think they could come out and show how the two triangles (on the overhead projector screen, might be "ongruent." Most students dismissed the possibility of the triangles being congruent from visual inspection, but Lisa wanted to attempt a demonstration of the idea. Her report of the interaction episode began with a feeling of delight at having been chosen by the teacher to go up to the front of the class. However, the delight quickly changed.

I: What were you doing there?

L: I was showing her on the screen what I meant. If you

- could take that corner to that corner.
 - I: I see.
 - L: And I didn't want to do it on the overnead so I just wanted to show her that way.
 - I: Why did you not want to do it on the overhead?
 - L: Because I didn't want anybody to see it because I knew I was wrong as soon as I done it.
 - I: Okay, so you had to go on with it. Did you mind that:
 - L: Well I was feeling upset because I had finally got up there, then I made a mistake, and it didn't look right and I just...I wanted to get back into my seat...I knew they [the other students] knew I was wrong and I was mad because I was wrong. (Lesson 2)

This traumatic episode for Lisa continued when the teacher used Lisa's error as a discussion teaching point in the lesson. As Lisa recalled, the discussion proved to be as encorrussing to her as when she was trying to demonstrate to the whole class the idea of congruency once she realized that it could not be done.

The teacher's views of the interaction episode were:

I had the impression at this point that liss knew she was wrong. Or that she knew it was not right because she giggled as she came up and she obviously heard the other children say "It can't be done." And yet I almost got the impression that she wanted to be the center of attention or almost playing the dumb role game. I don't know.

I thought it was good though. I thought the class in the end benefited from her mistake.

The teacher seemed to due accurately to lisa's desire for teing in the limelight, though probably perceived a different reason for Lisa wanting the attention. Whereas Lisa probably was endeavouring to enhance her public academic image, the teacher perceived Lisa to be staging some kind of show. The teacher definitely underestimated Lisa's embarrassment for public failure and her intense feelings of disappointment at herself. This was a reflection of Lisa's atility to conceal her feelings. The teacher instead capitalized on Lisa's error as an opportunity to reinforce teaching points and viewed the

episode from this perspective.

The two episodes described in the overall analysis up to this point demonstrate the potency of misperceptions. In the rapid interactions and exchanges of a normal classroom, the teacher seemed only to have time for a single interpretation of Lisa's behavior based on what she saw. This interpretation tended to be based on information the teacher possessed about Lisa and on her interpretations of Lisa's behavior from previous experiences.

The teacher seemed to be alert to Lisa's attempts to get attention. In a later episode involving a student-initiated private interaction, the teacher became annoyed:

It really did annoy me at the time and I think I spoke at her sharply. I said, "Come on, you have done questions like that before." It was one of the basics. She didn't know how to check for a flip or something or other...I thought, "Are you doing this deliberately to get some attention or...?" I could not believe that she'd forgotten something so basic. (Lesson c)

lisa claimed that she had forgotten:

Like I didn't know how to do it at that time, but I looked back, and then I did, and then I asked her, "I didn't know how to do it," and she goes, "Well, you have done it before."

Nevertheless, the teacher spent 50 seconds working through the exercise with Lisa. While she expressed satisfaction with the help she received, Lisa was sensitive to the criticism, as perceived in the teacher's use of voice, at the beginning of the interaction episode: "That's why I hate asking her. That's why you will see me turning around to Carol." These insights provided some explanations for why Lisa was observed to be interacting frequently during seatwork.

Lisa often initiated dyadic interactions with the tester because she needed help. The teacher's typical reaction to this need



was exemplified by the following verbal report:

I ended up going through a complete example with her, and watching her step by step and commenting as she went through it...I spend quite a bit of time with her. It seems to follow a pattern almost. I mean, almost every day I seem...she does have problems...I think by doing one with her like that and roing over it all again, it just sort of locks it all in.

The teacher often alluded to Lisa's learning problem of short memory recall. She also noted that often Lisa's difficulties were problems in reading directions. "I find if I just read the direction for her or go over with her, ask her to read it, that's half the problem" (Lesson i).

However, after viewing Lisa's behavior on the video recordings of the lessons, the teacher began to speculate on other reasons underlying Lisa's problems with the mathematics.

I'm thinking now that maybe part of Lisa's problem is that she doesn't exactly pay attention so closely in class. She's busy combing her hair. (Lesson R)

For this episode Lisa commented that when she did not understand she became bored. "I know I'm wrong so why pay attention?" and consequently combed her hair. Later, while viewing another lesson, the teacher commented:

I notice Lisa wandering around just after I explained it. You would think she would go ahead and work on another example... I'm wondering if part of her short memory recall isn't that she's not paying as close of attention or that her mind wanders while she's trying to learn.

These comments suggested the teacher was considering some possible modification of her thoughts about Lisa. Whereas she had formerly attributed Lisa's problem to a learning disability, the teacher now considered perceiving some of Lisa's problem to be a lack of effort.



The evidence presented indicated that the teacher held lower performance expectation for Lisa and that Lisa was aware of these expectations. The numerous public interactions between Lisa and the teacher reflected Lisa's desire to gain the teacher's attention, probably for the purpose of attempting to convey a favorable impression to her. By being seen to participate and willing to try, Lisa seemed to be responding to the socially approved effort ethic at the same time as maintaining a degree of self-esteem. However, the reality of public failure was nearly always present, hence a deep conflict existed for Lisa personally.

Private dyadic interactions between Lisa and the teacher also involved more than a content information exchange. The teacher provided a generous amount of assistance to Lisa which satisfied her greatly for she usually understood the task as a result of the help. While the help was a form of instructional compensation by the teacher, given the teacher's knowledge of Lisa's specific problem with learning, Lisa was nevertheless concerned about the image she was presenting to the teacher. As a consequence, she engaged in behaviors which were attempts to minimize the teacher's impression that she was not succeeding because of lack of ability or lack of effort.

Summary Analysis

The results of describing Lisa's behavior during the learning of mathematics have established some tentative relationships among and between Lisa's conception of self-performance, Lisa's causal perceptions of success and failure, Lisa's achievement-related

tenavior, the teacher's behavior towards Lisa, and the teacher's causal perception of Lisa's performance.

- Lisa wanted a good mark on the report form, preferably one which was better than that gained previously. There were indications of pressure from the home.
- 2. Lisa perceived that the espoused criterion for success was effort.
- 3. Despite expenditure of effort, Lisa perceived little or no effort-outcome covariation.
- A deeper belief held by Lisa was that the cause of failure
 was perceived to be lack of ability.
- 5. List perceived that she experienced failures of many forms. They occurred frequently and some were deemed significant because of their assumed importance for her learning progress and because of their perceived influence on peers.
- b. Lisa often expressed negative emotions, some of which included intense feelings of anxiety and disappointment.
- 7. Lisa was observed as a student who was mainly on-task. During seatwork, she was often observed interacting with other students.
- Lisa often found she could not understand much of the material to be learned.
- 9. Whenever whole class discussion sessions were prolonged, Lisa tended to engage in discreet distracted behaviors.
- 10. Lisa endeavoured to enhance her public academic image especially for the benefit of the teacher. However, she had

to balance this desire against the risk of embarrassment of public failure.

- 11. Lisa was able to maintain some degree of self-esteem for the benefit of herself, her teacher, her mother, and her peers, by engaging in self-protective behaviors which included misattributing her performance to lack of effort, mood, and/or external factors rather than to lack of ability.
- 12. List perceived the teacher held lower performance expectations for her and this would be realized in a lower mark on her report card.
- 13. Lisa was very concerned about what the teacher thought of ther. She engaged in various behaviors in an attempt to enhance the teacher's impressions of her.
- 14. Lisa engaged in various attention-seeking tehaviors, the intentions of which were sometimes misperceived by the teacher.
- 15. The teacher held lower performance expectations for Lisa. Incorporated within these expectations was an awareness of Lisa's problem with short memory recall.
- 1c. The teacher engaged in compensatory behaviors with Lisa especially when she needed help with her seatwork.
- 17. Occasionally, under the stress of normal classroom functioning, Lisa perceived elements of teacher criticism and annoyance toward her which she interpreted negatively. Consequently, Lisa often preferred to seek help from classroom friends, hence the frequent interacting with other students

during seatwork. At times Lisa was the target for corrective behavior during these interactions with other students, which Lisa again interpreted negatively. Lisa's reasons for avoiding asking the teacher for help pertained to the preserving of a more favorable impression.

- 18. The teacher began to consider modifying her attribution of disa's problem to lack of effort as much as to lack of ability, based on presumed lack of attention tehaviors by Lisa.
- 19. Lisa was an unsuccessful student because her approach to learning tended to be characterized by learned helplessness.

GREG

Hiveraphical Information

Greg was the second of two boys in his family which lived in their home in the neighbourhood of the school. He had always lived in Edmonton, mainly on the south side of the city.

Greg was aged 11 years and stated that his main interest was the family dog, an 11 month old German Shepherd. He had begun a rock collecting hobby following a recent visit to the site of petrified wood and bone fossils. Greg reported watching a considerable amount of television, with "Happy Days" and "Charlie's Angels" among his favorite programs. The family had travelled to San Francisco and to the British Columbia coast, as well as throughout Alberta.

While he commented that he fulfilled no duties around the

school, Greg nevertheless joined in school-organized extra-curricular activities. He felt the Grade 5 year had been a particularly good one for him as he believed he had learnt well, mainly through the help of a good teacher. His best subjects were mathematics, physical education, art, and language arts. He especially liked to write stories. Greg had two close friends in the class. He enjoyed their triendship because they treated him well in last year's class. Greg's response to the question, "Why do you come to school!" was, "To learn because parents don't know it all."

The teacher viewed Greg as a student with a new problems largely because his ability was not as great as the other students. She believed he knew he was not as bright as the others and perceived that Greg played on this somewhat:

He almost seems to play that role. I think he finds comfort in it, in a way. Probably in other years he's gotten away with not ioing as much work or working as hard because of this. I almost get the feeling he's using it, and I don't know exactly where to draw the line.

Some conflicting information was found in the formal descriptive data. In support of the teacher's comments, Greg achieved in mathematics a percentile score of 9 on a school district wide assessment. This indicated that Greg was a lower achieving student in mathematics. Meanwhile, Greg's total expectat: score of 147 on the forced choice, self-rating FAPS placed Greg among the students in his class who had the highest academic self-expectations. His expectations in mathematics sub-score of 24 out of a possible 28 was close to the class mean. Examination of his responses indicated that Greg expected to learn everything that would be taught, that he expected to be better than most kids and that he expected to enjoy math a great deal in the coming year. His self-rating of expectations for mathematics in the long term was consistent with these results. Greg's results on the PAPS were treated with caution given the reality of the other information, yet they raised immediately interesting questions as to why he should rate himself in this manner.

Greg's self-reporting of internal responsibility for success or failure on the IAR revealed that out of 34 situations he attributed 20 to internal causes and 8 to external causes. Because his internal responsibility for success score of 10 was higher than his internal responsibility for failure score of 10, it can be concluded that Greg tended to assume credit for causing good things to happen but was disinclined to accept the blame for unpleasant consequences or circumstances. Examination of his responses suggested that differential causal behavior toward success and failure largely occurred in situations comparing his input of effort with the teacher's explanations, the giving of help, and the teacher's lovel of concern. In success situations, Greg claimed internal responsibility but with failure situations he attributed blame to the external causes, namely, the teacher and those responsible for his learning development. In situations comparing his effort or ability with perceptions of tasks, other people's moods and views, and others' abilities, Greg claimed internal responsibility for both success and failure.

quantitative Analysis of Hehavior

Greg's behavior during time 10 lessons on mathematics was analyzed quantitatively according to both overt and covert aspects.

Overt Behavior

The results of the analysis of Greg's overt tehavior are presented in Tables 25, 20, and 27. Table 25 shows that the majority of Greg's overt behavior was coded as occurring during the lesson situations of teacher-led whole class instruction and, in seatwork, these situations accounting for approximately 47 percent and 45 percent, respectively, of the learning time. A high percentage of Greg's behavior in each lesson situation was observed to be attendingdistening. As Table 26 indicates, attending-listening behaviors accounted for about 92 percent of Greg's time during teacher-led whole class instruction, about 95 percent of student demonstration whole class instruction, and about 75 percent of seatwork. As a percentage mean of learning time across all learning situations, Table 27 shows that attending-listening behaviors were observed of percent of all behaviors during the 10 mathematics lessons.

Most of the remaining time Greg was coded as not attendinglistening. Table 27 indicates that across all lesson situations for all lessons the percentage mean for Greg's not attending-listening behaviors was about 11 percent. Tables 25 and 26 show that not attending-listening behaviors were more likely to occur during seatwork where Greg was coded to be not attending-listening on the average of about 15 percent of the time. Not attending-listening - 57

5
\sim
a
d.
~
F-

•

Percentage Distributions of Student Overt Behaviors in Each Lesson Situation Category for Each Lesson : Greg

Categories of Student Overt Behaviors Within Legan Stimations					Lessons	suc			 		l ercentage Mean	Within- Category Sum of
		\sim	3	ŧ	5	<i>ω</i>	~	ω	6			Percentage Means
<u>Teacher-Led Whole</u> Class Instruction			►						•			
Attending-listening Not attending-listening Interacting with other	37.5	59.0 3.3	3.0	t. 9t	30.8 1.9	24.5 3.9	46.0 6.0	51.2	†	00	43.7 3.4	
students Out of desk	1 1	9.1	1 1	/ I / I	8 I f 1	1 1	r , 1 1	1 1		¢ 1	0.2	
Student Demonstration Whole Class Instruction				· · -								
Attending-listening Not attending-listening Interacting with other	45.8	9.1	6.6	} }	5.1	 1 1		44	4 P	1 1	5.0 5.t	
students Out of dock	!		1	1	 t I		1	I J) 	+ 	1	8.2
Seaturnk				 ! !	1	1	1	1	 	 #	 1 1	
Attending-listening Not attending-listening Interacting with other	8.5. <u>9</u>	3.3	48.2	39.3	48.1		26.0 2.0	6.4	n. 4.		\$ 0. 9.9	
students Out of desk Supervised by teacher		6.1.1			6. C	; ; 5, 6 ;		1 I I 1 I I			- 10 A - 20	6.1
										ſ		

-

•

258

.

• . V

Table 26

ς. .

,

Percentage Distributions of Means for Student West Behaviors Within Each Category of Lesson Situations : Greg

Teacher-led whole class Attending-listening Instruction Not attending-listening Interacting with other students Out of dosk	92.4
	7.2 0.4
Student demonstration Attending-listening whole class instruction f. Not attending-listening function functing with other students Out of desk	94-9 5+:
Seatwork Attending-listening Not attending-listening Interacting with other students Out of desk Supervised by teacher	75.7 74.7 2.4 6.2 0.9

\$ 174 4

259

.

•

æ

*

Table 27

V N

٠

Ч., н С., н

Precentage Distributions of Student Overt Behaviors for Total Lesson Time : Greg

	Y

		ł			Les	Lessons					Ponontado
uategories of Student Overt Behaviors		~	3	オ	5	9	- ~ -	ω	6	10	Mean
Attending-listening	9:.7	9:.7 86.9	81.5	85.7	8: ·5 85.7 2 84.6 75.5	75.5	88.0	75.6 81.3	81.3	100	85.3
Not attending-listening	4.2	6.6	6.9	7.1	7.1 9.6	20.8	0 0	24.4	3.4:	1	:0.5
Interacting with other students	4.2	6.6	1	1	6.	:	1	5 1	1	1	·
e. Out of desk	1	,) ,)	6.9	7.1	3.9	6	1.0	l ł	2	1	2.8
Supervised by teacher	1	ł		1	i I	6.	;	1	•	1	4.0
•											

٠

1

260

•

. •

.

behaviors during teacher-led whole class instruction and student demonstration whole class instruction lesson situations were coded about 7 percent and 5 percent, respectively, of the time. The student overt behavior categories of interacting with other students, being out of desk, or being supervized by the teacher, were coded as occurring infrequently across all lesson situations.

In quantitative terms, Greg was observed to be on-task much of the time during most lessons, especially in the teacher-led and student demonstration whole class situations. He was most likely to be not attending-listening during seatwork. Tables 25 and 27 show that lessons 6 and 9 were days when Greg was not attending or listening about 15 percent of the time or more. Lesson 5 was a day when he was coded as not attending-listening during teacher-led whole class instruction if percent of the time. The daily variations in his overt behavior can only be explained from an analysis of dreg's covert behavior.

Covert Behavior

Table 28 presents the results of the analysis of Greg's interactive thoughts across the 10 mathematics lessons as determined by the use of GASSIT. Most of the interactive thoughts reported by Greg were self-performance thoughts and behavioral moves--self, which accounted for about 25 percent and 22 percent, respectively, of all thoughts. Approximately another 14 percent of Greg's thoughts were categorized as thinking processes for learning mathematics. Behavioral moves--student, self-performance--feelings, feelings, and non-task-related were categories of thoughts reported on an average

Table 28

ŧ

1

Percentage Distributions of Student Thoughts Over CASSIT Categories

Grag
••
Lesson
Each
$f \circ \mathbf{r}$

40

•

•

				,	Lessons						Fer-
Categories of Student									ľ		Centage
Covert Behavior	• •	2	3	1. 1	5	9	~	ත්	6	0	Mean
		-	((c	c t	č		L C			ע ר
Subject Matter	7.4	9.t	α. Γ	5.2	0.7.	1.7	i 1		1	1	
Cognitive Processes	7.4	24 . 1	X .2	:5.2	14.0	:6.2	6	:3.8	9.5	2.7	:3.9
Behavioral MovesSelf	14.8	24.1	33-34	9.6:	2:.:2		22.7	:0.3	28.6	30.8	21.9
Behavioral MovesStudent	7.4	6.9	11.	4.3	:0.5	10.8	:5.9	1	:4.3	:9.2	10.0
Behavioral MovesTeacher	7.4	-	2.8	4.3	5.3	3	ó. 8	3.5	ł	1 2	3. 8
Self-PerformanceThoughts	18.5	27.6		37.0	28.:	32.4	25.0	27.6	19.0	23.1	24.9
Self-PerformanceFeelings	1		1 •	13.0	2.0	:3.5	6.9	(•0;	4.8	19.2	8.5
Feelings	18.5	3.4	5.8 5	4.3	ŝ	2.7	4.5	:7.2	.6.	1	7.4
Non-Task-Related	18.5		î.	ar ar	5.3	1 1	9.	13.8		}	6.0

262

5

A...

.

٠

•••

1 7

of 5 to 10 percent of all thoughts. Relatively few thoughts in the categories of subject matter and behavioral moves--teacher were reported.

The interactive thoughts in six categories were further analyzed according to sub-categories of what the thoughts were about. They were behavioral moves--self, behavioral moves--student, behavioral moves--teacher, self-performance--thoughts, selfperformance--feelings, and feelings. Table 29 presents the results of this more intensive analysis.

Behavioral moves--self. As the observes, about 67 percent of Greg's behavioral moves--self thous the rained to attendinglistening. This could appear to support the findings in the overt analysis, as another 18 percent of the thoughts in this category reference to motives for attending. Only a small percentage of thoughts concerned the sub-categories of motives to avoid attending and avoiding or seeking to participate publicly.*

Behavioral moves--students Greg seemed to be aware of what other students were doing and hav well they were performing during the learning of mathematics. Table 29 shows that about 46 percent of Greg's behavioral moves--student thoughts were focused upon other students' successes and failures, while another 40 percent monitored aspects of their behavior. Relatively few thoughts were reported which were concerned with what other students were thinking. Thoughts expressed about interacting with other students accounted for only 9 percent of all thoughts in this category. .

Table 29

14

e)

5 m

· ~~

r g.

.

Within-Category Percentage Distributions of Sub-categories of Selected Student Thought Categories of CASSIT : Greg

.

Sub-categories of Selected CASSIT Categories of Student Thoughts	Within-Category ^P ercentage Distributions
ehavioral MovesSelf	,
Seek public participation	
Avoid public participation	2.63 .
	5.26
Attending-listening	67.11
Motive to attend	18.42
Motive to avoid attending	6.58
ehavioral MovesStudent	
Perceptions of other students' performances	45.71
Inferences of other students' thoughts	5.71
Interactions	8.57
Perceptions of other students' vehavior	40.00
ehavioral MovesTeacher	
Perceptions of teacher's instructional moves	13.33
Perceptions of teacher's reactions	13.33
Interactions	13.33
Perceptions of teacher's behavior	60.00
elf-EmformanceThoughts	
Self-assessmentsuccess	15.22
Self-assessmentfailure	21.74
Parceptions of task difficulty	19.57
Perceptions of task structure	
Self-attributions	20.65
	5.43
Self to spectations	7.61
Significant other	9.78
elf-PerformanceFeelings	
Anxiety	53.33
Morally neutralpositive	30.00
Morally neutralnegative:	16.67
eelings	
Positive /	45.45
Negative	54.55
	رر ۰ بر

264

Behavioral moves-teacher. Of the few behavioral moves-teacher thoughts reported by Greg, Table 29 reveals that most referred to his awareness of what the teacher was doing, these accounting for 60 percent of the thoughts in the category. Greg reported only a small number of thoughts about the teacher's instructional moves, her reactions to happenings in the class, and about his interactions with the teacher.

-4

Self-performance-thoughts. Greg's reported thoughts pertaining to his successes and failures accounted for about '37 percent of all his self-performance thoughts, with the failures being considered more frequently than the successes. Table 29 also indicates that another 40 percent of self-performance thoughts related to perceptions of the tesks in terms of their difficulty and their structure. Thoughts involving significant others accounted for a further 10 percent of the thoughts in this category, while selfexpectations and self-attributions were infrequently reported.

<u>Self-performance</u> <u>wellings</u>. Thought units coded in this category accounted for $z \rightarrow percent$ of all Greg's interactive thoughts. As Table 29 reveals, 53 percent of the emotions expressed involved anxiety, ranging from feeling scared to worry. Another 30 percent of Greg's self-performance feelings were morally neutral and positive in nature, while the balance of 17 percent were morally neutral and negative. No morally unneutral emotions were reported by Greg.

Feelings. Of the feelings which Greg expressed about classroom matters and happenings in general, more were negative in nature than positive. Table 29 indicates that feelings such as anger, embarrassment, boredom, anxiety, and dislike were expressed 55 percent of all feelings, while pleasure and amusement accounted for the balance of 45 percent.

Summary of Quantitative Analysis of Behavior

Greg was observed to be on-task for most of the time, cespecially during whole class discussion lesson situations. Some not attending-listening behaviors were recorded during seatwork. The majority of Greg's reported interactive thoughts focused upon the self, namely, his own performance, his behavior, and his feelings about his performance. While his behavior thoughts reflected a trend for sttending-listening, Greg's self-performance thoughts and feelings consisted of failure and negative tendencies, respectively. The attening-listening characteristic of Greg's thoughts were supported by the approximately 14 percent of all thoughts which were categorized as cognitive processes. The nature of the relationship between higher effort input on the one hand and lower outcome on the other may well be discerned from an attributional analysis of Greg's covert behavior.

Attributional Analysis of Self-Performance Covert Behavior

A significant proportion of Greg's mental activity during the learning of mathematics pertained to components of self-performance thoughts and feelings. These aspects of covert behavior contained sets of causal relationships and a base of underlying ideas, views, beliefs, emotions, and lines of reasoning.

Causal Explanations of Greg's Behavior

The transcripts of the interviews with lineg contained several chains of comments. These were collated int live clusters: approach to learning, perceptions of task, specific performance, public participation, and perceptions of other students' performances. Each cluster of Greg's chains of comments is described, with examples and anecdotes included.

(,

Approach to learning. When challenged with new or difficult material, Greg's behavior tended to follow a pattern:

I: How did you find this worksheet? G: Well, it was kind of hard in fact. I: How were you starting to feel then? G: Well, I was kind of mad but I kept on trying to make it work. Did **f**ou finally get it? 1: G: Oh no...I gave up finally. I: Do you usually like to give up when a thing gets tough. for you like that? Sometimes I give up and sometimes I don't. G: I: Why do you think you choose to give up? G: Well, when I feel like I can't do it and I try and try over and again, I finally give up because I'm mad at it. I: Why do you think you can't do that sort of thing? G: Maybe I don't try hard and I didn't try different things. (Lesson 5)

This example was typical of Greg's reported covert behavior when confronted with tasks which he could not handle. His comments indicated he expended much effort but when beaten he tended to give up, attributing this failure to insufficient effort. However, other situations of a similar nature suggested that a perceived lack of ability was recognized by Greg. On one occasion Greg reported becoming mad with a worksheet and began to worry about his performance. He attributed his failure in this instance to lack of competency.

Only when a classmate went to the teacher for help did Greg choose to do likewise. As a result of her verbal comments to him Greg felt a great deal better.

- I: In what way were you helped?
- G: Well she told me what my problem was.
- I: What was your problem?
- G: Like, my hand was shaky all the time. And the picture would slip sometimes.
- I: Okay, and how did that help you when you got back to your desk?
- G: Well, so that next time I could try to not make my hand shake a little bit. (Lesson 4)

While jreg reported still being mad with the worksheet, he no finger worried about his performance. In other words, the problem of failure remained but the perceived undesirable consequences had dissipated because of some kind of assurance Greg perceived he had received from the teacher. This interaction will be fully discussed in the overall analysis of dyadic interactions be creg and the teacher.

The examples described seemed to end the Greg's approach to learning. He frequently had to contend with failure situations in ". which he perceived little or no effort outcome covariation. There were indications that he attributed his failure to lack of ability and lack of effort, and experienced a variety of negative emotions accordingly. Behaviors and performances of these kinds would appear to be symptomatic, at least to a partial degree, of a learned helplessness approach to learning. However, at times, Greg showed signs of motivated concern toward his self-performance.

Following the announcement of his test result, Greg was attending closely to the teacher's comments. While his result was one of the lowest the class, he nevertheless was positively ಸರಚ
motivated.

I: When she first began speaking, what were you thinking?
G: Well, I was thinking about things I got wrong, and Iwas looking at them...I wanted to see what I did wrong. (Lesson 10)

Greg reported he followed the teacher closely throughout this extended discussion session.

Overall, while Greg displayed certain signs of learned helplessness, there were other indications which precluded totally categorizing his approach to learning in this way.

<u>Task perceptions</u>. Greg's verbal comments about his learning of mathematics were characterized by numerous references to uncertainty toward understanding the subject matter and to task difficulty. His cognitions were sometimes accompanied by an expression of an emotion and some causal explanation.

I: What were you thinking?

G: I was thinking about the answer to her question and I think I said "Yes." Well, I muttered it to myself.
I: You never put up your hand though.
G: No.

- I: Why not?
- G: Well, I wasn't quite sure so I was a bit scared.
- I: In what way were you not quite sure?
- G: well, like this was the first time I had ever worked on this. (Lesson 2)

In this situation, Greg's response to the teacher's question. would have been incorrect. Uncertainty about his understanding was accompanied by some anxiety and was attributed to newness of the . subject matter. Greg often claimed that he was reluctant to participate publicly in the discussion because "it was my first time doing this stuff" (Lesson 3).



Greg frequently reported instances of not fully attending, especially during the whole class situations. While he sustained attention for periods of time, he often commented that his mind began to wander once the sessions became prolonged: "I was kind of tir and I wasn't too quite sure of the answer. . . I was kind of with my ruler" (Lesson 5). On different occasions Greg reportioning distracted by what other students were doing, by daydreaming in oy concerns beyond the classroom. Even during the test he found himself distracted by what others were doing. The evidence suggest that Greg tended to be distracted when he did not have a clear perception of the task.

At times Greg perceived initially that a task was easy but found it harder than he had anticipated: - G: Well, like, the first question I did. I had to do it over again, 'cause I didn't get it exactly right. I: Who decided for you to do it over again?

G: I did. I was thinking I should erase it and do it over again.

I: -Why did you want to erase it and do it over again?

G: Well, so I could get it right. (Lesson 2)

In addition, Greg pointed out that he wanted the teacher to see that his work was right when he handed it in for marking.

On a few occasions, Greg's initial perception of an easy task

proved to be well founded:

G: Well it [the worksheet] looked kind of easy.
I: How did you find it?
J: Well, it was okay.
I: How do you feel you have got on with the worksheet?
G: Pretty good.
I: Why do you think you have done pretty good?
G: Well I was going kind of pretty fast. Like, I followed her...I knew what to do because I was listening. (Lesson 3)

C.)

١

Greg attributed his clear perception of task to effort during the whole class instruction lesson situation.

His initial perception of the unit test relieved him of some deep anxieties. Prior to the distribution of the test materials, Greg reported: "I was scared and I was hoping she would have forgotton about the test...I don't like tests all that much because I'm scared that they will be hard" (Lesson 9). Greg associated hard questions with a poor performance. However, when he first looked over the test, Greg realized that it was as easy as he had hoped and expected little or no trouble. At the end of the test Greg felt "pretty good" because he believed he had nearly all of it correct. "I will tell her [my Mom] that I had my maths test today and that it wasn't too hard" (Lesson 9). These perceptions of the unit test by Greg are interesting, given that he scored 43 percent--the second lowest mark of all the students in the class.

<u>Specific performances</u>. Greg often made reference to specific successes and failures which tended to be accompanied by the expression of an emotion. While working on a worksheet the teacher had publicly <u>asked</u> him a question.

I: What were your thoughts when you heard your name called?

- G: I was kind of scared because I didn't know what she wanted me for because I didn't think it was about the maths sheet.I: What went through your mind at that time?
- G: Well, I was thinking about the answer I drew. Like the cross...I felt good because she accepted my answer...I was right and she agreed.
- I: Now just after she had finished talking with you, can you remember what you were thinking and feeling at that time?
- G: I was feeling good because it had made me feel better. And I was right. (Lesson 7)

•

On another occasion, when the teacher was returning work-

Sheets, Greg felt anxious while waiting for his paper.

G:	I was kind of shaky. Like, if I did good or bad.
I:	What went through your mind when you actually had the
	paper in your hand?
G:	I was kind of happy with the work I did, and some things.
I:	Why was that?
G:	Well I did kind of good in mathon the sheet.
ĹΙ:	How well did you do? Well, on the first question I didn't do very good, and the
G:	Well, on the first question I dian't do very good, and the
	last one. The rest I did pretty good.
I:	Did you think that was about as good as you could do at
	this time?
G:	Well, no! I thought I could do better.
l:	Why do you think you didn't do as well as you could have
	done
G:	Maybe I didn't try too hard or I wasn't paying attention
	mayber (Lesson 5)

This example typified a number of chains of comments, particularly in terms of the doubtful attribution of failure to effort. As the transcript sample also indicates, Greg perceived success when he judged he had obtained as many correct responses as needed to allay the fear of failure. His early concern before receiving back the worksheet suggested that the possibility of failure definitely existed in his mind. Therefore, Greg's threshold for success or failure seemed closely linked to his feelings toward self-performance.

These notions were brought sharply to focus with the results of the unit test. Prior to the announcement of the test results, Greg reported:

- G: I wasn't feeling too good.
- I: Why not?
- G: Well, I was kind of scared I might get the lowest mark maybe. (Lesson 10)

Again Greg considered the possibility of perceived failure. He was very concerned about the teacher's reactions and his Mom's reactions if he gained the lowest mark or a bad mark.

- I: Greg, how did you get on in the test?
- G: I had 43 percent.
- I: And were you surprised at the result?
- G: Yes a bit.
- I: Why were you surprised?
- G: Well I did good sort of . Like I didn't get all ...why, I got some right but sometimes I got half of them wrong, right.
- I: Why do you think you got about half and half?
- G: Well like, maybe I was paying attention a bit on math and thinking of different things like the diorama...I also didn't study for it.
- I: How do you feel about the result generally?
- G: Not too bad. (Lesson 10)

Greg's attribution of his performance to effort or lack of effort again seemed doubtful. Also, he reported that he viewed his test result as a partial success.

- I: What were you thinking when you first looked at your score?
- G: Well, I looked at the 43 percent and then I looked at the blackboard to see how high it was from the top. I didn't feel too bad., (Lesson 10)

While Greg's remarks probably reflect a hidden concern about his performance, he seemed relieved that he had not gained the lowest mark in the class. This may well have proved to be the basis of his partial success.

Overall, Greg experienced few real successes but frequently faced the prospect of perceived failure. However, by accepting a lower threshold level for differentiating between success and failure, which seemed closely linked to feeling good or feeling bad, he seemed able to cope with his all-round lower level of achievement. Greg reported attributing his lower performance level to lack of effort, but these causal explanations, by the manner in which they were expressed, tended to be unconvincing, suggesting that Greg might have perceived other factors to be responsible.

<u>Public participation</u>. Greg preferred not to participate publicly. On several occasions he reported he was unwilling to demonstrate an idea in front of the class:

G: I kind of get shy when I'm up there in front of the classroom.

- I: Why do you become shy?
- G: Well, like, everyone's looking...watching you. That makes me kind of feel funny. (Lesson 5)

Greg indicated in an earlier lesson that he preferred not to be the "irst one up at the front of the class to demonstrate an idea but, if the teacher had called on him, he would have gone through with the demonstration. In some instances his main reluctance to participate seemed to focus on his feeling of perceived uncertainty toward the subject matter because this was his first exposure to motion geometry. More likely, however, Greg preferred not to volunteer to participate in order not to place himself in a situation of potential public failure. When commenting about a practical joke that was played on a boy who was demonstrating an idea in front of the class, Greg stated: "I wouldn't feel all that good if I was him, because I might think that they were laughing at my answer...that I was getting it wrong" (Lesson 1). Greg noted that he would prefer the other students to laugh at something he was wearing than laugh at the quality of the ideas he was demonstrating.

Greg tended to volunteer to contribute an idea during whole class discussion only when he was absolutely certain of the answer. However, Greg's hand raising was characterized either by putting his hand half up or by putting his hand up a little later than others,



usually when he observed the teacher was on the point of selecting from other volunteers.

- I: You put your hand up there. You put it up a little later than others. Why was that? G: Well. maybe 'cause they they be a little later
 - Well, maybe 'cause they thought of the answer before I did. (Lesson ?)

Greg reported not being disappointed when another student was chosen and felt that his slowness at putting up his hand and his desk placement near the back of the room contributed to this non-selection by the teacher. He believed the teacher selected students to respond on the basis of first hands up and sometimes because she noted that a student had not answered for some time. Greg commented that the teacher did not seem to pick on some students more than others.

Overall, Greg appeared reluctant to volunteer to demonstrate ideas in front of the class or contribute ideas toward the class discussions. While he frequently claimed he was unsure of whether he would be correct or not, largely because of unfamiliarity with the subject matter, he also seemed to engage in behaviors designed to avoid participation. His expected feelings of nervousness and shyness when in front of the class were emotions which he perceived might contribute negatively to his public image.

<u>Perceptions of other students' behavior</u>. Whenever other students were demonstrating ideas in front of the class Greg tended to have little or no desire to take their place, but he did use their ideas and answers as a basis for comparison with his own.

- I: Now, Lisa goes up and works on this. What were you thinking and feeling then?
- G: That's when she put the mark with those squares and then

275

Ł

she did it from the bottom to the top of the other one. And that's when I was sure it was wrong. You couldn't do it. Then I was quite sure I was right. (Lesson 2)

Where Greg perceived success when another student was unsuccessful he tended to have good feelings. However, he felt differently when a student whom Greg perceived to be less successful at maths succeeded where he failed.

- G: Well, I was...like he had his all right...like all of them and I didn't have all of them right. So it didn't make me feel too good.
- I: Would you like to explain this feeling that you didn't feel too good?
- G: Well, I felt kind of...well I didn't feel good because like he got his right and maybe I didn't trace mine that good. Maybe I was going too quickly in tracing. (Lesson 6)

In the pre-test interview Greg did report that he was

concerned about how other students would perform in the test. "Well, I mostly think how good they are" (Lesson 8). During the test Greg became aware of other students finishing early.

- I: What thoughts did you have at that time?
- G: Well, I was wondering if they went fast or if they...if I went slow. (Lesson 9)

Greg had expected some of those who finished to be the first through the test, but some who finished early did surprise him. He was able to learn from the manner in which the teacher announced the results how others performed on the test.

Greg's Basic Ideas and Beliefs

<u>Self-expectations</u>. Greg seemed to hold generally lower self-expectations for his performances. At times these selfexpectations seemed to be a function of his perceptions of the tasks such as exemplified in a pre-lesson interview.

- I: How do you think you will get on in maths today?
- G: Fairly good. It's not too hard.
- I: Why do you think you'll get on pretty good? *
- G: Well, because I'm here [had been absent earlier in the week] and she explains it well:
- I: What is there about today's maths that makes you think you will do fairly well?
- G: Well, from the other work, it is kind of easy. (Lesson 1)

As the pre-test interview showed, terms used by Greg such as "fairly

good" or "pretty good" referred to about a 50 percent pass rate.

- I: How do you expect to get on in the test?
- G: Maybe good.
- I: What do you mean by maybe good?
- G: I would say maybe about half and half.
- I: Why do you think you are going to get about half and half?
- G: Well, 'cause there are some hard questions like ...
- I: How do you know they are hard?
- G: Well from the worksheets maybe, because that's where it's all coming from. (Lesson 9)

Greg's disinclination to participate publicly also underlined the expectations he held for himself. He did not want other students 'to know his level of performance or ability and preferred to use avoidance processes where possible. When Greg needed help from the teacher he reported:

- G: I like to wait when there is hardly anyone there, so that I'll feel better.
- I: Why does it make you feel better?
- G: Well like, kids...they might laugh at the question I have to ask for. (Lesson 1)

Greg's uncertainty about many of his responses to the questions asked of the class by the teacher probably added to his belief of lower expectations.

A clear indication of Greg's self-expectations was revealed in his comments when awaiting the handing back of his unit test results. "I was kind of scared I might get the lowest mark, maybe . . . I wasn't thinking I would get the lowest mark but maybe a little close to it" (Lesson 10). With his test result, Greg commented, "Well, I would say it's about what I usually get."

The direct evidence and interpretations of a variety of his comments tend to support the conclusion that Greg held lower performance expectations for himself in the learning of mathematics. While he tended to report that he attributed his lower performance to lack of effort, the evidence would suggest other causal factors were perceived to be responsible. Greg often commented that he tried very hard but still failed.

<u>Greg's perceptions of the teacher's performance expectations</u> of himself. Greg often expressed concern about what the teacher might be thinking of him. He tended to believe she perceived him to be "in-between bad and good in maths" (Lesson 4) which he considered to be a fair reflection of his performance. When he provided the correct response to a public dyadic interaction between the teacher and himself, Greg felt particularly good because he believed the incident would help her to think he was doing well. "I think it might give me a better mark on my report card" (Lesson 7). This incident typified Greg's desire for the teacher to notice that his work was correct.

His thoughts around the unit test situation provided insights into what he perceived the teacher's impression of him was like.

- I: What sort of thoughts do you think she is going to have about you and motion geometry after tomorrow's test?
 - G: Maybe good...She might want me to go more than half and half...maybe all right or a little more than a half. (Lesson 8)

These thoughts reflected the reality of the teacher's expectations matching his own, yet Greg also implied that she would like him to

achieve better. On the day of the test Greg felt he had managed to succeed with most of the test; yet, as he handed it in: "Well, I was kind of scared...I was kind of worrying what she would think...me done that fast" (Lesson 9). After the test was marked, Greg reported:

G: She might think I haven't been trying too hard or , something, maybe.

I:

ţ

Σ.

I: Detrou think she would have been surprised about the mark" G: No: (Lesson 10)

Greg commented that the teacher's views on his for the in mathematics were about the same as the ideas he here bout his performance.

Greg seemed to be of the opinion that the teacher would ascribe her lower performance expectations for him to a lack of effort. Earlier in the lessons Greg had commented that if he had answered incorrectly one of the teacher's questions:

- G: I wouldn't feel too good.
- I: For what reasons?
- G: Well, like, I wouldn't have the answer and she might think I was fooling around, maybe. (Lesson 7)

Overall, Greg perceived the teacher held lower performance expectations for him and that these were possibly based on an attribution of lack of his effort.

<u>Perceptions of other students' performances</u>. Greg seemed to perceive clearly the achievement hierarchy of students in the classroom.

- I: Why do you think they did better?
- G: Well there is lots of smart kids in the classroom.
- I: Do you know who the smart kids are?
- G: Some of them.
- I: How do you know they are smart?
- G: Well, like they are good in other things, like maybe reading, language arts...they answer a lot of questions

- and stuff. They get a lot right. I: Do you know who the not so smart kids are in the room"
- G: Yes.
- I: Why do you think they are not so smart?
- G: Well, maybe they don't get all the questions right or maybe or sometimes they don't pay attention very much...they didn't try too hard. (Lesson 5)

Greg had noticed that the students who were good at mathematics tended to answer more questions and especially the difficult questions. While he did not expect to score the top mark in the test, he did guess successfully the two likely candidates from among the class. Again, Greg perceived that less successful students were those who expended insufficient effort.

Perceptions of tasks and activities. Greg reported activity sessions such as completing worksheets to be the most important part of the lessons in mathematics. As he explained, "When we do our worksheets then I know what is going to be coming up ahead." He preferred worksheets which were somewhere in between hard and easy.

- I: Why do you prefer those particular ones?
- G: Well, so the easy ones could help you understand it and the harder ones make you think a lot more. (Lesson 8)

His rationale for this view was that it was best if a person did not have to think hard all the time nor have to not think at all throughout the maths lessons.

In terms of the total unit, Greg felt that he would remember the worksheet activities more than anything else that happened during the 10 lessons. But the most important aspect of the whole unit on motion geometry for Greg was the unit test. Marks were given for the tests which he realized counted toward his report card.

280 -

On several occasions during the interviews Greg made reference to the report card. When he was worried about his performance on a worksheet he commented:

- G: I was worried, like...I was thinking that I would get lots of them wrong.
- I: Why would that worry you?
- G: Well...like, when the test came, when I do the test, I wouldn't do too good on it.
- I: Why would you not like to do badly on a test?
- G: So I could get a better mark on my report card. (Lesson 4)

Greg reported in the pre-test interview that he did care about how well he did on the test because of what it meant for his report card. He even wished he could get more correct than he expected for this reason. On the day of the test Greg felt he concentrated more than he usually did for a worksheet in order "to get better marks on my report card." The motivating effect of the unit test and its implications for the report_card seemed to be prominent throughout the entire mathematics unit.

A Characterization of Greg's Conception of Self-Performance

Greg was selected for study because he was judged to be a less successful student. A characterization of his conception of selfperformance was interpreted from the available evidence. The major elements of this characterization were:

1. Greg's approach to learning consisted of several elements of learned helplessness. He perceived little effort-outcome covariation, believing that failure tended to be controlled by external factors. However, on a few occasions, he displayed elements of a positive, functional approach to 281

۰. .

learning, but these occasions were usually short-term in duration.

- 2. Greg's perception of tasks influenced his achievement-related behavior. Where he perceived the task as easy and to be structured as a skill task, Greg attributed his successes and failures to the amount of effort expended. But, when perception of task was not clear, he attributed the frequent failures to a considerable lack of effort. Extreme lack of effort, in Greg's terms, appeared to be synonymous with lack of ability.
- 3. Greg found success to be satisfying and pleasurable, especially since he experienced it infrequently.
- 4. Greg experienced intense negative feelings of anxiety and anger with significant failures.
- 5. Greg's threshold for perceived success and failure seemed to be a function of his feelings about his performance rather than a function of any predetermined level of achievement. If his low achieving performance was perceived by Greg to be better than expected, better than some feared level of achievement, or better than some unwished for class ranking, he felt good and viewed it as a relative or partial success. The converse appeared to apply when a low achieving performance was viewed as a relative or partial failure.
- Greg sought to avoid participating publicly. He seemed to adopt avoidance processes designed to prevent other students from learning of his level of performance.

8

282

)

- 7. Greg was aware of other students' performances which he used as a basis for comparing his own performance.
- 8. Greg held lower performance self-expectations.
- 9. Greg believed the teacher held lower performance expectations for him. He was concerned about her impressions of him.
- 10. Greg perceived clearly the classroom achievement hierarchy in relation to himself. This served to reinforce his lower self-expectations.
- 11. Greg was concerned about his report card. Because he held lower self-expectations and because he perceived the teacher held lower performance expectations of him, Greg was worried as to how he could achieve the good mark on his report. He was afraid that the teacher might perceive him as one who exerted little effort, as revealed by his performances on the worksheets and the unit test. Greg believed that effort expenditure was the causal criterion of success. His anxiety in the learning situation appeared to be centered on this dilemma -- he wanted to present a favorable image to the teacher of his work input and yet he could not achieve the results to support the effort exerted. Consequently, Greg sought to conceal his real level of performance from the teacher, from his peers, and even from himself. He preserved a degree of self-esteem by adjusting his threshold of perceived success as well as engaging in avoidance processes with other people in the chassroom. These face-saving approaches enabled him to mask his real level of performance and to often view hise

results through positive eyes. Wherever possible he sought to avoid diminishing the teacher's impression of him.

Analysis of the Dyadic Interactions Between Greg and the Teacher

The dyadic interactions between Greg and the teacher were analyzed according to their overt and covert aspects.

Analysis of the Overt Behavior

The results of the analysis of the interaction sequences between Greg and the teacher are reported in Table 30 and in Figures 13 and 14. Table 30 shows that Greg and the teacher interacted dyadically 13 times during the 10 mathematics lessons. Only two of these were public in nature, both teacher-initiated. Most of the interactions occurred at the private level, five being teacherinitiated and six being student-initiated. Duration of the private interactions totalled 2 minutes 43 meconds, comprized of 1 minute 5 meconds for teacher-initiated interactions and 1 minute 38 meconds for student-initiated. On three of the days there were no dyadic interactions of any kind between Greg and the teacher.

Figure 13 indicates that with the two public teacher-initiated interactions, Greg was selected as a non-volunteer to respond to one product and one choice question. His answers were correct, to which the teacher provided feedback that was neutral but extended in nature.

The nature of private interactions between Greg and the teacher is presented in Figure 14. Two of the five teacher-initiated interactions were related to behavior and involved issuing of warning

Table 30

Ç

Frequency Distribution of Dyadic Interactions Between Greg and Teacher

.

•

Type of					Les	Lessons					, F
Interaction ·		- 4	~ ~ ~	- t	ź	9	6	ω	6	0.	lotal
Teacher initiated public interaction	ı	I	ı	I		,		ı	I	I	F .)
Student initiated public interaction	I	I	I	ı	,	I	I	1	J	I	I
Teacher initiated private interaction ^a	• •	- 1	r	I	24	۰ .	I	I	I	I	5
Student initiated ^b private interaction ^b	• •	1 	3	• •	I	١	- 1	,	~	I	Q
Total:	. .	• •			3		~	1	5	ŀ	::

285

•

,

v

.



Figure 13. Nature of Public Dyadic Interactions Between Greg and Teacher and Their Proportions

.

				Student	+ +
Type of Interaction	Initiated By	Interaction Related to	on Teacher A Reaction		Lon
		Behavior	0.40	0. 33 No Error 1.00 0.67 No Error 1.00	
	Teacher 0.45	Mork	0.60 Coserved	0.33 0.17 0.33 + + + No Response	1.00 se 1.00
Frivate 0.85		Content com 0.17	0.17 Frocess	. 00	
u	Student 0.55	Content Q.	0.66 Affirm 0.66 Of Process 0.66 Of Office O	0.12 0.12 0.38	0.33 0.33
		Proc Q.	0.17 <> Affirm Frocess	0.50 Ask Q. 0.50	1.00
					•
				C	į



and criticism by the teacher. The work-related teacher-initiated interactions involved brief periods of observations of Greg's work by the teacher during desk supervision. Two of these interactions were recorded as observations only, while the third interaction involved a verbal exchange of content information. Most of the dyadic contacts initiated by Greg concerned the mathematical content of the learning. These interactions tended to be short term, for which the teacher provided a variety of feedback behaviors such as affirmation, negation, process explanations and giving of clues.

In summary, Greg's interactions with the teacher were largely private in nature and tended to be brief. Aside from three interactions, the dyadic contacts were of 10 seconds duration or less, involving brief exchanges of communication. Apart from the two behavior incidents, the teacher's reactions were neutral. On workrelated interactions she tended to affirm, provide process feedback, or give clues.

The Overall Analysis of the Dyadic Interactions

Some of the dyadic interactions were analyzed by an overall examination of the observable data, the teacher's covert behavior and the student covert behavior. With Greg, the very small number of interactions meant that only a sample of behaviors could be described. Nevertheless, the items provide additional relevant data in the relationship between Greg's conception of self-performance and the teacher's thoughts about him.

During lesson 1 Greg initiated an interaction with the teacher

explaining that he did not understand the worksheet task. Greg reported that he had not wanted to go to the teacher because of what she might think about him, yet finally he felt compelled to handle the task in this way. He approached her near the blackboard and when the interaction commenced they were alone. Greg felt fine about the teacher's explanation until another student, whom he perceived to be smart at mathematics and confident in manner, approached waiting to talk with the teacher. From that point on Greg wanted to sit down, to escape from conveying an impression to the peer that he needed any kind of help. He was particularly troubled when the teacher, perceiving him to be now confused and interpreting this as still not understanding, began to reteach the ideas on the blackboard. This served to embarrass Greg even further so he withdrew prematurely from the interaction. As Greg confided in the post-lesson interview, he still did not understand the worksheet task.

r

Meanwhile the teacher had been puzzled by Greg's behavior: I did not know if I was not making my questions clear, but it seemed every question I asked him he came up with something totally off course.

I said, "Now what were we talking about today? What has the lesson been on?" and he didn't even...he said "seg...ments." And he was even stumbling on the words, and I thought, "My goodness! What's gone wrong here? He hasn't been paying attention or something." (Lesson 1)

The teacher's first reaction was to check her own explanation, hence she made recourse to the blackboard. When she noted that Greg finally knew what the lesson had been about, she asked:

"What is your worksheet on?" "Oh yes, rays," and he turned and left and of course we hadn't even talked about the question that was actually bothering him in the first place. (Lesson 1)

The teacher had often noticed this aspect of Greg's behavior, "He thinks he understands things too quickly!" Greg's explanation, involving the desire to withdraw from the presence of the other student, was not perceived by the teacher. Her inclination was to attribute the cause of the behavior to lack of effort during the earlier listening session. Begause she had to deal with another student, she soon forgot the incident. Greg, on the other hand, had retreated feeling bad about the situation. He suspected the teacher was more aware of his lower level of performance, he perceived he had failed to save face in front of the peer, and he still had not resolved the original difficulty.

On another opcasion, Greg approached the teacher and initiated the interaction with the comment, "Look what I've got--it looks horrible!" (Lesson 4) Despite frequent attempts and much effort he had trouble with using the Plexiglas to good effect. He now sought help. Greg felt satisfied with the teacher's comments about trying not to be so shaky and left the interaction no longer feeling worried. The teacher perceived the cause of Greg's problem to be a lack of good coordination for this was similar to his difficulty with handwriting:

I didn't want to knock him down really and sort of say, "That's terrible," or something. So I said, "It's a bit shaky," or something like that.

This indicates Greg's attitude though. He believes himself to be dumb. You know, he thinks he is not smart. And he is usually the first one to admit it. And I think he is believing he is not smart and he doesn't even attempt some things. (Lesson 4)

Analysis of this interaction episode raised a number of

290

underlying explanations of behavior. While the teacher was sensitive to Greg's situation and acted accordingly, her general views about Greg's overall performance seemed to conflict with those of Greg. She perceived that Greg believed he was lacking in ability but she tended to infer that greater effort by Greg would overcome the problems he was experiencing. Greg recognized the high value placed on effort by the teacher.

In the interaction episode, Greg felt he had expended considerable effort for no return. He perceived the cause of the problem was either a lack of ability or some external factor such as the nature of the task or teacher explanation. Greg's problem was one of compromising this realization with the expected effort-oriented perspective of the teacher. Yet following the interaction Greg reported feeling better even though he was still mad with the task. Apparently Greg had interpreted the teacher's feedback comment to him as not criticizing him for lack of effort. While Greg might have pondered as to why he could not do the task, he felt the teacher's impression of him had not been **dimin**ished by this perceived failure experience. He perceived the teacher's performance expectations for him had been maintained.

A third dyadic interaction between Greg and the teacher further underlined the teacher's views about Greg. Greg was busy working on a worksheet with the class functioning quietly when he inadvertently knocked a ruler to the floor. The teacher immediately asked Greg a question. I: You were looking around the room when he dropped something. Was that the trigger for you to ask him that question?

Tr: No, I think it was a combination of two things. I hadn't spoken to him yet at that class and also I felt he was not paying attention. I thought this might jar him back into...But he obviously had been doing something because he had an answer. (Lesson 7)

Greg became a little scared when he heard his name because he was uncertain as to the reason for the interaction. However, this was an interaction episode which proved to be a successful experience for Greg. He felt that the teacher's impression of him would be enhanced and that ultimately, "I think it might give me a better mark on my report card." Greg also believed that if he had failed she would attribute that to lack of effort. Because in this incident the teacher had considered that Greg's input of effort might have been questionable, the participants in the dyadic interaction apparently were thinking along similar channels.

Overall, the evidence suggests that the teacher held lower performance expectations for Greg and that these were based on a perceived lack of effort. Greg seemed aware of these expectations. His problem was to convince the teacher he was trying hard which he perceived to be the basis of her expectations for him. From the few dyadic interactions which were amenable to detailed analysis, Greg seemed very conscious of the kind of impression the teacher might have of him and attempted to present as positive an image as possible.

Summary Analysis

The results of describing Greg's behavior during the learning of mathematics have revealed some possible relationships among and . . .

between Greg's conceptions of self-performance, Greg's causal perceptions of success and failure, Greg's achievement-related behavior, the teacher's behavior towards Greg, and the teacher's causal perception of Greg's performance.

- 1. Greg wanted to 'gain a good mark on the report card.
- 2. Greg perceived that success in achieving this goal was dependent on the expenditure of effort.
- 3. Greg recognized that the teacher valued highly the exertion of effort.
- 4. Greg exerted considerable effort for little or no outcome return.
- 5. Greg perceived the teacher held lower performance expectations for him.
- 5. Greg perceived the teacher based these lower performance expectations on a lack of effort.
- 7. Greg was viewed as a student who was mainly on-task.
- 8. The teacher perceived Greg held lower performance selfexpectations which he attributed to lack of ability.
- 9. The teacher held lower performance expectations for Greg based on attribution to lack of effort.
- 10. The teacher appeared to "communicate" these expectations to Greg through differential behaviors and response opportunities for different levels of students.
- 11. Greg appeared to engage in behaviors which were designed to convince the teacher he was expending considerable effort and that his lack of success was attributable to other factors.

12. Greg was a less successful student because in many respects he was learned helpless in his approach to learning.

A Summary Comparison of Lisa and Greg

This chapter has described the achievement-related behavior of two "less successful" students, Lisa and Greg. Both studies were observed mainly to be on-task most of the time, especially during teacher-led and student demonstration whole class instruction? Behaviors other than attending-listening were more likely to be ded during seatwork. In these situations Lisa was often observed to be interacting with other students, while Greg's behavior was often coded as not attending-listening. In quantitative terms, the interactive covert behavior of the two students was similar in pattern except for one difference. Greg's thoughts seemed to be more task-omiented than Lisa's, whereas Lisa's thoughts were slightly more self-oriented than Greg's.

Lisa and Greg both desired a good grade on their report card. They seemed to believe that the teacher--the person who distributed the report grades--espoused effort as the criterion of success. Consequently, Lisa and Greg exerted much effort yet experienced little effort-outcome covariation. This often resulted in intense negative feelings and causal perceptions of failures to factors other than effort. Both students displayed tendencies of a learned helpless approach to learning. By perceiving tasks to be structured such that success on them would be determined by factors external to their control, Lisa's and Greg's self-conceptions of performance were characterized by failure perspectives and negative consequences.

Overall, Lisa and Greg differed only slightly in how they coped with these undesirable self-conceptions of performance. Both seemed aware that the teacher held lower performance expectations toward them and both believed the teacher based these expectations on lack of effort. As a consequence, Lisa and Greg engaged in behaviors designed to protect their levels of self-esteem, such as presenting favorable work-oriented images to the teacher. In addition, Lisa often misattributed her performance to factors other than lack of ability as a means of attempting to conceal her failure. Greg chose to imply at least partial success whenever possible as his attempted means of concealing failure. Both students frequently engaged in avoidance behaviors if they considered their public image to be at risk. 295

ŕ

Chapter VII

SUMMARY, CONCLUSIONS, IMPLICATIONS, AND RECOMMENDATIONS FOR FURTHER RESHARCH

Summary of the Study

Purpose

The purpose of the study was to seek some understanding of the causal perceptions and behaviors of four Grade 6 students and their teacher as they relate to the expectancy effect in the classroom. The study focused on questions pertaining to the characteristics of student behavior during the learning of mathematics, the conceptions of self-performance which students appear to develop in achievementrelated situations and, from an attributional perspective, the phenomena which characterize teacher expectancy effects. In addition, the study also sought to examine the effectiveness of stimulated recall as a research tool when used in naturalistic settings.

Methodology

Ten consecutive mathematics lessons were videotaped in a Grade 6 classroom for use with follow-up stimulated recall interviews. Four target students reported their interactive thoughts and described their thoughts and feelings pertaining to self-performance in mathematics. The teacher also reported her interactive thoughts. Some pre-performance and post-performance interviews were also

Overt process data were also obtained from the videotapes. Students' overt behaviors and dyadic interaction behaviors between the teacher and each student were coded.

The research design was intended to yield data which provided insights into the attributional behavior of classroom participants and possibly contribute to an explanation of the expectancy phenomena.

Analysis Techniques

The analysis of the data proceeded in three main phases. Student behavior, both in terms of its overt and covert aspects, was coded according to category systems developed for the research by inductive means. The category system for analyzing student overt behavior and the content analysis system of student interactive thoughts facilitated the quantification of student behavior.

Student self-performance covert behavior was analyzed from an attributional perspective. Causal explanations of student behavior were described according to identified chains of comment. As well, the characteristics of a student's underlying ideas, views, beliefs, emotions, and lines of reasoning were described. From this attributional analysis, a characterization of each student's conception of self-performance was discerned.

Dyadic interactions between each of the students and the teacher were analyzed at the observable level by using a revision of the Expanded Brophy-Good Teacher-Pupil Dyadic Interaction Classroom Observation System. These data, together with the verbal reports of their covert behavior from each of the interaction participants, facilitated an overall analysis of the interactions. Some expectancy

phenomena were revealed in this analysis.

A summary analysis for each student was developed to indicate the kinds of relationships found to exist between and among aspects of student achievement-related behavior and teacher behavior.

١

Descriptions of the Case Studies

A collation of the summary analyses for each student is shown in Table 34. This is intended to provide nume bases for the drawing of conclusions in the next section. When presented as significant features of the case studies in relation to the major elements of the analyses undertaken in the study, comparisons between the four students could be made. However, discussions of the perceived comparisons between Tina, Mike, Lisa, and Greg are treated fully in the presentation of evidence leading to the drawing of some of the conclusions and will not be attempted from the information contained in Table 31.

Conclusions

Because the study was limited to the investigation of a small number of student conceptions of self-performance as they relate to the expectancy effect in one Grade 6 classroom, it is not possible to generalize beyond the limits inherent in the study. However, certain conclusions can be reached concerning student bermwior in the classroom, the causal explanation of that behavior, and how the explanations of behavior seem to relate to the prevailing performance expectations held by the teacher and the student. The conclusions are presented in the form of answers to the research questions with

	Types of Data	Analyzed	Overt Beliavlor	Interactive Ma Covert th Behavior mo	Causal Explanations as to peepeepeepeepeepeepeepeepeepeepeepeepee
õ		Tina	80% attending-listening. Balancemainly not attending-listening, especially during teacher- led and student demonstration whole class instruction.	Mainly self-performance thoughts and behavioral movreself. Characterized ty positive self-oriented elements.	Mastery-oriented approach to learning; significant success and minor failure attributed to effort; postive affective consequences; often tasks perceived to be easy, perceived to be easy, perceived to be easy, mend public partici- pation for face-saving reasons.
Talle 31 Collation of Summary Analyses of the Case Studies	Case 2	Mike	89% attending-listening. Balance - madnly not attending-listening, especially during teacher- led and student demonstration whole class instruction.	Mainly self-performance thoughts. Gharacterized by task-criented elements.	Mastery-ortented approach to learning: significant success attributed to ability, minor fallure to effort; often tasks perceived to be easy, because review; usually unemotional; unconcerned tegarding public participation.
of the Case Studies	Case Studies	Llca	MGK attending-listening. Falance-rainly inter- acting with other students, especially during seatwork.	Mainly self-performance thoughts and behavioral movesself. Characterized by negative self-oriented elements.	Learned telpless approach to learning; minor success and failure attributed to effort, frequent signifi- cant failure to ability; often intenso negative emotions; unclear perceptions of task perceptions of task face-saving reasons.
1 Sec.		Greg	85% attending-listening. Balancemainly not attending-listening. especially during seatwork.	Mainly self-performance thoughts and behavioral movesself. Characterized by some task-oriented elements and negative personal elements.	Tendency toward a learned helpless approach to learning; mincr success and failure attrituted to effort, froquent signifi- cant failure probably to ability or external factors; negative affective consequences; unclear perceptions of tasks; avoided public participation for face- saving reasons.

. 299

continued)
્છ
31 (
Table

Types of Data		Case S	Case Studles	
Analyzed	Tina	Mike	L18a	gerg
Baal Under.ying Covert Behavlor	Effort the criterion of success; higher perform- arce self-expectations; aware that teacher held higher performance expectations for her hased mainly on effort attribution; aware of classroom achlevement hierarchy; report card grade deemed significant.	Higher performance self- expectations based on ability; aware that teacher held higher performance expectations for him based mainly on ability; aware of class- room achievement hierarchy; report card grade deemed significant.	Lower performance self- expectations based largely on ability; recognized effort as the criterion of success but little effort-outcome covariation; perceived that wher held lower perf : mance expectations for her based mainly on effort; report card grade deemed significant.	Lower performance self- expectations based probably on ability; perceived that teacher held lower performance expectations for him based mainly on effort; the espoused criterion of success but little effort- outcome covariation; report card grade deemed significant.
Teacher- Student Dyadic Interaction Behavior	Sought to enhance or maintain the teacher's effort-based higher performance expectations in order to achieve good report and protect self- esteem.	Expected and achieved success including a good report; his successful performance sustained the teacher's higher expectations of him.	Sought to gain the teacher's attention to give impression of effort input; attempted to conceal failure-prome performance; tried to .rotect self-esteem.	Warted teacher to perceive him as an effort-oriented student; attempted to conceal failure-prone performance; tried to protect self-esteem.

which the study was concerned. A further conclusion concerning differential teacher behaviors for different student performance expectations held by the teacher is also reached.

1. What are the characteristics of students' behavior during a lesson?

a) Categories of student overt behavior

Examination of the video recordings of the Grade 6 mathematics lessons revealed initially a variety of student overt behaviors. Some of these pertained to different postures, varying degrees of attention, varying kinds of distractions, the kinds of ongoing interactions, novements around the room, and so on. Some overt behaviors reflected on-task tehavior while others reflected off-task behavior. Some overt behaviors tended to be reported in a descriptive manner because of observer uncertainty as to whether or not they reflected on-task or off-task behavior. As a consequence, student overt behavior was categorized as attending-listening, not attendinglistening, interacting with other students, out of desk, and supervised by the teacher.

Therefore, it can be concluded that several categories of student overt behavior could be discerned.

b) Frequency of student overt behaviors

All students were observed to be mainly on-task. Mike was recorded to be attending-listening 88 percent of the time, Lisa and Greg 85 percent of the time, and Tina 80 percent of the time. For the balance of the time, most students were observed to be off-task. Tina was not attending-listening for 14 percent of the time, Greg 10 percent of the time, Mike 11 percent of the time, and Lisa 3 percent of the time. Lisa was observed to be interacting with other students for about 7 percent of all learning time.

Therefore, it can be concluded that a major proportion of the student vert behaviors observed were attending-listening behaviors, a sma portion were not attending-listening behaviors, and other overt behaviors occurred infrequently.

c) <u>Relationships between student overt</u> <u>behaviors and lesson situations</u>

Three proad categories of lesson situations were identified: teacher-led whole class instruction, student demonstration whole class instruction, and seatwork. In all lesson situations the students were observed to be mainly attending-listening. However, differences between the successful students and the less successful students were discerned. Successful students were more likely to engage in not attending-listening overt behaviors during teacher-led and student demonstration whole class instruction. Within this trend the proportions for not attending-listening were higher with Tina and Mike during student demonstration whole class instruction than during teacher-led whole class instruction. Less successful students were more likely to be engaged in behaviors other than attending-listening during seatwork. For Lisa, these other behaviors tended to be interacting with other students, whereas Greg tended to be not attending-listening.

Therefore, it can be concluded that while attending-listening behaviors were prominent in all lesson situations, successful students tended to engage in off-task behaviors more during whole class instruction, whereas less successful students tended to engage in behaviors other than attending-listening during seatwork.

d) Categories of student covert behavior

Examination of the transcripts of the stimulated recall interviews with the students revealed initially a variety of covert behavior which included interactive and non-interactive aspects. Only the interactive thoughts were categorized. A diversity of interactive thoughts were coded according to the broad trends of self-oriented thoughts and feelings, learning of mathematics thoughts, other person-oriented thoughts, feelings in general, and non-task-related thoughts. The nine main categories of student interactive thoughts were subject matter, cognitive processes, behavioral moves--self, behavioral moves--student, behavioral moves--teacher, self-performance --thoughts, self-performance--feelings, feelings, and non-task-: ated.

Therefore, it can be concluded that several categories student covert behavior could be discerned.

Within a number of categories of student interactive thoughts, sub-categories were identified to facilitate a more intensive analysis of those thought units relevant to the intentions and directions of the research. The sub-categories tended to separate the objects of, the intentions of, and the nature of the thoughts and feelings pertinent to each category. Those categories which were subcategorized were behavioral moves--self, behavioral moves--student, behavioral moves--teacher, self-performance--thoughts, selfperformance--feelings, and feelings. Therefore, it can be concluded that several sub-categories of selected categories of student interactive thoughts could be discerned.

e) Frequency of student covert behaviors

All students reported self-performance--thoughts most frequently. These ranged from Greg's 25 percent of all thoughts to Mike's 31 percent. Behavioral moves--self thoughts were the second most frequently reported interactive thoughts by all students, ranging from Mike's 14 percent of all thoughts to Lisa's 23 percent. The third most frequently reported category reported by the students was behavioral moves--student, which ranged from Lisa's 9 percent of all thoughts to Tina's 14 percent. Cognitive processes category of thoughts was reported 14 percent and 12 percent of all thoughts, respectively, by Greg and Mike but only 7 percent and 6 percent, respectively, by Lisa and Tina. Other categories of thoughts which tended to be reported 4 to 10 percent of all thoughts by all students were self-performance--feelings, feelings, behavioral moves--teacher, and non-task-related. Subject matter thoughts were the least frequently reported thoughts, ranging from Mike's 6 percent of all thoughts to Tina's 2 percent.

The self-oriented categories of thoughts, namely, selfperformance--thoughts, behavioral moves--self, and self-performance-feelings, accounted for most of all students' thoughts, ranging from Mike's 48 percent to Lisa's 58 percent.

Therefore, it can be concluded that the most frequently reported categories of student interactive thoughts were 304

1.
self-performance--thoughts and behavioral moves--self.

f) Student self-performance covert behavior

The student self-performance--thoughts category was subdivided according to aspects of self-assessment, perception of tasks, self-expectations, and significant others. The most frequently reported interactive thoughts in this category were perceptions of task difficulty and perceptions of task structure, which accounted for 29, 40, 50, and 53 percent, respectively, for Lisa, Greg, Tina, and Mike. Self-assessments of success and of failure thoughts were reported 24, 28, 37, and 40 percent, respectively, of all selfperformance thoughts by Mike, Tina, Greg, and Lisa. Self-expectations and self-attributions thoughts ranged from Greg's 13 percent of all self-performance thoughts to Mike's 18 percent. Apart from Lisa's 17 percent of all self-performance thoughts, the sub-category of significant others was reported 10 percent or less by the other three students.

Therefore, it can be concluded that a large proportion of self-performance--thoughts which students reported were perceptions of tasks while a smaller proportion were self-assessments of success and failure.

Self-performance covert behavior also included selfperformance feelings. These were reported infrequently by students, ranging from Mike's 3 percent of all interactive thoughts to Lisa's and Greg's 8.5 percent each.

Expressions of anxiety were the most frequently reported emotions, especially by Lisa and Greg. Morally neutral feelings were reported by all students less frequently, while only Tina reported feelings which were sub-categorized as morally unneutral.

Therefore, it can be concluded that a large proportion of self-performance--feelings which students reported were expressions of anxiety, while morally neutral emotions were less frequently expressed.

g) Comparisons between the self-performance covert behavior of high achievement/high motivation students and low achievement/low motivation students

Several comparisons between the self-performance covert behavior of successful students and less successful students can be drawn. Of all self-performance thoughts, perceptions of tasks were reported by the successful students, Mike and Tina, 53 percent and 50 percent, respectively. These percentages of thoughts were higher than Mike's and Tina's 24 percent and 28 percent, respectively, of thoughts reported as self-assessments of success and failure. The less successful students partly reversed this trend in their reported self-performance thoughts. Greg's and Lisa's reported perceptions of tasks accounted for 40 percent and 29 percent, respectively, of all their self-performance thoughts. These were close to or lower thar. Greg's and Lisa's self-assessments of success and failure thoughts which were reported as 37 percent and 40 percent, respectively, of all thoughts in this category.

Tina and Mike both reported a higher percentage of selfassessment--success thoughts than self-assessment--failure thoughts. Greg and Lisa's self-assessment thoughts reversed this trend with 306

more self-assessments of failure being reported than self-assessments of success.

Other trends in the comparison of the self-performance covert behavior between successful and less successful students were discerned. Mike and Tina, as successful students, reported thoughts referring to self-expectations of 15 percent and 12 percent, respectively, of all self-performance thoughts, but infrequently referred to self-attributions and significant others. Lisa and Greg, as less successful students, reported thoughts referring to significant others of 17 percent and 10 percent, respectively, of all self-performance thoughts, but infrequently referred to selfattributions or self-expectations.

More self-performance feelings were expressed by less successful students than successful students. Both Lisa's and Greg's reported expressions of self-performance emotions accounted for 9 percent of all their interactive thoughts, whereas Tina's and Mike's expressions of self-performance emotions accounted for 4 percent and 2 percent, respectively. Anxiety was the self-performance feeling most frequently reported by Greg and Lisa, with higher proportions of morally neutral negative feelings being experienced as well. Mike and Tina reported very few expressions of anxiety. Their reported selfperformance feelings referred to mainly morally neutral emotions.

Therefore, it can be concluded that the successful students, compared with the less successful students, tended to report more perceptions of tasks, more self-assessments of success, and more references to self-expectations; whereas the less successful students,

compared with the successful students, tended to report more selfassessments of failure, more references to significant others, and more expressions of self-performance feelings, especially anxiety.

2. What are the conceptions of self-performance that students appear to develop in classroom achievement situations?

a) The ideas, beliefs, views, emotions, lines of reasoning of a student's conception of self-performance

From the interviews with the students, the researcher endeavoured to gain an impression of the conceptions of selfperformance held by each student. The major components of these conceptions were found to refer to several common ideas, beliefs, views, emotions, and lines of reasoning. One obvious component centered on performance, with notions of success and failure and causes of success and failure. Effort was perceived by all students to be the criterion of success, while ability was perceived as a supportive reason. Other causes reported included mood, nature of the task, and teacher's explanation.

Feelings were often expressed by the students in relation to performance. A common emotion reported was anxiety, ranging in degree from a slight concern through worry to fear and despair. Morally neutral emotions such as pleasure, satisfaction, disappointment and boredom, and morally unneutral emotions such as pride, envy and shame were also expressed with references to varying degrees of intensity.

Student perceptions of the tasks occurred frequently. Whether a task was easy or hard and whether the working of a problem was understood or not seemed to be central concerns of the students. Beliefs about the preferred difficulty level of tasks and which facets of the lesson students considered important were other notions identified by students.

All students commented about the achievement hierarchy of students for the learning of mathematics in the classroom. Not only were the relative performance capabilities of other students recognized but, as well, the students reported the cues to which they attended for identifying the achievement places of other students. Each of the four students perceived his or her place in the hierarchy and this related closely to the performance self-expectations held by each student.

All students made frequent referral to the end of unit test and especially the mark on the report card. With three students the mark on the report card tended to be considered in association with the parent(s) and the home. The pervasiveness of the influence of the report card seemed to transcend most self-performance covert behavior.

The teacher's roles in the student's conception of selfperformance seemed to be crucial. Most of the students expressed concern about what the teacher thought of them. They were anxious that the teacher held as high performance expectations for them as possible. All students perceived the teacher valued effort as the criterion of success. They all perceived the teacher to be the person who distributed the report marks.

Fear of failure was reported both directly and indirectly by

309

most studies. This usually was associated with self-esteem. Fear of public failure, which could involve being laughed at by peers or diminuition of the teacher's impression of himself, tended to result in the student engaging in what Goffman (1967) referred to as face-saving tactics. Fear of private failure and its potential effect on the report card created further pressure on a student's performance. Hope of success was seldom reported by the students.

Therefore, it can be concluded that a student's conception of self-performance tended to include the following components: notions of success and failure and their causal determinants of mainly effort and, to a lesser extent, ability; emotion of anxiety and, to a lesser degree, a range of affective states; perceptions of tasks; perceptions of performance self-expectations and the achievement hierarchy of the classroom; mark on the report card; teacher's performance expectations of the student; and fear of failure.

b) The relationships among the ideas, beliefs, views, emotions, and lines of reasoning of a student's conception of self-performance

From the components of the students' underlying selfperformance covert behavior several relationships were discerned. The most pervasive relationship seemed to pertain to what Doyle (1978) referred to as the performance-grade exchange. Students seemed desirous of successfully completing tasks in the most efficient manner possible in order to place themselves in an advantageous position for gaining a good mark on the report card. Of necessity, students perceived the teacher to be the mediating influence in achieving this goal and they tended to adapt their behavior.with a view to presenting



themselves favorably.

An associated relationship was the students' approach to learning. All students accepted that effort was the espoused criterion of success and consequently expended effort accordingly. However, while two of the students realized the desired outcomes for their effort expenditure, the other two students perceived little or no effort-outcome covariation. As a consequence, those who achieved their goal adopted a mastery-oriented approach to learning, while those who failed to achieve their goal tended to adopt a learned helpless approach to learning.

The approach to learning adopted by the student tended to be a function of his perception of the task. Where the task was perceived by the student to be easy and the means by which to complete the task clearly understood, the student tended to be mastery-oriented. If the student perceived the task to be difficult and the means of completing it were not understood, then the student tended to be learned helpless.

Perceptions of success and failure seemed to be associated with positive and negative emotions, respectively. These feelings were the more intense when associated with public performance and with significant performance, such as the unit test.

Perceptions of other students' behavior and performance appeared to be used by the student for comparing his own behavior and performance. These comparisons seemed to contribute to the kind of performance self-expectations formed and held by the student.

Performance self-expectations were generally based on a

self-perceived causal factor. Mike tended to base his self expectations on ability, whereas Tina perceived effort to be the underlying cause. Lisa and Greg, both less successful students, referred to effort frequently but there were indications that they also based their performance self-expectations on ability.

Whether or not a student wished to participate publicly seemed to reflect a relationship with the student's perceived public image. Mike's level of self-esteem enabled him to participate publicly as easily and readily as he participated privately. Intervening factors seemed to be involved with the other three students who engaged in what doffman (1967) referred to as various kinds of face-work. Tina and Lisa, to a varying degree, were concerned with enhancing the beacher's impression of themselves at the same time as risking the embarrassment of public failure. Freq was particularly concerned¹⁹with protecting his public image, hence he adopted avoidance tactics.

Therefore, it can be concluded that a student's conception of self-performance tended to include the following relationships among and between the components: the performance-grade exchange; the approach to learning; the perceptions of the tasks and the approach to learning; performance and emotions; other student performances and self-expectations; self-expectations and attributions of performance; and public participation and personal status.

c) The origins of these beliefs, views, emotions and lines of reasoning of a student's conception of self-performance.

The evidence suggested that the nature of a Grade 6 student's Sonception of self-performance reflected the background of the student 31.

and his previous school experience. Two related pivotal notions seemed to be involved. The "all important" grade on the report card appeared to link directly the student's classroom performance and the home. Students tended to want a good report mark to satisfy the expectations of the parent(s). Society, including the parent, the teacher, other students, and the self, espoused effort expenditure as the major criterion for achieving this good report mark. Where editre-steeme covariation was perceived by the student in the past, so his conception of self-performance was developed accordingly. Mike's and Tina's conceptions were characterized by success and positive feelings, while lisa's and long's conceptions were ensared rised by failure and negative feelings. Their current view of the perior makes arena of the classron is speared to be shaped from perspectives derived from previous grades on the report curd and the associated parental reaction to them.

Therefore, it can be concluded that the nature of a Grude i student's conception of self-performance tended to originate largely from previous report card performance and its associated causal eriterion of effort.

d <u>Causal perceptions of success and failure</u> and conceptions of self-performance

Tina accepted the effort criterion of success, whereas Mike attributed his significant success largely to higher ability. Where he succeeded with learning which he perceived to be easy, Mike attributed his performance to review or familiarity with the subject matter. While apparently not experiencing significant failure, Mike 3:3

attributed his infrequent and minor failures to lack of effort. Mike appeared to be uninvolved emotionally and unconcerned about other students in the classroom. As a consequence, his conception of self-performance was very task oriented.

Tina's conception of self-performance tended to involve more personal elements. She was concerned about her public image and strived to be viewed as an effort-conscious student. She attributed her significant successes to effort but her routine successes to a combination of effort and ease of task, especially if it was review. Tina attributed her infrequent and minor failures to lack of effort.

• Lisa and Greg attributed their failures initially to lack of effort. However, both perceived other factors to be involved. Lisa reported back of ability and Greg inferred lack of ability as the reason for not achieving the outcome expected from the amount of effort expended. Both students' conceptions incorporated personal elements, especially in terms of face-saving tactics as mechanisms for retaining some level of self-esteem. A component of these face-saving tactics appeared to involve misattributions. Lisa and Greg seemed to want to avoid attributing their failure to lack of atility by attributing it to lack of effort, mood, or external factors. Where Lisa and Greg exterienced success, they both attributed this mainly to effect to partly to task difficulty.

Therefore, it can be concluded that causal perceptions of success and failure influenced significantly student conceptions of self-performance. 3:4

e) <u>Characteristic qualities of a student's</u> <u>conception of self-performance</u>

From the discussion to this point it has been possible to discern some overall qualities of a student's conception of selfperformance. The drive or motive of the report card appeared to be salient. This seemed to be linked to the student's approach to learning which, in turn, gave rise to the particular emotions experienced. The student's causal explanation of his performance seemed to influence the manner by which he behaved in the classroom. These behavioral reactions tended to reflect the nature and level of self-esteem of the individual student.

The set of qualities which appeared to be characteristic of a student's conception of self-performance seemed to resemble, in a generalized form, the causal framework for the chains of comments referred to in the attributional analysis of behavior. The expression of a cognition tended to be accompanied by the expression of u. emotion. The subsequent causal explanation of behavior tended to reflect some underlying idea or belief, and often gave rise to the adoption of some consequent behavior.

Therefore, it can be concluded that the characteristic qualities of a student's conception of self-performance appeared to be attributionally based.

3. From an attributional perspective, what phenomena characterize the expectancy effect?

a) <u>Causal perceptions of students and their</u> effects on achievement <u>behavior</u>

The causal perceptions of performance by the successful students, who were described as mastery-oriented, were reported earlier. In relation to achievement behavior, whenever Tina and Mike experienced failure they simply applied more effort to the task and, as a result of this persistent behavior, achieved the desired outcome. They approached all tasks, including the unit test, confident of success, and usually sustained a prolonged intensity of effort until the tasks were completed.

The causal perceptions of performance by the less successful students, who were described as learned helpless, were also reported earlier. Both Lisa and Greg, when faced with a difficult task, persisted for a time, then gave up. While they often approached their routine assignments readily, on some occasions the the elesitant. This hesistancy was more profound when approaching the unit test. Both students viewed the test as an anxiety-loaded task. During assignments, both Lisa and Greg tended to be less intense in their work effort and were more likely to be engaged in behaviors other than attending-listening. This appeared to be related to their perceptions of the task.

Therefore, it can be concluded that a student's achievementrelated behavior is determined by his causal perceptions of success and failure.

b) The relationship between teacher's behavior and students' causal perceptions

The evidence from the four case studies indicated that the students were aware of the performance expectations which the teacher held for them. Furthermore, the students perceived the bases by which the teacher formed and maintained the expectations. Mike 'perceived the teacher's high performance expectations of him were based largely on his higher ability. Tina perceived the teacher's higher performance expectations of her were based on her expenditure of effort. With both students, some of the teacher's instructional moves conveyed to them the teacher's expectations held toward them. The amount of extra time given to other students during seatwork and not to them, and the selective use of students to publicly participate were examples of this. Both the successful students recognized the value the teacher held towards effort. Tina often pretended to give the impression of attending at times when the lesson became dull in order to maintain her perceived image of an attending student. Overall, the teacher's behaviors reinforced the successful students' attributions of performance to effort and ability.

Both Lisa and Greg recognized the value placed on effort by the teacher and both perceived the teacher's lower performance expectations held toward them were based on attributions to insufficient effort. For example, Lisa sometimes perceived the teacher to criticise her performance. Because she had exerted considerable effort, Lisa attributed her lower achievement to factors other than effort, namely, lack of ability. Greg, in recognizing the teacher's behaviors as espousing effort, also faced dilemmas when he expended much effort for little return. He sought to attribute the cause of failure to factors other than effort. These situations indicated that teacher behaviors which extolled the virtues of effort often resulted in the less successful student attributing his failure of performance to lack of ability or, occasionally, to other external factors.

Lisa and Greg were also aware of the teacher's differential behaviors when interacting with other students. Selective use of some students to respond to different questions and even the number of response opportunities offered to some students were recognized by the less successful students during whole class instruction. The overall effect on Lisa especially, and to a lesser extent on Greg, was an awareness of their ranking in the achievement hierarchy of the classroom.

Therefore, H can be concluded that students' causal perceptions of success and failure were influenced by teacher behaviors which extolled the virtues of effort and by differential behaviors when interacting with students.

c) <u>Causal perceptions by teachers and their</u> effect on behavior toward students

The evidence suggests that the teacher's higher performance expectations for the successful students were based largely on ability and some effort for Mike, and largely on effort for Tina. Perceiving Mike to be a student of higher ability resulted in the teacher often calling upon him when she wanted to change the pace or the direction of a lesson. She found she could always rely on Mike to contribute or demonstrate the significant idea during discussion. She seldom interacted with him privately. With Tina, the teacher perceived a student who needed very little help with her work. Consequently, when Tina did ask for help the teacher felt sure that the problem would be minor and that Tina would already have expended considerable effort in an attempt at a solution. She tended to make the assumption that Tina would "catch on" to the merest clue.

The teacher's lower performance expectations for Lisa appeared to be based largely on ability, whereas for Greg they were based largely on effort. Because the teacher was aware of Lisa's learning problem she tended to provide compensatory behaviors. She frequently interacted with Lisa, helping her to understand directions and often worked through an example of a problem with her. However, the teacher later began to suspect part of Lisa's problem to be a lack of effort. Teacher feedback containing criticism was only coded following this comment by the teacher. With Greg, the teacher seldom interacted privately but when this occurred it was largely intended for checking his progress. On occasions she felt it necessary to check whether or not Greg had been attending.

. The teacher differentiated her behaviors toward the lower performance expectation students according to her causal perception of their failure. Where she perceived failure to perform was based on a stable factor, as in Lisa's case, she provided additional academic support. However, with Greg, to whom she attributed failure largely to the unstable cause, her behavior was less academically oriented and more attention-demanding in nature.

<u>ر</u>..

Therefore, it can be concluded that the teacher's differential behavior toward students seemed to be determined by her causal perception of their success and failure.

d) Consensus between the teacher and student regarding the causal perceptions of the student's successes and failures.

Four sets of causal perceptions were observed to prevail in the learning situation: the student's causal perceptions of his performance; the student's perception of the teacher's causal perceptions of the student's performance; the teacher's causal perceptions of the student's performance; and the teacher's perception of the student's causal perception of his performance. With Mike there appeared to be consensus with all sets that his performance was attributable largely to ability and some effort. Similarly, all sets of causal perceptions attributed Tina's performance largely to effort.

Therefore, it can be concluded that, for the successful students, a consensus was found between the teacher and the students regarding the causal perceptions of the students' performances.

With Lisa, failure was perceived by her to be attributable largely to lack of ability. However, she perceived that the teacher attributed her lower performance to lack of effort. The teacher meanwhile attributed Lisa's failure to ability. While no direct evidence was found as to what the teacher perceived Lisa attributed her failure, the discrepancy in causality sected to exist with Lisa's interpretation of the teacher's behavior. With the teacher beginning to modify her impression of causality toward lack of effort, Lisa's interpretations probably will be reinforced. Greg perceived his failure to be attributable to factors other than effort, probably lack of ability. In his view, the teacher perceived his failure to be caused by insufficient effort. The teacher did perceive Greg's performance to be attributable largely to lack of effort but perceived that Greg attributed his failure performance to lack of ability. Both Greg and the teacher perceived the other's attributions of Greg's performance correctly, but lack of consensus in the ascription of failure generated the pressure in the learning situation. Greg perceived the teacher to be in the power role and so adopted a passive profile. The teacher perceived Greg as a quiet student who seldom participated and was, therefore, inclined to be distracted.

Therefore, it can be concluded that, for less successful students, a lack of consensus was found between the teacher and the students regarding the causal perceptions of performances.

To the researcher, the variation in consensus of attribution of performance for successful and less successful students constituted . one probable explanation of the expectancy effect. Variations in teacher behaviors and in student behaviors appeared to be based on this central point.

Therefore, it can be concluded that the issue of consensus between the teacher's and students' causal perceptions of student performance was a key mechanism in the explanation of the expectancy effect.

4. How effective are stimulated recall procedures when used as a research technique for observing the covert behavior of classroom participants?

a) <u>Difficulties or problems in the use of</u> stimulated recall methodology

The use of stimulated recall methodology with elementary school students was a comparatively new venture. Not surprisingly, a number of problems and difficulties arose in the course of the study.

The research design contained a familiarization phase and most aspects of this phase which pertained to stimulated recall proceeded as planned. The technical difficulties were overcome, the obtrusive effects of the presence of the researcher and the video recording equipment were reduced, the selection of target students accomplished, and the establishing of rapport between the researcher and the target students was undertaken. This included the researcher, of necessity, naving to learn the Grade 6 vernacular. However, an unforeseen difficulty was one which focused on the unusual experience for students to discuss and report to an adult thoughts about their beliefs, ideas, and values which pertained to their performance in the classroom. The rehearsals overcame this problem to some extent for two of the students but it was not until several days had passed in the data gathering phase that the researcher became aware of ar. increase in the research value of the other two students' reports of their self-performance covert behavior. While the researcher had been alerted to the need for choosing students who were more verbal and chatty and who were audible on a cassette tape, the experience seemed to suggest taking account of this further individual difference variable of conversing about the self with an adult.

The researcher, as interviewer, was attentive to the need to use mainly neutral questions and comments in eliciting student verbal reports of covert behavior. Because of the intentions of the study, the researcher used some leading questions when the student's recall of thoughts seemed complete. The researcher found, however, that withholding the use of leading questions during the stimulated recall sections of the interviews became difficult with two of the students who tended to recall brief thoughts.

Interviewee preparation was considered very important. The researcher now believes that the details of interviewee preparation, as outlined in Appendix B, should be undertaken with each student before each interview. While students differ in the need for this preparation, some discussion before each interview appears to be necessary. The researcher was surprised at the few occasions in which students stated they could not recall their interactive thoughts.

Most mathematics lessons were of 45 to 60 minutes duration. The lessons were too long for students to sit and recall thoughts; the researcher found that after 30 minutes the effectiveness of the stimulated recall interview diminished. Therefore, 30 minutes would appear to be an upper time limit for interviews with Grade 6 students. Such a limit would be helpful given the concern of withdrawing a student from the regular classroom learning sessions for the interviews.

Therefore, it can be concluded that with certain refinements few, if any, major problems exist in the use of stimulated recall methodology with Grade 6 students.

b) The potential of the stimulated recall methodology for investigating student interactive thoughts

The immediate feature observed about student interactive thoughts was their tendency toward brevity. Students recalled initially only a few thoughts at each stimulus point and often these appeared to be unsophisticated in substance. Usually students would focus on a single idea which they might describe briefly. Only on some occasions would student verbal reports in responses to the researcher's question extend beyond six sentences. Nonetheless, student interactive thoughts tended to be coherent, logical, and readily applicable to content analysis. The data derived can be described as rich, interesting and extensive in scope. They appeared to reflect the full range or array of thoughts experienced by the student during the lesson.

Students reported without hesitation interactive thoughts which ordinarily one might have deemed indiscreet or embarrassing to the student. Their reports reflected a spontaneity and candidness which initially had been unexpected. However, the richness of the data diminished when the mood of the student was $n_{\rm eff}$ appropriate for the needs of the interview. This seemed to occur rarely given the number of interviews conducted over the 10 day observation phase.

Therefore, it can be concluded that the potential of stimulated recall methodology for investigating student interactive thoughts appeared to be profound though, as yet, not fully realized.

5. The teacher's covert behavior which pertained to performance expectations of the students

From the stimulated recall interviews conducted with the teacher, several general comments about the teacher's performance expectations of students were derived from the data. These included her views about the class overall, an awareness of her use of differential behaviors with different students, her potential to modify performance expectations of individual students, and her views of the teaching-learning process as they pertained to expectations.

The teacher did compare the class this year with the Grade 6 she taught at another school the previous year. She commented that the class consisted of varying degrees of ability and so the progress through the curriculum guides was expected to be slower. The students were perceived to vary markedly in their familiarity with the unit topic of motion geometry; it was either review or new material. However, she wanted the class to proceed as a total group. When the unit test results were revealed the teacher reported that the students who scored low were expected because their work had been careless throughout the lessons. Some good students were observed to have not achieved as well as expected, while other students surprised her with their good results.

In the course of teaching the teacher was aware of some differential use of process teaching behaviors for different students. Often when she wanted to change the pace of the lesson or redirect it back to her preconceived plan, the teacher reported calling upon students whom she perceived or presumed would provide the desired response. Alternatively, when the teacher wanted to reinforce an idea she reported calling upon students whom she perceived or presumed would provide an incorrect response. Through use of process feedback to these students, the teacher believed she might be clarifying the point for the benefit of the whole class. The teacher also reported her awareness of definitely changing her pace of speech and development of idea according to her views of the student's performance capability. She was aware that under the normal pressures of teaching she often only chose volunteer students to respond. Because of this the teacher later realized that some students did not publicly participate in the lesson. Finally, with some students, the teacher was aware that any difficulty perceived to experience with their work was usually in the order of needing an assurance that they were "on track." Consequently, she adopted differential behaviors according to whether or not this observation might be valid for each individual student.

The teacher reported that she often found herself modifying her opinions of students according to her current perceptions of their performance. One student whom she expected to perform well performed poorly. She perceived no slackening in his effort input, hence she became concerned about the reason for this drop-off in performance. When it became known that the student was bothered by geometry the teacher reported being relieved. Her original attribution of his performance to ability and effort remained intact for she now explained away his poor performance to nature of the task, an external factor. The teacher twice reported that when she perceived a student

to be "on the tall" early in the lesson she endeavoured to encourage this unexpected behavior. Finally, the teacher did report holding a student in mind when she was marking the test, although she believed this did n t sway her marking at contentious points.

The teacher expressed some views of teaching-learning as they pertained to student performance expectations. Both direct and indirect evidence was obtained that the teacher placed a high value on effort. Her behavior when returning test results confirmed the evidence. The teacher's justification to the students for an encine the names of those who obtained the highest marks was to reaction considerable attention and listening undertaken by the sturint. throughout the lessons.

The teacher often expressed a sensitive concern about or behavior in relation to the students. She reported that she did of like to think any student thought that she had ignored him. Likewise, she endeavoured to take care to avoid referring to the relationship between activities and happenings in the elaboroom and test results and report cards.

Therefore, it can be concluded that the teacher was aware of her performance expectations held towards students and was aware accordingly of some differential process behaviors for different students.

Implications for Teaching and Learning

The implications of this study of the conceptions of student self-performance and the expectancy effect seem to be numerous. The four case studies presented underline the uniqueness and idiosyncracies of each individual classroom participant. Each student is a different human being engaged in the process of learning and, especially, in just being a person. By viewing the achievementrelated situation of the regular functioning classroom through different students' eyes, several insights emerged which should have implications for teaching and learning. While these realizations may not necessarily reflect anything new educationally, they do emphasize basic aspects of human functioning in the learning process which may have been overlooked in the drive for superior curricula and advanced instructional techniques.

The evidence from the study would indicate that the society, the school, the home, and even the individual students espoused effort as the overriding criterion for success. The adage seemed to exist that so long as anyone worked hard they would achieve success. Parents, the teacher, and the individual students accepted the notion that the quality of the outcome derived depended on the amount and kind of effort expended. Where effort and outcome tended to covary for a student, that student appeared to experience positive selfperformance feelings and functioned in a success-oriented climate. However, where effort and outcome tended not to covary for a student, that student appeared to experience negative self-performance feelings and functioned in a failure-oriented existence. The successful students appeared to view their outcomes as upholding the generally high value placed on effort. The less successful students tended to find themselves in a dilemma. They believed that they had exerted all possible effort, yet the societal value had not been realized. The study would support the effort expenditure, for it provided evidence that less successful students were attending-listening as much as, if not more than, the successful students. In addition, less successful students perceived both directly and indirectly that the teacher and other students attributed their failures to insufficient effort. Inevitably, the less successful student attributed his failure to factors other than effort, usually lack of ability. The situation for less successful students, therefore, appeared to be educationally bleak.

The implications for teaching and learning in this situation seemed to focus upon the nature of the tasks. The study indicated that the less successful students were motivated to learn but often found the task beyond them. Despite considerable effort, these students were unable to perceive tasks to be structured as ones which they could solve by their own efforts and abilities. Had the tasks been clearly perceived by the less successful students, there seemed every reason to expect that effort-outcome covariation would have followed. The implication for teaching, therefore, would be one of adapting instructional tasks to meet the task perception needs of the various students in the classroom. If the tasks were perceived to be manageable by all students then the espousing of the effort criterion of success would seem to be more appropriate.

Ð

In situations where all students in the class were learning the same material at the same rate at the same time, both the successful students and the less successful students were disadvantaged. The successful students viewed the tasks as usually easy, usually review from previous years, and expressed boredom and disappointment. The continued, long term effect of these views might result in negative attitudes to learning. The less successful students were disadvantaged for the different reasons described earlier.

Ey accommodating differing student success thresholds through the adaptation of instructional tasks, teachers would be obliged to accept the notion of students functioning at differing levels of ability. An eclectic approach to instruction in the regular classroom might be one means of achieving this. The successful students would be appropriately challenged, while the less successful students would experience more success which they could attribute to effort rather than to factors other than effort.

The study revealed that all students were willing to work and learn at school. The report card seemed to be the motivating factor. In a sense, the pervasiveness of the report card in the thoughts of students was surprising, yet it confirmed Doyle's (1978) study which referred to the performance-grade exchange character of the classroom. Successful students were very keen to retain their good merks on the report, while less successful students were very keen to improve their report card marks. In three of the four cast studies the researcher detected that either subtle or direct pressure was exerted from the home for the student to achieve a good mark.

The report card seemed to be the ultimate though seldom visible goal of student behavior in the learning process and the implications of this for teaching seemed far-reaching. The purpose of the report card may need to be questioned. Students were aware that the taking of a unit test was directly related to the report card, therefore, the purpose of tests also may need to be reviewed. In addition, students expressed concern about what the teacher's impressions of them might be. All students saw the teacher as the person who distributed the marks on the report cards and consequently . appeared to deem it necessary to impress the teacher. Evidence showed that the students' intentions were to enhance their report card prospects. While teachers and administrators would most likely emphasize that elementary classroom learning is designed for higher goals, the reality of three students' conceptions of the goal of learning cannot be overlooked.

 $\overline{\chi}$

The study indicated that a student's feekings of self-esteem mediated his achievement-related performance. Where these feelings were positive, then the student tended to perform with success. Student self-esteem attributed to high ability seemed, for the student, to be related to more confidence and greater immunity from emotional concerns and awareness of other students than when student self-esteem was attributed more to effort.

Low levels of self-esteem seemed to be associated with face-saving tactics. The study indicated that misperceptions and misattributions by the teacher and less successful students tended to occur when a student was engaged in protecting his self-image. In the normal hustle of a regular classroom the teacher tended to make judgements about student behavior and performance based on insufficient information. Reference to stereotyped explanations tended to result in inappropriate teacher behavior. Where the teacher did have full information she acted appropriately. The implication for teaching is one of becoming aware that a causal explanation normally can be found for most if not all student behavior. Hopefully this study has provided insight into some of the causes of student behavior which, in turn, could add to the information base which teachers might use in their decision making.

A great deal of literature has been presented about teacher performance expectations of students and how these might be "communicated." This study showed clearly that students are aware of the performance expectations which teachers hold toward them. Furthermore, the students accurately perceived the attributional basis of those expectations. The study also indicated that students engaged in behaviors which were designed to preserve or enhance the performance expectations which the teacher held towards them and there was evidence which suggested that not only did these behaviors succeed, but students perceived them to succeed. Such an all-round degree of awareness by the student might surprise teachers, especially those who are sensitive and strive to be fair. Students tended to be very active classroom participants even though outwardly they may spend much time seated passively in a desk. However, the findings illustrate the importance of the social psychological dimensions of classrooms in a student's mind. The nature and extent of student interactive thoughts which focus on self-performance considerations compared to subject matter and learning considerations may need to be taken into account more by teachers. Teachers need to be well versed in the prevailing group processes of classrooms and this also has implications for teacher education programs.

Finally, the impression is gained from the study that classroom life for successful students is more happy, more positive, and more conducive to learning than it is for less successful students. The evidence to support this impression was both varied and extensive. Less successful students experienced personal and learning pressure, complete with intensity of feelings, to a degree virtually unknown or not experienced by the successful students. The reasons for the situation appeared to range from the less successful students' inability to feel relaxed with the subject matter to perceptions of ability differences among students. While the teachers in most classrooms may be unaware of the scale of the problem, a very real need seems to exist for less successful students to be relieved of some of the personal and learning pressure. Teaching behaviors which accentuate the achievement hierarchy of the classroom appear to. contribute directly to this pressure. The implications for teaching seem considerable, for teachers wish - redress the situation may have to revise significantly many cherest and long held behaviors and ideas.

Recommendations for Further Research

٩,

This study was exploratory in the sense that attempts were made to describe students' conceptions of self-performance and, through an attributional analysis, hopefully to arrive at contributory explanations of the expectancy effect phenomena. Very little is known about student thought processes during learning, especially interactive thoughts concerning self-performance. The research line has considerable scope for development and this study seems to suggest numerous directions for further inquiry. Some of the directions recommended are presented below.

Student Interactive Thoughts

- Student interactive thoughts in a normal classroom are only becoming known. How students process information while learning and how they integrate this with conceptions of self-performance could contribute to theorgen of learning and teaching.
- 2. The study has revealed that self-oriented thoughts and feelings form a significant component of student interactive thoughts. This data base, however, is precariously thin and a real need exists for replication studies across grade levels, across subject matter, and across student ability levels.
- 3. The origins of student conceptions of self-performance would appear to be significant for teachers and counselors. This study was able to indicate some very general trends, hence the need for a detailed study of how students developed the correction of selfperformance which they have acquired.

- 4. The study indicated certain components of the student's conception of self-performance were viewed to be more significant than others. An investigation of which hierarchy of components was associated with which kind of student should prove helpful for teachers.
- 5. No attempt was made in this study to investigate the comparative characteristics of student conceptions of self-performance on criteria other than successful and less successful students. Other variables, such as sex and socio-economic status, may reveal variations of a different order.
- b. The impact of success and failure on students seemed to vary. The bases and origins of these variations should be worth investigating for they might provide an important information base to teachers in their decision making related to feedback behaviors.
- 7. Student emotions, though a corollary to a number of other factors, appeared as a significant component of student interactive thoughts. Extensive studies into the impact of this intervening variable could help teachers to better monitor the emotional health of students and the influence on student emotions of teacher behaviors.
 - 6. The significance of students' conceptions of self-performance on their learning probably cannot be overlooked. A study which would investigate relationships between student product measures and student conceptions of self-performance may provide more realistic causal explanations of student performance which, in turn, might suggest more appropriate instructional directions.

9. Goffman (1967) has elaborated on face-work and this study found evidence that self-protective tendencies of this kind were adopted by students. Detailed analysis of the reasons underlying the use of these tendencies, the different kinds of tendencies adopted and the impact on student behavior of continued use of self-protective tendencies could help teachers to better understand the effects of some of their teaching behaviors.

Teacher-Student Dyadic Interactions

Dyadic interactions seem to be crucial in the "communication" of performance expectations. From the episodes described in this study, a micro-analysis of many episodes should yield rich and valuable explanatory data. A need exists for longitudinal analysis of teacher-student dyadic interactions over an extensive time period and across a number of subject areas. Such an analysis might reveal a greater array of subtle behaviors and intentions of both the teacher and student, for these seem to hold the clues in the transmission of interpersonal impressions.

The dyadic interaction episodes between the teacher and the student seemed to be fraught with misperceptions and defensive attributions. A detailed study of these behaviors may help teachers to become more aware of the reasons underlying a student's behavior.

The conclusion was reached in the study that the issue of consensus between the teacher's and the student's causal perceptions of the student performance was probably a crucial point in the explanation of the expectancy effect. For such a conclusion to be accepted in a general form, more investigation of the consensus issue is recommended.

The Teacher's Interactive Thought Processes

This study has emphasized the student perspective of the interaction process of the expectancy effect. Student conceptions of self-performance, including performance self-expectations, are now viewed to be significant elements of the interaction process. A study which examined closely the teacher's covert behavior in relation to the interaction process would seem necessary. No attempt at an explanation of the expectancy effect would seem to be complete without such a study.

-

.

BIBLIOGRAPHY

338

•.

BIBLIOGRAPHY

- Adams, G. R., & LaVoie, J. C. The effect of student's sex, conduct, and facial attractiveness on teacher expectancy. <u>Education</u>, 1974, <u>95</u>, 76-83.
- Ames, C. Children's achievement attributions and self-reinforcement: Effects of self-concept and competitive reward structure. Journal of Educational Psychology, 1978, 70(3), 345-355.
- Anderson, H. H., & Brewer, H. M. Studies of teachers' classroom personalities: I. Dominative and socially integrative behavior of kindergarten teachers. <u>Applied Psychology Monographs</u>, 1945, No. 6.
- Anderson, D. F., & Rosenthal, R. Some effects of interpersonal expectancy and social interaction on institutionalized retarded children. <u>Proceedings of the 76th Annual Convention of the</u> <u>American Psychological Association</u>, 1968, 4, 479-480.
- Andrews, G. R., & Debus, R. L. Persistence and the causal perception of failure: Modifying cognitive attributions. <u>Journal of</u> <u>Educational Psychology</u>, 1978, <u>70</u>(2), 154-166.
- Atkinson, J. W. <u>An introduction to motivation</u>. Princeton, N.J.: Van Nostrand, 1964.
- Bar-Tal, D. Attributional analysis of achievement-related behavfor. Review of Educational Research, 1978a, <u>48</u>(2), 259-271.
- Bar-Tal, D. Perceptions and behaviors of pupils and teachers: An attributional analysis. Unpublished paper, 1978b. To appear in I. H. Frieze, D. Bar-Tal, & J. S. Carroll (Eds.), Attribution theory: Application to social problems. San Francisco: Jossey Bass (in press).
- Bar-Tal, D., & Saxe, L. Teachers' information processing: The effect of information about pupil on teachers' expectations and affect. <u>Psychological Reports</u>, in press.
- Battle, E. S., & Rotter, J. B. Children's feelings of permal control as related to social class and ethnic group, Journal_of <u>Personality</u>, 1963, <u>31</u>, 482-490
- Beez, W. Influence of biased psychological reports on teacher behavior and pupil performance. In M. W. Miles &
 W. W. Charters, Jr. (Eds.), Learning in social settings.
 Boston, Mass.: Allyn & Bacon, 1970.

Ş.

-

Bell, L. G., Wicklund, R. A., Manko, G., & Larkin, C. When unexpected behavior is attributed to the emvironment. Journal of Research in Personality, 1976, 10, 316-327.

.

- Bennett, S. N. <u>Recent research on teaching: A dream</u>, a belief and <u>a</u> model. Unpublished paper, University of Lancaster, 1978.
- Berger, J., Conners, T. T., & Fisek, M. H. (Eds.). Expectation states theory: A theoretical research program. Cambridge, Mass.; Winthrop Publishing, 1974.
- Bloom, B. S. Thought processes in lectures and discussions. The Journal of General Education, 1953, 7(3), 100-109.
- Blumenfeld, P. C., Hamilton, L., Wessels, K., & Falkner, D. "You can," "you should," and "you'd better": Teachers' attributions regarding achievement and social behaviors. Paper presented in A. Chun (Chair), <u>Success and failure attributions and student behavior in the classroom</u>. Symposium presented at the meeting of the American Psychological Association, San Francisco, 1977.
- Forko, H., & Shavelson, R. J. Teachers' sensitivity to the reliability of information in making causal attributions in an achievement situation. Journal of Educational Psychology, 1975, 70(3), 271-279.
- Fradley, G. W. Self-serving tiases in the attribution process: A re-examination of the fact or fiction question. <u>Journal of</u> Personality and Social Psychology, 1978, <u>30</u>(1), 30-71.
 - Braun, C. Teacher expectations: Sociopsychological dynamics. Review of Educational Research, 1976;446(2), 195-213.

ч

Brown, S. S., & Aderman, D. On the relationship between empathy and the actor versus observer hypothesis. Journal of Research in Personality, 1977, 11, 340-346.

Bremme, D. W. <u>Accomplishing a classroom event: A microethnography</u> of <u>first circle</u>: Unpublished paper, Harvard Graduate School of Education, Learning Environments, 1976.

Brophy, J. E., & Evertson, C. M. Low-inference observational coding measures and teacher effectiveness. 1973. ERIC microfiche ED 077 879.

Brophy, J., & Evertson, C. <u>Process-product correlations in the Texas</u> <u>teacher effectiveness study: Final report</u>. Austin, Texas: Research and Development Center for Teacher Education, 1974.

Brophy, J., & Evertson, C. Learning from teaching: A developmental perspective. Boston: Allyn & Bacon, 1976a.
Eropny, C. E., & Evertson, C. M. The student attribute study: A prel. inary report. 1970: ERIC microfiche ED 121 799.

Brophy, J., & Evertson, C. Teacher behavior and student learning in second and third grades. In G. L. Boricr. (Ed.), <u>The appraisal of</u> <u>teaching: Concepts and process</u>. Reading, Mass.: Addisor.-Wesley, 1977.

Erophy, J. E., & Good, T. L. <u>Teacher-child dyadic interaction</u>. <u>A</u> <u>manual for coding classroom behavior</u>. Report Series No. 27. Austin: Texas University, Research and Development Center for Teacher Education, 1969.

Frophy, J. E., & Good, T. L. Teachers' communication of differential expectations for children's classroom performance: Jone tehavioral data. Journal of Educational Psychology, 1970, 21, 3:5-374.

Erophy, J. E., & Good, T. L. <u>Teacher-student relationships</u>: Causes and consequences. New York: Holt, Kinenart & Winston, 1974.

Erustein, S. C. Learned nelplessness. Journal of Instructional <u>Fsychology</u>, 1976, <u>5</u>(2), c-ic.

Sarter, R. Locus of control and teacher's expectancy as related to achievement of young school chi. men. Unpublished doctoral dissertation, Indiana University, 1969.

Chapman, J. Fersonal communication. University of Alberta, Department of Educational Psychology, 1979.

Chapman, J. W., & Boersma, F. J. Projected academic performance scale. Edmonton, Alberta, 1975.

Claitorn, W. Expectancy effects in the classroom: A failure to replicate. Journal of Educational Psychology, 1969, 00, 377-303.

Clark, C. M., & Leterson, F. L. <u>Teacher stimulated recall</u> of <u>interactive decisions</u>. Paper presented to the AERA, Wasnington, 1976.

Clark, C. M., & Yinger, R. J. Hesearch on teacher trinking. Curriculum Inquiry, 1977, 7(4), 479-304.

Conn, L. E., Edwards, C. E., Rosenthal, R., & Crowne, D. Ferceptions of emotion and response to teacher's expectancy by elementary school children. <u>Fsycholog</u>: <u>El Reports</u>, 1968, <u>42</u>, 27-34.

Content, R. L. An analysis of teacher thought processes, beliefs, principles during instruction. Unpublished doctoral Sertation, University of Alberta, 1975a. Conners, R. D. Using stimulated recall in naturalistic settings--some <u>technical procedures</u>. Occasional Paper Series: Technical Paper No. 78-2-1. Edmonton, Alberta: University of Alberta, Centre for Research in Teaching, 1978b.

· ·

- Cooper, C. Information processing by teacher and pupils during <u>mathematics instruction</u>. Doctoral provide University of Alberta, 1975.
- Cooper, H. M. Controlling personal rewards: Professional teachers' differential use of feedback and the effects of feedback on the student's motivation to perform. Journal of Educational Esychology, 1977, 62(4), 419-4-7.
- Cooper, H. M., & Baron, R. Academic expectations and attributed responsibility as predictors of professional teachers' reinforcement behavior. Journal of Educational Esychology, 1977, 1977, 1977.
- Douper, H., Baron, K., & Lowe, C. The importance of race and social class in the formation of expectancies about academic performance. <u>Journal of Educational Psychology</u>, 1975, <u>67</u>(2), 312-319.
- Cooper, H., & Burger, J. M. <u>Internality, stability and personal</u> <u>efficacy: A categorization of free response academic attributions</u>. Unputlished manuscript, University of Missouri, Columbia, 1970.
- Dooper, H. M., & Seymour, G. E. <u>Classroom context and student atility</u> as influences on teacher personal control and success expectations. Technical Report No. 121. Columbia, Missouri: University of Missouri, Center for Research in Social Behavior, 1977.
- Cornbleth, C., Davis, C., & Button, C. Expectations for pupil achievement and teacher-pupil interaction. <u>Social Eucation</u>, 1974, <u>28</u>(1), 54-58.
- Crandall, V., Katkovsky, W., & Crandall, V. Children's beliefs in their own control of reinforcements in intellectual-academic achievement situations. <u>Child Development</u>, 1965, 36, 91-109.
- Crano, W. D., & Mellon, F. M. Causal influence of teachers' expectations on children's academic performance: A cross-lagged panel analysis. Journal of Educational Psychology, 1978, 70(1), 39-49.

a

- Dalton, W. The relations between classroom interaction and teacher rating of pupils: An explanation of one means by which a teacher may communicate her expectancies. <u>Peabody Papers in Human</u> <u>Development</u>, 1969, <u>1</u>(6).
- Darom, E., & Bar-Tal, D. <u>Causal perceptions of pupils' success or</u> failure by teachers and pupils: A comparison. Unpublished manuscript, Tel-Aviv University, 1977.
- Deaux, K. Sex: A perspective on the attribution process. In J. Harvey, W. Ickes, & R. Kidd (Eds.), <u>New directions in</u> <u>attribution research</u> (Vol. I). New York: Wiley, 1976, pp. 335-352.
- Deaux, K., & Farris, E. Attributing causes for one's own performance: The effects of sex, norms, and outcome. Journal of Research in Personality, 1977, <u>11</u>, 59-72.
- deCharms, R. Personal causation. New : rk: Academic Press, 1965.
- deCharms, R. Enhancing motivation. New York: Irvington, 1975.
- Diener, C. I., & Dweck, C. S. An analysis of learned helplessness: Continuous changes in performance, strategy, and achievement cognitions following failure. Journal of Personality and Social Psychology, 1978, 36(5), 451-462.
- Doyle, W. Learning the classroom environment: An ecological analysis of induction into teaching. Paper presented to the AERA, New York, 1977.
- Doyle, W. Student mediating responses in teaching effectiveness: An interim report. Paper presented at the AERA, Toronto, 1978.
- Doyle, W., Hancock, G., & Kifer, E., <u>Teachers' perceptions: Do they</u> make a difference? Paper presented at the AERA, New York, 1971.
- Dunkin, M. J., & Biddle, B. J. <u>The study seteaching</u>. New York: Holt, Rinehart and Winston, 1974.
- Dusek, J. Do teachers bias children's learning? <u>Review of</u> Educational Research, 1975, 45(4), 661-684.
- Dusek, J., & O'Connell, E. Teacher expectancy effects on the achievement test performance of elementary school children. Journal of Educational Psychology, 1973, 65, 371-377.
- Dweck, C. S. The role of expectations and attributions in the alleviation of learned helplessness. Journal of Personality and Social Psychology, 1975, 31, 674-685.

- Dweck, C. S., Davidson, W., Nelson, S., & Enna, B. Sex differences in learned helplessness: II. The contingencies of evaluative feedtack in the classroom, and III. An experimental analysis. Developmental Psychology, 1978, 14(3), 268-276.
- Dweck, C. S., & Goetz, T. E. Attributions and learned helplessness. In J. H. Harvey, W. Ickes, & R. F. Kidd (Eds.), <u>New directions in</u> <u>attribution research</u> (Vol. II). New York: Wiley, 1978, pp. 158-181.
- Dweck, C. S., & Repucci, N. D. Learned helplessness and reinforcement responsibility in children. Journal of Personality and Social <u>Psychology</u>, 1973, 25, 109-116.
- Elashoff, D. J., & Snow, R. E. <u>Pygmalion reconsidered: A case study</u> in statistical inference : <u>Reconsideration of the Rosenthal</u> <u>Jacobson data on teacher expectancy</u>. <u>Belmont</u>, <u>California</u>: Wadsworth, 1971.
- Entwisle, D., & Webster, M. Raising children's performance expectations. <u>Social Science Research</u>, 1972, <u>1</u>, 147-155.
- Erlwanger, S. Case studies of children's conceptions of mathematics. Unpublished doctoral dissertation, University of Illinois, 1974.
- Evertson, C. M., Anderson, L., Edgar, D., Minter, M., & Brophy J. Investigations of stability in junior high school math and English classes: The Texas junior high school study. R & D Report No. 77-3. Austin: University of Texas, Research and Development Center for Teacher Education, 1977.
- Fay, B. <u>Social theory and political practise</u>. London: Allen & Unwin, 1975.
- Fedigar, 1. <u>School-based elements related to achievement</u>. Monograph prepared for the Alberta Department of Education, 1978.
- Fiedler, M. L. Bidirectionality of influence in classroom interaction. Journal of Educational Psychology, 1975, 67(6), 735-744.
- Finn, J. D. Expectations and the educational environment. <u>Review of</u> Educational Research, 1972, 42, 387-410.
- Firestone, G., & Brody, N. Longitudinal investigation of teacherstudent interactions and their relation to academic performance. Journal of Educational Psychology, 1975, 67(4), 544-550.
- Flanders, N. A. Personal-social anxiety as a factor in learning. Unpublished doctoral dissertation, University of Chicago, 1949.

÷ 1

345

- Flanders, N. A. Teacher influence in the classroom. In E. J. Amidon & J. B. Hough (Eds.), Interaction analysis: Theory, research and application. Reading, Mass.: Addison-Wesley, 1967.
- Fleming, E. S., & Antonnen, R. G. Teacher expectancy or my fair lady. American Educational Research Journal, 1971, 8, 241-252.

Frieze, I. H. Causal attributions and information seeking to explain success and failure. Journal of Research in Personality, 1970, 10, 293-305.

- Gage, N. L. Paradigms for research on teaching. In N. L. Gage (Ed.), <u>Handbook of research on teaching</u>. American Educational Research Association. Chicago: Rand McNally, 1963.
- Gaier, E. L. Memory under conditions of stimulated recall. The Journal of General Psychology, 1954, <u>50</u>, 147-153.
- Galper, R. E. Turning observers into actors: Differential causal attributions as a function of empathy. Journal of Research in Personality, 1976, 10, 328-335.
- Gay, G. Teachers' achievement expectations of and classroom interactions with ethnically different students. <u>Contemporary</u> <u>Education</u>, 1975, <u>46</u>, 166-172.
- Getzels, J. W., & Thelen, H. A. The classroom group as a unique social system. In N. B. Henry (Ed.), <u>The dynamics of instructional groups</u>. 59th Yearbook of the NSSE, Part 2, 1960, pp. 53-82.
- Glaser, E., & Strauss, A. The discovery of grounded theory: Strategies for qualitative research. Chicago: Aldine, 1967.
- Goethals, G. R. An attributional analysis of some social influence phenomena. In J. Harvey, W. Ickes, & R. Kidd (Eds.), New directions in attribution research (Vol. I). New York: Wiley, 1976, pp. 291-310.
- Goffman, E. The presentation of self in everyday life. Garder City, New York: Doubleday, 1959.
- Goffman, E. <u>Interaction ritual</u>. Harmondsworth, Middlesex, England: Penguin, 1967.
- Good, T. L., & Brophy, J. Behavioral expression of teacher attitudes. Journal of Educational Psychology, 1972, 63, 617-624.
- Good, T. L., & Brophy, J. Looking in classrooms. New York: Harper and Row, 1973.

- A.

Good, T. L., & Grouws, D. A. Teaching effectiveness in fourth-grade mathematics classrooms. In G. D. Borich (Ed.), <u>The appraisal of</u> <u>teaching: Concepts and process</u>. Reading, Mass.: Addison-Wesley, 1977.

- Haigh, N. J. Teacher expectancy: Accuracy, change and communication. New Zealand Journal of Educational Studies, 1974, 9(2), 97-112.
- Hallworth, H. J. A teacher's perception of his pupils. Educational Review, 1962, <u>14</u>, 124-133.
- Hanse, R. D., & Stonner, D. M. Attributes and attributions: Informing stimulus properties, actors' dispositions, and causes. Journal of Personality and Social Psychology, 1978, 36(6), 657-667.
- Harre, R., & Secord, I. F. <u>The explanation of social behavior</u>. Oxford: Blackwell, 1972.
- Harvey, J. H. Attribution of freedom. In J. Harvey, W. Ickes, & R. Kidd (Eds.), <u>New directions in attribution research</u> (Vol. I). New York: Wiley, 1976, pp. 73-96.
- Harvey, J. H., Ickes, W., & Kidd, R. (Eds.). <u>New directions in</u> <u>attribution research</u> (Vol. I). New York: Wiley, 1976.
- Harvey, J. H., Ickes, W., & Kidd, R. (Eds.). <u>New directions in</u> attribution research (Vol. II). New York: Wiley, 1975.
- Heider, F. The psychology of interpersonal relations. New York: Wiley, 1958.
- Holsti, O. R. Content analysis. In G. Lindzey & E. Aronson (Eds.), <u>The handbook of social psychology</u> (2nd Ed.), Vol. II. Reading, Mass.: Addison-Wesley, 1968, pp. 596-692.
- Holsti, O. R. <u>Content analysis for the social sciences and</u> <u>humanities</u>. Reading, Mass.: Addison-Wesley, 1969.
- Hudgins, B. B., & Ahlbrand, W. P. Some properties of formal teacher and pupil classroom communication. <u>Psychology in the Schools</u>, 1970, 7, 265.
- Hughes, L. C. An experimental investigation of the effects of pupil responding and teacher reacting on pupil achievement. <u>American</u> <u>Educational Research Journal</u>, 1973, <u>10</u>(1), 21-37.
- Jackson, P. W. Life in classrooms. New York: Holt, Rinehart & Mark Winston, 1968.
- Jackson, P. W., & Lahaderne, T. Inequalities of teacher-pupil contact. Psychology in the Schools, 1967, 4, 204-211.

Jenkins, D. H. Interdependence in the classroom. Journal of Educational Research, 1951, 45, 137-144.

- Jensen, A. How much can we boost I.Q. and achievement? <u>Harvard</u> <u>Educational Review</u>, 1969, <u>39</u>, 1-123.
- Jeter, J. Can teacher expectations function as self-fulfilling prophecies? <u>Contemporary Education</u>, 1975, 46(3), 161-165.
- Aohnson, S. M., & Bolstad, O. D. Methodological issues in naturalistic observation: Some problems and solutions for field research. In L. A. Hamerlynck, L. C. Handy, & E. J. Mash (Eds.), <u>Behavior change: Methodology</u>, concepts, and practice. Champaign, Illinois: Research Press, 1973, pp. 7-58.
- Jones, E. E., & Davis, K. From acts to dispositions: The attribution process in person perception. In L. Berkowitz (Ed.), Advances in experimental social psychology (Vol. 2). New York: Academic Press, 1965, 219-226.
- Jones, E. E., & Nisbett, R. E. The actor and the observer: Divergent perceptions of the causes of behavior. In E. Jones, D. Kanouse, H. Kelley, R. Nisbett, S. Valins, & B. Weiner (Eds.), Attribution: Perceiving the causes of benavior. Morristown, N.J.: General Learning Press, 1972, pp. 79-94.
- Jose, J., & Cody, J. J. Teacher-pupil interaction as it relates to attempted changes in teacher expectancy of academic atility and achievement. <u>American Educational Research</u> Journal, 1971, 5, 39-50.
- Joyce, B. Variables, designs, and instruments in the search for teacher effectiveness. Technical Report 75-10-4, <u>beginning</u> <u>teacher evaluation study</u>. San Francisco, California: Far West Laboratory for Educational Research and Development, 1975.
- Kash, M. D., & Borich, G. D. Teacher behavior and pupil sel: -concert. Reading, Mass.: Addison-Wesley, 1978.
- Kelley, H. H. Attribution theory in social psychology. In D. Levine (Ed.), <u>Nebraska Symposium on Motivation</u> (Vol. 15), Lincoln, Nebraska: University of Nebraska Press, 1907.
- Kerlinger, F. Foundations of behavioral research (2nd Ed.). New York: Holt, Rinehart & Winston, 1973.
- Kester, S., & Letchworth, G. Communication of teacher expectations and their effects on achievement and attitudes of secondary school students. <u>The Journal of Educational Research</u>, 1972, <u>66</u>, 51-55.

. .

King, L., & Smyth, J. Some fromising variables in the study of teacher effectiveness. Monograph prepared for the Alberta Teachers' Association, 1978.



- Kuiper, N. A. Depression and causal attributions for success and failure. Journal of Personality and Social Psychology, 1978, <u>36(3)</u>, 236-246.
- Kukla, A. Attributional determinants of achievement-related behavior. Journal of Personality and Social Psychology, 1972, <u>32</u>, 100-174.
- Lefcourt, H. M. Locus of control: Current trends in theory and research. Hillsdale, N.J.: Lawrence Erlbaum Associates, 1976.
- Lewin, K., Lippitt, R., & White, R. K. Patterns of aggressive rehavior in experimentally created social climates. <u>Journal</u> of <u>Social Psychology</u>, 1939, <u>10</u>, 271-299.
- Lockheed, M. E. Some determinants and consequences of teacher expectations concerning pupil performance. 1975. ERIC microfiche, ED 127 371.
- Long, B. H., & Henderson, E. H. Certain determinants of academic expectancies among southern and non-southern teachers. <u>American Educational Research Journal</u>, 1974, <u>11</u>, 137-147.
- Lovell, K. The growth of basic mathematical and scientific concepts. London: The University of London Press, 1902.
- Lowe, C. A., & Hansen, R. D. Motivational influence of behavior desirability on actors' and observers' attributions. <u>Social</u> <u>Behavior and Personality</u>, 1976, 4(1), 17-25.
- MacDonald, B. The case study as a method of educational inquiry. Unpublished manuscript, University of East Anglia, Norwich, 1978.
- MacKay, D. A., & Marland, F. W. Thought processes of teachers. Paper presented to AERA, Toronto, 1978.
- Magoon, A. J. Constructivist approaches in educational research. Review of Educational Research, 1977, 47(4), 65:-093.
- Marland, P. W. <u>A study of teachers' interactive thoughts</u>. Unpublished doctoral dissertation, University of Alberta, 1977.
- Marliave, R. Observable classroom variables. Technical Report I-2. Beginning teacher evaluation study. San Francisco, California: Far West Laboratory for Educational Research and Development, 1976.
- McDonald, F., & Elias, T. <u>Beginning teacher evaluation study</u>. Phase II Technical Summary, final report. Princeton, N.J.; Educational Testing Service, 1976.

- Meichenbaum, C., Bowers, K., & Ross, R. A behavioral analysis of teacher expectancy effect. Journal of Personality and Social Fsychology, 1969, 13, 306-316.
- Merton, R. K. <u>Social theory and social structure</u>. Glencoe, Illinois: Free Press, 1949.
- Miller, A. G. Constraints and target effects in the attribution of attitudes. Journal of Experimental Social Psychology, 1970, 12, 325-339.
- Monson, T. C., & Snyder, M. Actors, observers, and the attribution process: Toward a reconceptualization. <u>Journal of Experimental</u> <u>Social Psychology</u>, 1977, <u>13</u>, 89-111.
- Moreno, J. L. <u>Who shall survive</u>? Washington, D.C.: Nervous and Mental Disease Fublishing Company, 19**34**.
- Muttart, D. G. A study of teacher-pupil interaction as it relates to differential teacher expectations and selected teacher characteristics. Unpublished doctoral dissertation, University of Alterta, 1977.
- National Institute of Education. <u>Teaching as clinical</u> information processing. National Conference on Studies in Teaching, No. 1. Washington, D.C.: Author, 1975.
- Nicholls, J. G. Effort is virtuous, but it's better to have ability: Evaluative responses to perceptions of effort and ability. Journal of Research in Personality, 1976, <u>10</u>, 300-315.
- Nolar, F. The composing processes of able sixth grade writers. Unjulished doctoral dissertation, University of Alberta, 1978.
- Nowicki, S., & Strickland, B. A locus of control scale for children. <u>The Journal of Consulting and Clinical Psychology</u>, 1973, 40, 148-155.
- O'Connell, E., Dusek, J., & Wheeler, R. A. A follow-up study of teacher expectancy effects. <u>Journal of Educational Psychology</u>, 1974, 66, 325-328.
- Piaget, J. The child's conception of the world. Paterson, N.J.: Littlefield, Adams & Co., 1960.
- Pidgeon, D. A. <u>Expectation and pupil performance</u>. London: National Foundation for Educational Research, 1970.
- Rappaport, M. M., & Rappaport, H. The other half of the expectancy equation: Pygmalion. Journal of Educational Psychology, 1975, 67(4), 531-536.

2 -

- Regan, D., & Totten, J. Empathy and attribution: Turning observers into actors. Journal of Personality and Social Fsychology, 1975, 32, 850-856.
- Rosenshine, B. <u>Teaching behaviours and student achievement</u>. London: National Foundation for Educational Research, 1971.
- Rosenshine, B. Classroom instruction. In N. L. Gage (Ed.), <u>The</u> <u>Isychology of teaching methods</u>. National Society for the Study of Education, 75th Yearbook. Chicago: University of Chicago Press, 1976.
- Resenshine, F. Review of teaching variables and student achievement. In G. D. Borich (Ed.), <u>The appraisal of teaching: Concepts and</u> process. Reading, Mass.: Addison-Wesley, 1977.
- Kosenshine, B., & Furst, N. The use of direct observation to study teaching. In R. M. W. Travers (Ed.), <u>Second handbook of researce</u> on teaching. Chicago: Rand McNally, 1973.
- Rosenthal, R. The pygmalion effect lives. <u>Psychology Today</u>, 1973, 7(4), 50-03.
- Hosenthal, R. On the social psychology of the self-fulfilling prophecy: Further evidence for pygmalion effects and their mediating mechanisms. New York: MSS Modular Publications, 1974.
- Rosenthal, R. Experimenter effects in behavioral research (and Ei.). New York: Irvington, 1976.
- Rosenthal, R., Baratz, S. S., & Hall, C. M. Teacher behavior, teacher expectations, and gains in pupils' rated creativity. <u>The Journa</u> of Genetic Psychology, 1974, <u>144</u>, 115-121.
- Rosenthal, R., & Jacobson, L. <u>Pygmalion in the classroom</u>. New York: Holt, Rinehart & Winston, 1965.
- Ross, L. The intuitive psychologist and his shortcomings: Distortions in the attribution process. Unpublished manuscript, Stanford University, 1976.
- Rothbart, M., Dalfren, S., & Barrett, R. Effects of teacher's expectancy on student-teacher interaction. <u>Journal of Educational</u> <u>Psychology</u>, 1971, <u>62</u>, 49-54.
- Rowe, M. Wait time and rewards as instructional variables, their influence on language, logic, and fate control. <u>Journal of</u> Research in Science, 1974, 11(4), 291-308.
- Rubovits, P. C., & Maehr, M. L. Pygmalion analyzed: Toward an explanation of the Rosenthal-Jacobson findings. <u>Journal of</u> <u>Personality and Social Psychology</u>, 1971, <u>25</u>, 210-218.

Ryans, D. G. <u>Characteristics of teachers: Their description</u> <u>comparison</u>, and <u>appraisal</u>. Washington, D.C.: American Council on Education, 1960.

.

- Schmuck, R. A., & Schmuck, P. A. <u>Group processes in the classroom</u>. Dubuque, Iowa: Wm. C. Brown, 1975.
- Schroeder, D. A., & Linder, D. E. Effects of actor's causal role, outcome severity, and knowledge of prior accidents upon attributions presponsibility. Journal of Experimental Social Psychology, 1976, <u>12</u>, 340-356.
- Seligman, M. E. P. <u>Helplessness</u>: On depression, development and death. San Francisco: W. H. Freeman, 1975.
- Shavelson, R., & Dempsey-Atwood, N. Generalizability of measures of teaching behavior. <u>Review of Educational Research</u>, 1970, <u>40</u>, 553-011.
- Shaver, K. J. <u>An introduction to attribution processes</u>. Cambridge, Mass.: Winthrop, 1975.
- Shaver, K. G. <u>Principles of social psychology</u>. Cambridge, Mass.: Winthrop, 1977.
- Smith, F. J., & Luginbuhl, J. E. R. Inspecting expectancy: Some laboratory results of relevance for teacher training. <u>Journal of</u> <u>Educational Psychology</u>, 1970, <u>68</u>(3), 265-272.
- Smith, M. Attribution, achievement motivation, and task difficulty in physical education. Paper presented at the Third Annual B.C. Conference on the Teaching of Physical Education, University of Victoria, B.C., 1977.
- Smith, L. M., & Geoffrey, W. <u>The complexities of an urban classroom:</u> <u>An analysis toward a general theory of teaching</u>. New York: Holt, Rinehart & Winston, 1968.
- Snyder, M. Attribution and behavior: Social perception and social causation. In J. Harvey, W. Ickes, & R. Kidd (Eds.), <u>New</u> <u>directions in attribution research</u> (Vol. I). New York: Wiley, 1976, pp. 53-72.
- Snyder, M. L., Stephan, W. G., & Rosenfield, D. Attributional egotism. In J. Harvey, W. Ickes, & R. Kidd (Eds.), <u>New directions</u> <u>in attribution research</u> (Vol. II). New York: Wiley, 1978, pp. 91-120.
- Soar, R. S. <u>Final report, follow-through classroom process</u> <u>measurement and pupil growth (1970-71)</u>. Gainesville, Florida: University of Florida, Institute for the Development of Human Resources, 1973.

Sohn, D. Affect-generating powers of effort and ability self attributions of academic success and failure. Journal of Educational Psychology, 1977, <u>69</u>(5), 500-505.

- Stake, R. The **case** study method in social inquiry. <u>Educational</u> <u>Researcher</u>, 1978, 7(2), 5-8.
- Stallings, J. A., & Kaskowitz, D. <u>Follow-through classroom</u> <u>observation evaluation, 1972-3</u>. Menlo Park, California: Stanford Research Institute, 1974.

.

- Steele, C. M., & Woods, L. Trait attributions and defense against insult from a dissimilar other. Journal of Research in Personality, 1977, <u>11</u>, 318-328.
- Storms, M. D., & McCaul, K. D. Attribution processes and emotional exacerbation of dysfunctional behavior. In J. Harvey, W. Ickes, & R. Kidd (Eds.), New directions in attribution research (Vol. I). New York: Wiley, 1976, pp. 143-164.
- Thorndike, R. L. Review of pygmalion in the classroom. American Educational Research Journal, 1968, 5, 708-711.
- Iravers, R. M. W. (Ed.). Second handbook of research on teaching. A project of the American Educational Research Association. Chicago: Hand McNally, 1973.
- Van Manen, M. An exploration of alternative research orientations in social education. <u>Theory and Research in Social Education</u>, 1975, 3(1), 1-28.
- Webb, E. J., Campbell, D. T., Schwartz, R. D., & Sechrest, L. <u>Unottrusive measures: Nonreactive research in the social sciences.</u> Chicago: Rand McNally, 1966.
- Webster, M., & Sobieszek, B. Sources of evaluations and expectation states. In J. Berger, T. T. Conners, & M. H. Fisek wis.), <u>Expectation states theory: A theoretical research program</u>. Cambridge, Mass.: Winthrop Publishing, 1974, 115-150.
- Weiner, B. (Ed.). <u>Achievement motivation and attribution theory</u>. Morristown, N.J.: General Learning Press, 1974.
- Weiner, B. An attributional approach for educational psychology. Ir. L. Shulman (Ed.), <u>Review of research in education</u> (Vol. 4). Itasca, Ill.: F. E. Peacock, 1976.
- Weiner, B., Frieze, I., Kukla, A., Reed, L., Rest, S., & Rosenbaum, R. M. <u>Perceiving the causes of success and failure</u>. New York: General Learning Press, 1971.

- Weiner, B., Russell, D., & Lerman, D. Affective consequence's of causal ascriptions. In J. Harvey, W. Ickes, & R. Kidd (Eds.), <u>New directions in attribution research</u> (Vol. II). New York: Wiley, 1978, pp. 59-90.
- Weinstein, R. S., & Middlestadt, S. E. <u>Student perceptions of teacher</u> <u>interactions with male high and low achievers</u>. Unpublished manuscript, University of California, Berkeley, 1978.
- Willis, B. J. The influence of teacher expectation on teacher's classroom interaction with selected children. Unpublished doctoral dissertation, George Peabody College, 1970.
- Willis, S., & Brophy, J. E. Origins of teachers' attitudes toward young children. Journal of Educational Psychology, 1974, <u>66</u>, 520-529.
- Wilson, S. The use of ethnographic techniques in educational research. <u>Review of Educational Research</u>, 1977, <u>47</u>(1), 245-265.
- Winne, P. H., & Marx, R. W. Reconceptualizing research on teaching. Journal of Educational Psychology, 1977, <u>69</u>(6), 668-678.

ithall, J. The development of a technique for the measurement of social-emotional climate in classrooms. Journal of Experimental Education, 1949, 17, 347-361.

- Withall, J., & Lewis, W. W. Social interaction in the classroom. In N. L. Gage (Ed.), <u>Handbook of research on teaching</u>. American Educational Research Association. Chicago: Rand McNally, 1963.
- Wortman, C. B. Canal attributions and personal control. In J. Harvey, W. Ickes, & R. Kidd (Eds.), <u>New directions in</u> <u>attribution research</u> (Vol. I). New York: Wiley, 1976, pp. 23-52.
- Yinger, R. J. <u>Fieldwork as a basis for theory building in research on</u> <u>teaching</u>. Paper presented at AERA, Toronto, 1978.

APPENDICES

•

APPENDIX A

.

THE INTELLECTUAL ACHIEVEMENT RESPONSIBILITY SCALE



THE INTELLECTUAL ACHIEVEMENT RESPONSIBILITY SCALE

The Intellectual Achievement Responsibility Scale (IAR) developed by Crandall, Katkovsky, and Crandall (1965) provides a measure of student locus of control. The IAR contains two sets of questions; half pertain to attributions of responsibility for success (I+) and half to attributions for failure (I-). The total IAR score is a composite of internal attributions for both success and failure.

٠.

The IAR Scale

<u>.</u> .	Li a teacher passes you to the next grade, would it probably be
I+	a. tecause she liked you, or
2. . I+	When you do well on a test at school, is it more likely to be a. because you studied for it, or b. because the test was especially easy?
3.	When you have trouble understanding something in school, is it usually
I ~	a. because the teacher didn't explain it clearly, or b. because you didn't listen carefully?
	When you read a story and can't remember much of it, is it usuallya. Because the story, wish't well written, orb. because you weren't interested in the story?
5.	Suppose your parents say you are doing well in school. Is this likely to happen
I+	a. because your school work is good, or b. because they are in a good mood?
	Suppose you did better than usual in a subject at school. Would it probably happen
I+	a. because you tried harder, or b. because someone helped you?
7.	When you lose at a game of cards or checkers, does it usually happen
I-	a. because the other player is good at the game, or b. because you don't play well?

8. Suppose a person doesn't think you are very bright or clever a. can you make him change his mind if you try to, or Ib. are there some people who will think you're not very bright no matter what you do? 9. If you solve a puzzle quickly, is it a. because it wasn't a very hard puzzle, or I+ _____t. because you worked on it carefully? 10. If a boy or girl tells you that you are dumb, is it more likely that they say that _____a. because they are mad at you, or I- ____ b. because what you did really wasn't very bright? 11. Suppose you study to become a teacher, scientist, or doctor and you fail. Do you think this would happen I- ____ a. because you didn't work hard enough, or b. because you needed some help, and other people didn't give it to you? 12. When you learn something quickly in school, is it usually I+ ____ a. because you paid close attention, or b. because the teacher explained it clearly? 13. If a teacher says to you, "Your work \$s fine," is it a. something teachers usually say to encourage pupils, or I+ _____ b. because you did a good job? 14. When you find it hard to work arithmetic or math problems at school, is it I- ____ a. because you didn't study well enough before you tried them, or b. because the teacher gave problems that were too hard? When you forget something you heard in class, is it a. because the teacher didn't explain it very well, or I- ____ b. because you didn't try very hard to remember? 16. Suppose you weren't sure about the answer to a question your teacher asked you, but your answer turned out to be right. ' Is it likely to happen a. because she wasn't as particular as usual, or I+ ____ b. because you gave the best answer you could think of? When you read a story and remember most of it, is it usually 17. _____a. because you were interested in the story, or I+ | b. because the story was well written? If your parents tell you you're acting silly and not thinking 18. clearly, is it more likely to be I- _____a. because of something you did, or _____b. because they happen to be feeling cranky?

19. When you don't do well on a test at school, is it a. because the test was especially hard, or I- _____b. because you didn't study for it? 20. When you win at a game of cards or checkers, does it happen _____a. because you play real well, or I+ b. because the other person doesn't play well? If people think you're bright or clever, is it 21. a. because they happen to like you, or I+ _____b. because you usually-act that way? 22. If a teacher didn't pass you to the next grade, would it probably be a. because she "had it in for you," or I- _____b. because your school work wasn't good enough? Suppose you don't do as well as usual in a subject at school. 23. Would this probably happen I- ____ a. 'because you weren't as careful as usual, or b. because somebody bothered you and kept you from working? 24. If a boy or given lls you that you are bright, is it usually I+ ______a. because they like you? 125. Suppose you became a famous teacher, scientist or doctor. Do you think this would happen a. because other people helped you when you needed it, or I+ _____ b. because you worked very hard? Suppose your parents say you aren't doing well in your school 26. work. Is this likely to happen more I_____a. because your work isn't very good, or b. because they are feeling cranky? 27. Suppose you are showing a friend how to play a game and he has trouble with it. Would that happen a. because he wasn't able to understand how to play, or I- _____b. because you couldn't explain it well? When you find it easy to work arithmetic or math problems at 28+ school, is it usually _____a. because the teacher gave you especially easy problems, or I+ ____ b. because you studied your book well before you tried them? 29. When you remember something you heard in class, is it usually I+......a. because you tried hard to remember, or b. because the teacher explained it well? ¢

358 .1

- If you can't work a puzzle, is it more likely to happen 30.
 - I- _____a. because you are not especially good at working puzzles, or

____b. because the instructions weren't written clearly enough?

- If your parents tell you that you are bright or clever, is it 31+ more likely
 - _____a. because they are feeling good, or I+
- b. because of something you did?

44

- Suppose you are explaining how to play a game to a friend and 32+ he learns quickly. Would that happen more often
 - I+ ____ a. because you explained it well, or b. because he was able to understand it?
- Suppose you're not sure about the answer to a question your 33. teacher asks you and the answer you ve turns out to be wrong. Is it likely to happen
 - _ a. because she was more particular than usual, or b. because you answered too quickly? ₹.
 - If a teacher says to you, "Try to do better," would it be a. because this is something she might say to get pupile? to try harder, or I-____b. because your work wasn't as good as usual?

: ۲

Validity and Relievility .

Crandall et al. (1965:100) are cautious in their statements of validity and reliability. Modest reliability correlations are presented, namely, test-retest correlation of 0.65 and internal consistency correlation of 0.60 for students in the grade six range. However, Crandall et al. (1965:107) could only provide what they describe as significant but low construct validity correlations in the 0.20-0.30 range for the same students. Despite these data and other supporting psychometric findings as summarized by Lefcourt (1976:69-71), the IAR is frequently used in educational and psychological research, probably more so than other similar scales such as Battle and Rotter's (1963) Children's Picture Test of Internal-External Control and the Nowicki-Strickland Locus of Control Scale (Nowicki & Strickland, 1973).



GUIDELINES FOR STIMULATED RECALL INTERVIEWS

The guidelines for stimulated recall interviews were adapted from those developed by C. Cooper (1978) and Marland (1977).

Discuss with the Student the Goals of the Research

Little is known about the student's thought processes during instruction. The goal of this research is to find out the <u>thoughts</u>, <u>feelings</u>, <u>moment-to-moment reactions</u> and <u>perceptions</u> of the student during the instructional process (that is, cognitive interaction, classroom events, math content, instructional strategies, the teacher, the class, etc.). Viewing the videotape of the lesson facilitates recall of thoughts and feelings. Since individual student talk constitutes such a minor portion of classroom interaction, even the most perceptive observer cannot detect his/her emotional and intellectual reactions to classroom events; hor can the student's enjoyment of the lesson, understanding of the content and awareness of many classroom events always be inferred from his/her verbal and

It is considered that a study of these processes could yield insights which would assist in the development of theories of instruction which in turn could lead to the improvement of the eacher education program.

"How well the student behaves or how well he achieves is not the focus of the interview. Impress upon the student that the stimulated recall interview is not a memory test nor a test of any kind. He/she should consider the (taped) lesson as an ordinary lesson and behave or react as he naturally would.

Prior to the Interviews with the Student or the Teacher:

- 1. Preview the videotape of the lesson to identify those segments which appear most significant for investigating the thought processes of the student or teacher.
- 2. Arrange the interview setting so that the interviewee can look directly at the monitor and not be distracted by the interviewer.
- 3. Give the student or teacher time to become familiar with the operation of the VTR, have the student or teacher operate the VTR and spend some time in viewing the videotape, particularly if the student or teacher has never viewed himself on videotape before.

- 4. Explain the purpose of the stimulated recall interview and indicate clearly what is required of the student or teacher. Stress that the role of the student or teacher is to recall the thoughts, feelings, and moment-to-moment reactions experienced <u>during</u> the lesson and to indicate when a conscious decision was made and the reasons for making that decision.
- 5. Explain to the student or teacher that he/she will probably form new impressions of the lesson and of the events which occurred during the lesson while viewing the tape. Ask the student or teacher to try and distinguish during the interview between the thoughts and feelings experienced during the lesson and those experienced after the lesson or when watching the videotape.

Discuss with the Teacher the Goals of the Research

At the present time very little is known about the thought processes of those involved in the classroom teaching-learning situation. These processes are the focus of interest of this research project. The goal of this research is to find out what information the teacher and the students use during instruction and how they process this information. The interactions between the teacher and the students are of special interest.

It is considered that a study of these processes could yield insights which would assist in the development of theories of teaching and learning and which would lead eventually to the improvement of teacher education programs and school curricula. How well the lesson was taught and how well the student learned is <u>NOT</u> the focus of the interview.

Role of the Teacher/Interviewee

The method used in this research project to obtain data on teachers' information processing during instruction is called "stimulated recall." Asking teachers to recall after a lesson the thoughts and feelings they experienced while actually teaching the lesson has not proved very satisfactory. Recall of thoughts and feelings is facilitated when teachers are shown a videotape of the lesson. Seeing events in the lesson on videotape helps to trigger or stimulate recall--hence the term "stimulated recall."

Whereas it is possible to have people in some professions "think out loud" about their professional duties because they are not interacting with other people, it is not possible to do this with teachers because it would interfere with the instructional process.

We know that the mind works faster than the voice. As teachers interact with children in the classroom they--

a

- · become aware of many more classroom events than can be inferred from their verbal and nonverbal behavior;
- react to classroom events emotionally and intellectually in ways which even the most perceptive observer could not detect because they are internal--many reactions are not revealed to the observer;
- make numerous decisions about what to do and say next or at some future point in the lesson, or what not to do or say. The atternative courses of action considered, the reasons for the final choice of action are frequently not declared or revealed; the observer is not privileged with this "inside" knowledge and with the various rationales.

As the seacher relives the lesson by viewing the videotape, he is invited to provide a detailed account, to talk aloud, about:

- (a) thoughts, feelings, moment-to-moment reactions;
- (b) <u>concious choices</u> (that is, when you chose to do or say the thing rather than other things, or when you chose to "say or do nothing), the <u>alternatives</u> you considered before making a choice, and the <u>reasons</u> for choosing to do or say that particular thing.

Note:

- 1. You may stop and start the tape as often as you wish.
- 2. The interviewer may also stop the tape on some occasions to ask you if you can recall your thoughts, feelings, reactions, etc., in relation to certain classroom events.
- 3. The interviewer's role is simply to assist you to recall what you thought and felt during the lesson.
- 4. As you view the tape you will probably form new impressional of the lesson and of events which occurred during the lesson, and think of other things that you might have said or done. Try to distinguish during the interview between the thoughts and feelings you had during the lesson and those you had after the lesson or when watching the videotape; ensure that the interviewer is aware of the distinctions too.

If you have any questions, the interviewer will be pleased to discuss these with you prior to the interview.

Role of the Interviewer

In the stimulated recall session with the stident or teacher, the role of the interviewer is to assist the student or teacher to recall and verbalize the covert thoughts and feelings experienced during the lesson which has been videotaped. To facilitate as complete and as accurate recall as is possible the interviewer must:

- try to establish a relaxed, friendly, supportive atmosphere prior to and during the interview;

- try to facilitate and encourage self-discovery; **4**t is important for the interviewee to believe that he/she is capable of telling about inner processes without the interviewer telling the interviewee what they were;
- avoid making interpretations of, and judgments about, what appears on videotape; ask questions requiring elaboration or clarification but avoid questions answerable by "yes" or "no";
- assume a respectful set towards the student or teacher and the videotaped material; communicate to the interviewee that he/she is being taken yery seriously;
- keep the student's or teacher's attention focused on the TV image; refrain from unnecessary activity as such activity may actually interfere with recall;
- encourage the interviewee to talk; don't have the student or teacher become so engrossed in listening to you that the person forgets what he/she is reliving; the interviewee is the authority--you are that person's interested student;
- be patient; give the interviewee a chance to become involved in reliving the recorded lesson;
- immerse yourself in the interviewee's communication rather than trying to figure out what to say next;
- keep the student's or teacher's discussion focused on what transpired in the actual videotaped lesson and, in particular, on the student's or teacher's covert thoughts, feelings, and the sources of these; conscious decisions and reasons for making those decisions;
- stop the tape (if the student or teacher has not already done so) at points in the lesson where it appears likely to be profitable for purposes of this research and at the following points identified during a preview of the videotape:
 - 1. When the teacher asks a question of the student (or another student).

(.,

- 2. When the student's (or another student's) answer to a question is part-correct or incorrect.
- 3. When a student-initiated question (relevant) occurs.
- 4. When a student-initiated comment (relevant) occurs.
- 5. When the teacher responds to a student's answer.
- o. When there is a behavior-related teacher-afforded warning.
- 7. When there are non-verbal cues suggesting that the teacher (or student) is anxious, annoyed, perplexed, excited, enthusiastic, excited, etc.
- 8. When the lesson is not running smoothly.
- 9. When the student interacts with other students.

ask probing questions to facilitate maximum djsclosure by the student or teacher, for example:

- * What were you thinking, feeling at that point? Why did you say, do? Did you have any reasons for saying, doing? Did you understand what the teacher was saying, doing? What did you think the teacher was wanting, thinking,? Can you recall any other kinds of thoughts your mind? Were there any fantasies (daydreams) going your mind?
- Was there anything that you did not want to want? Was there anything that you wanted to do at that time?
- Note: questions should be brief and should create an intense awareness in the student or teacher of himself/herself. Avoid questions which are suggestive of, or imply criticism, incredulity, disagreement, disapproval, etc.
- check frequently that the student or teacher is differentiating between interactive thoughts and feelings and those subsequently formed.

APPENDIX C

PRE-TEST AND POST-TEST INTERVIEW SCHEDULES

PRE-TEST AND POST-TEST INTERVIEW SCHEDULES

.

Pre-Test Interview Schedule

	PRE-TEST AND POST-TEST INTERVIEW SCHEDULES	
	Pre-Test Interview Schedule	
1 •	What is coming up tomorrow? (the test)	
2.	How do you expect to get on?	
3.	Why do you think you will do?	
·+ •	What is there about the unit that makes you think you will and	
5.	How do you expect to feel? Why?	
ι.	Do you ever think of how others might be doing in the test?	
7.	How do you know how well they are doing?	
8.	Do you think they wonder about bow well you are doing?	
9.	Do you think the teacher will know how well you are doing?	
10.	What do you think she will think of you and motion geometry after the test?	
11.	Why do you think she will think this of you?	
12.		
•	Post-Test Interview Schedule	
1.	How did you get on in the test?	
2.	Were you surprised at the result? Why? Why not?	
3.	Why do you think you got?	
	What reasons would you give for getting	
4. E	How do you feel about this result?	
5. 4	Do you have any idea of how others got on?	
6.	How do you know how they got on?	
7.	How do you feel about them knowing your result?	
8.	What do you think the teacher thinks of you after this test?	
9.	Will the teacher be surprised about you and the math test?	
	You have a pretty good idea of how well you are doing in maths. Do you think the teacher has the same idea about you and maths as you do, or are her ideas different?	

APPENDIX D

SUMMARY OF CATEGORIES IN THE REVISION OF THE EXPANDED BROPHY-GOOD TEACHER-PUPIL DYADIC INTERACTION CLASSROOM OBSERVATION SYSTEM

.

1

44

• *

,

SUMMARY OF CATEGORIES IN THE REVISION OF THE EXPANDED BROPHY-GOOD TEACHER-PUPIL DYADIC INTERACTION CLASSROOM OBSERVATION SYSTEM

The summary of categories in the low inference observation system is adapted from Marland (1977). The main differences from the system used by Marland pertain to the private dyadic teacher-student contacts.

The major aspects of classroom life coded by this system are represented by the four cells in the diagram appearing below. Within each cell are the sul-categories of those four aspects which are then further broken down into still smaller units.

	Public Response Opportunities	Private Dyudic Teacher-Pupil Contacts
Teacher Afforded	۸.	C. I. Work-related II. Personal III. Procedure-related IV. Fehavior-related
Student Initiated	B. I. Student initiated questions II. Student initiated comments	D. I. Work-related II. Personal-related

ъ

A. Teacher Afforded Response Opportunities

The three key aspects of this category of classroom event are:

- (a) they are public interactions between the teacher and a child, intended to be monitored by the class or group with which the teacher is working;
- (b) they occur when the teacher asks a question requiring either a vertal or nonverbal response;
- (c) only one child makes the response.

For each response opportunity that is coded, information has to be checked off in each of four subcategories: (1) type of response opportunity; (2) level of question asked; (3) quality of child's answer; (4) nature of the teacher's feedback reaction.

(1) Types of response opportunity

Predesignated (PRE):	teacher names the child first and then asks a question;
Non volunteer (N. VOL):	teacher asks a question first but calls for a response from a child who has not raised his hand;
Volunteer (VÓL):	teach or asks a question first and invites a response from a child with hand raised;
Called out (CALL):	teacher asks a question but a child calls out the answer before the teacher has a chance to select a respondent; the teacher nevertheless responds to the child who called out the answer.
'evel of question asked	,

(2) Level of question asked

•

•

Process (PCSS):	question requiring student to integrate facts or show knowledge of their relationships;
Product (PROD):	question for which a specific correct answer is sought;
Choice (CHOIS):	question requiring an answer to tense selected from one of the alternations presented;
Sel: Reference (SELF REF):	question requiring child to make a non-academic contribution to the classroom discussion. This type of question has then to be further classified as <u>subject-matter related</u> (SUB) or non <u>subject-matter related</u> (NON SUB) and then whether it requires the child to show a <u>preference</u> (PREF) or to give information about his past <u>experience</u> (EXP).
Opinian:	question requiring student to take a position on an issue or to predict the outcome of an experiment or hypothet- ical situation. If the child gives <u>no response</u> (NR) this is coded. On the other hand if the child does respond, the teacher's reaction to

the answer is coded: if it is praised (‡), criticized (]), ignored (0), accepted (ACPT), integrated (INTEG) into the ongoing discussion, or if the teacher disagrees (DISAG) with the child's opinion.

(3) Quality of child's answer

The child's answer is coded as correct (+), partially correct (+), incorrect (-), or no response (NR) but, if the child indicates that he doesn't know, this item of information is also coded (DE).

(4) Nature of the teacher's feedback response

teacher's reaction to the child's response has been tegorized as terminal or sustaining. Reaction which is terminal, that is, it has the effect of terminating the interaction with the child, could be one of nine types. The teacher may praise (‡), criticize (=), ignore (0), give a simple affirmation (AFFIRM) or simple negation (NEGATE), give process feedback (PCSS), give the correct answer (GIV ANS), ask another (ASK OTH) child for the answer, or the answer may be called out (CALL) by another student. Reaction which is sustaining, that is, it has the effect of prolonging the interaction, could be one of four kinds. The teacher may repeat the question (REFT w), rephrase the question or give a clue (REP or CLU), ask a new question (NEW w), or repeat the student's statement (REF SS).

Student Initiated Response Opportunities

I. Student initiated questions

This category of response opportunity is used if the student asks the teacher a question regarding the subject matter under discussion or some other matter. If the student <u>calls out</u> (CALL) the question without prior teacher approval, this point is coded and also if the question is <u>relevant</u> (REL) or <u>irrelevant</u> (IRREL). Two kinds of teacher reaction to the question, <u>praise</u> (‡) and <u>criticism</u> (Ξ), are coded if they occur, and also types of teacher feedback. The teacher may provide <u>no feedback</u> (0) (that is, ignore the question), <u>delay</u> (DELAY) her answer, <u>not accept</u> (NACPT) it into the discussion, provide a <u>brief</u> or <u>long</u> answer or she may <u>redirect</u> (RDRCT) the question to another student. Three other categories, <u>praise</u> (‡), <u>criticism</u> (Ξ), and <u>warning</u> (WARN), are provided if the teacher makes a reaction related to the student's behavior in initiating the question.

II. Student initiated comments

The details surrounding a student initiated comment that are coded are very similar to those for a student initiated question. All but three teacher response categories, <u>brief</u>, <u>long</u>, and <u>redirect</u> (RDRCT) are retained. They are replaced by another three. The teacher may <u>accept</u> (ACPT) the student comment, <u>integrate</u> (INTEG) it into the class discussion, or may use it to shift the direction of the class discussion.

3. Teacher Afforded Dyadic Contacts

1 Teacher afforded contacts (work-related)

0

These are instances when the teacher makes private contact with an individual child about his work. Several features of these contacts are coded. The <u>duration</u> of the contact is timed. The contact may be one in which the teacher just <u>observes</u> (OBSERV) without entering into verbal interaction. The teacher may <u>praise</u> (\ddagger), <u>criticize</u> (\ddagger), <u>ignore</u> (0), give a simple <u>affirmation</u> (AFFIRM) or a simple <u>negation</u> (NEGATE), give <u>process feedback</u> (PCSS), or <u>give the correct answer</u> (GIV ANS). Alternatively the teacher may sustain the interaction. He may <u>repeat the question</u> (REPTW), <u>rephrase</u> the question or give a <u>clue</u> (REP or CLU), ask a <u>new question</u> (NEW w), or <u>repeat</u> the student's statement (REP SS).

11. Teacher afforded contacts (personal)

These contacts do not involve either work content or procedure but are of a strictly personal nature.

III. Teacher afforded contacts (procedure-related)

Within this category a distinction is made between those instances when a teacher seeks a <u>favor</u> (child helps in running the classroom) and those in which the request has to do with getting the child ready to work. The latter are coded as <u>management</u> (MANAG). <u>Thank you</u> (THANKS) is coded in the teacher thanks the child following the management or favor request.

IV. Teacher afforded contacts (behavior-related)

This category is used whenever the teacher makes some comment on the child's classroom behavior. They are subdivided into <u>praise</u> (‡), <u>non-verbal intervention</u> (NVI), <u>warnings</u> (WARN), and <u>criticism</u> (=). Errors which the teacher makes when warning a child are also noted. Three kinds of errors, target errors (TARG), <u>timing errors</u> (TIM), and <u>overreactions</u> (OVERT) are coded. The <u>no error</u> category is used whenever the teacher does not make one of the three errors. Provision also exists for the coder to record his <u>uncertainty</u> (?) if he is not sure that an error has occurred.

Student Initiated Dyadic Teacher-Pupil Contacts (referred to as Child Created Contacts on the coding sheets)

I. Child created contacts (work-related)

This type of contact may be initiated by a content comment (CONT C), a content question (CONT Q), a procedural comment (FROC C), or a procedural question (PROC Q). The teacher's feedback to the child is coded as either terminal or sustaining. The teacher may praise (1), criticize (2), ignore (0), give a simple affirmation (AFFIRM) or a simple negation (NEGATE), give process feedback (PCSS), or give the correct answer (GIV ANS). Alternatively, the teacher may sustain the interaction. He may give process feedback (PCSS), give a clue (GIV CLU), ask a new question (NEW Q), or repeat the student's statement (REP SS). Where sustaining feedback in the form of a question is provided by the teacher, the student may give a correct response (+), a partially correct response (+), or an incorrect response (-). For other teacher sustaining feedback the student may ask a further question (ASK Q), make an appropriate comment (GOMMENT), or simply acknowledge the teacher reaction (ACKNOW).

II. Child created contacts (personal-related)

•

In this category there are two first-order divisions, <u>experience</u> (EXP) sharing and <u>procedural</u> (PRCC). All experience sharing contacts are personal ones in which the student contacts the teacher to tell him something which is not related to either classroom work or procedure. The teacher's response is coded as either <u>acknowledged</u> (ACK) (that is, the contact is acknowledged by the teacher) or <u>delay</u> (that is, the teacher indicates she is unable to listen or talk to the pupil at that time).

A procedural contact occurs when the pupil is making a request, offers to do an errand, or reminds the teacher of something. The teacher's reaction is coded as grant or non-grant (N GRANT) (teacher has or has not granted the request) or as <u>delay</u>.

APPENDIX E

POST-OBSERVATION OPEN-ENDED STATEMENT COMPLETED BY THE TEACHER

.

1

•

÷

374

*

POST-OBSERVATION OPEN-ENDED STATEMENT COMPLETED BY THE TEACHER : A VERIFICATION OF VALIDITY

Following the observation phase, the researcher requested the teacher to write down everything she knew about or perceived regarding the intention and direction of the research. She was invited to classify her thoughts into (a) definite knowledge, (b) strong additional hunches, and (c) possibilities (no matter how slight the bases for thinking these). The following is the teacher's statement of her thoughts.

As requested in your letter of November 21, I submit the following as my knowledge or perception of your research project:

The project proposed to look at the thought processes and observable behaviour of the teacher and students during regular mathematics lessons. Lessons would be video-taped over a two-week period, then used to stimulate recall of thoughts during an interview. The information that was obtained from video-taping and interviews would later be coded and classified and analyzed; hopefully this would lead the researcher to some clues concerning thought processes during teaching and learning and how these thoughts influence us to do or think certain things. The use of stimulated recall had already been researched and proven viable in this type of study.

The study involved a two-week period of familiarization. During this time, the researcher made use of classroom observation, personal interviews, casual conversation and inventories to acquaint himself with the participants. To ensure that the video-taping would take place in as normal a setting as possible and that participants would act as naturally as possible, the video-equipment was utilized during this time, followed up with individual interviews. In short, "trial-runs" of the daily procedures of the next two weeks were carried out.

During the two weeks of actual taping, the researcher recorded math. lessons, usually in their entirety, meanwhile logging key points and interactions with the video meter. Outside the classroom, the researcher previewed the lesson, selecting suitable segments that would be used later to stimulate thought-recall. A variety of individual and small group interviews were conducted and recorded. Later the researcher would review the interviews, transcribing, coding and analyzing the data.

Signed: Teacher

November 23rd, 1978.

APPENDIX F

AMPLE OF LESSON OVERVIE


SAMPLE OF LESSON OVERVIEW : LESSON &

1

.

Ŧ

Time	VTR Counter	Indicators, Interactions, Questions
10:08	000	Tr: We will have some practice making some flip - images. Students move to get tracing paper
	010	Studends move to get the start
	; 016	Tr: Students given warning to hurry
	020	l di la constante di la consta
	021	Tr: Please wait and we'll do these together '
	025	Tr contact with Roger
10:10	· 0 3 0	Tr: Let's review as to how we use the Plexiglas.
	040	Tr: Holding it straight up and down
10:11	- 049	Tr: Can you with your finger trace along the Tr: Try and find the very most top point of the
10:12	059	Tr: Try and find the very most top point of the
	060 067	Trevor: Student question.
	070	
,	070	Tina: Student question.
	41-	Arthur.
10:13=	077	Tr to Trevor
	, 080	
	08 3	Rodney to teacher
	·	Tr with Cathy
	0.07	Tr with Cheryl Lisa out of desk
	087	LISA OUL OF LESK
10.15	090	New instructions to Rodney
10:15	, 097	Tr to Cherry]
	100	Rodney up with Trevoralso interacting with Tina
	107	Tr to Denis
	110	Lisa call out, Tr accepts.
	• .	Tr gives extra instructions to those finished.
	117	Richard to tr.
	120	Tr to Cheryl Greg to tr, "Look what I've got. It looks
	120	horrible!"
10 : 17	125	L. Tr to Denis Tr to the BBexplaining re sending a message using the Plexiglas.
	130	
	135	Thelma to teachershe had gone on ahead. Ricky to teacher.
	;	· · · · ·

377

[VTR	
Time	Counter	Indicators, Interactions, Questions
,		
	■ 140 141	Andrew to teacher Thelma given another worksheet.
	144	Tr with Lisa
	1 50	
	156	l a
0:20	160	Tr: Okay, jet's try the next example with tracing paper.
	163	Tr to Cathy.
	168	Denis: Student question, "Which one are we doing?"
	170	
1	178	Tri How are we going to check congruency?
• ,	180 188	Sandra responds.
	1 90	Tr: Now you are ready to flip your image.
	190	Denis: Student question.
	195	Tr used Denis' comment Rodney out of desk.
	_ 200	Tr to Roger
	203	Tr to Denis .
		Tr to Thelma
	207	Tr to Disa
	210 212	Trevar out of desk to Tina
	214	
0:23 [±] /2	215	Tr to Ria.
	217	Tr: Which did you prefer, the tracing paper or the
	220	Plexiglas.
	223	Tr: Let us do some checking with the Plexiglas.
	225	Prevor and Tina still interacting.
0.35	230	
0:25	233	Lisa call out. • Thelma to the teacher.
	238	Tr: All right. Go ahead and use whichever method you prefer.
	240	
	249	Denis: Calls for tr's attention across the room.
	250	
	253	Tr to Richard.
	260	•
0:27	261	Tr to Rodney
10:27	i i	
	270	Tina out of deak for tracing paper.
	277	
	280	Tr to Richard. Mike waiting by tr.
10:29	293	Mike talks with teacher.

•

ľ

1.1

•

.

۰.

٩

APPENDIX G

CONTENT ANALYSIS OF STUDENTS' INTERACTIVE THOUGHTS (CASSIT)

379

.

CONTENT ANALYSIS OF STUDENTS' INTERACTIVE THOUGHTS (CASSIT)

This content analysis system has been developed to enable the categorization of students' interactive thoughts and feelings during the learning of mathematics. The structure and format of the system, and its underlying rationale, reflect the direct influence of Conners' (1978a) and Marland's (1977) systems for analyzing teachers' interactive thoughts. However, the categories of the system have been generated from examination of the transcripts of the students' interviews and, in part, from a consideration of the major research questions of the study.

Researchers have long recognized the probable interdependence of students' task oriented and self-oriented thoughts and feelings while learning. An analysis of this covert behavior could suggest bases for, and provide cues to, a better understanding and appreciation of student learning in the normal classroom.

<u>Data</u>

Transcripts of students' interactive thoughts during a lesson contained the data to be analyzed. The transcripts were prepared from audio recordings of post-lesson interviews with each of the students. The post-lesson interviews involved largely the use of stimulated recall methodology. While viewing a video recording of a recently completed lesson, each student was asked to report his thoughts and feelings during the lesson. The vertalized recall of thoughts occurred at each of a number of stimulus points in the lesson chosen by either the interviewer or the student. Normally, on each occasion, the videotape was stopped. Following the student's reporting of thoughts and feelings during a particular stimulus point in the lesson, the interviewer often questioned the student further regarding some of the causal explanations and perceptions of his overt and covert behavior.

The data available in the transcripts, therefore, consists of more than just students' interactive thoughts. Data of a noninteractive form include:

- 1. Subsequent disclosures pertaining to causal explanations and perceptions of a student's behavior.
- 2. References by the student to thoughts or feelings experienced prior to the lesson (preactive data).
- 3. Thoughts and feelings which a student had after the lesson or during the post-lesson interview (postactive data).

4. Discourge during which the student endeavoured to explain an incident, situation, or circumstance for the benefit of the interviewer.

Coding

'There are two major steps in the content Analysis system:

- 1. Identification of interactive data.
- 2. Unitization. This second phase of the system usually involves two stages:
 - (a) identification of units of interactive data, and
 - (b) placement of units of interactive data into categories. •

Step 1 : Identification of Interactive Data

The first step in using the content analysis system is to identify those thoughts and feelings that the students report as having occurred during the lesson. These thoughts and feelings are the interactive data and are to be distinguished from the non-interactive lata, that is, those thoughts and feelings that occurred before or witer the lesson. Non-interactive data is not to be coded.

Before each interview the student was reminded that the interviewer was seeking recall of only the thoughts and feelings which occurred during the lesson. In addition, the student was asked to indicate when non-interactive thoughts were being verbalized. While the student was able to contribute in this way in some cases, the infferentiation between interactive and non-interactive data remains largely the task of the coder.

However, in the transcripts, many clues make easy the task of distinguishing interactive from non-interactive data. Examples of clues which indicate interactive data are:

- S: I was just thinking about . . .
- S: I was sort of feeling
- I: What were you thinking then?
- S: I was kind of laughing at him in my mind. He looked be
- I: When you first saw the worksheets what were your thoughts at that time?
- S: I'd probably get through it in short time. It usually is quite easy.

>

1

Likewise, many clues in the transcripts indicate noninteractive data:

- S: I can't remember going like that . . .
- S: I think I was thinking a bit about Richard
- S: . . . and I was kind of upset last night because I thought . . .
- I: You were saying?
- S: Me and Ruth have a habit of putting our pencils in our mouths.

<u>Guidelines</u>. In addition to the obvious clues for distinguishing between interactive and non-interactive data, there is available a set of appropriate guidelines developed by Marland (1977). The guidelines specified below are modified slightly from Marland's set. They should be used especially when the clues illustrated above are less clear or not present. In the examples of non-interactive data provided with each guideline, any included interactive data have been underlined.

Guideline 1

Label as non-interactive those sections of the transcripts in which the student is recalling what he was saying or doing, or what he had said or done, rather than what he was thinking.

Example:

Examples:

- I: Can you remember what you were thinking at that time?
 - S: About the fight. I was looking at what they were doing.
 - S: I was just finishing the first part when <u>I heard</u> her say, "Example B."

Note 2: When the student is describing or referring to a learning experience and his description contains reasons,

E: . . so I looked in my notes, my book to see what it looked like.

Note 1: Statements indicating that the student saw, heari, or sensed what another student or the teacher was saying or doing are to be coded as interactive since it is clear that the student was processing other people's behavior during that segment of the lesson.

purposes or consequences, then the data are to be coded as interactive.

Example:

S: Well...like just before I went up, and we were all talking like, because we were talking about the picture like, we couldn't get it exact. So he went up first to ask so that I decided that I might as well go up to find out.

Guideline -

Label as non-interactive those sections of the transcripts in which the student is showing awareness of what he was doing rather than of what he was thinking.

Example:

5: I can't remember. But I remember me shaking my head, 'cause I saw her looking at me.

Guideline_3

Later as non-interactive those sections of the transcripts in which the student is engaged in general discussion about learning, and situations in the learning process that sometimes arcse, where it is obvious that the discussion is not related to the interactive thoughts of the student.

Example:

3: You know, I know symmetry, and she...I would just ask what the question was again and she would tell me and then I would know.

Guideline 4

Label as non-interactive those sections of the transcripts in which it appears the student is providing a reason, explanation, or rationale for what he was doing or saying.

Example:

S: I don't like eating, but sometimes things are so boring that I feel like doing something, so I eat. Sometimes I draw, but that draws my attention from the board and then I might miss something.

Guideline 5

Label as non-interactive those sections of the transcripts in which the student summarizes, restates, reviews what he or the interviewer said previously.

Example:

- S: I got it wrong a couple of times and went to another picture.
- I: Yes. So you...
- S: Went onto an easier question and just left the hard one.

Guideline o

Label as non-interactive those sections of the transcripts in which the student discloses causal explanations and perceptions of behavior in response to extended questioning by the interviewer. These disclosures tend to consist of personal justifications, rationalizations, motives, or emotions that underlie student interactive thoughts during learning. At the time of a lesson these processes would probably function in a subconscicus or reflexive manner, given the speed and complexity of most covert behavior.

Example:

- I: What do you do with the questions that are hard?
- S: Well, I try to figure them out, or go up to the teacher or raise my hand.
- I: How do you feel when you have to go up and see the teacher?
- S: Well, I like to wait when there is hardly anyone there. So that I'll feel better.
- I: Why does it make you feel better?
- S: Well, kids...I'm kind of shy a bit still.
- I: And you were going to say about the kids.
- S: Well, like they might laugh at the question I have to ask for.

Guideline 7

Label as non-interactive those sections of the transcripts where the student indicates any uncertainty about thoughts and feelings being interactive.

Examples:

S: Well, I think I was kin: p: . . . S: I guess I was thinking

Guideline 8

When classifying parts of the transcripts as interactive or non-interactive, look for contextual clues. Often decisions about the nature of sections of the data can only be reached after examining clues found in lines prior, or subsequent, to the section under scrutiny.

Guideline 9

When in doubt, classify the section of the transcript in question as non-interactive.

Step 2 : Unitization

According to Marland (1977), the unitization, phase of a coding system usually involves two stages, namely, the process of unitization or segmentation and the process of categorization. Segmentation consists of dividing the data source into units, such as a paragraph, sontence, or thought. Categorization involves placing each unit into the of several discrete, clearly defined categories.

As Marland (1977:304) points out, "the distinction between the two stages of segmentation and categorization is [often] blurred tocause the unit selected is best understood in terms of the categories. In fact, the categories largely establish the unit." Therefore, in coding the data of this study, the procedure to be used involved one of examining the transcripts and determining each unit on the halfs of a section of the data complying with one of the categories of interactive thoughts. The unit is only established when a segment of the transcript matches with the characteristics of a category.

The Unit. The unit of analysis in the content analysis system is what Marland (1977) and Conners (1978a) refer to as a thought or invational unit. Conners (1978a:352) defines this unit as "a remark or peries of remarks, which expresses a more or less complete idea, and serves a specified function." Consequently, the thought unit may comprise a single work, a part of a sentence, or an entire paragraph, so rong as it is centered on a single thought.

4

Each single thought unit is to be clearly delineated when coding the interactive data by enclosing in parentheses.

Example of segmentation of student thoughts:

- I: What were you thinking at that time?
- 5: (It's a crazy looking diagram) and (Crystal was trying to say that you could) [make a slide between two diagrams'. (There was no way you could.) (Just by looking at it you could tell.) (And she had to try it out,) (and I knew). ("Oh Crystal, you're nuts! There is no way you could make anything out of that like a slide.") (And I felt like going up there and changing it.)

Conners (1978a:353) refers to "false starts" and "mazes" as two common speech characteristics. However, because they do not meet the criteria of a single thought unit as defined, they are not to be unitized. Ĩ.

Examples of "false starts": (Examples are underlined)

- S: Well...I was only...(I was trying it out in my mind.)
- S: Well, I was going to, like say that, I didn't have, ... <u>Plexiglas you know</u>...(since it's quite thick and sometimes you have to move it around.)

Example of "mazes": (Examples are underlined)

S: (I was afraid I'd get laughed at if I got it wrong.) Like, because, when they, if I really want, because I wasn't sure, I didn't put up my hand.

<u>Categories</u>. From examination of the transcripts, nine categories of students' interactive thoughts and feelings have been created. Each category consists of a set of thought units with one or more distinct characteristics. Consequently, a thought unit may be categorized only once, given the discrete nature of the categories. The nine categories of interactive thoughts and feelings are: 1) subject matter, 2) cognitive processes, 3) behavioral moves--seli, 4) behavioral moves--student, 5) behavioral moves--teacher, 5) self-performance--thoughts, 7) self-performance--feelings, 8) feelings, and 9) non-task-related.

Given the purpose of this study the categories of behavioral moves--self, behavioral moves--student, behavioral moves--teacher, self-performance--thoughts, self-performance--feelings, and feelings were subcategorized.

1. Subject Matter

2

r.

Units in which a student's thoughts are focused specifically on the content and skills of mathematics are to be coded as "subject matter." Thoughts expressing ideas associated with the concepts," operations and applications of mathematics are typical of this category.

Example: (Shown in parentheses)

S: See, she asked if they were alike, (and just like they are all the same in size and that's it, they are congruent.)

When a student reports simply that he was thinking about math, the math question, or makes other general references to the subject area, then the units are not normally coded as subject matter. Such references usually occur in the context of being on-task, that is, attending, and are to be coded as "behavioral moves--self."

Care must be taken to exclude from the subject matter category any reference by the student to his thought processes involved when

386

learning the mathematical concepts, operations, and skills. These are coded as "cognitive processes." Similarly, units of student perceptions of task structure and of task difficulty are not to be coded in this category, but rather as a subcategory of "selfperformance--thoughts."

2. Cognitive Processes

A "cognitive process" is the unit in which the student reports a thought process involved in learning the subject matter. The thought process must include a mental action performed by the student as he engages in the task of learning. As covert behavior, these mental actions may be accompanied by overt behaviors, such as verbalizing, writing, and acting. Specifically, cognitive processes range from perceiving or being aware, to knowing or remembering, to understanding, to reasoning or justifying.

References to thinking how to do a task (a question, a problem, a worksheet) are included in this category.

Examples: (Shown in parentheses)

S: I remember that (I was thinking, I was doing on a piece of paper how it could be different.)

S: (I was trying to think how to do it.)

3. Behavioral Moves--Self

A "behavioral move--self" is a category in which the student reports his thoughts about an action he was performing, had performed, or was considering performing in relation to the learning process. Such an action must reflect a personality orientation, usually in the form of personal decisionmaking. Thought units should refer to actions involving a motive, desire, intention, interest, or some other self component.

A pair of subcategories of behavioral moves--self includes those references to avoiding or seeking public participation.

Example: (Shown in parentheses)

S: See I put up my hand...(I wanted to do the question.)

A further pair of subcategories refer to attending-listening and not attending-listening.

Examples: (Shown in parentheses)

S: (I was listening to what she was doing.)

I: What were your thoughts at that time?

S: Maths...(I was thinking about maths.)

Example: (Shown in parentheses)

S: (I didn't want to listen to the maths lesson.)

4. Behavioral Moves--Student >

A "behavioral move--student" is a category in which the student reports his thoughts and feelings about an action involving other students. The thought units may refer to either the other student as a person, the other student's behavior, or both.

Subcategories of behavioral moves--student include perceptions of other students' performances, inferences of other students' thoughts, perceptions of other students' behavior, and a direct interaction with another student or students.

Examples: (Shown in parentheses)

- S: (I saw that he had got his all right.)
- S: She was looking at me and (I could tell she was thinking she was stupid.)
- S: (Jenny was working hard) so (I decided to ask her what was going on.)

Thought units in which a student reports a self-performance thought in relation to an involvement by a significant other student are excluded from this category, and are to be coded as "selfperformance--thoughts."

5. Behavioral Moves--Teacher

A "behavioral move--teacher" is a unit in which the student reports his thoughts and feelings about an action involving the teacher. The thought unit may refer to either the teacher as a person, her behavior, or both.

Subcategories of such actions by the student include a sensory experience or a cognitive experience such as a perception, interpretation, or evaluation of the teacher's behavior, perception of the teacher's instructional moves, and a direct interaction with the teacher.

Examples: (Shown in parentheses)

- S: (I wondered why she began to talk quietly to us.)
- S: (I saw that she was asking all the kids without their hands up.)

Thought units in which a student reports a self-performance thought in relation to an involvement by a significant other teacher are excluded from this category, and are to be coded as "selfperformance--thoughts."

6. Self-Performance--Thoughts

Units in which the student is thinking about <u>his</u> performance behavior and outcomes are to be coded as "self-performance--thoughts." The essential element of these work-related units is self, where the student is (i) reflecting upon how well he has achieved thus far, (ii) monitoring his work performance of the moment, and (iii) contemplating his expectancies of future behaviors and outcomes.

Some subcategories of self-performance thought units emphasize aspects of self-evaluation. This cluster of subcategories consists of references by the student to self-assessment of success, selfassessment of failure, perceptions of task difficulty, and perceptions of task structure. Perceptions of task structure are those units which reflect how a student views the task before him. Does he believe success or failure on the task will depend on his own level of competence and input of effort (for example, "I could see how to do it") or does the student believe that success or failure on the task is controlled by factors external to the self (for example; "I couldn't see how to do it")? Elements of personal judgement of competence are contained in a task structure perception.

Examples: (Shown in parentheses)

S: (I thought I would get it all right.)

S: (I knew I didn't do too well and I got three wrong.)

S: (The worksheet was pretty easy.)

Thoughts which are causal explanations of self-performance behaviors and outcomes are to be coded as the subcategory of self-attributions. Likewise, thought units which are references to a student's future performance behavior and to his outcome expectancy of success or failure are to be coded as the subcategory of selfexpectations.

Examples: (Shown in parentheses)

S: (I was trying) but I just couldn't get it.

S: (I don't usually get it right.)

Another subcategory of "self-performance--thoughts" units relates to the contributing effect of significant others on performance behavior and outcomes. The primary element of these thought units is self, that is, the mind of the student is focused upon <u>his</u> performance behavior or <u>his</u> performance outcome. The significant other element relates to this <u>self</u> by way of acting as a perceived probable consequence, a causal explanation, or a guideline for self-evaluation. These units of thoughts often relate closely to the appropriate cluster of units in the category of "self-performance-feelings."

Examples: (Shown in parentheses)

- S: (If I got a lot wrong they would laugh at me.)
- S: (I didn't want anyone to see it because I knew I was wrong as soon as I done it.)
- 7. Self-Performance--Feelings

Units in which the student reports an emotion pertaining to <u>his performance behavior and outcomes are to be coded as "self-performance-feelings.</u>" As expressions of affective states in relation to self-performance, these units are to be coded as:

- (a) anxiety, e.g., concern, worry, fear;
- (b) morally neutral--positive, e.g., happiness;
- (c) morally neutral -- negative, e.g., unhappy, sad, disappointment;
- (d) morally unneutral -- positive, e.g., pride;
- (c) morally unneutral--negative, e.g., shame, embarrassment, jealousy.

Examples: (Shown in parentheses)

S: I said, ("Oh, I'm gone completely.")

S: (I was really happy with my result.)

- S: When I was wrong (I felt bad).
- S: (I felt proud for now my Mom could brag about me.)
- S: (I felt so stupid when I knew I was wrong.)

8. Feelings

The transcripts contain reports by students of the emotions they experienced during the lesson, other than those experienced in association with self-performance. These other emotions are expressed about many classroom happenings and usually take the form of pleasure, surprise, annoyance, frustration, apprehension, and anxiety.

Subcategories of (a) positive emotions and (b) negative emotions are to be used.

Examples: (Shown in parentheses)

- S: (I was really excited.)
- S: I didn't feel like listening (because it was getting boring.)

9. Non-Task-Related

Student thoughts which do not pertain to the learning of the subject matter are coded in this category. In general, the units reflect off-task thoughts such as investigator awareness, matters external to the classroom and the lesson, and self-involved and other student distractions.

Examples: (Shown in parentheses)

- S: (I was watching the camera to make sure that you douldn't pick me up, and suddenly you got me.)
- S: (I was thinking about the weekend, then Hallowe

Coding Guides

- 1. Using the guidelines provided, decide whether the data interactive or non-interactive.
- 2. Place parentheses around each thought unit within the set of interactive data.
- 3. Assign each interactive thought unit to a category. Note: Coding guides No. 2 and No. 3 usually will occur concurrently.
- 4. Where repetitive interactive data are noted, unitize with the parentheses but do not categorize. Mark with asterisks so as not to include them when quantifying the thought units.
- 5. Where one unit appears within another unit, place brackets around the inner unit and categorize separately.

Example:

S: (I wanted to question him [because I didn't understand it] and I wanted to ask him to help me.)